## Time, space, and intimacy in the

## everyday rhythms of Chinese

## mobile game players

# Yumeng Dou

## PhD

# **University of York**

Sociology

September 2023

## Declaration

Thanks to my supervisors Dr.David Beer, Dr.Richard Tutton. And TAP meeting member Dr.Haley McAvay.

Thank to all the interviewees who participated in this research.

Thanks to my family, friends, and the cat in the coffee shop.

I declare that this thesis is a presentation of original work and I am the sole author. This work has not previously been presented for a degree or other qualification at this University or elsewhere. All sources are acknowledged as references.

I acknowledge that I received assistance from professional proofreader to proofread this assessment in line with the Policy on Acceptable Assistance with Assessment

#### ABSTRACT

This study explores the impact of freemium mobile games on players' daily lives. Using online semi-structured interviews and Lefebvre's rhythmanalysis theory as the primary methodological framework, the research recruited 48 interviewees from mainland China during 2019-2020. Based on the interview responses, this thesis examines how mobile games schedules affect individual's perceptions of time, space and intimacy. First, the study demonstrates how in-game mechanisms or systems create in-game rhythms and how these rhythms integrate into players' everyday lives. Second, it shows how mobile games reshape and re-organise people's attention and time in daily life, and how these rhythms of the digital space interact with the existing rhythms of everyday life that already exist in physical space. Finally, the simultaneous engagement of players with both the rhythms of mobile gaming and daily life offers players limited freedom. The choice to adapt to the rhythms of mobile gaming can be regarded as an attempt to escape from the demands of non-game life. Ultimately, this thesis aims to contribute to the study of mobile gaming within the Chinese context, particularly in the field of sociology, and to provide insights for mobile game players and designers who are interested in this topic.

## **Content Page**

Chapter 1 Introduction	6
1.1. Freemium Mobile Games	17
1.1.1 Brief History of the Freemium Game	17
1.1.2 Freemium Mobile Games in the Chinese Market	20
1.1.3 Features of the Chinese Mobile Game Players	
1.2 The Social and Policy Context of Freemium Mobile Games	33
1.3 Conclusion	40
Chapter 2 Literature Review	41
2.1 Introduction	41
2.2 Game Studies on Playing Time	
2.3 Game Studies on the Life Space	63
2.3.1 The Early Game Space Research	63
2.3.2 Research on Hybrid Reality-Games Space	71
2.3.3 Lefebvre's Space theory to the Video Game Area	80
2.4 Online Intimacy	86
2.5 The Influence of Video Game Mechanics on Players' Daily Life	96
2.6 The Game Avatar and the Embodied Experience of Gaming	104
2.6.1 The Avatar in Gaming	104
2.6.2 Game Embodiment	
2.7 Development of Lefebvre's Rhythmanalysis Theory on Daily Life Research	125
2.8 Conclusion	135
Chapter 3 Methodology	137
3.1 Introduction	137
3.2 The Participant Information, Recruitment, and the Development of Rhythman	alysis
Method	
3.2.1 The Interview Structure and Questions	142
3.2.2 The Interview Process	143

3.2.3 The Build of Rhythmanalysis Method	148
3.3 Code and Data	150
3.3.1 Attribute Codes	150
3.3.2. Index Codes	152
3.3.3. Analytic Codes	157
3.4 The Researchers' Internalities and Limitations	161
3.5 Language Translate and Intercultural Context	166
3.6 Conclusion	168
Chapter 4 Establishing Gaming Rhythms within Everyday Life: From the in-screen World	to
the out-screen Physical World	169
4.1 Introduction	169
4.2 Linear and Cyclical Rhythm Performance in Mobile game	174
4.2.1 Linear Rhythm Performance on Mobile Gaming	174
4.2.2 Cyclical Rhythms Performance in Mobile Gaming	184
4.2.3 The Intertwining and Combination of Cyclical and Linear Rhythms	190
4.3 Polyrhythmia, Multitasking and The Automation of the Mobile Game	213
4.4 Create 'Your' Character—The Production of In-game Digital Avatar Rhythm	223
4.5 Conclusion	231
Chapter 5 Combining Rhythms, Eurhythmia and the Synergy of Multiple Rhythms: How	
Multiple Rhythms Work Together in a Hybrid World	234
5.1 Introduction	235
5.2 Switches in Mobile Gaming: Translating Between Digital and Physical Worlds	240
5.3 The Role and Function of Mobile Game Rhythms in Everyday Life	247
5.3.1 Gap-filling and Re-use of the Daily attention: How Mobile Gaming Rhythms	3
Can Reorganise Time	248
5.3.2 Game Rhythm for Creating the Collective Rhythm	263
5.3.3 Leaving, Escape, or Negotiation: The mobile game Rhythm Against Daily	
Repetition	274
5.4 Conclusion	287

Chapter 6 Arrhythmia, Discipline and Industrial Structural Conflicts in Muti-Rhythms of Mobile
Gaming and Everyday Lives
6.1 Introduction
6.2 The Performance and the Reason for Rhythm Conflict
6.2.1 Digital Feeding, Bodily Discipline and Disruptions
6.2.2 Player-Game Assemblage and Dual Presence in Digital-physical Worlds 314
6.2.3 City Rhythm and Migration of Spatial Rhythms on the Human Body
6.3 Conclusion
Chapter 7 Conclusion
Reference:

### Chapter 1 Introduction

This thesis aims to provide insights into the mobile gaming experiences of Chinese people and how those experiences influence their daily routines. As digital media becomes increasingly pervasive, many people have developed a new daily rhythms that enable them to navigate between virtual and physical worlds. In other words, they inhabit the hybrid space of physical and digital worlds.

With this context, this thesis aims to apply Lefebvre's (2013) rhythmanalysis method for exploring the possible influence of the mobile game to the player's daily life. Unlike traditional gaming methods, the mobile phones, as a typical digital space, provides access to gaming environments at any time and space. This ubiquitous access integrates mobile gaming behaviours interaction with physical activities in complex and diverse ways. Rather than being a crosssection of modern life, the gaming experience is entwined with the player's daily live, grounding itself over extensive period. The portability of smartphones enables gaming to occur either simultaneously or interspersed with many daily activities. Therefore, any examination of the gaming time or gaming space must consider interlacement with other aspects of people's lives, which necessitates the use of analytical tools that have a mesh-work feature rather than viewing these elements in an isolated cross-section.

The rhythmanalysis presents an advantage in analysing a mesh-work structure. It does not 'isolate an object, or a subject, or a relation', but rather "grasp a moving but determinate complexity " (Lefebvre, 2013, p. 12). According to Christiansen and Gebauer (2019), the core argument of

rhythmanalysis is "thinking distinct issues together" (p. 7). Based on Lefebvre's theoretical framework, cyclical and linear rhythms, along with bodily and social rhythms are mixed. His analysis is always linked to the composite nature of multiple rhythms. This thesis also focuses on the complexity and intersection of the digital and physical worlds.

Before applying for a sociology PhD degree, I held an official position as a video game journalist and operated a personal social media account with around 20,000 followers, where I reviewed popular video games. My role as a game writer involved discussing popular freemium games, attending industry conferences, and writing reports for my readers. Through this experience, I observed that the Chinese mobile game market heavily focused on revenue figures to represent a game's status. Industry professionals primarily focus on showcasing metrics such as the number of players, monthly active users, monthly revenues, retention rates and the players' willingness to pay.

As an insider in the game industry, I do not underestimate the importance of these metrics, such as player logins, revenue, and the time spent gaming. However, while sitting in the main conference hall during the annual report presentation of one of China's largest mobile game companies, I realised that game studies need to include more than these figures. There should be a more comprehensive framework for understanding the significance of these games to players and how they integrate into their daily lives.

Over the past decade, with the rise of smartphones, the transition from payfor-download to the free-to-play model has dramatically transformed players' daily experiences. The current China has the largest number of mobile gamers

in the world (Statista, 2021a, 2021b). Despite this, limited research exist on the impact of mobile games on individuals' lives and their psychological experiences of daily gaming. This gap raises the following questions: How do free-to-play games affect the players' lives in various aspects? Therefore, this thesis aims to deepen the current understanding of players' daily experiences of mobile gaming.

Video games allow participants to use avatars to interact with other in-game objects (e.g. items, scenes, characters) within the digital environment. Caroux et al. (2015) identified the core of player-game interaction as the dynamic between the player, facilitated by a tool (e.g. joystick, motion detection sensor, touchscreen), and a display screen (e.g. a games console, television, smartphone, or tablet). The terminal responds to the commands sent by the player and presents new information. This player–video game interaction is "the key point that differentiates games from other types of cultural media" (Caroux et al., 2015, p. 367). This interaction includes concepts such as engagement, immersion, presence and perceived realism (Caroux et al., 2015).

With the context of mobile games, this interaction can be considered as the precise measurement and rapid feedback of the player's behaviour. By completing daily tasks prescribed by the game, players earn rewards from the system. Building on the traditional video game task design, mobile games have introduced a set of in-game objectives for the player, often referred to as daily goal or daily task system. When players switch into the digital space constructed by the mobile game, they are presented with a series of pre-designed targets. Each time a player completes a designed goal, they receive

a corresponding reward from the game system. Completing these daily tasks generates resources (in-game currency, experience points, or materials) to feed the in-game character. For many players, as revealed in the interviews for this research, the development of in-game characters is the primary motivation to keep daily gaming.

With the advancement of gaming devices, the payment models for digital games have also developed, leading to the introduction of the freemium model. Unlike the traditional payment model for console or personal singleplayer games, where players pay once for a piece of game content that usually presents a relatively complete story, without the need for additional expenses during gameplay, the mobile gaming sector predominantly adopts the freemium model. This model allows basic access to games for free while enabling players to spend an almost unlimited amount of currency on in-game items, such as characters, weapons, and clothing. The model is designed to create an endless array of items for users to purchase for benefits, it I also means that it needs continuously engaging tasks to keep players involved in the gaming world. According to one of this research's interviewees who had rich experience in mobile game design, it challenging for a typical mobile gam e company to create large volumes of non-repetitive playable content. Thus, companies tend to rely on repetitive, cyclical, and probabilistic content to sustain the mobile gaming experience.

This thesis also investigates the potential for building online social relationships through mobile games. Social relationships are typically defined as situations where 'two or more people coordinate with each other so that their action, affect, evaluation, or thought are complementary' (Fiske, 1998).

Sociality is explained as the social relationships that pervade all aspects of an individual's life. Matsuo et al. (2017) showed the potential of video games to adopt the four principles of sociality: a modeling theory, verbal instructions (coaching), operant conditioning, and rehearsal.

These principles were proposed by King and Kirschenbaum (1992) to improve players' social skills. First, modelling refers to video games providing a nonplayer character as a role model to supervise the player's behaviour. Second, verbal instructions relate to the instant feedback that video games provide to players. Third, operant conditioning differentiates correct from incorrect player behaviours and gives rewards or penalties for corresponding in-game actions. Finally, rehearsal provides an environment for players to practice the given behaviour until they have learned the skill. Matsuo et al. (2017) also confirmed that practising/experiencing behaviours with sociality in-game enhances players' sociality in real life. In other words, simulated experiences in games can contribute to nurturing players' social skills in the real world. In the context of mobile gaming, the interaction between in-game characters and human players may simulate real-world relationships, potentially influencing players' perceptions of daily life.

To understand the role of gaming in players' everyday experiences, this thesis investigates the following three research questions:

1. How does playing freemium mobile games shape how users organise their daily lives?

2. How does the location of players use freemium games shape their perception of space?`

3. How do the process of playing these games and the in-game interactions shape players' intimate relationships?

Beyond these questions, this thesis specifically addresses the ordinariness of game playing. According to Crawford (2012), video gaming in everyday life is not merely a leisure activity confined to isolated times and locations; it is connected to social life even when the player is disconnected from the game console and screen. Crawford (2012) posited that gaming is "a source of memories, dreams, conversations, identities, friendships, artwork, storytelling and so much more" (p. 143). His work emphasised video games as an ordinary daily activity, possibly a 'mundane activity' (p. 148) that is part of everyday life. In his view, playing is not only a novel, exciting leisure activity requiring total concentration but also an ordinary, mundane routine that is closely interconnected with the rest of life.

In this context, traditional game design elements such as 'interaction', 'immersion', and 'sociality' should be re-visited in the context of accessible entry-exit gaming environments. These elements influence how mobile games function as a form of 'ordinary daily activity' (Crawford, 2012, p. 148) that interacts and negotiates with other daily activities and can significantly impact people's schedules, particularly in terms of time management and the use of interstitial time (Chess, 2018). Players may develop a tendency to multitask, allowing the game to run concurrently with other daily activities. Crawford (2012) illustrated this through a participant's experience, where the participant enhanced a character's skills in a game by physically modifying the controller to automate actions while engaging in other tasks, such as reading. This instance illustrates how player distraction can be attributed to subjective

choices. Semi-automation enables the game system to execute specific actions automatically and deliver results after a set period when players issue a command (e.g., engaging in combat). This mechanic restricts player participation during certain phases of gameplay, making distraction and the integration of mobile gaming into daily routines a widespread phenomenon among players of games with semi-automated mechanics. For players, these daily tasks are predictable and lack novelty, making their completion a mundane part of daily life. Gaming becomes an endless cycle of repetition and anticipation of what will happen next. This everydayness, rooted in ordinary life, is a key aspect of concern in this thesis.

Furthermore, this thesis also explores daily game-related rhythms and how they intertwine with mundane and ordinary daily routines. The respondents' perceptions of play and other life commitments, such as work, reading, cooking, and caring for a young baby, hold equal importance from the perspective of this thesis. Crawford (2012) argued that the use of video game technology and its integration into individuals' routines and patterns is influenced by pre-existing social hierarchies, with evidence suggesting that women tend to play in fragmented time slots due to longer working hours and heavier daily responsibilities compared with men. The thesis describes players' experiences of 'interstitial time' gaming. In these cases, mobile gaming and game-related rhythms were balanced alongside other daily life commitments, such as commuting, dining, intervals between lectures, and toilet breaks. These intertwined fragments demonstrate the interplay between game-related and non-game rhythms.

To address the research questions, I recruited 48 interviewees from online gaming forums in China and conducted one-to-one online interviews with them. Based on the data collected from these interviews, the thesis is organised into six main chapters and a conclusion chapter:

Chapter 1 briefly introduces the necessary background information on the history, market, and social environment of China's freemium mobile game industry. This thesis aims not only to focus on those who are already players but also for the readers who are unfamiliar with games and gamers. Thus, the thesis offers the latter group of readers a straightforward yet fundamental introduction to the Chinese mobile game industry in the opening chapter.

The first section of this chapter introduce the history and features of the freemium model and the profile of mobile game players in the current China context. It provides a quick review of the development of the freemium game model and present the typical popular mobile games which is usually mentioned in the interview data. Secondly, this chapter also introduces the China government policy and public opinion on video games. For a long period, the China mainstream opinion usually kept a relatively negative attitude toward the game and gamers, this public negative attitude may influence the player's time and space arrangement of gaming, in particular on avoiding playing in public spaces and the side of their parents.

The Chapter 2 reviews of the current literature on video game research about people's time usage, space, and intimacy. Chapter 3 discusses the research methodology used in this study; Due to the impact of the COVID-19 pandemic from 2019 to 2020, this thesis primarily employed online interviews as its main

data collection method. In total, 48 interviewees were recruited, and each of interview lasted between 30 and 210 minutes of semi-structured conversation. Chapter 3 first outlines the basic characteristics of the 48 interviewees and describes the process of data collection. It then introduces the coding method and the code analysis process. This chapter also addresses the impact of the researcher's role as an insider observer. The researcher's experience in the gaming industry provided an insider perspective on the analysis process and the interviewees also trends to gives the researcher credence due to the shared gamer identify. This section also discusses the existential influence of the researcher's insider role and perspective.

Chapters 4, 5, and 6 explore the themes of time, space, and intimacy within the daily rhythms of Chinese mobile game players from the perspective of rhythm analysis. Chapter 4 discusses the concept of 'daily tasks' in mobile games. Freemium mobile games usually require players to dedicate specific amounts of playing time and to perform certain in-game actions; the game system provides an accurate, executable action list for the players. If players do not adhere to this schedule, they risk losing the opportunity to earn unique rewards within the gaming community. For many mobile games, certain ingame rewards are only available on specific days of the week or at specific times of the day. Hence, these regulated in-game actions are interwoven with the players' daily routines outside the game. The influence of freemium mobile games on the process of organizing daily lives is further discussed in Chapter 4. This Chapter also explains the rhythm production process facilitated by common in-game mechanisms. It also introduces both linear and cyclic

rhythm performance of the mobile games as well as the role of in-game avatars in rhythm production.

Additionally, Chapter 4 discusses the automation mechanisms in mobile games that lead to a multitasking trend among players. Many mobile games do not require the player's full attention. In most player vs. environment mobile games, players usually could perform other daily activities while playing. This could influence the player's engagement in the digital world. According to Caroux et al. (2015), video game engagement is associated with immersion and presence, or perceived realism. Immersion is " a psychological state characterised by perceiving one's self to be enveloped by, included in, and interacting with an environment that provides a continuous stream of stimuli and experiences" (Stanney and Salvendy, 1998, p. 156), while presence is defined as " the subjective experience of being in one place or environment, even when one is physically situated in another" (Witmer and Singer, 1998, p. 225). In the context of mobile games with automation mechanisms, these mechanisms caused an intermittent gaming experience, diminishing immersion and presence compared to console or personal computer (PC) gaming.

Compared with consoles and personal computers, smartphones eliminate the restriction of users' geographical space to one meter in front of the device, thus extending the playing field from specific locations such as family rooms, personal bedrooms, and gaming halls to public spaces such as subways, buses, restaurants, and company offices. Consequently, this thesis hypothesises that the experience of playing impacts players' perceptions of public spaces. Chapters 5 and 6 cover the discussion on game spaces.

Mobile game players often build friendships with strangers in-game or strengthen existing relationships (such as with roommates and colleagues) by playing together. Besides person-to-person relationships, freemium mobile games also aim to build intimacy between players and non-player characters, as well as between players themselves. Games employ precise metrics to measure levels of intimacy, which are abstract in the real world but quantified and expressed accurately within the game world. Some players develop emotional attachments to game characters as if they were real-life entities. In the self-organised mobile game community, fan art by creative players attributes personality traits to these game characters. The portability of mobile games reduces the time cost of interacting with in-game characters compared with console and PC games. The continuous updates of mobile games offer the possibility of building long-term relationships between the player and the character.

The advancement of gaming technology, especially in imaging, has enabled the creation of game characters with appealing appearances. Game companies often endow virtual characters with lifelike traits, particularly in games targeted at female audiences, where 'falling in love with the character' becomes a key marketing strategy. Chapters 5 and 6 also explore the potential for intimacy between players and non-player characters within the mobile gaming context, as well as how in-game relationships may affect players' personal lives and their relationships outside the game. Specifically, Chapter 5 discusses the combination between daily life rhythms and mobile game rhythms. It mainly focuses on three key aspects of eurhythmia between in-game rhythms and daily routines rhythms outside gaming: the gap-filling

and reuse of daily attention, the creation of collective rhythms, and the player's 'pseudo-disengagement' from daily repetitions outside the game.

Chapter 6 examines the arrhythmia that experienced by mobile game players. These include the conflicts between the rhythms dictated by human biological and psychological needs; rhythms arising from social relationships with both human and non-human characters; rhythms influenced by geographical spaces; and rhythms generated by mobile game mechanisms.

Through the discussions in these three chapters, this thesis aims to offer insights into the daily gaming lives of people in China, as well as assist mobile game players from various countries in understanding their own experiences with mobile games.

### 1.1. Freemium Mobile Games

### 1.1.1 Brief History of the Freemium Game

The freemium business model allows the software to be downloaded freely from the app store but includes payment options within the in-app user interface. This business model can be traced back to the 1980s when computer software companies such as Adobe offered a 'light' version of their softwarem which required a registration key to unlock full features (Wagner et al., 2014). The term 'freemium' first appeared in 2006a, when Fred Wilson provided a detailed description of his preferred business model, which one reviewer suggested naming 'freemium'. On his article, Wilson (2006a) defines this term as following:

'Give your service away for free, possibly ad supported but maybe not, acquire a lot of customers very efficiently through word of mouth, referral networks, organic search marketing, etc., then offer premium, priced value-added services or an enhanced version of your service to your customer base ' (Wilson, 2006a, p. 1).

The 'freemium' model became widely used in online games, originating in South Korea and quickly spreading in Asian markets around 2005 (Sina Game, 2011, 2012). In this model, access to the game is free, and the game company's revenue relies on selling support items. According to Sina Games data (2012), in 2008, 45 of the 50 most popular online games adopted the item-based (freemium) model, and it took only about three years for the freemium model to dominate the Chinese online gaming market.

Within this context, the acquisition of mobile game items or rare characters has, to a considerable extent, adopted a form of gamblification (Brock and Johnson, 2021a). Typical examples include loot boxes (players can spend in-game currency to purchase boxes that randomly generate in-game props) or online slots and roulette. As Ross and Nieborg (2021) have pointed out, game companies design the acquisition of in-game rare rewards in a manner akin to social casinos, with roulette and lotteries becoming common methods for generating rare items. Brock and Johnson (2021a) citing Ross and Nieborg (2021)'s research and clarified that "the experience of playing slots, poker, blackjack and roulette is becoming a mobile experience; one which draws on existing games design monetisation strategies, such as engagement, retention, informality and the push/pull of casual gaming sessions" (p. 8). This monetisation method in mobile games signals significant changes in game

related consumption and player experience, which may also have influence players' daily lives— an issue this thesis addresses.

Nowadays, video games that adopt the freemium business model are often called free-to-play or pay-to-win games. These terms refer to the same group of games, which represent a significant portion of the current game market, However, the meanings of these terms vary depending on the context. Table 1 outlines the distinct differences between the application of these terms.

Table 1: Concepts of freemium, free-to-play, and pay-to-win games

Name	Feature
Freemium business model	The game offer basic services on the free version and provide the premium added services on the paid version ( <i>Wilson,2006a</i> ).
Free-to-play	Free-to-play games can be acquired and played free of charge while players are encouraged to buy virtual goods during gameplay (Alha et al, 2014, p.1).
Pay-to-win game	'A game where paying real money gives players a significant advantage over those who don't spend any money' (Stegner, 2021, p. 1).

According to Appshopper (2019), most popular mobile games adopt the freemium business model. In 2019, these freemium mobile games generated about US\$87.1 billion and accounted for 80% of digital game revenues in the global market (Statista, 2019). Thus, the freemium model has become the dominant approach in mobile gaming.

### **1.1.2 Freemium Mobile Games in the Chinese Market.**

In 2005, the massively multiplayer online role-playing game (MMORPG) Legend of Mir 2 developed in South Korea and operated by Shanda Games in China, became the first to adopt the freemium model in the Chinese market (Sina Game, 2012). However, the historic freemium game that propelled this business model into the public discussion was the Giant Interactive Group Inc's Zhengtu. This game reached a peak of 2.1 million concurrent online players, becoming one of the most popular online games of its time. Zhengtu became a famous game in Chinese gaming history for it introducing the freemium concept to both designers and players. In the public discussion of Zhengtu and Giant Interactive Group among game industry employees, a well-known game journalist Command likened Zhengtu's freemium business model and other online games to a battle between 'soldiers with modern thermal weapons and the ancient people used to physical combat' (Zhihu, 2014).

Unlike traditional Massively Multiplayer Online Role-Playing Games (MMO games), such as World of Warcraft, where players can only purchase game time, Zhengtu allowed, and sometimes encourage the purchase of the in-

game items with real-world currency, part of items only be purchased by outgame currency. This approach effectively broke the independence of the digital playing space, known as the 'magic circle' (Huizinga, 1938/2016), a fundamental concept in traditional game design philosophy. Consequently, the boundary between in-game and out-of-game experiences became blurred and indistinct. Huizinga (1938/2016) defined the 'magic circle' as a play space marked off beforehand, either physically or ideally, that separates gaming from non-gaming activities. This division segments 'temporary worlds' from the 'ordinary world'. Juul (2008) further explained the 'magic circle' as 'the boundary that players negotiate' (p. 62) and used it as a metaphor, likening it to a 'puzzle piece' that fits into different interfaces. In either sense, the exchange of in-game and out-of-game currencies reflects the monetisation of game elements and reshapes the gaming space.

Further, this exchange of out-game currency for in-game items laid the foundation for the freemium model, which has persisted from personal computer online games to mobile games. Over time, mobile games have supplanted PC online games as the dominant force in the Chinese gaming market. As of 2019, 95% of gamers in China play mobile games (Niko partners, 2019). By 2018, China had 598 million game players, predominantly is mobile gamers, contributing US\$15.63 billion to the global mobile gaming market (Niko Partners, 2019). The number of mobile gamers and the annual revenue were projected to increase by statistical agencies (Niko Partners, 2019). While there is not an exact percentage indicating how many mobile games adopt the freemium business model, as of 10th May 2023, all 30 of the top-selling games in the App Store's China region utilise this model. It is,

therefore, reasonable to conclude that the freemium model has become a staple in the Chinese mobile market.

This section next introduces common types of freemium mobile games and their representative examples in the current Chinese market. The representative games come from the interviewee-mentioned list. Based on the type of in-game payment, this thesis categorises these popular games into different categories. The first category is player versus player (PVP) games, where player-to-player interaction forms the core of daily gameplay. This category includes three types: multiplayer online battle arena (MOBA) games, shooter games, and asymmetric multiplayer games (Table 2). In the three types, in-game payments have little or no impact on the competition between players. The content of the freemium service focuses more on aspects such as the characters' appearance, actions, voice, and model. There is also no significant difference between the abilities of paid and free characters.

Player versus Player game		
Туре	The Most Popular or the most Representative Example	Game Description

Table 2 :Representative Player versus Player mobile game

The Multiplayer Online Battle Arena (MOBA) Games:	Honor of Kings	This game is a 5v5 MOBA game (similar to Dota and League of Legends). Each battle involves ten players split into two teams, with victory achieved by destroying the enemy's crystals.
Shooter Games	Game for Peace	Game for Peace is similar to a mobile version of Player Unknown's Battlegrounds. Players assume on the role of a soldiers and compete against each other.
Asymmetric Multiplayer Games	Identity V	In Identity V, players are divided into two or more groups of unequal numbers to compete. Different groups have unique objectives and gameplay mechanics.

The second category is the Massively multiplayer online(MMO) game category, which is a genre that has long existed in the traditional PC online gaming. This category involves two types of games: MMORPGs and MMO strategy war games (Table 3).

Massively multiplayer online game			
Туре	The Most popular or the Most Representative Example	Game Description	
Massively Multiplayer Online Role-Playing Game	Fantasy Westward Journey	Player assume roles and participate in the in- game events.	
Massively Multiplayer Online Strategy War Game	Immortal Conquest	Players act as lords and command the wars	

Table 3: Representative massively multiplayer online mobile game

For MMO role-play games, freemium items typically focus on objects such as weapons, equipment, pets, etc. Meanwhile, in massively multiplayer online strategy war games, payments can involve a three-party trade among employer players, employee players, and the game company. These types of games often include guild gameplay, where players may organise an external company structure involving dozens or hundreds of people to improve their ingame experience. The third category is ACG (Animation, Comics, and Games) game. This category includes games that are commonly referred to as '二次元' by Chinese players. A characteristic feature of this category is that the games usually develop an attractive in-game characters and monetisation through the sale of those characters or character-related items as the primary revenue source. The ACG category includes the following types: open-world games, tower defence games, action games, and turn-based role-playing games (Table 4).

ACG game		
Туре	Most Popular or the most Representative Examples	Type Description
Open-World Games:	Genshin Impact	The game features a vast map with multiple exploration locations, stories, and monsters. Such games focus more on player interaction with the in-game world.

Table 4 : Representative ACG Mobile Games

Tower	Arknights, The Path	Players act as commanders,
Defence	to Nowhere	strategizing the positions and actions of
Games		the in-game characters to prevent the
		enemies from reaching designated
		positions.
Action	the Honkai Impact	These games focus more on real-time
Games	3rd, Punishing: Gray	interactions between in-game
	Raven.	characters and the monsters.
Turn-Based	Onmyoji	Players manage of in-game characters
Role-Playing		and direct them in battle. Characters
Games		and enemies alternate actions in a
		turn-by-turn order.

A common feature among these games is their focus on producing attractive in-game characters and mainly selling these characters to players. In ACG mobile games, the freemium items mainly pertain to in-game characters.

The fourth category is female-oriented games. This category involves two main types of games: romance simulation games and dress-up games(Table 5). Unlike general ACG games, which tend to increase the character numbers, female-oriented romance simulation games typically feature a limited number of characters, normally four to five. ACG games focus on selling new characters and encourage players to build personal relationships with them. On the contrary, female-oriented romance simulation mobile games often emphasise the development of long-term relationships between players and characters. Thus, games in this category are more inclined to sell items to unlock new memory, scenes or events for developing relationships with the ingame characters.

Female-oriented game		
Туре	Most Popular or Representative Examples	Type Description
Romance simulation Games	Mr Love: Queen's Choice; Light and Night; Tears of Themis	Playes will develop a romantic relationships with the in-game characters.
Dress-up Games	Love Nikki-Dress UP Queen, Shining Nikki	Players could collect the clothes and accessories to dress up the in-game character

Table 5: The representative Female-oriented mobile game

Romance simulation games focus on developing relationships with in-game characters. Events between the player avatar and in-game characters are often represented by 'cards', the players must collect these cards of character to read the story between the corresponding character and avatar. The event, story, or relationship becomes the main item of consumption for players. And

in dress-up games, the freemium item mainly consist of clothes or accessories for the avatar (normally depicted as a young beautiful girl) of player.

In addition to these four categories, other types of mobile games are included in the 'others' category. Two main types of mobile games are put in this category: idle games and casual games (Table 6). The term 'idle' refers to a mechanism where in-game materials automatically accumulate over time. This mechanism is often integrated into various game types. In such games, freemium functions include the button of 'skip the waiting time' or 'increase the speed of waitting time two or threefold '. And the casual games normally include a series of challenge levels, with each level taking limited time or steps to complete, where the player can purchase item to help for overcome challenge.

The others		
Туре	Most Popular or the most Representative Examples	Type Description
Idle Games	One Hundred Scenes of Jiangnan	Players act as the mayor/administrator of a city/shop/building. The in-game units produce materials within a fixed period.

Table 6 : The Other Representative Mobile Games

Casual Games	Match-Three Games	These games normally include a series of challenge levels, with each challenge taking a liimited time or steps to complete

Regardless of the game type, there are normally two different methods to acquire in-game items. The first method is fixed-price selling, where virtual items are assigned a specific price, payable in either in-game or real-world currency. Fixed-price selling mechanisms are commonly applied to items that are low of value, common, or frequently used in-game. The second method of in-game acquisition is the Gacha model, also known as as the loot box mechanism. This approach allows players to spend a specific amount of virtual currency, after which the game system provides random feedback. Game study scholars Johnson and Brock (2019) described it as a new method of game monetization, as them said " where players spend real currency to purchase digital containers which yield an unknown set of items when opened. These can enhance a player's social status or gameplay ability but getting the desired item(s) is never certain" (p. 3). Over time, mobile game companies have developed a minimum reward protection mechanism to ensure players receive the items they desire after a certain number of attempts, commonly referred to as '保底' in Chinese. Gacha items are typically rarer or of higher value, and Gacha boxes are often the main source of income for most popular mobile games.

Virtual currency can be accumulated through in-game behaviour, including but not limited to completing quests, participating in events, and daily log-ins. However, the most efficient way to obtain virtual currency is still through purchasing it with real-world currency. The extent to which players are willing to pay for freemium content significantly influences the gaming experience they receive. From the popular mobile games introduced in this section, players may spend the real-world currency to acquire uniqueness, compete with others, completing challenges, foster in-game character relationships; or save time.

### 1.1.3 Features of the Chinese Mobile Game Players

According to a recent game industry report (China Game Publishing Committee, 2016, 2017, 2018, 2019), Chinese mobile game players exhibit the following characteristics:

**Age:** Young people are more likely to play mobile games. In 2012, 79.2% of mobile game players were under the age of 30 years old (China Internet Network Information Center, 2012). MobTech (2018) report showed that the main demographic of mobile game users is aged between 25-34 (40%), with

only 9% of middle-aged and older individuals (over 45) engaging in mobile gaming.

**Gender:** Males constitute a higher proportion, and the female mobile player base is growing. From 2015 to 2019, the number of female gamers increased from 230 million to 300 million. Female gamers in China accounted for 46.2%, with a consumption of 52.68 billion yuan (China Game Publishing Committee, 2019). In 2012, 61.6% of mobile game players were male (China Internet Network Information Center, 2012); however, as time has passed, an increasing number of females have shown interest in mobile games.

**Education:** The mainstream mobile game player typically has a high school diploma. According to the China Internet Network Information Center, the most common education level among mobile game players was a high school diploma (2012, 39.7%; 2013, 36.7%; 2014-2015, 30.2%). Also, MobTech (2018) revealed that high school diploma holders represent 42.6% of users.

Due to the widespread adoption of the freemium business model in the Chinese mobile gaming market, it is plausible to suggest that the characteristics of freemium mobile game players closely align with those of mobile gamers in general.

Furthermore, the spending patterns of freemium mobile players differ from those associated with traditional video games. Traditional game sales—via cartridges, discs, or digital codes—are typically set at a uniform price across a country or region, meaning each player pays the consistent price for a game.

However, the consumption model of freemium mobile gamers is closer to a pyramid, where a small number of player willing to pay high prices for additional services contribute significantly to a mobile game's revenue. In Western contexts, these spenders are called 'whales' (Carmichael, 2013), while in China, they are referred to as 'Huge R', with 'R' representing the first letter of the Chinese currency.

As an example, Table 7 introduces *Honor of Kings*, one of the most popular mobile games in the Chinese market, to show the typical revenue structure of mobile games. In this mobile game, more than half of the paying users spend less than 100 yuan (about 14 dollars) each. Meanwhile, 3.1% of users have contributed more than 5000 yuan (714 dollars) to the game in the last 30 days (Talking Data, 2017).

Payment Amount	Student Group	Working Group
Under 100 yuan	56.8%	56.7%
100-500 yuan	24.6%	24.7%
500-1000 yuan	7.5%	7.5%

1000-5000 yuan	8.0%	8.1%
Above 5000 yuan	3.1%	3.1%

Note. Source: Talking Data (2017).

Unlike traditional console or PC games, where each player can have a similar experience, some mobile games treat players differently based on their ingame spending. According to the consumption amount, many mobile games categorise players into different consumer levels, ranging from VIP 1 to VIP 12 or higher. A higher VIP level typically indicates that players have more privileges both in and out of the game (such as priority on customer service).

## 1.2 The Social and Policy Context of Freemium Mobile Games

In 1995, a famous game review article titled 'Crow, Crow, Caw' highlighted the reasons behind the lag in the Chinese gaming industry:

(1) The public perception of the gaming industry is unclear, and Chinese parents generally struggle to accept their children playing video games. Authoritative media have labelled video games as 'spiritual opium', and schools have issued letters to parents urging them discourage the student from playing video games.

(2) Chinese game industry lacks standardization management, and the market is rife with fierce competition. There is a disconnect order between production and sales. Due to piracy and vicious competition, game software

revenue cannot support the developer's reproduction. The profits and copyrights of the game manufacturers remain unprotected.

(3) The video gaming is a complex industry requiring development teams to include directors, choreographers, artists, programmers, sound engineers, and musicians, etc. Training these professionals on a large scale poses a significant challenge in the current Chinese context (GameSoftware, 1995).

The first and second arguments from this article remain relevant in current Chinese mobile gaming market. Even now, Chinese gamers face negative stereotypes and parental pressure, and the sustainability of mobile gaming companies, especially the small and medium-sized companies, in uncertain.

Three government policies have significantly impacted the development history of the Chinese video game industry. The first significant policy was the ban on the sale of game consoles. In June 2000, China's General Office of the State Council issued a document titled < The Special Management of Electronic Game Business Sites> which pointed out that:

Electronic games business place is overmuch, and some areas lack supervision and management..... it is serious harm to teenager's growth and disturbs the social order and public security. The electronic games business has become a public nuisance.

The production and sale of video game equipment and accessories for domestic use shall be stopped after this document's publication. Any enterprise or individual shall no longer enable the production or sale of electronic game equipment and accessories for domestic use.

(China Ministry of Culture, etc, 2000)

The ban on the sale of game consoles lasted approximately 15 years. On January 24, 2014, China Shanghai implemented a regional policy to allow the sale of game consoles (China State Council, 2014). A year later, the nationwide ban on game consoles was fully lifted after 15 years, following the issuance of the notice from the China Ministry of Culture (China Ministry of Culture, etc., 2015).

This policy's significantly influenced the gaming industry, as seen when comparing China's game market renvenue structure with other countries. In 2019, the Chinese mobile game market generated 158.11 billion yuan in revenue, it constituting 68.5% of game industry;s total revenue. Meanwhile, console and PC games respectively accounted for just 0.3% of revenues (China Game Publishing Committee, 2019).

According to the Interactive Software Federation of Europe (2021), 54% of players engage with consoles and 52% with PCs (Figure 1). However, the data for 'playing on smartphones or tablets' stands at 62%, slightly higher than the former categories, indicating no significant disparity among the three options. In the global market, mobile gaming contributes \$68.5 billion of the total \$152.1 billion, representing 45% of the entire gaming industry (Figure 2). Meanwhile, console and PC games account for 32% and 23%, respectively (Newzoo, 2019).

These figures highlight how the Chinese game market significantly differs from Europe and the global market. In China, mobile gaming occupies a substantially larger market share and generates more revenue compared with the rest of the world, while console and PC gaming penetration remains
relatively low. The Game Machine Sales Ban policy may have led more people to familiarise themselves with video games through smartphone device, and as a result, the freemium model had become the predominant approach to player's understanding for in-game consumption.

Figure 1 : Revenue of difference devices in the European Game Market (Source: Interactive Software Federation of Europe's report, 2021)



Figure: 2 Revenue of different devices in the global game market (Source: Newzoo, 2019)



Before the ban on game consoles was lifted, a portion of the population bypassed the policy restrictions by purchasing game consoles overseas or acquiring pirated game CDs. Pirated game CDs are extremely cheap to produce, easy to distribute on the street, and difficult to bring under regulatory scrutiny; in contrast, licensed games lack effective distribution channels due to policy restrictions. As a result, pirated, affordable, and potentially problematic video games have dominated the market, which hovers on the border between legality and illegality. Since the ban targets the 'production and sale' rather than individual purchases for entertainment, it is illegal for the vendors to supply games, but not for gamers to buy them (China Ministry of Culture, etc., 2000). This market for the purchase and sale of games has persisted in mainland China for a significant duration.

The second policy is the game license number policy (China National Press and Publication Agency, 2021). This policy mandates that game companies must submit their game content to the government for publishing permission. On 29th March 2018, the Chinese government suspended the distribution of game license numbers, and a suspension that lasted about eight months until the 29th of December. In 2017, the Chinese government authorised the publication of 9,369 games. By 2019, this number had fallen to fewer than 2,000. A similar scenario unfolded from August 2021 to March 2022. In 2022, around 70 to 80 games were permitted for publication each month (China National Press and Publication Administration, 2022).

At present, predicting the impact of the game publishing policy on the video game market is challenging. People's Daily Online (2019) revealed that 9,705 gaming companies went out of business in 2018, and in 2019, the number rose to 18,710. This concern is not unreasonable—as of now, no reliable sources have clarified the standard criteria for game licensing, leaving game developers uncertain about which games can be published and which cannot. This uncertainty publishing standards lead to high risk for the development of mobile games companies, especially for small and medium-sized enterprises.

This policy, alongside the console ban, has created division among those discussing the gaming industry in China. The game license number policy has led to many globally popular video games not being legally available in Chinese stores, with only a fraction of console games obtaining licenses. However, mainland Chinese citizens have found ways to bypass restrictions on purchasing games by adopting foreign online nationalities, such as Argentine or Russian, when interacting with game providers, such as Steam. Nintendo e-shop or PlayStation shop. While such practices are not typically deemed illegal (the worst consequence is the potential suspension of an

online account), they should be viewed as a privilege contingent on one's perception, education, second language proficiency, or information-gathering skills.

The third policy is the notice to prevent minors from being addicted to online games. This notice published on November 21, 2019, by the China National Press and Publication Agency (2021), introduced three main provisions: First, all online game users are required to use valid identity information for game account registration. When playing for more than one hour, they must provide their real name and personal ID information for system verification. Second, online game companies are prohibited from offering game services to minors from 22:00 to 08:00. The duration of game services available to minors must not exceed 3 hours per day during holidays and 1.5 hours on school days. Third, online game enterprises are forbidden from providing paid game services to users under the age of 8, and the amount minors can top up each month will be strictly limited.

To comply with this notice, each player must submit their identification number within the game to confirm that they are over 18 years of age (China National Press and Publication Agency, 2021). Players from other countries cannot legally access games offered in the Chinese market, and those who have verified their legal identity will receive an in-game notification if they play for 3 or more hours per day. This notice demonstrates the government's stringent attitude to regulating for mobile games, albeit in a subtle manner.

Besides that, the preference of people to play mobile games may be related to the current working hours system in China. Many workers, including several

interviewees in this study, face the serious issue of excessive overtime in their daily lives. Some companies have adopted the '996' working hours system, which entails 'starting at 9 am, finishing at 9 pm, and working 6 days a week' (BBC News, 2021). This results in a 72-hour workweek, leaving employees with little time for leisure activities.

Furthermore, the employees also experience a loss of time beyond the official working hours boundary (Nowotny, 2018). Modern information technology provide the ability to people remain connected to their work even outside of work hours. In China, whether for work or personal communication, the same software (WeChat) is used, making it challenging to disconnect from work by closing the information channel. This situation makes escaping the work-life imbalance even more difficult.

This state of overwork diminishes the possibility of engaging in leisure activities that demand significant time and concentration. However, smartphone leisure can meet the need for short and convenient entertainment. The data from this thesis also reveal that several respondents played mobile games during working time, which proved the adaptation of mobile games to employees during their limited leisure time.

# **1.3 Conclusion**

This chapter provides a brief overview of Chinese mobile gaming and mobile game market. In the following chapters. this thesis elaborates on the literature pertaining to time usage, space, and intimacy within the field of gaming, with a particular exphasis on the players' life experiences. To explore the complex relationship between mobile gaming and players' everyday lives, this thesis

employed Lefebvre's (2013) theory of daily rhythms as the analytical framework to analyse the data provided by the 48 interviewees' semi-structured interviews. It examined both the rhythms of non-gaming social life and those of mobile gaming.

### **Chapter 2 Literature Review**

### 2.1 Introduction

As introduced in the previous chapter, mobile games allow players to connect to the game world anytime. Traditionally, the time and space for video games were relatively fixed, with gameplay restricted to a few square meters in the living room or bedroom. This spatial limitation also imposed a time restriction, as players typically played indoors. However, due to the ban on the sale of gaming consoles in China from the 2000s to the 2010s, portable game machines could not be legally sold. Thus, smartphones have become the primary means for gamers to expand their gaming space beyond the home. As the availability of smartphones increased, gaming spaces expanded from the confines of the home to public places such as cafes, bus stops, restaurants and corporate offices.

Hjorth and Richardson (2009) noted that although mobile devices are used outside the physical confines of the home, perceptions of home and place still influence users. According to them, the understanding of the home is both a 'geo-imaginary' and a 'socio-cultural precept' (p. 371). This spatial flexibility also supports temporal flexibility, allowing the gaming space in the mobile

gaming context to expand more extensively into the social sphere than during the traditional era.

In his framework, Lefebvre (2013) outlined the basic pattern of poly-rhythms between commoditised media from afar and the human being as a subject across spatial distances. He described how individuals can join rhythms produced by media from afar and rhythms of their local geospatial environment simultaneously. These rhythms can function in conjunction across spaces. Lefebvre (2013) referred to 'modern television or radio' (p. 47) as examples of instantaneous media that require the user to turn them on to receive a message. However, with advancements in communication technology, the mode of receiving information has gradually shifted from immediate to time delayed. Warde et al. (1998) introduced the concept of 'hypermodern convenience devices' to describe how modern machines affect people's schedules. They used the examples of email and video recorders, both of which provide temporal flexibility. Whereas people could only receive information in real time by telephone, email allows the exchange of information at a time freely determined by the recipient. These 'hypermodern convenience devices' enhance an individual's ability to reorganise and reallocate time, thus redistributing the act of receiving information throughout daily life (Warde et al., 1998). In the context of this research, the gaming phone represents a type of hypermodern convenience device. It enables players to reorganise their leisure time and receive gaming information according to a reorganised flexible schedule.

Given the flexibility of mobile gaming in both spatial and temporal contexts, this literature review chapter is divided into eight sections. Section 2.1

discusses how mobile gaming has evolved to allow players greater spatial and temporal flexibility and outlines the literature review structure. Sections 2.2–2.4 cover game studies on playing time, life space and intimacy. Section 2.5 addresses the risks associated with game design, specifically how it trains or disciplines players and invades their daily lives. These risks are referred to as 'dark patterns' in game design by Zagal, Björk and Lewis (2013). Section 2.7 introduces Lefebvre's (2013) rhythm analysis method, which is applied in the data analysis chapter.

Section 2.2, which explores studies on playing time, begins with Lizardi's (2012) argument regarding the perpetual commodification of games. The method of earning benefits from games, often referred to as game monetisation, was detailed by Johnson and Brock (2020). Their research outlines the evolution of game monetisation strategies. Before the digital era, early monetisation methods included vending, renting gaming space or equipment, taking a percentage of revenues in gaming venues (similar to casinos taking small commissions on bets), selling retail goods within the game space and establishing in-game currencies (such as the currency used in Japanese pachinko machines).

With the rise of gaming arcades, game monetisation evolved as players bought tokens to access game time or challenges on machines provided by the operator. Subsequently, with the introduction of game cassettes, players could purchase a cassette, take it home and play it on their own machines. This shift marked a change in ownership from arcades to individual players. Initially, due to logistical and distribution constraints, these cassettes were typically sold as complete products. Subsequently, game expansion packs

and downloadable content (DLC) emerged as significant additions to basic game content. These additions served as precursors to microtransactions, where players spend out-of-game real-world money to access supplemental content within the game.

In today's gaming environment, the most popular method of purchasing games is through online stores (e.g. Steam). With advancements in Internet technology, game supplement packs have evolved from large content additions to smaller, more frequent updates. Johnson and Brock (2020) suggested that the transition from physical to online game purchases and updates has led to the concept of 'games as a service' and the proliferation of gambling systems within game design. Freemium mobile games, the focus of this research, exemplify the concept of games as a service, transforming game monetisation from a single purchase to providing players with commodities that optimise their gaming experience.

Lizardi's (2012) research on DLC posits that continuously purchasable content represents a 'perpetual form of commodification' (p. 33), which he considers exploitative of video game players. He cited the game Rock Band, which published 2000 downloadable tracks, to illustrate a 'cyclical form of commodification' (p. 33). According to Lizardi (2012), this model involves players "continually coming back and paying for the same product over and over again" (p. 33), a strategy akin to the freemium model and microtransaction systems used in contemporary mobile games.

Lizardi (2012) also identifies two methods by which game companies exploit players, both explicitly and implicitly. The first method is the exploitation of

user-generated content, where players use developer-provided tools to create new levels, music, or other customised content. This method not only expands the content but also extends the life of the game. The second method involves using players as free testers. Players identify issues during gameplay and report them to the company through online channels, effectively replacing the role of traditional game testers.

This subsection introduces the concept of 'immaterial labor' (Lazzarato,1996)'. and its application in defining in-game behaviours. This process can manifest as advertisements, online tests (Zhang and Fung, 2014) and guild labor (Zhang and Fung, 2014) in contemporary gaming. Kücklich (2005) introduced the concept of 'playbor', where the boundaries between 'labor' and 'play' are blurred. Thus, the gaming behaviour has transformed into a combination of work and play (Kücklich, 2005), Later, Dyer-Witheford and De Peuter (2009) further explore the development of playbor in the video game history.

Section 2.3, which explores studies on the life space, begins with the early research on 'Multi-User Dungeons (MUDs)'. MUDs are " programs that accept network connections from multiple simultaneous users and provide access to a shared database of 'rooms', 'exits' and other objects" (Curtis and Nichols, 1994, p. 193). MUD games typically involve text-based interaction, where players type commands to interact with items or other players. These games create social spaces independent of physical space, governed by their own unique rules, in line with Johan Huizinga's (1938) concept of the 'magic circle'.

The 'magic circle' is an independent space with unique rules and boundaries, separate from the non-game physical world. It is a "special place in time and

space created by a game" (Salen, Tekinbaş and Zimmerman, 2004, p. 95). However, in the context of mobile games, the boundary between the physical and digital worlds becomes blurred. Research by Karpashevich et al (2016) demonstrated that the in-game world and the real world provide mutual value and meaning to each other.

Mobile games increasingly incorporate real-life elements, such as social time and player geographic locations. Mobile gaming locations should be regarded as hybrid space as described by De Souza e Silva (2006). Therefore, the latter part of Section 2.3 focuses on the concepts of 'overlapping realities' (Atkinson and Willis, 2007) and 'hybrid spaces' (De Souza e Silva, 2006). For De Souza e Silva (2006), hybrid spaces are interconnected spaces, mobile spaces and social spaces that mix physical and digital spaces. Subsequently, the section explores Lefebvre's (2013) space theory and its application in the context of video games.

Section 2.4, which explores game studies on online intimacy, examines how online games provide socialisation and support to players. According to Yee (2006), most players build online relationships with others, which can help them explore their identity (Osborne, 2012), establish social capital (Huvila et al., 2010), gain social support (Trepte et al., 2012), and share trust (Yee, 2006). However, these relationships can also lead to a sense of loss (Domahidi and Quandt, 2015), control (Yee, 2006) exploitation through immaterial labour (Zhang and Fung, 2014). Gaming can also enhance existing social relationships between players. Wen, Kow and Chen (2011) indicated that playing together can reshape family relationships, redefine dialogue environments and create a more relaxed atmosphere for conversations.

People with existing social ties may deepen and explore new aspects of their existing relationships within the context of online gaming (Bengtsson et al., 2021). Furthermore, players may develop social relationships with non-human characters (NPCs) within games. Levy (2009) and Edirisinghe and Cheok (2016) demonstrated the potential for humans to form relationships with robots. Similarly, Waern's (2011) research based on the single-player game *The Dragon Age* illustrates how players can develop romantic feelings for non-human in-game characters.

Section 2.5 focuses on the effects of video game mechanics on players' daily lives. It introduces the application of Skinner Box in the freemium game and the 'dark game design patterns' proposed by Zagal, Björk, and Lewis (2013). In this model, game designers deliberately create negative experiences for players to earn profits.

Section 2.6 explores research related to game avatars and the embodied experience of gaming. It begins with the work of D'Aloia (2009), Klimmt, et al. (2009) and Boellstorff (2010), which define game avatars as extensions of the player's body (D'Aloia, 2009) and as a merging of the player's self with the game protagonist (Klimmt et al., 2009). Further, the section examines the concept of embodiment in video games, particularly focusing on the work of Keogh (2014).

Section 2.7 introduces the rhythmanalysis method, which serves as the main theoretical framework of the thesis. The section begins with the concepts of linear rhythm, cyclical rhythms and Henri Lefebvre's (2013) theory of everyday life. It then explains how the experience of mobile gaming can be understood

through the lens of everyday life theory. Section 2.8 concludes this chapter by summarising the integration of real-life elements into the virtual world of mobile games, the implications of hybrid spaces and the potential risks and dynamics of digital labour and in-game relationships.

# 2.2 Game Studies on Playing Time

Lizardi (2012) argued for the possibility of video games as cyclical commodities. He used the traditional video game series Rock Band as an example to illustrate the endless monetisation cycle through the sale of DLC. The Rock Band is a music video game in which players assume the role of a rock star and play various music tracks. Besides the free content included in the main game, new playable songs are added weekly at a cost of \$1.99 each. In this context, DLC refers to game content supplementary packs that are continuously updated and sold repeatedly for a certain period after the game's release. According to Lizardi (2012), during the lengthy production cycles of video games, companies plan content that requires additional purchases. In this framework, the main game and the DLC purchased by the player are deliberately split by the producer. As Lizardi (2012) noted, "These maps represent the admission on the part of the game developers that the product purchased initially was already inherently incomplete, and it was knowingly sold that way" (p. 40). Although the 'incomplete' which Lizardi (2012) described may have controversial, this deliberate segmentation indicates that game producers aim to continuously commodify game content and extract additional benefits from consumers who have already paid.

Lizardi's (2012) work demonstrated how games provide a 'cyclical form of commodification' to keep players engaged. The music video game releases new purchasable discs almost every week, described as a cyclical commodity to ensure players consistently pay for a game they already own. Lizardi (2012) noted that "Users are lulled into a false sense of free play only to have the bare bones given to them with everything else under monetary lock and key" (p. 37). The sale of freemium items aims to lock players into a cycle of consumption, where they pay to 'enhance' their enjoyment of the game with 'small additions' (Lizardi, 2012, p. 37). This aligns with Johnson and Brock's (2020) concept of 'games as a service', where players are provided with basic gaming services but must pay to access more advanced features. In the case Lizardi (2012) presented, players who have already paid for the basic game may feel compelled to continue paying to achieve a 'complete' version, becoming completionists. Lizardi (2012) argued that this cycle of game commodification is "exploiting consumers and their free/immaterial labor" (p. 33).

The concept of 'immaterial labour' was introduced by Maurizio Lazzarato (1996) and defined as "the labor that produces the informational and cultural content of the commodity" (p. 1). According to Lazzarato (1996), immaterial labor has two different aspects: 'informational content' and 'cultural content'. Both are integral to the sphere of labor. Informational content refers to "the changes taking place in workers 'labor'" (Lazzarato, 1996, p. 1), while cultural content involves "a series of activities that are not normally recognized as "work". such as "defining and fixing cultural and artistic standards, fashions,

tastes consumer norms, and, more strategically, public opinion" (Lazzarato, 1996, p. 1).

Immaterial labour does not produce physical goods but instead fosters social relationships for users. It "enlarges, transforms, and creates the 'ideological' and cultural environment of the consumer" (Lazzarato, 1996, p. 82). Terranova (2000) described it as "a specific mechanism of internal 'capture' of larger pools of social and cultural knowledge" (p. 38). In the context of the digital economy, immaterial labour is widespread, involving not only 'productive' activities such as website design but also activities not traditionally considered labour, such as online chatting, storytelling and replying to emails, etc.

In online gaming, players assume dual roles as both producers and consumers. Zwick et al. (2008) described this phenomenon as "the ideological recruitment of consumers into productive co-creation relationships hinges on accommodating consumers' needs for recognition, freedom, and agency"(p. 185). Ritzer and Jurgenson (2010) described it as "a new form of capitalism" (p. 22). Traditionally, the relationship between game designers and players was unidirectional and one-time: the consumer would purchase a game from a shop, pay for it and the transaction ends, with all of the game content created by the game company and players having no right to alter it.

However, the Internet and online gaming have transformed this relationship. Online games often use the Player vs. Player (PVP) mechanism, where the gaming experience is based on competition with other players. While the game itself may be free, players must invest money and time to win these

competitions. In this context, players' emotions, experiences and time become commodified services, almost invisible within the structure of online games.

In the online gaming industry, methods such as advertisements, online tests and guild labour can commodity players' leisure time. In many mobile games, players are forced to watch advertisements to access game content. This game time commodification mirrors Smythe's (1981) argument of mass media, where the audience of media is sold to advertisers as a commodity, effectively trading users' time for advertising revenue. Moreover, online games often encourage players to act as free testers and allow gameplay in incomplete stages (Zhang and Fung, 2014; Lizardi, 2012). Thus, what was traditionally compensated labour has been transformed into voluntary activities under the guise of play and leisure.

Guilds and other self-organising groups of players can prompt players to engage in immaterial emotional labour. Game guilds are usually formed by volunteers with shared interests and goals. As Shen et al. (2014) stated, "players work with each other to achieve common goals" (p. 2). This collaboration fosters opportunities for social relationships. According to Shen et al. (2014), the risk of a weak tie between two players in the same guild is half that of a weak tie between players in different guilds. They described that guilds are "player-created social institutions" (p. 9) that divide the vast space of online games into smaller, more intimate units. In this context, the game industry can capitalise on players' affective productivity and social relations and transform them into measurable and manageable forms of labour (Zhang and Fung, 2014).

Zhang and Fung's (2014) interviews revealed a common phenomenon within game guilds: guild leaders may negotiate with game companies using personal or business connections. In the early stages of a game's development, a guild relations manager from a game company would typically contact guild masters or leaders to establish cooperation agreements. These agreements often include cash back for guild leaders based on the number of players recruited or the total amount of money spent by players introduced to the game by the guild. Each game guild member is given a card to track their in-game spending and activities, allowing guild leaders to receive benefits from their members' in-game consumption. This arrangement between guild leaders and game companies can lead guild members to feel "sold out by their GM (Game Master) to game companies" (p. 44).

Emotional connections between guild members become a significant motivation for them to remain engaged with the game. As an interviewee in Zhang and Fung's (2014) study expressed, 'If I were to play by myself, there would be nothing to do by now'. However, the guild system keeps players busier through constant communication with others—whether leading, helping or just chatting.

Lukacs et al. (2010) explored how playing online games such as World of Warcraft (WoW) can transform from a fun activity into a burden or obligation for guild members. In WoW, 'raiding' is a primary activity after levelling, involving between 10 to 25 players and lasting anywhere from 45 minutes to 12 hours. Most guilds establish a weekly raid schedule to help members coordinate their daily activities and ensure timely participation in the game. Interviewees in Lukacs et al. (2010) highlighted the feelings of burden and

obligation associated with these schedules. For example, a mother who had to leave guild activities to care for her newborn baby explained,

I left [guild]. I just got a baby and was unable to make the raid times regularly. Kind of sucks—I had a lot of friends in the guild, but I cannot play with them unless I make the raids. (p. 174)

In the context of this thesis, the regular, weekly updated raid schedule established by gaming guilds represents a cyclical rhythm, while players' relationships and emotional connections, as illustrated by Lukacs et al. (2010), provide motivation for following this rhythm.

Competition between guilds also plays an important role in keeping players engaged. Zhang and Fung's (2014) research showed that game companies often invite rival guilds to sponsored 'guild tournaments' to encourage competition among members of different guilds. As Zhang and Fung's (2014) interviewees explained, when players desire to win 'guild wars', they need to purchase 'weapons and medicines'. This deliberately created competition generates hatred between players and increases their need to purchase ingame items for competition. Consequently, the need for in-game items keeps guild members active and motivated to follow the in-game rhythm, which is a focus of this thesis.

Furthermore, competition within in-game guilds can significantly increase stress for member players. Snodgrass et al. (2016) showed that guilds often require members to prioritise guild activities over their offline lives. Guild events, such as defeating a boss, require the presence of a majority of members. Thus, during these events, members need to log in at agreed times,

which can place considerable pressure on participating members. As one of Snodgrass. et al.'s (2016) interviewees described, "For example, I am not exaggerating this at all, let's say I'm in a very hard-core raiding guild. They raid 4 days a week, 4 hours a day, you need to be there at the instance, on time" (p. 63). Snodgrass et al. (2016) described this regular and repetitive behaviour of participating on time in guild events as 'player rhythm', which this study focuses on. To win in-game competitions, gaming guilds establish rules for their members, such as logging in on time and playing for extended hours. These guild rules help to catalyse the rhythms of a game's players.

As guilds become central to 'generating, reproducing, and sustaining affective labor' (Zhang and Fung, 2014, p. 46), the emotional connections between members also boost the monetisation of their in-game playing time. Zhang and Fung's (2014) research mentioned that most benefits generated from game members are divided between game companies and the core leadership of the guild. This often leads to dissatisfaction among guild members, as expressed by one interviewee: "Guild members are no longer comrades; we become tools for leaders to profit and get rich " (Zhang and Fung, 2014, p. 46).

According to Jarrett's (2021) online ethnographic research, the emotions and sentiments players invest in playing time are integral to the digital economy. Jarrett (2021) described that "Affect is converted into economic value for commercial games publishers" (p. 104). His research showed that some players feel 'obliged to donate' by paying for the enjoyment the game provides. In this context, players and developers form an ongoing, mutually beneficial relationship, with differences in scale, motivation and power often ignored

(Jarrett, 2021). However, the way the emotions of online game players are monetised can be more subtle and indirect. According to Lukacs et al. (2010), friendships among guild members can motivate participation in in-game activities, which may indirectly generate profits for the game company.

The role of the player has evolved from a leisure participant to a provider of labour, a transition marked by a longstanding ambiguity between 'play' and 'labor'. The blend of 'play' and 'labor' emerged from Modding, where players create additional content—commonly referred to as 'mods' or 'modifications'— and distribute it to others. Modding represents the free production and distribution of digital content for commercial games, primarily undertaken by highly engaged players who volunteer to contribute content to the game.

Kücklich (2005) pointed out that, within the cultural context of video games, there is an implicit rule that activities related to games should be a voluntary, non-profit-oriented activities. However, the leisure activities of modders have been commodified by the gaming industry. Kücklich (2005) described this phenomenon as 'playbour' (also called playbor), a mixture of 'the relationship between work and play (2005, no pagination)', which is "simultaneously voluntarily given and unwaged, enjoyed and exploited" (Terranova, 2000, p. 32). Consequently, playbour is 'unclassifiable in traditional terms of work and leisure' (Kücklich, 2005, no pagination). Broadly speaking, the concept of 'playbour' is similar to 'produsage' (Bruns, 2008), where users assume the roles traditionally associated with producers, regardless of their awareness of this role.

Dyer-Witheford and De Peuter (2009) explored the concept of 'playbour' and identified four stages of it: micro-development, modding, massively multiplayer online (MMOs) games and machinima. In the micro-development stage, labour in gaming occurred only within the game itself, involving tasks such as programming, creating graphics and composing music. Some video games, such as Tetris<sup>1</sup>, were freely created by hackers but eventually became commodities of speculative capital, marking the original form of 'playbour'.

The next stage is 'modding', where players make changes to games, such as adding new missions, maps, characters, or even entirely new gameplay mechanics. For example, the PC game Warcraft 3 includes a map editor that supports player-developed gameplay modifications. Sun and Lin (2023) observed the evolutionary design of gameplay on the Warcraft 3 mod platform. Players can create modifications (mods) based on the game design of Warcraft 3, resulting in gameplay that can be quite different from the original game. A famous example is DotA, which "replaced regular Warcraft 3 gameplay to create the MOBA" (Sun and Lin, 2023, p. 313). These mods should be viewed as products of player labour. As Lindstrom (2019) noted, mod creators "deserve the same respect and ownership rights as other artists" (p. 843). However, under current copyright laws, mods are typically distributed free of charge to avoid copyright infringements.

At this stage, the phenomenon of 'playbour' overlapped with volunteer work and involved only a few players. However, the blend of play and labour quickly expanded to a broader scale during the MMOs stage. A typical example of MMO playbour is 'gold farming', which is prevalent in developing

<sup>&</sup>lt;sup>1</sup> Tetris is an early puzzle video game.

countries. Gold farming involves collecting valuable in-game goods and then selling these virtual items for real money (Ahmad et al., 2009). Tai and Hu (2018) defined 'gold farmers' as a specific type of 'worker-generated content', linking it to digital labour and co-creative labour. Interviews with gold farmers revealed long hours, boredom and repetitiveness in their job, although they rarely complain. Tai and Hu (2018) demonstrated that repetitive in-game tasks can generate economic value, which can be measured and exchanged in the real-world market outside the game.

The repetitive in-game work of common players is often referred to as 'farming' or 'grinding'. Rettberg (2008) described the endless questing in WoW as follows:

When a quest giver asks you to complete a quest, you are shown the rewards you will be able to choose between when you have finished. In this way, a quest is a promise—I will give you this if only you do that. Each promise is graded by difficulty and sorted by zone so that the player's quest log contains a collection of possible activities and possible rewards. You have only to choose between them—if you are able to kill all those monsters and find all those places, items, and people... Quests are more organised than most todo lists, providing automatic crossing-off of completed items. They are as endless as to-do lists as well (p. 176).

The tasks described by Rettberg (2008) are tools created by game designers to communicate with players. Early game designs did not have clear quest instructions for players. Games such as Tetris or Galaxian used scores to represent player achievements, while games such as Donkey Kong and

Super Mario used levels to encourage players to continue exploring. As game design evolved, the concept of the 'quest' became a standard convention around the mid-1990s. Aarseth (2004) defined 'quest' as a scenario where "the player-avatar must move through a landscape to fulfill a goal while mastering a series of challenges" (p. 368).

In his analysis, Rettberg (2008) emphasised the endless list of tasks within online games. He adopted Brooks' (1984) perspective on the narrative of "anticipation of retrospection as our chief tool in making sense of narrative" (1984, p. 23) to analyse the infinite nature of in-game tasks. Brooks (1984) highlighted that, while readers desire to reach the end of a story, reaching the end also means there is no more narrative left to consume. Based on Brooks' (1984) opinion, Rettberg (2008) argued that the quest system in WoW represents a unique achievement where : "Humans have finally succeeded in creating something that we can desire endlessly, have entirely and never consume" (p. 176).

When quests are endless, the time cost to players becomes also endless. For example, Rettberg (2008b) noted that he had to kill thousands of furbolgs (a type of monster in the game) over one hundred hours to achieve honorary status in Winterspring (a game zone). This process is often called 'grinding' in MMO games. Thompson (2008) described his experience of grinding in the following terms:

I had to spend the next eight hours planted at my desk, repetitively clicking through the same tasks as if I were an industrial robot making car parts on

the Chrysler line... Then I did it again. And again. And again. Until 3 a.m., actually (Online source, para. 5).

From the perspective of this thesis, the long hours of work by gold farmers and the repetitive clicking by players reflect the game-designed rhythms imposed on players' bodies. Rettberg (2008) argued that the highly repetitive labour involved in advancing a WoW character is mindless and mirrors Taylorism:

The majority of the play involved in advancing a World of Warcraft character is mindless and repetitive to the extent that it verges on Taylorism. There is an assembly-line mentality to many of the quests, many of which involve killing a staggering number of a certain type of beast or enemy (grinding), over and over again. There is little more novelty involved in grinding than there would be in welding two sections of a fender together, over and over again, all day long (p. 30).

This 'grinding' reflects players' negative experiences with the endless quests described by Rettberg (2008). At the same time, Ruffino' s (2021) work on the idle game AdVenture Capitalist demonstrated how the game can run automatically without human intervention. In this game, a player acts as an investor; once the basic setup is complete, the game can continue to operate on its own. Players earn income from their investments, which they can then reinvest to generate more money. The more players earn, the faster their earnings grow. Theoretically, players can return to the game hours or even days later and find that the game has continued to run without their participation. More importantly, Ruffino (2021) argued for the feasibility of

players enjoying the game AdVenture Capitalist through 'non-game' behaviour, stating, " the game is played when the app is closed" (p. 214). In this case, the fun of the game is not only derived from the player's direct interaction with the game but also from the player's use of the video game as a medium for capital appreciation, exploring identity and the simulation of capitalism. The mechanics of AdVenture Capitalist highlight the meaninglessness of the player's labour in the game and point to the endlessness of the game process and the gradual dissolution of the game's purpose through endless repetition. The repeatability and cyclicity discussed by Rettberg (2008) and Ruffino (2021) are also key features of rhythmanalysis that this thesis focuses on.

At the same time, game studies scholars have proposed the idea of gaming as a craft to explain gaming time. Brock and Fraser (2018) argued that gaming activity includes 'thoughtful and engaged practice' (p. 1221). Games provide a series of puzzles and challenges for players to solve, allowing them to develop skills, learn strategies and engage in iterative mental processes to overcome these challenges. According to Brock and Fraser (2018), repetition in games can be interpreted as an intellectual activity where players think, reflect, adjust their actions and pass puzzles to achieve goals.

The process of playing and reflecting on the gameplay is a creative practice for developing strategies and skills. Therefore, the time spent playing Dota 2 should not be viewed merely as a way to achieve accomplishments through repetition and abstraction from reality. Brock and Fraser (2018) argued that computer gaming offers a similar experience to improving skills as in traditional crafts. The interaction between players' hands and the mouse,

screen and controls, mastery and understanding of the rules and communication with others reflect mental activity and demonstrate gaming on a computer as a form of craft.

Furthermore, Brock and Johnson (2022) argued that video game controller behaviour represents the earliest stage of the craft consumer process. The concept of 'craft consumption', introduced by Campbell (2005), refers to products that are "both 'made and designed by the same person', with the consumer typically bringing skill, knowledge, judgment, and passion to the activity, motivated by a desire for self-expression" (Campbell, 2005, p. 23). Brock and Johnson (2022) used the examples of Super Mario Maker and Dark Souls to illustrate the link between craft consumption and console games. Players use in-game items or mechanisms to demonstrate their creativity and improve their skills, thus the experience of in-game "confers sovereignty to the consumer" (p. 609).

Based on this research, defining repeated in-game behaviour requires careful consideration. The three-way transaction model of playing time, in-game items, and out-of-game currency, along with the corresponding game farming phenomenon, demonstrates the potential for exploiting non-professional players' personal leisure time. Meanwhile, the research on gaming as craft consumption highlights the potential for players to use the in-game process as a means to develop creativity and skill.

Besides the concern of in-game repetitive experience, the development of gaming platforms has led to a new characterisation of players' time use. The rise of Social Networking Game(SNS) has seen the 'time management'

element evolve from an offshoot of traditional gaming to a popular model of online Social Networking games (Chess, 2018). SNS is a model of gaming embedded in social media, such as Facebook and Twitter, however, it usually runs on mobile devices as well.

Chess's (2018) research raises the 'push and pull' phenomena on gaming time, in her description, it is " the player is expected to play for a short period of time over several points during a typical day, in order to play effectively" (p.107). In there, 'push and pull' refers to the process of players repeatedly switching between in-game and out-of-game worlds. Based on this feature, Chess (2018) suggests a characterisation of SNS casual games: the game will usually be completed within 15 minutes; if the player wants to play longer, they need to pay real-world currency. This feature hints at the commodification of game time and the invisible influence of game design mechanisms on players, who remain within the game for a period of time that is almost entirely under the pre-scheduling of the game system.

In this model, players' engagement time is referred to as 'Interstitial time' by Chess (2018). She analyses four typical SNS games—respectively FarmVille 2: Country Escape (FV2:CE), Restaurant Story 2 (RS2), Kim Kardashian: Hollywood (KK:H), and Clash of Clans (CoC)—to understand players' time management in casual games. Chess (2018) emphasises the 'low-risk ' element of these casual games, which refers to that the casual games do not substantially interfere with players' daily lives, and that players are usually empowered rather than disempowered <sup>2</sup>in gaming. This type of casual

<sup>&</sup>lt;sup>2</sup> In contrast to the casual games Chess (2018) concerned, in some traditional 'hardcore games', such as the Dark Souls series, players are offered difficult bosses or mazest, and the penalties for death or--

gameplay focuses on the interstitial between players' entering and leaving the online game, with a variety of different guises, such as crop ripening, cooking dishes, acting shows, etc...however, the ultimate aim of the interstitial mechanisms is to limit the player's engagement to a short period, with narrative excuses for waiting ruled real-world time. Under these conditions, player and the casual game will develop a kind " intimate and ongoing relationship" (Chess, 2018, p.113), and the nature of this relationship is directed towards players' regular repetition of 'push and pull' by the game mechanisms over the course of cycle, as Chess's (2018) account points out that: "The process where players start something and must return to complete that task creates a rhythm of play that taps into feminine leisure styles" (p. 115).

### 2.3 Game Studies on the Life Space

# 2.3.1 The Early Game Space Research

Early research on game spaces began with the MUDs (multi-user dungeons). The term 'dungeons' originates from the well-known tabletop game Dungeons and Dragons and refers to an 'adventure'. MUD games are often considered prototypes for many popular online games. These games are typically textbased, and players interact with items or other players by typing commands into an input field.

failure is strict. Casual games are usually likely to provide 'light' penalties for failure to players, Chess (2018) explains it as a feminine leisure style which is 'positive and pleasant fictions' (Juul, 2010).

>look Corridor The corridor from the west continues to the east here, but the way is blocked by a purple-velvet rope stretched across the hall. There are doorways leading to the north and south. You see a sign hanging from the middle of the rope here. >read sign This point marks the end of the currently-occupied portion of the house. Guests proceed beyond this point at their own risk. -- The residents >go east You step disdainfully over the velvet rope and enter the dusty darkness of the unused portion of the house.

Figure 3 : An example of MUDs (Curtis, 1998)

Typically, players can create one or multiple online characters and freely choose their appearance, species (which may not be human) and gender. Using simple programming languages, players can create game spaces (often called 'rooms') and design their own game rules. In an early significant study on MUDs, Bartle (1990) described MUDs as 'imaginary world', but a welldesigned MUD game should be 'believable'. Everything in the game world should operate similarly to the real world, allowing players to use the same mindset—reacting and responding as they would in reality. Therefore, emotional responses to adventure events in MUD games can affect players in the real world.

Bartle (1990) outlined several reasons why people immerse themselves in games. First, gaming is 'fun', and provides players with a lasting and effective sense of excitement (often described as 'flow' in game studies). Another

reason is convenience. Beyond that, games offer a strong sense of realism without requiring players to suffer real-world consequences. However, many players are strongly drawn to the emotions generated by the game. To some extent, players merge with their game characters, and experiences in virtual games become naturally integrated into their life experiences. Thus, perceptions in the game space can be considered equivalent to experiences in the physical geographic space.

Bartle (1990) asserted that players project their emotions onto their characters. Therefore, the experiences of an in-game character (also called an avatar) is influence the player outside the screen. He described this phenomenon as 'believable' (Bartle,1990, Chapter 7.2 Why Do People Play, paragraph 10)', rooted in the simulation of real-world rules in MUD games. As he argued:

'[Mud games] work the same way as the real world, then the players use the same mind-set as if they were in the real world, and hence emotional response to events in the MUA world are as if they were affecting the player directly in the real world (Bartle,1990, Chapter 7.2, Why Do People Play, paragraph. 10)'

In summary, as Bartle (1990) noted, 'The game's virtual reality becomes (temporarily) the player's reality' (Bartle, Chapter 7.2 Why Do People Play, paragraph 56). The player's actions in virtual reality are conceptual symbolic experiences that overlap with the physical world and influence the player's memory and emotions. Bartle articulated the relationship between virtual reality and real life as follows: 'Virtual reality does exist; not in the same way

as real life, but as a conceptualization which can have an effect on people in real life' (Bartle, 1990, Chapter 7.2 Why Do People Play, paragraph 24).

In further game studies, human-computer interaction researcher Katherine Isbister (2016) adopted the 'grounded cognition' theory in psychology to support a similar argument. Grounded cognition posits that the human brain compares current experiences with past experiences, regardless of whether they occurred in the real world or were created by media. Based on this comparison, people generate a series of emotional or cognitive feedback responses. Therefore, when players see or hear (or form a mental picture of) someone's feelings in a social setting, the brain is tricked into believing it is a real social experience. The adoption of grounded cognition explains that game players can enrich their own life experiences by playing different roles in the digital world.

Game scholars often refer to the phenomenon of experiencing near real-world experiences in a digital game space as 'perceived game realism' (Malliet, 2006) or 'gamic realism' (Sommerseth, 2007). Sommerseth (2007) provided an elaborate mechanism for the relationship between the player's body and realism in video games. He defined the game experience as 'half-reality' (p. 766) and "depending on iterative feedback between a real body and a virtual world" (p. 766). Inspired by Heidegger's (1966) notion of the ready-to-hand and Merleau-Ponty's (2002) description of tool use, Sommerseth (2007) regarded the game joypad as a means to access the game experience. Sommerseth (2007) argued that if " the joypad functions as an extended part of the player's body, video games may be seen as a subset of the player's experienced reality" (p. 767).

In Sommerseth's (2007) opinion, the feeling of control over the in-game avatar provides the ability to affect the virtual environment. Therefore, the process of gaming is not a mimetic representation akin to cinematic conventions but an embodied experience, which complements and supports Bartle's (1990) argument about the perception of reality.

Similarly, Curtis (1998) emphasised the properties of virtual space in MUDs and described them as spaces and networks that provide multi-user access to a shared database. Through text input, users can interact with other users or the environment. The MUD space, where users gather, is an artificially constructed place with many social attributes and mechanisms similar to physical spaces. Curtis(1998) described it as "a kind of virtual reality, an electronically-represented "place" that users can visit " (p. 1). Curtis (1998) believed that many social attributes and mechanisms of physical places also apply to virtual spaces.

Building on Curtis's opinion, Reid (1994) further explained that a MUD is a medium for communicating through another's imagination. Virtual reality is a construct in the human mind; it does not exist in the machine that generates the images but in the minds of those who treat these manifestations of imagination as real.

In the 1990s, some researchers viewed MUD games as an emerging social field. Curtis and Nichols (1994) introduced the concept of 'social virtual realities', which they defined as "software systems that allow multiple people to interact and communicate in pseudo-spatial surroundings" (p. 194) to build a virtual world showcasing the capabilities of MUDs. Muramatsu and

Ackerman (1998) pointed out that the MUD world represents a highly simplified social world where social interactions, including conflict and cooperation, are designed into the game world for entertainment. The MUD experience is thus a managed social life. For instance, players in Illusion (a MUD world) have auction spaces and courts. Auctions occur on auction channels where players can sell, bid, auction and purchase items through the system code. In court scenarios, inhabitants of the virtual world assume roles such as criminal, victim, judge and juror. This phenomenon aligns with the concept of 'pseudo-spatial surroundings' described by Curtis and Nichols (1994). The court and judge in virtual space function as a conceptualised simulation, similar to their real-life counterparts.

Besides that, Muramatsu and Ackerman (1998) clearly demonstrated the social structure of MUD games. While other reports, such as Bartle's (1990), also explained this structure, they did not do so as systematically. The MUD society has two main social levels. The first level consists of 'mortals', who are the most active players. The second level consists of 'immortals' (also referred to as 'caretakers' or 'wizards' in other literature), who do not play the game but create, administer, maintain and organise its content. These immortals are the rule makers. In literary terms, the immortals control the fate of the 'mortals', with the ability to change levels and attribute points, items or other data about the virtual world. In later games, mortals became 'players' while immortals took on the roles of game managers and designers. Each role carries corresponding rights and responsibilities. As Muramatsu and Ackerman (1998) stated, "The MUD is deeply socially stratified, with players both informally segregated by game levels and formally segregated by their

power to control the system" (p. 97). Mortals focus on gameplay, while immortals are responsible for managing, maintaining and keeping order in the game world. In the MUD space, roles confer authority or power. Immortals have privileges such as changing character attributes, deleting characters and being invisible to mortals. This disparity in power underscores the social stratification present in the digital space.

Another key focus in MUD studies is the generation of user identity. According to Turkle (1994), the MUD space allows players to present themselves as characters distinct from their 'real self'. All social interactions occur in a virtual space, and everyone is anonymous. Hence, MUDs provide a great platform for players to take new identities, offering a 'second chance' to explore the nature of the self and rethink personal questions.

A popular notion in MUDs is that 'you are who you pretend to be', Turkle (1994) considered this notion to have 'mythic resonance' (p. 161). The online role of the player can be not only an 'aspect of self' (p. 161) but also a "separate self of game space" (p. 161). In other words, the game space provides an environment for players to realise their fantasies, allowing them to develop a separate self—an aspect of personality independent from their real-world roles during the online gaming process. Turkle's (1994) interviews corroborated this finding: MUD games facilitate self-reconstruction and reconfiguration of intimacy. For instance, Julee, a 19-year-old who dropped out of Yale due to a strained relationship with her mother following an abortion, played the role of a mother in a MUD game. In the game, her 'daughter' betrays her, as scripted in the game. Within this digital space, the characters of 'mother' and 'daughter' engage in a profound conversation—a discussion

that, although necessary, never occurred in real life. This virtual interaction allowed Julee to empathetically understand her mother's perspective, providing her with an opportunity to reevaluate their relationship.

Bruckman (1992) described MUDs as being "organised around the metaphor of physical space" (p. 5). She incorporated Erikson's (1993) classical psychological insights from *Childhood and Society* into her MUD analysis. Erikson (1993) observed a four-year-old girl deeply engaged with playing toy house using a simple block structure that included a living room, kitchen, bedroom, bathroom and garage. In this play space, toy figures of parents and a child were positioned—the girl lies in the bedroom with the mother beside her, and the father in the garage. When faced with real-life dilemmas, the young girl crafted solutions within this toy house space, at one point discarding the doll that represented her mother and lavishing the male doll with gifts from other toys.

Erikson (1993) stated that this 'toy situation', a space filled with dolls and toy cars, provided a safe environment for the little girl to express her emotions. It represents an 'unreality' that allows her to reveal and process her feelings. Bruckman (1992) extended Erikson's (1993) 'toy situation' to cyberspace, specifically the MUD programming world. She argued that it does not matter whether the psychological play space is made of wooden blocks or blocks of text, both are imaginative spaces created to address real-life issues.

Generally, early MUD game research focused on two main directions: investigating the social systems and organisational forms within MUD societies and exploring how players explore their identities in online game

environments. During this stage, MUD spaces were viewed as independent from real spatial environments yet serving a function similar to geographic space. In other words, it acts as a 'Magic circle'.

# 2.3.2 Research on Hybrid Reality-Games Space

The previous subsection introduced the concept of the independent game space known as the 'magic circle'. This subsection discusses how this magic circle is broken, or in other words, how the boundaries between game spaces and reality become blurred or overlap. Digital marketing scholar Molesworth (2009) viewed video game experiences as escapes from routine daily life. Players may seek nostalgia, experience desired dreams that are unattainable in reality, explore worlds depicted in movies or novels, or embark on entirely new adventures. Video games serve as a long-term daily dream that provides comfort and an escape from the real world. However, the intersection of realistic elements and game elements complicates the sustained achievement of this 'long-term daily dream'.

The myth of the 'magic circle' is often associated with Johan Huizinga's (1938) book named 'Homo Ludens, which plays a significant role in early leisure and game studies. Although Huizinga's (1938) era did not have video games, his definition of 'gaming' implied that games occur in designated spaces independent from the outside world. These game spaces are " temporary worlds within the ordinary world, dedicated to the performance of an act apart" (p. 10). Ideally, the 'magic circle' is completely separated from real life, with its own rules of operation, making it a "special place in time and space created by a game" (Salen, Tekinbaş and Zimmerman, 2004, p. 95). Castronova
(2008) described that the magic circle acts as a shield that protects the inner world, likening it to a membrane with countless holes through which people share values, facts, emotions and meanings.

As an important contributor to the 'magic circle' concept, Zimmerman (2012) emphasised that the 'magic circle' is a theoretical model—it functions as an ideal model in game design and should be understood from a game design perspective. In a sociological sense, games and gaming behaviour are not isolated from daily life.

The simulation in digital games represents a conceptual reproduction rather than a photographic copy. Australian sociologists Atkinson and Willis (2007) introduced the idea of 'overlapping realities', suggesting that players are situated within a mediated relationship between the simulated and real-life environments. The experience of playing video games provides players with new perspectives on the cities they inhabit and shapes their perceptions of various people (such as gang members and people of colour). When memories of playing are triggered by familiar scenes, players may briefly slip transition from real-world contexts to those of the game.

Atkinson and Willis (2007) drew on Baudrillard's (1998) concepts of simulation and hyperreality. In their framework, video games such as Grand Theft Auto (GTA) 3 reconstruct players' perceptions of the city based on a gamemediated simulation system. The game, set in an American city, allows players to assume the role of a criminal (or someone who becomes a criminal). With advanced game production technology, GTA's imagery closely resembles real-world urban environments. The simulation is not an exact

replication of the real-world physical environment but is designed to enhance player perception. The game environment does not reproduce the actual operational logic of a city but re-presents it based on social concepts, spatial imagination and social ecology. For example, Atkinson and Willis (2007) described the game's depiction of the city's gangs as follows: "ethnic and urban stereotypes are deployed to create a clear sense of demarcated gang spaces, amplifying the social/physical connection and social ecology of the city" (p. 838).

Consequently, perceptions formed in the game world also influence players' feelings about real-life urban environments. Atkinson and Willis (2007) recruited young players of Grand Theft Auto <sup>3</sup>(GTA) to discuss their interpretations of the urban environments depicted in GTA. During the time of Atkinson and Willis's (2007) article, direct crossovers between game and reality were rare. They used 'in-game music' to bridge the connection between real life and game experiences. For one of Atkinson and Willis's (2007) interviewees, game music became a key link between the real world and the gaming experience. In GTA, part of the gameplay involves driving in a virtual city, with game music presented through in-car radio. Consequently, when players listen to their car radios while driving in the real world, the overlap of in-game and real-world experiences creates a powerful sense of hyperreality.

In this context, in-game music (and its mode of delivery) acts as a bridge from the emotional in-game experience to the player's own local context. An

<sup>&</sup>lt;sup>3</sup> The background of GTA is on a American city. In this game, the player takes the role of a criminal (or people who would become a criminal). With the advanced technology of game producers, the image of GTA is similar to the real-world city environment.

interviewee in Atkinson and Willis's (2007) study described the 'slipping' feeling from reality to the game when listening to game music. He felt as though he was in the space of Bell Street [a place in the game] rather than his physical room. This slipping experience demonstrates that the player's perceptual experience of space transcends geography. When the sense of ingame spaces overlaps with the non-game world, the gamer operates in a ludodrome, defined as "a mediated space between immersion in urban simulation and a real world that is simultaneously generated, destabilized and blurred by the effect of such gameplay" (Atkinson and Willis, 2007, p. 818)

Modern video game design has increasingly linked games with reality. Location-based mobile games (LBMGs), such as Ingress, *Pokémon Go, and Harry Potter: Wizards Unite,* utilise the mobile phone's positioning system as part of the game mechanics. These games often employ a mobile device typically a smartphone—to superimpose virtual graphics onto the screen, merging the virtual game world with the urban space the player inhabits daily.

Players tend to experience an overlap of the virtual and real worlds, although most claim to be aware of the virtual world through the screen while playing (Karpashevich et al., 2016). Stenros et al. (2012) adopted a qualitative approach to study pervasive games (games that blend the digital and physical worlds) to gain firsthand material. The sample interviews occurred during gameplay, *Ingress*, a typical location-based mobile game, features two factions: Enlightened and Resistance. Players from these factions compete to capture portals, which are based on real geographical locations, and link them to establish control fields. Two researchers joined the Resistance group, and one joined the Enlightened. Other players met during the game volunteered to

participate in interviews. In total, 25 interviewees participated in Karpashevich et al.'s (2016) study. The interviewees generally viewed physical locations as keys to the virtual world, with most of the experience occurring on the screen. They did not care about the actual situation of real locations. For these players, real-world landmarks become mere buttons in the game world, losing their real-life function but retaining in-game significance. However, this does not imply a complete separation between the virtual and real worlds. First, real-world landscapes provide significant advantages for mobile game experiences. Second, Karpashevich et al. (2016) found that the in-game and real worlds provide mutual value and meaning. Players often choose game portal locations (important game resources) based on personal interests, such as wedding venues. These portals also provide value to real locations in the minds of experienced players. Third, game rules and rewards motivate players to take action in the real world, such as visiting a dark forest at night or climbing a mountain.

Frith (2013) considered that LBMGs are built on 'hybrid spaces' (De Souza e Silva, 2006), which are defined as spaces that merge social connections, digital information and physical space. These mobile spaces are created by users who connect to the Internet with others. The appearance of hybrid spaces is facilitated by the capabilities provided by mobile phones, enabling users to be 'always-on', and 'enfolding remote contexts inside the present context' (De Souza e Silva, 2006, p. 262).

De Souza e Silva (2006) defined hybrid spaces as interconnected spaces that are connected, mobile and social. Initially, hybrid space was conceptualised as a blend where the boundary between digital and physical spaces blurred.

With the advent of the Internet and the ubiquity of smartphones, teenagers born into this era naturally accept digital spaces. The portability of Internetconnected smartphones embeds hybrid spaces into users' daily lives. Unlike previous generations, the new generation does not consider 'leaving' physical space to enter cyberspace; instead, these two spaces, traditionally seen as separate, are naturally linked together, forming what De Souza e Silva (2006) called hybrid spaces.

The creation of hybrid spaces is attributed to the mobility of digital devices. The transformation of the interface from static to mobile redefines the public's perceptions of digital spaces (De Souza e Silva, 2006). As De Souza e Silva (2006) stated, "Users are no longer required to sit in front of their computers, but rather, they move around in urban spaces—which are already social public spaces"(p. 270). With the evolution of communication devices, the Internet has become 'portable'. This shift has also impacted the video game industry. In the early days of the game industry (1970–1999), games were single-player and the interface was static, typically being arcade games or personal computers. After 1999, online games appeared, allowing multiple players to share the same gaming space in the virtual world via the Internet. With the popularisation of smartphones, players currently have the flexibility to play anytime and anywhere, integrating gaming spaces seamlessly into daily life.

Hybrid spaces can also function as social spaces by transforming physical spaces into potential multi-user environments within a cyber context. Traditional online communities do not rely on users' actual geographic locations. However, the transition from static to mobile interfaces allows users

to connect with others in shared virtual spaces while in physical public spaces. These virtual spaces are based on users' common geographical areas, but the connections between participants are facilitated by digital technology rather than physical human contact.

For example, PUBG Mobile, a multiplayer online shooting game, supports an invitation mechanism with online friends and also offers a matching method based on real geographical locations. Players can invite (or be invited by) nearby individuals to join the competition. Another example is *King's Glory*, where players can earn titles of honour for their heroes based on their geospatial location, such as being ranked the greatest in a particular city or district.

With the advancement of mobile internet technology, the concept of 'space' has become increasingly complex. Hayles (2002) noted that 'There is no longer a homogeneous context for a given spatial area, but rather pockets of different contexts in it' (as cited in De Souza e Silva's personal communication with N. Katherine Hayles, November 19, 2002). This idea of enfolded spaces is exemplified by the popular location-based mobile game Pokémon GO. In this game, players need to visit real geographic locations to capture different Pokémon. While engaging with the game, they do not share consistent information with the surrounding people. According to a survey by Colley et al. (2017), 33% of players reported colliding or almost colliding with objects such as signs, poles, or other people, and 11% admitted to endangering themselves by forgetting real-world safety rules, such as crossing the road without looking.

Mobile games create environments that overlap with real geographic settings, allowing players to coexist in both the Pokemon habitat and the busy city streets. This dual existence means that players navigate a compound context while moving through urban environments. Current research suggests that this composite presence of multiple spaces also implies multiple rhythms based on hybrid spaces. In Colley et al.'s (2017) study, signs, poles and other people on urban streets represent a default geographic rhythm of urban space. In contrast, actions such as challenging gyms and capturing Pokémon in Pokémon GO follow a game-space-based rhythm. The collisions reported by respondents in Colley et al.'s (2017) study indicate an inherent conflict between these multiple rhythms within the hybrid space of geographic and digital environments. Players of Location-based mobile games may experience arrhythmias, as described in Lefebvre's (2013) rhythmanalysis framework, due to the dual demands of their surrounding environment and mobile gaming.

For residents of hybrid spaces, smartphones provide the ability to transition into virtual spaces anytime and anywhere, making virtual socialising in real spaces possible. 'Hybrid spaces' offer users the flexibility to switch freely between virtual and real space contexts, empowering them to reshape their living spaces with digital software, mobile games and other technologies.

The redefinition of the space where people live is multifaceted. First, the traditional function of public space is redirected and enhanced with additional value from digital spaces. Karpashevich et al. (2016) found that players of location-based mobile games often use their personally significant locations within a geographic environment to set game portals, which in turn provide an

emotional link to those real-world locations. Second, hybrid spaces alter the way users interact with both digital devices and real environments; players are motivated to explore and find a new ways to interact with urban spaces. Lastly, hybrid social spaces provide a new model for users to build relationships, where digital sociability and communication occur within real-world spaces.

Mobile devices enable users to be present at all times, allowing them to log into virtual spaces from social public spaces, workplaces, or homes. This spatial flexibility suggests that virtual rhythms generated by digital games can overlay the rhythms generated by geographic spaces. In a sociological sense, the overlay of physical and digital spaces is not merely a functional combination but also a merging of rules generated by these spaces. In De Souza e Silva's (2023) description of hybrid space, she cited the game *Can* You See Me Now? from Blast Theory. In this game, players are divided into two groups: one group sits in front of a computer using avatars, while the other walks through city streets. Players on the streets can 'catch' their virtual counterparts if they come within five meters of each other. This system establishes two necessary conditions for the grasping action: proximity of position and execution of the action. Proximity of position happens in physical spaces, while execution of the action occurs in digital spaces. The urban walking players follow the rhythms of the physical space (e.g. traffic lights and buses) while navigating the city and executing interactions according to the rules of the game space.

Overall, this section of the literature examines the interaction between digital game spaces and physical spaces. This research aims to explore the

interplay between the two through the subjective perception of participants (players).

# 2.3.3 Lefebvre's Space theory to the Video Game Area

The early studies applying Lefebvre's (2013) theory to the gaming area, such as computer games researcher Aarseth's (2001) work, primarily focus on the nature of computer space. For Aarseth (2001), cyberspace (including video game space) is constituted by signs, and the experience of game space is dependent on the receiver's bodily experience and their experience of real space. In his subsequent papers on game space, Aarseth also pointed out that the space of the game world is often distorted and modified to enhance player enjoyment. This view is supported by his student, Løvlie (2007), whose thesis showed that, in war simulation games, it is challenging for players flying in the sky and those fighting on the ground to collaborate effectively on a battlefield. When digital space becomes a copy of real space, the game mechanics may not function well. For Aarseth (2001), computer games are a kind of allegory of space—they attempt to portray space in more realistic ways, but ultimately, the reproduction of spatial relationships in video games "rely on their deviation from reality in order to make the illusion playable" (Aarseth, 2001, p. 169).

Based on his opinion that 'computer game space' is constituted by signs, Aarseth (2001) introduced Lefebvre's space theory into game space analysis. According to Lefebvre (2013), space is socially constructed and can be distinguished into representational space, representations of space and spatial practice. Space is not only a physical 'empty area', but also a

performance of productivity and production relations. Physical space, abstract space and social space coexist on the same site.

Aarseth (2001) expressed caution in applying Lefebvre's system to game analysis, noting it as 'a type of spatial representation he [Lefebvre] did not anticipate' (p. 163). He suggested that Lefebvre's theory in the cyber context requires "a much longer refinement and adaptation" (p. 163). However, Aarseth argued that computer games, as a spatial practice, have dual properties: representations of space (a formal system of relations) and representational spaces (symbolic imagery with a primarily aesthetic purpose). The representation of digital space is "not itself spatial, but symbolic and rulebased" (p. 163), which aligns with Curtis and Nichols's (1994) description of MUDs as 'pseudo-spatial surroundings' (p. 194), emphasising the digital representation of physical space.

In subsequent research, scholars have focused more on the connection between game space and the broader social context. A notable example is Crawford's (2015) research on sports games, which argues that video games are 'themed' spaces similar to fast-food restaurants and theme parks. The 'theme' involves using "advertising, branding and other corporate efforts to stimulate consumer demand" (Gotham, 2005, p. 227) to create consumer experiences. For instance, Disneyland employs extensive ornamentation and spatial design to craft specific experiences for visitors. The context of the 'theme' is rooted in 'nonplaces', which lack their own history and identity, such as the shops or airports, and are often given a 'theme' tor create novel experiences.

Crawford (2015) introduced Theodor Adorno's culture industry theory to explain how the 'theme' influences cultural production, His analysis of the top 100 best-selling games in the UK reveals that only a few games are original. Most are reboots, remakes of old games, or sequels to existing series. This 'theming' of space decoration extends to game design. In the Chinese mobile gaming market, the standardisation and pseudo-individualisation described by Theodor Adornoare are even more evident.

As themed spaces, game areas are packaged and sold to players. In these spaces, everything is quantifiable, including traditionally non-quantifiable aspects like an athlete's body. The record of player in-game behaviour is referred to as gameplay metrics, as described by Drachen et al. (2013, p. 23): "When players interact with something or are exposed to something, their information is recorded in five categories: 'What is happening? Where is it happening? At what time is it happening? Additionally, when multiple objects (e.g., players) interact: to whom is it happening?" (Drachen et al., 2013, p. 23). Real-world spaces are imitated in more orderly, controllable patterns to provide an exciting experience for players. Similarly, Brock's (2021) research on the PC game Defense of the Ancients 2 (Dota 2) highlights the concept of themed space. He noted that the number of actions a player performs per minute, along with metrics such as 'kills', 'deaths', 'assists', and the amount of gold and experience gained per minute, are used to measure a player's skill in the game. In this context, the player's interactions with other players are represented within an e-sports-themed space, governed by the game's measurement systems.

Similar to theme parks and shopping malls, game spaces blend leisure and commercial activities, Players can pay to obtain in-game goods or reduce the waiting times. The detailed in-game mechanics will be explained in Chapter 4.

Other scholars have attempted to identify Lefebvre's spatial triad theory in the context of game studies. Denham and Spokes (2020) recruited 15 participants to explore Red Dead Redemption 2 (an action-adventure game) to explain the video game context through Lefebvre's concept of space. First, representations of space are 'conceptualized space', which is created by planners, urbanists, technocratic subdividers and social engineers: it is the dominant space in society. In Denham and Spokes's (2020) research, the game director, scriptwriter, 3D modeller and game artists replace the roles of planners and urbanists. Undoubtedly, any space within the game is the product of careful design: thus, " every crevice of RDR2's spaces are intrinsically 'conceived'" (Denham and Spokes, 2020, p. 5)

This sense of space design is always present in the player's gaming experience. For instance, St. Denis, a state capital in the game map, is a bustling town but seems unwelcoming to participants. Only one participant did not immediately leave the city after a necessary commodity exchange. Another participant explained that it's because St. Denis doesn't seem friendly; it's for well-dressed rich people, not for wilderness cowboys like the player's avatar. This top-down, spatially-designed message is present in every corner of the game.

Second, 'representational space' also works in the game space. For Lefebvre, it is the idea, theory and vision that overloads in the area, related to sign and

symbolic representational comprehension. In Denham and Spokes's (2020) study, the majority of participants (14/15) reported that specific memories were evoked, recalling films or film characters without any prompts from the researchers.

Third, in the context of Lefebvre, spatial practices are defined as the daily routines and realities of urban life. According to Denham and Spokes (2020), the spatial practices of players involve how they "explore and take advantage of manipulating this filmic world in experimental ways" (p. 11).

Besides providing a possible explanation of Lefebvre's spatial triad theory in game space, another significant contribution of Denham and Spokes (2020) is that they exhibit the intertextuality of participant experiences. Some participants mentioned the Rockstars(the production company of Red Dead Redemption 2), while others recalled memories of real-life movie stars, which are independent of the game space.

Beyond the video game area, other scholars have contributed to meaningful discussions of Lefebvre's theory of space production in the internet context. Prey (2015), for example, bases his research on music streaming, emphasising the portability of 'perceived space'. For Prey (2015), mobile streaming takes sounds out of the home or office, allowing listeners to "reclaim lived space through privatizing the public realm" (Bull, 2005, p. 354). At the same time, Prey (2015) points out that the space created by mobile streaming must contend with the public environment: sunshine or traffic noise inevitably affects the space shaped by streaming media. Similar to Denham and Spokes (2020), Prey's (2015) research shows that media messages

construct the user's perceptual space. Music can evoke of memory or experience, further shaping the user's perception of local space. As Prey (2015) states, "the 'digital space' is always embedded in, and interwoven with, 'physical space'"(p. 7).

Lefebvre (2013) described how media affects the rhythm of life: "Just by having a modern television or radio, you can hear and/or see images and receive messages from afar, by pressing a button or turning a dial. And beyond the mountains and seas" (p. 48). In this passage, actual events from distant places are inserted into the daily rhythm of local life through messages transmitted by media. The media offers the possibility of adapting to multiple rhythms, allowing individuals to participate in other rhythms through information beyond their current geographical context.

Lefebvre's framework explains the basic pattern of polyrhythms between commodified media from afar and humans as subjects across spatial distances. The rhythmic producer, as a supplier of the commodity, provides segmented rhythms, and the almost costless flow of information enables the producer to compartmentalise media information. This allows the recipient to adapt to rhythms in their current environment rather than experiencing them in a geographical and spatial sense. As Lefebvre (2013) stated:

Producers of the commodity information know empirically how to utilise rhythms. They have cut up time; they have broken it up into hourly slices. The output (rhythm) changes according to intention and the hour. Lively, light-hearted, in order to inform you and entertain you when you are preparing yourself for work: the morning. Soft and tender for the

return from work, times of relaxation, the evening and Sunday (Lefebvre, 2013, p. 48).

The amount of user data that TV or radio media can feedback to manufacturers is far less than what mobile games can record. Mobile games can track usage length, participation frequency in specific in-game activities and even subtle details like the number of times a player is killed by a boar. More detailed feedback data allows algorithms to create a more precise perceptual space for mobile gamers.

In this basic model, Lefebvre (2013) used the example of 'modern television or radio', which are instantaneous and require the user to turn them on to receive messages. Smartphones being more portable than televisions or radios, mean that the perceptual space generated by games can permeate almost all social spaces.

#### 2.4 Online Intimacy

In Isbister's (2016) view, when players immerse themselves in their in-game roles (avatars), the characters they play can collaborate to share a rich and genuine gaming experience, which makes the video game space create a kind of 'real social interaction' (p. 52). Isbister (2016) described that "the game worlds they create may be imaginary, but the social dynamics are not" (p. 53). However, building in-game relationships is based on structures present in the virtual world and does not a simply mirror real-world interactions (Huvila et al, 2010). In video games, the representation of a personal image is more related to game skills, character avatar appearance and the behaviour governed by game rules, rather than to real-world identity (gender, age, or physical stature,

etc.). Thus, the way people build relationships in the online games differs from how intimacy develops in reality.

For online game players, building intimate relationships in a game is a common experience. Yee's (2006) report showed that 66.3% of male players and 81.4% of female players have made strong friendships with people they first met in an online game, with many considering their online friends to be better than their in-person friends. Yee (2006) and Osborne (2012) argued that people have a higher tendency to self-disclose in an online game environment, which may lead to the development of higher-quality intimate relationships.

Role-playing in online games often features strong anonymity. According to Osborne (2012), this anonymity provides a safety measure for players to explore aspects of their identity. Her research on gender disparity in gaming included a participant named R033, who played a male role despite being biologically female. R033 used this male character to conceal her homosexual relationships. She explained that playing the game character gave her the opportunity to 'explore personas and actions that I am not allowed to comfortably explore in "real life" (Osborne, 2012, para. [5.21]). This anonymity is supported by two main elements: pseudonymity and characterisation. Pseudonymity guards against the risk of being recognised in real life, while characterisation provides a natural mask for self-disclosure. This mask allows players to explore alternative identities and provides a sense of security, making them feel safer when revealing personal aspects of themselves.

The impact of in-game relationships on players is a controversial topic. Huvila et al. (2010) offered a relatively optimistic view; They adopted an early questionnaire about social capital in the real world to determine whether the virtual environment of Second Life<sup>4</sup> helps build social capital. Their results showed that Second Life is a profoundly social environment that enables residents to generate and contribute to social capital. The researchers categorised participants into two groups: producers and non-producers. Producers are individuals who consider themselves contributors to the virtual world. Huvila et al. (2010) found that the social capital indicators for producers were always higher than for non-producers. In their study, Huvila et al. (2010) concluded that online game organisation can function as "kind of spontaneous and voluntary community, or rather a compound of communities" (p. 311). Their research showed that the more players contribute to 'producing' something within the game, the more social capital they earn.

Yee (2006) also suggested that in-game relationships may have higher quality than real-life intimate relationships. He explained that players create idealised projections of each other, as he said:

We focus on all the things we do agree on and the sense of compatibility is enhanced even though this would not have been the case if this meeting occurred in the real world. In other words, many relationships that would never have even begun in the real world have a far better chance of developing online (p. 8).

<sup>&</sup>lt;sup>4</sup> An online video game that allows people to create virtual representations of themselves.

Yee (2006) also believed that in-game relationships encourage players to engage in self-disclosure, leading to deeper connections: "We get to know each other much more, we truly tell each other what we think/feel and you really create this amazing bond with one another. It's much less superficial than some RL friendships can be" (p. 11).

However, some scholars have argued that players carefully select what they self-expose to their online friends. Steinkuehler and Williams (2006) described online games as a 'third place' for social interaction, in addition to the traditional environments of home and workplace. The concept of 'third place' originally came from Oldenburg (1999), who defined the home and the workplace as the first and second places, respectively, with the third place being social surroundings distinct from both. Steinkuehler and Williams (2006) identified the online gaming space as such a third space. They described that the tone of communication in this third space is playful and casual (Steinkuehler and Williams, 2006). For example, when a guild member attempts to introduce a serious topic, the other members quickly steer the conversation towards lighter and more enjoyable subjects. This 'playful mood' often prevents players from engaging in emotional and weighty conversations.

Online games provide players with sociality, social support (Trepte et al., 2012) and a safe space for exploring one's true self (Osborne, 2012). However, not participating in game activities can lead to negative feelings, including a sense of loss (Domahidi and Quandt, 2015), a feeling of being control (Yee, 2006) and a sense of guilt towards fellow guild members (Hommadova Lu and Carradini, 2020). For instance, if a player cannot join a guild activity at the scheduled time, the player may feel ashamed and guilty towards other guild

members. The potential negative effects of video games are often related to the players' real-life situation. Players who lack social support in real-life are more likely to become highly engaged in video games (Domahidi and Quandt, 2015).

In addition, in-game relationships can also facilitate a form of immaterial labour. A notable example is the relationships within in-game guilds. Early research on game guilds often described them positively, highlighting their role in helping players develop social capital, expand social networks and explore new friendships, romantic relationships and community ties (Williams et al., 2006; Ang and Zaphiris, 2010). However, Zhang and Fung (2014) suggested that guilds may be alienation from being purely social places to mediating sites that form new labour relations between game players and game companies. As mentioned earlier in this chapter, guild managers sometimes use guild members as bargaining tools in negotiations with game companies and may receive benefits when a guild members spend money in the game.

Of course, trades between guild administrators and game companies do not always happen. Depending on the game structure, guilds can vary significantly in size, from small groups of 2–3 people to large groups of hundreds. The scale of the guild directly impacts the negotiation strategy of the game company. Many guilds place no demands on players, and organisers rarely engage in direct management. Thus, the conclusion drawn by Zhang and Fung (2014) that the strategies "companies have adopted to capitalize on the guild community have transformed the early guild culture of mutual support and collaboration into one of antagonism and contradiction" (p.

46) is not entirely accurate. This scenario mainly applies to part of large-scale guilds, which represent only a small percentage of the total player population. In other words, the early guild culture of mutual support and collaboration coexists with the competitive guild environment, forming a complex ecosystem rather than a simple dualism.

Zhang and Fung's (2014) research was focused on online games in China's political and cultural environment. Since the late 1970s, China's development has heavily relied on cheap labour, state monopolies and foreign investment (Hong, 2008). This reliance has stifled innovation across industries, including the game industry. Piracy and Shanzhai (a Chinese term referring to counterfeit, imitation, or parody products) have made it difficult for high-quality products to survive in the fiercely competitive gaming market. According to Cao and Downing (2008), many online games are mere copies and low-quality knockoffs of other successful examples. In such a highly capitalised, profit-driven game vendor environment, ethical considerations are often neglected. In this framework, guild members may experience being commodified, with their in-game relationships exploited to increase the value of these commodities.

In another aspect, online games also offer opportunities to redefine or develop existing relationships. According to Yee's (2006) survey, 19% of MMO players had played games with their family members (excluding romantic partners or spouses). The experience of playing together can reshape family relationships, Wen, Kow and Chen (2011) found that family members visited each other's virtual farms 3 to 24 times a day, while other forms of communication (e.g. face-to-face talks, phone calls, or emails) occurred only once or twice a week.

One participant ( a 36-years-old daughter) in Wen, Kow and Chen's (2011) study stated, "This game [QQ Farm] opened a door to us to become farmers —the role we couldn't be in the practical world, it was exciting that we can plant and harvest just like a real farmer...We found more things to talk about and found more joy in our conversation" (p. 10). Thus, the content of online games can strengthen ties between family members and reshape the existing relationships.

Current research supports the idea that online games can strengthen family bonds in multiple ways. For example, Wen, Kow and Chen (2011) clarified that online games create a relaxed space for family members to engage in non-conflictual dialogue, helping to sidestep potential intergenerational value conflicts. Additionally, Boudreau and Consalvo (2014) conducted a study on Social Network Games (SNG) involving 163 respondents and 11 in-depth interviews. They revealed that players are more likely to help family members, even at the expense of their own gaming progress. Their research also suggests that SNGs provide a leisure space that helps family members engage together and feel connected with each other. Further, Wang, Taylor and Sun (2018) surveyed 361 parents through Amazon's Mechanical Turk. Their results showed that the frequency of family members playing digital games together positively correlates with family satisfaction and closeness. Families with poor communication benefit more from playing together than those with effective communication. Furthermore, Musick, Freeman and McNeese (2021) interviewed 20 individuals who had played modern digital games with their parents or children. They identified three main positive impacts of gaming on family relationships: "augmenting parent-child

relationship with friendship, promoting close connection by removing physical and emotional barriers, and facilitating quality time" (p. 15). They argued that digital games, as a common activity, can benefit the parent-child relationship and promote a 'democratized' (p. 20) family life.

Additionally, gaming together can increase intergenerational communication. Pecchioni and Osmanovic (2018) recruited college students and asked them to choose an immediate family member over the age of 55 to participate with them. The experimental group consisted of two subgroups: the first group had 182 participants who were instructed to play video games with their immediate family members for at least three hours a week, while the second group consisted of 88 participants who were asked to engage in conversations with their selected family member for the same duration. After the 6-week experiment, the first group reported stronger family ties, greater self-other overlap and a deeper understanding of each other's lives, knowledge and thoughts. The second group reported less significant effects in these aspects. This evidence suggests that playing games together can serve as an intergenerational bridge to maintain and enhance relationships between generations.

Pearce et al.'s (2022) research during the COVID-19 pandemic demonstrated how playing video games together can help family members cope with both communal and individual stress. They interviewed 27 families (including 33 parents and 37 school-aged children) who played Animal Crossing: New Horizons. The 90-minute interviews revealed that video games play a significant role in relaxation, anxiety relief and stress management. Playing games together was viewed as a communal coping behaviour that helped

family members deal with pressures. Respondents indicated that family members felt a responsibility to spend time together and needed a way to connect and pass the time. Communal gaming activities allow family members to connect with each other and with those who are physically separated, enabling them to 'be together in games' (p. 785).

In maintaining and developing existing relationships within the context of online gaming, real-time presence becomes less important. In Wen, Kow and Chen's (2011) study, farm development in a game did not require real-time participation, as family members could visit each other's virtual farms even when the other was not online. De Grove (2014) suggested that, in online game friendships, the discourse of the relationship is more important than the time spent together. This argument indicates that online games reshape both the methods and content of information transmission between parties in existing relationships, allowing these relationships to develop. According to Bengtsson et al. (2021), 'experienced game player' interviewees felt more relaxed in relationships facilitated by screen interactions compared with face-to-face interactions. This finding supports the idea that transitioning from reality to online practices may lead to changes in social relationships and how these relationships are maintained and developed in the digital age.

The relationship between humans and in-game NPCs can also significantly influence players' in-game experiences. Isbister (2016) believed that rich and lively interactions with NPCs can create an anthropomorphic affective experience. She conducted an experiment focused on desert survival, where participants were informed that their plane had crashed and they needed to rank the importance of various items, such as a compress kit, a book, and a

raincoat. During the task, an NPC would try to convince participants to change the rank of these items using different strategies, including body postures and phrases of advice. According to the strategy of body postures and diction, the NPCs were categorised into four types: consistent dominant cues, consistent submissive cues and two mixed types (dominant body with submissive phrasing, or submissive body with dominant phrasing). The results showed that participants who interacted with the mixed-type NPCs were less likely to change their rankings, a trend similar to human-to-human interactions. In reallife social environments, consistency between verbal and body language is linked to honesty and trust. This human perception of consistency or inconsistency also applies to communications between humans and NPCs. Isbister (2016) explained that "to the extent that NPCs display human-like actions and reactions; we engage them using social norms and intuitive emotional responses" (p. 22).

Waern (2011) and Ganzon (2019) further explored how human players may develop relationships with non-human characters (NPCs) instead of other humans. Waern (2011) analysed 62 articles related to players' romantic experiences with NPCs in the game Dragon Age. Many players admitted to falling in love with their targeted characters, despite fully understanding that these characters are fictional. Waern (2011) stated that the 'love effect' is embedded in players' daily lives: some players wrote that they 'regularly dream about (the non-player character), had trouble sleeping that night'. Waern (2011) referred to this phenomenon as 'bleed-in', meaning that "the player shares the emotions of the character" (p. 5). These findings underscore

how deeply virtual interactions can resonate with human emotions and shape players' perceptions and experiences outside the game.

In the context of mobile gaming, Ganzon (2019) highlighted the emotional labour experienced by female players in the mobile game Mystic Messenger. This game requires players to communicate with various in-game characters and develop romantic relationships with them. Ganzon (2019) argued that female players "consent to give their own time and money for the performance of emotional labor within in-game relationships" (p. 147), suggesting that this setup predisposes female players to develop emotional relationships with non-human characters.

However, the emotional connections between humans and NPCs may pose risks to human players. Southerton and Tomlinson's (2005) research on the Health and Lifestyle Survey in the UK showed that individuals may suffer from time disorders due to feelings of obligation. The realistic emotional experiences between NPCs and players may lead to a risk of humans developing a sense of obligation towards digital roles. This potential risk is discussed further in subsequent chapters of this thesis.

# 2.5 The Influence of Video Game Mechanics on Players' Daily Life

This section first introduces quest design in video games before exploring the effects of video game mechanics on players' daily lives. Espen Aarseth (2004) defined a,quest as "the player-avatar must move through a landscape to fulfill a goal while mastering a series of challenges" (p. 368). In most MMO games, quests typically require the player to complete a series of tasks, including killing a certain number of monsters, delivering messages, or helping

someone achieve a goal—in simple terms, a to-do list. The concept of the 'quest' became a standardised feature of game design around the mid-1990s. It was not an inherent mechanism but rather a tool created by game designers to communicate with players. Early game designs did not include clear quest instructions. For example, in games such as Tetris or Galaxian, player achievement was represented by scores, while games such as *Donkey Kong and Super Mario*, used the level system to encourage players to continue exploring through progressively challenging levels. Modern game design provides clear guidance on how to complete quests; game systems provide extensive assistance for players, including step-by-step guidance and ingame automatic navigation.

In the 1930s, American psychologist Burrhus Frederic Skinner (1972) designed a mechanical device for animal experiments. This device, known as the Skinner box, features a lever on one side of the box wall and a small hole beside the lever. When an animal presses the lever, a grain of food immediately falls into the box from the hole. After several repetitions, the animal forms a conditioned reflex to press the lever to obtain food. These conditioned reflexes are not natural but can be acquired through training. Some commentators have criticised the World of Warcraft (and many online games) as being akin to a Skinner box. Yee (2003) explained how 'shaping' in operant conditioning experiments works in the context of online games. Operant conditioning includes several schedules of reinforcement for animal behaviour: fixed interval schedules, fixed ratio schedules and random ratio schedules.

The fixed interval schedule means that animals receive a reward (such as food) at regular intervals, regardless of whether they perform the expected behaviour. In the context of the Skinner box, this means that, for example, a mouse gets a reward every 5 minutes, regardless of whether it presses the lever. Although Yee (2003) considered fixed interval schedules ineffective and rarely used in online games, after 21 years, popular video games have come close to applying this model, often referred to as 'idle game' or 'idle mechanics'. The first idle game, Progress Quest, was created in 2002. Many elements of Progress Quest were similar to those found in the popular online game Endless Quests at that time. However, the significant difference was that the idle online game did not require players to take any actions-no clicking, dragging, or controlling the direction was necessary. The game had an automatic progress bar at the bottom of the screen, and when the bar filled automatically, players would receive equipment, items, experience, or whatever they would get in traditional online games. The creation of Progress Quest appears to mock traditional MMO games. The effort that players put into completing game quests and achieving milestones in traditional online games was simplified to an automatic progress bar. In other words, the role of the game player was no different from that of an automatic progress bar or a robot that functions autonomously.

In other examples, some idle games put little operational pressure on the player, but this does not necessarily imply reduced engagement with the game. Cutting, Gundry and Cairns (2019) highlighted a significant limitation of traditional game research: it often considers only single sessions of play from start to finish, without accounting for the frequency or regularity of play and

how gaming becomes integrated into daily life. Cutting, Gundry and Cairns (2019) surveyed 1,972 participants, with 67.5% reporting that they opened idle games 'many times a day'. Many participants had been playing for an extended period, with 38.9% playing this game for 2–4 weeks and 24.9% for 1–3 months. In the game they studied, Neko Atsume, 'waiting' is an integral part of the gameplay. Specifically, it is a cat-collecting game, after players use the currency to buy items (toys or food), cats appear in the yard once the game is closed. As a result, the game progresses only when it is not being actively played.

The most notable finding from this survey is the redefinition of game engagement time. The authors highlighted that traditional gaming engagement was previously understood as active gameplay, but players can also engage with the game through out-of-game interactions, such as sharing it on social media. They introduced 'checking frequency' as a metric for analysing player engagement alongside 'time spent'. This research demonstrates that in the design of idle model, a player's non-gaming time is also considered part of game engagement and can become integrated into the gaming world's time cycle. Although the fixed interval schedule may be considered an ineffective method of behaviour training in Skinner's studies, it remains applicable to online gaming because idle games fill daily time gaps. Idle games prompt people to frequently check the game for automated progress, although this behaviour does not lead to an actual increase in ingame revenues.

In Skinner's experiments, the fixed ratio schedule proved to be more effective. It involves rewarding the animal after a specified number of actions are

completed. This method could potentially be more applicable to other gaming scenarios where direct interaction is required to earn rewards. Yee (2003) considered experience points systems in online games as fixed ratio schedules. This system functions as a character reinforcement mechanism, where players earn experience points by completing in-game quests or defeating enemies. Once players accumulate a certain amount of experience points, their characters level up. Skilled players can roughly estimate how many experience points they need to reach the next level and translate this into specific actions, such as defeating a set number of monsters.

For the game designer, the 'specified number of actions' for geting reward may become the method of game monetisation. As Zagal, Björk, and Lewis (2013) introduced 'dark game design patterns', which are "used intentionally by a game creator to cause negative experiences for players which are against their best interests and likely to happen without their consent " (p.7). Whether a game design pattern is considered 'dark' often relates to the time commitment required from players. Karlsen (2018) considered that games which encourage players to 'waste time (and money) on mindless activities or into excessive gameplay or addiction' belong to dark game design patterns. Games such as The WoW, FarmVille, and many online games exhibit similar characteristics. Under this design pattern, layers may be unaware of how much time or money the game requires, often spending more than expected. A typical example is the 'pay to skip' system, commonly found in role-playing mobile games. In these games, the fighting or collection process is often boring and repetitive. Players must pay to speed up or skip these autoplay animations. The design of these repetitive autoplay animations intentionally

creates a payable option to entice players to skip them. In there, the 'specified number of actions', such as fighting or collection the specific times, were artificially designed as intermediaries for game monetisation.

Another example of the fixed ratio schedule in games is the concept of ingame multi-layered and overlapping goals, which is emphasised by Yee (2003). Most games establish clear in-game objectives, allowing players to estimate the distance between their current status and potential rewards. Rarely do players focus on a single target at a time; instead, they pursue multiple goals such as gaining experience points, earning in-game currency, completing quests, collecting equipment or props, and developing skills. These goals coexist and often overlap, with the reward from one goal quickly contributing to another. Therefore, players are always close to achieving a goal and receiving a reward, yet they are seldom far away from obtain all of their aimed rewards. Each time a player completes a single objective, the system immediately assigns a new one. While individual goals are easy to accomplish, the overall network of goals creates an endless cycle.

Finally, in Skinner's experiments, the most effective reinforcement schedule is the random ratio schedule. In this schedule, the relationship between receiving food and the number of lever presses is completely random. The mouse knows that pressing the lever will result in food, but it cannot predict how many times it needs to press the lever. Yee (2003) explained the random increase of skill points and the unpredictable penalties of death to illustrate the application of a random ratio schedule in video games. Several years later, around 2010, the introduction of gacha/loot box systems in games provided another example of a random ratio schedule.

The loot box is a random reward system that emerged with the transition of online games from time-based to item-based monetisation. Li, Mills and Nower (2019) described the loot box as a system where "players can buy loot boxes or keys to open loot boxes, which, in turn, may contain valuable ingame items to enhance gameplay" (p. 27). However, this description is not entirely accurate. Keys for loot boxes can be acquired through various means, including real-world purchases or in-game currency, and may also be obtained randomly in-game. In most cases, the first few keys to loot boxes are provided free, likely as a strategy by game designers to familiarise players with the mechanism of the loot box. The similarity between loot boxes and gambling has been widely debated among game studies scholars. In their review article, Brock and Johnson (2021) highlighted how this phenomenon reshapes video game production and consumption. Johnson and Brock (2020) argued that the unpredictable reward element of loot boxes is a longstanding design feature in game development. The core distinction that fuels the gambling controversy is the ability for players to use real-world currency to purchase these boxes. They described the blurred boundary between gambling and digital gaming experiences as the 'gambling turn' in the monetization of digital games.

Brock and Johnson's (2021) research further explores the evolving consumption patterns at the intersection of gaming and gambling. They noted that this emerging field sits at the crossroads of three distinct disciplines: consumerism, game studies, and gambling studies. As they explained, 'consumption scholars address economic, sociological, and psychological phenomena, game studies scholars foreground humanities approaches and

close textual readings, while gambling studies scholars pursue models of pathology and deviance' (p. 9). The interdisciplinary nature of this field, coupled with the rapid development of the gaming industry, calls for more nuanced and critical approaches to studying the gaming-gambling intersection.

In early research on the relationship between online game patterns and the Skinner Box, Yee (2003) analysed the three operant conditioning schedules separately. However, in contemporary free-to-play game design, some mechanisms are closer to a combination of multiple patterns. The Battle Pass represents a new type of monetisation approach for online gaming that has gained minimal scholarly attention. It originated with Dota 2's eSports tickets and was widely promoted by Fortnite. The prototype of the Battle Pass model, as seen in Fortnite's new season update on December 14, 2017, has since become a popular game design strategy (as of July of 2024). According to Joseph (2021), the Battle Pass is a seasonal business timetable that indicates "when games will be updated with content and events" (p. 69). This new game monetisation strategy provides players with the opportunity to earn rewards for completing specific actions within a time limit. A typical Battle Pass offers two parallel reward paths: the first is free, and the second requires payment. Each time a player completes a quest (or wins a battle), they receive Battle Pass points, which accumulate on the path to rewards. Many of these rewards include loot boxes. The Battle Pass represents a hybrid of a fixed-interval schedule and a random ratio schedule. Players are aware of how many points they need to progress to the next stage, and part of the rewards content is loot boxes keys. Joseph (2021) viewed the Battle Pass as a showcase for purchasable items and an incentive for players to engage more deeply in the

monetisation of the game. He noted that gameplay and monetisation strategies are increasingly intertwined, demonstrating "how games are now shop" (Joseph, 2021, p. 81).

Yee (2003) believed that the online game environment functions as an interconnected network of Skinner boxes system, which he refers to as a Skinner Network. In his view, individual Skinner boxes are not particularly interesting, but the Internet and online games link all of these boxes together to form a social network. Each Skinner box is tailored to the player's needs and reinforcement schedule, and the owners of the boxes can interact with each other. His description bears a resemblance to *the Matrix,* where everyone is isolated in their nutrient vat yet able to interact with others in a digitally constructed world.

The essence of the Skinner box is a training system that motivates behaviour through rewards. The reward and 'level up' systems of online games are essentially endless. Thus, behavioural training never ends. For skilled players, the cost of switching from one game to another is relatively low, which means their trained experience is not lost but rather transfers from one game to another. This trained ability is what Keogh (2014) calls 'embodied literacy'. The following section continues exploring research on game embodiment and examines how these digital experiences shape player behaviour and skills across various gaming platforms.

## 2.6 The Game Avatar and the Embodied Experience of Gaming

#### 2.6.1 The Avatar in Gaming

When new players first log into a mobile game, they are usually required to create a character. Within certain limits, a player can adjust attributes such as gender, appearance and sometimes race and background. After these adjustments, the player can usually customise the full name of the character. This adapted character often becomes the character that the player controls. In this process, the character created by the player is often referred to as an 'avatar'.

Game studies showed that avatars have a close relationship with the human self. According to D'Aloia (2009), the avatar is:

"an extension of the player's own body and entertains a prosthetic relationship with it; it incorporates the player and disciplines his/her body. It is the embodied manifestation of the player's engagement with the game world; it is, at the same time, a reflection of ourselves and an envoy of ours in the parallel world in order to conquer new territories and retake possession of the lost world" (p. 51).

Klimmt et al. (2009) described that players perceive a game's main character as "a merging of their self and the game protagonist" (p. 354) rather than as an independent social entity. This view is reflected in the descriptions provided by the interviewees in this thesis, where the merging of the human self with the game avatar appears nearly indistinguishable.

Further, Boellstorff (2010) explored the role of avatars in the game world and established a framework for understanding the player's virtual body. Boellstorff (2010) emphasised that a game avatar does not merely represent a human body; rather, it forms a type of embodiment that constructs

emplacement within a virtual world. The presence of the avatar is an embodiment of one's presence in the digital world; it is "the locus of perception and sociality" (Boellstorff, 2010, p. 507). According to this perspective, the digital avatar and the virtual world constitute each other. The virtual body (avatar) is the bearer of all digital activities which make human interaction in the digital world possible. Expanding on this concept, Boellstorff (2010) cited Merleau-Ponty's (1962) opinion on phantom limbs to draw a parallel. Merleau-Ponty's (1962) stated,

To have a phantom arm is to remain open to all the actions of which the arm alone is capable; it is to retain the practical field which one enjoyed before mutilation. The body is the vehicle of being in the world, and having a body is, for a living creature, to be intervolved in a definite environment, to identify oneself with certain projects and be continually committed to them. (Merleau-Ponty, 1962, p. 81–82).

This parallel highlights that just like a phantom limb, the avatar serves as a conduit for sustained engagement and identity in the virtual world.

Expanding Merleau-Ponty's (1962) view of the digital world, avatars imply the intervolvedness of a virtual environment. In other words, the performance of embodiment involves actions intertwined with the practical field. The imagined limb or avatar is not merely a conceptual entity but carries an individual's ability to perceive and interact with the external world. The body self-constructs the person's being-in-the-world. Boellstorff (2010) quoted Heidegger's term of 'dwelling' to analyse Merleau-Ponty's (1962) view of 'being-in-the-world'. Merleau-Ponty (1962) explained embodiment as "a thing

in objective space, but as a system of possible actions, a virtual body with its phenomenal "place" defined by its task and situation. My body is wherever there is something to be done" (p. 250). According to this argument, Boellstorff (2010) used the term 'indexicality' to support the idea that virtual bodies influence or change the digital world. Indexicality links the actual bodies and the subject's emplacement in perception and social landscapes and articulates the dwelling relationship between embodiment and place. This argument proves the significance of virtual bodies or 'being in the world' in digital form and reconfigures or concretises longstanding notions of humanity. From the perspective of phenomenology, the presence of digital avatars connects human embodied experiences in both the physical and game worlds. The following subsection details studies on game embodiment.

### 2.6.2 Game Embodiment

Research on embodied experience can be traced back to Sudnow (1983). His book written during the era of arcade games, describes a time when game screens featured simple lines, blurred photos and minimal text—far from the rich, high-fidelity imagery of modern games. However, Sudnow (1983) used an almost immersive style to describe his interactions with other participants and games. Sudnow (1983) described that players in front of their TVs, like commanders, intercepted missiles, produced bombings, guarded cities and saved the world from a room in their house close to the kitchen.

Sudnow (1983) discussed how through practice, players develop in-game techniques to achieve their goals. Actions such as timing missile launches or hitting a ball grow into microscopic, concrete, perceptible skills embedded in
the player's body. The performance of human organs is transmitted from a three-dimensional physical space to a miniature screen world, displayed with the image, line, or movement of in-game items. Actions to cope with specific game mechanics become 'embedded' (Sudnow, 1983, p. 101) into the human body and transform into natural reactions to corresponding game situations. This process constructs the learning of gaming skills in the out-of-game human body.

The body records various behaviours related to the body, including eye movements, finger trajectories, heartbeats and blood flow. These behaviours are crucial for the player's in-game character improvement, which is achieved through a continuous 'Repetition–Feedback' process. This process helps the player's body adjust to a model that enhances interactions with the game. Sudnow (1983) described this process as 'programming' (p. 46) for the human player. He regarded gameplay and game narratives as "scripted interactions with various sorts of computer-generated events" (p. 46). Through interaction with these events, human body organs, such as eyes and fingers, learn complex skills through long-term repetition.

Sudnow's (1983) argument about the process of the body developing gamingrelated skills is similar to the concept of the 'symbiotic circuit' argued by Friedman (1995) in his article about game interactive textuality. In this article, Friedman (1995) described the feeling of being 'lost' inside a computer game as "flowing through a continuous series of decisions made almost automatically, hardly aware of the passage of time, the player forms a symbiotic circuit with the computer" (p. 8). In this framework, Friedman (1995) regarded the computer as an "organic extension of one's consciousness" (p. 8)

and suggested that players may also feel themselves as an extension of the computer. The process of a player taking an in-game role—such as mayor or industrialist—reflects how players internalise the game program into their consciousness.

In a subsequent essay, Friedman (1998) expanded and clarified the specific processes of cyborg consciousness. He defined the interaction process between humans and computers as "computer games teach structure of thought" (Friedman, 1998, p. 136). In this process, players are encouraged to learn and predict computer logic to outperform it. While Sudnow's (1983) concept of 'programming' focuses more on bodily movements, Friedman's (1998) argument emphasises how games shape mental structures, cognition and perception. Friedman (1998) argued that player pleasure comes from entering a 'computer-like mental state' (p. 136) and replacing sentient cognition with 'thinking like a computer' (p. 136). He described this interaction between player and computer as 'melding'.

Building on the 'symbiotic circuit' concept presented in 1995, Friedman (1998) introduced the term 'cybernetic circuit' to describe the 'player-game' relationship—a constant feedback loop that stimulates continuous play. Within this framework, the computer responds immediately to the player's inputs, and the player reacts to the computer's responses, creating a perpetual loop that blurs the distinction between the player's consciousness and the computer's logic. The act of logging into the game is viewed as the moment when a player transcends their physical body. Friedman (1998) described this process as "alienation" (p. 138) for refusing and recognising the physical experience, which also leads to rethinking cyborgs (Haraway, 1985), a hybrid

of machine and organism. The connection between a game and a player creates a cybernetic link that maintains the cyborg state, expanding new kinds of stories and new forms of understanding, unlike conventional media. Building on Friedman's (1995, 1998) ideas, Giddings (2007) inherited the term 'cybernetic circuit' and argued that "to play a digital game is to plug oneself into a cybernetic circuit" (p. 1). He proposed that a cyborg functions as an 'augmented body' that extends human nerves, senses and motor actions from bodily systems to the "prosthetic devices and environments of controllers, dance mats, and virtual worlds" (p. 5).

Keogh (2014) posited that analysing video games through the lens of 'gameplay as a cybernetic circuit' is one of the 'most fruitful' approaches. He conceptualised the 'text' of a video game as a cybernetic circuit that bridges both actual and virtual worlds. Keogh (2014) also viewed this cybernetic circuit as primarily serving the embodied pleasure of the player, whose body is "distributed during play across actual and virtual worlds via the video game hardware" (p. 4). Video gaming represents a mixed state of the player and the game. Thus, the game player should not be understood as a separate entity from the game, but rather as "a component of the video game text" (p. 15), "redistributed across a network of information and actors and materialities" (p. 15). During gameplay, the player "embodies a hybridized body, incorporating flesh, hardware, and virtual objects and beings into their corporeal schema" (p. 15).

Following this premise, the player and the game must be analysed as a singular, inseparable whole. Swink (2009) described how an action game's assault rifle feels 'meaty' and a driving game car feels 'heavy'. The outputs of

a video game are received by the player's physical senses (sight, hearing and touch), and then the player's bodily actions are inputted into the game hardware (e.g., keyboard and controller), changing the game's audiovisuals.

The player's feelings during gameplay are described as "an embodied pleasure caught in an intimate coupling of bodies" (Keogh, 2014, p. 16). This pleasure arises from the actions on the game screen and the player's perception of reality in the physical world. Hence, Keogh (2014) used the phrase "meshing of materially different bodies into a single, cyborg body"(p. 16) to describe the symbiotic relationship between the player's flesh and the video game. He emphasised that video game analysis should not isolate the player's physical body nor focus solely on the world within the screen. Instead, the text of the game must be understood from a 'distributed, cyborgian, embodied perspective' (p. 16).

Four years later, Keogh (2018) proposed a more detailed and comprehensive framework for understanding the gaming body and embodied experience. Keogh (2018) described the confusion experienced by players oscillating between the systems of human flesh and game hardware, lights and sounds. He stated:

For the video game, as for me, just who I am in this collection seems to flicker between bodies and between worlds. I am both lota and myself; I am both here and there... This virtual world is not my world. It is not, truly, a world at all. Yet I perceive it as if it were a world, and I engage with and feel some sense of presence in this world as if I somehow enter it. But I don't enter it. (Keogh, 2018, p. 2)

This narrative demonstrates the dual existence of the player, existing simultaneously in both physical and digital worlds, with a sense of embodiment that spans both on-screen and off-screen environments.

Keogh (2018) described the video game experience as an 'embodied textuality' (p. 21), referring to "the synthesized embodied experience of audiovisual design, video game hardware and the player's physical body constitute the site of meaningful engagement with the video game" (p. 21). The embodiment in video games is reflected in the mutual relationship between player and game, existing within the 'player-game' assemblage, rather than as separate and independent subjects and objects. The player is not merely a manipulative entity but rather exists symbiotically with and is in a sense created by the video game. From Keogh's (2018) perspective, neither the player nor the game exists as a pre-existing whole; rather, both are defined by their relationship to each other. In the dynamic and ongoing 'player-game' relationship, the shared frame between them shapes the player and is shaped by the player simultaneously. The human body is not stable and essential but is an essentially unstable intercorporeal entity continuously mediated by interactions with other human and non-human bodies. The human being-in-the-world comprises the intercorporeal assemblage of objects (such as video games) and the body. The experience of embodiment arises from the ongoing process of integration with multiple corporealities. Just as a blind person uses a cane to interact with the external environment, electronic devices (such as computers and mobile phones) are incorporated into the player's self and become part of the being-in-the-world.

One key argument regarding this integration is the combination of input devices (for games) and the human organism. Keogh (2018) noted that while playing video games, his thumb 'effortlessly' and 'unconsciously' moves from 'X'<sup>5</sup> to 'Square' to command his character to move, jump and kill enemies. Meanwhile, his housemate struggles to comprehend how he accomplished this. When Keogh (2018) tried to observe his own actions on the gamepad, he realised that these movements were unconscious. His thumb would "roll down and then to the left" (p. 76) without him consciously directing it, while his brain focused on in-game actions such as jumps and spins rather than the physical movements of his fingers on the gamepad. The connection between 'in-game activities' and 'specific gestures entered on specific devices' should be recognised as an ability which Keogh (2018) referred to as 'embodied literacy'. The process of a game player learning how to interact with a video game through repetitive practice corresponds with Henri Lefebvre's (2013) description of 'dressage' (this thesis further discusses the related content in the analysis chapters).

Video games teach and train players in a set of perceptual and interaction methods (skills) to the player through repeated practice until the player considers them normal. Specifically, the buttons on a gaming device (such as circle and triangle) do not have a logical connection<sup>6</sup> with the in-game actions. The process through which players establish a link between these buttons

<sup>&</sup>lt;sup>5</sup> Buttons on a video game controller typically include four buttons on the right side of the joystick: O, X, Triangle, and Square.

<sup>&</sup>lt;sup>6</sup> In the current era, the logic of the gamepad varies between Europe, Japan, and Hong Kong. The same in-game action, such as jumping or attacking, may correspond to different buttons in the same game across these regions. As Keogh (2018) noted, "In the late 1990s and early 2000s, it was common for PlayStation games made in Western countries to use the X on the gamepad as a confirmation button in menus and Triangle as a cancel button. Japanese video games, however, used O as a confirmation button and X as a cancel button" (p. 94).

and their corresponding in-game actions through repeated practice is 'dressage' described by Lefebvre, rather than occurring 'naturally'. In this process, the game device, such as a gamepad, becomes integrated into the schema of the human body, effectively merging technology with physical intuition.

Chess (2005) conducted research based on the video game GTA to demonstrate the process of player learning. She argued that the in-game disciplining mechanisms operate through three aspects: space, time and the docile body. First, in both the digital and physical worlds, the player's perception can be affected by the game. In the game world, players must navigate within the limitations of the game map and narrative. Completing a mission grants the player more freedom to unlock additional content. Simultaneously, the player's physical body must remain stationary, staring at the PC or TV screen. Second, the game design uses time as a metric to measure a player's skill. The time taken to complete a mission serves as a stable measure to distinguish between good and poor performance. Third, repeated in-game practice within the game trains players to learn the functions of the buttons on the control device. When players need to act in the game, they must press the corresponding button with precision. Through this process, players' bodies learn the rules of the game. In her article, Chess (2005) argued that video games train players to use their bodies (hands, fingers, eyes and ears) to achieve game goals. The method of playing is not 'automatically intuitive', so the player must learn the standard gestures to "acquire the necessary automatisms" (Chess, 2005, p. 84).

This process of training and growing the ability to interact with the game is similar to the concept of 'embodied literacy' proposed by Keogh (2018). For Keogh (2018), the term 'embodied literacy' refers to a "knowledge that has sunk beneath consciousness" (p. 91), an ability to link specific gestures, actions and tool interactions with in-game events. The gameplay logic across different video games resembles a 'performative grammar' (Keogh, 2018, p. 91). Experienced players can easily interpret the text presented by different games through a conventional spatial syntax of the input device. It is a bodybased language and grammatical system that must be mastered through learned experience. As Keogh (2018) argued,

it is to perceive virtual spatialities and physicalities through the resistances, textures, distances, and movements of plastic under thumb and finger. Such literacy does not come naturally but is instilled through the ritualized repetition of behaviors and the sensitization of the body. (p. 107)

Keogh (2018) further emphasised the importance of repetition and discussed the concepts of cyclical and linear rhythms from Lefebvre's (2013) rhythmanalysis framework. He describes gaming as a constant, cyclical process interspersed with linear repetition (Keogh, 2018). This includes "failing and trying again, the repeated switching on and off again of digital buttons to perform the same stop–start actions on one level after another" (Apperley, 2010, p. 40). The player's body adapts and incorporates gaming within these cyclical processes and linear repetitions.

Although Keogh (2018) primarily focused on the process of players dying and trying again during gameplay, he clarified that the focal circuit between the player's body and the video game hardware is "always already situated within particular material and social contexts" (p. 147). This circuit is "a subsystem that branches out and is situated in much broader networks and rhythms. The experience of time in video game play is always already embedded in and mediated by the rhythms of the player's everyday life, not detached from them" (p. 147).

When the time in a video game 'progresses' through repeated rhythms, it always wraps around itself (Keogh, 2018). The experience of gaming is, therefore, part of the experience of daily life. According to Keogh (2018), the player does not 'step into' a new body or world but enters a 'partial becoming' (p. 199). Playing video games involves a "distributed, immanent, and partial lived" (p. 199) experience. The player's body undergoes a cyborgian embodiment through the digital and physical worlds, with the cyber avatar and physical flesh in an imperfect and unstable form.

After Keogh's (2018) argument, Brock (2022) discussed phenomenology in game studies, introducing four case studies to understand the gaming body. The first case study includes Keogh's (2018) experience of playing Audiosurf<sup>7</sup> and how playing this music game translated into embodied practices. The second case study involves James Ash's (2013) research on how gaming environments cultivate a feeling of captivation and apply Heidegger's concept of attunement and Stiegler's applications of retention in the gaming body. The

<sup>&</sup>lt;sup>7</sup> Audiosurf is a music-adapting puzzle racer game that allows players to create unique experiences with their own music. The player can influence the shape, speed, and mood of each ride by choosing specific tracks (AudioSurf on Steam, 2008).

third case study, by Brock and Fraser (2018), investigated gaming craft and how players develop skills or 'craft' through the movement of their hands and fingers in the game Defense of the Ancients 2<sup>8</sup>. The final case study by Bennerstedt and Ivarsson (2010) examined how players coordinate through avatars in MMO games.

In his article on Call of Duty 4<sup>9</sup>, Ash (2013) demonstrated how game mechanics encourage players to develop skills and knowledge. Ash observed the growth of two players from inexperienced beginners to competent and skilled players. Ash (2013) proposed that the game provides an organised and commodified form to the body's sensual and perceptual relations. The user's body occupies a central role in Ash's (2013) analysis. Building on Heidegger's concept of 'attunements', which refers to "the fundamental ways in which we find ourselves disposed in such and such a way" (1995, p. 67) and "how according to which one is in such and such a way" (1995, p. 67), Ash (2013) further explained attunements as "complex assemblages of bodily capacities and cognitive processes, which work together in skilled gameplay" (p. 14). In his analysis, this process is realised through the development of bodily and cognitive skills educated by game mechanics. The user adopts both somatic and analytic attunements during gameplay. Somatic attunements are reflected in the player's development of hand-eye coordination and control of the muscle groups in the hands to move the crosshair to an exact point on the screen. Analytic attunements appear when users move their avatars to specific parts of the map based on intuition nurtured by past experiences.

<sup>&</sup>lt;sup>8</sup> Defense of the Ancients 2, commonly referred to as Dota 2, is a multiplayer online battle arena game. The core gameplay involves two teams of players competing against each other on a map.

<sup>&</sup>lt;sup>9</sup> A first-person shooter game set in a modern-day context.

Thus, when the user tries to defeat others or avoid being defeated, they draw upon both types of attunements.

Brock and Fraser's (2018) gaming craft theory focuses on how players "apprehend, name, and reflect on their practices" (Brock, 2022, p. 15) and how players develop creativity and strategy through craft labour. In Brock and Fraser's (2018) analysis based on Defense of the Ancients 2, they described how players understand the rules of the game through the interaction between their bodies and input devices (mouse and keyboard). By reflecting on and learning from mistakes, players coordinate the information obtained from the game screen with their bodily movements to adjust and improve their personal actions. Thus, gaming skills result from creative thinking and extensive practice involving precise movements of fingers, shoulders and eyes.

In another research perspective, Bennerstedt and Ivarsson (2010) analysed text-based chat in the MMO game Lord of the Rings Online. Through 90 hours of participant observation of player group activities, they found that players cooperate on missions in the virtual world through a series of situated practices. Their research focused on how online actions, which do not originate from the physical body, are recognisable in online group activities. During gameplay, players follow a shared order based on situated practices that allow them to recognise the meaning of specific actions in given in-game scenarios. For instance, when a player's avatar displays a specific kneeling position, other players immediately understand it signifies setting up a trap, prompting them to stand behind the trap setter, reorganise and maintain group formation. Similarly, when an avatar in the group draws a weapon and

appears ready to fight, other players immediately look around and search for the enemy's location.

Bennerstedt and Ivarsson (2010) pointed out that players often regard specific in-game actions as 'precursors of unfolding events' (p. 223) and take subsequent actions based on their projections of possible future scenarios without verbal communication. They argued that players can project the next action using various resources, not limited to turn-taking in conversations or physical body actions. This skill may depend on a player's proficiency in the game or the amount of time invested in it. However, in general, people can learn to use communication systems that are independent of their physical bodies.

Different research studies on game embodiment have shown that research is a complex and multifaceted field. These studies covered various aspects of game embodiment study, including players' bodily perceptions and engagement in gaming, the interaction between input devices and players' senses, gaming craft and the development of gaming skills. The popularisation of mobile gaming has also introduced new contexts for related research. Farman's (2009) article on the embodied experience of mobile gaming, particularly in location-based games, demonstrated that the daily experience of mobile gaming involves "navigating the correspondence or disjunction between the physical landscape and the digital landscape" (p. 4). In this context, the player's presence becomes a discontinuous patchwork between the virtual and real worlds. The active play with the 'proprioceptivesemiotic' body embodies the socio-cultural texts produced simultaneously with the hybrid digital-physical space. This body is defined as "phenomenology

converging with the inscripted body-as-text in a world of stimuli and signs" (p. 5), existing in the hybrid augmented space landscape.

Hjorth and Richardson (2009) emphasised the concept of '(tele)presence' in mobile gaming embodiment and engagement experiences. A person may simultaneously interact with others in reality and live in the game. Daily gaming occurs within the gaps between productive and telic or goal-oriented activity. Hjorth and Richardson (2009) argued that mobile phones are a device to achieve the 'body-in-waiting' proposed by Bissell (2007). In public spaces, mobile games create a state of constant interaction with the screen, which allows players to reach a common social understanding with strangers to avoid direct interaction, while still maintaining 'environmental knowing' and an 'open' approach to the external environment to respond to external changes. Mobile gaming collaboration reveals a kind of labour that can fill temporal gaps in daily life. Further, Keogh (2018) proposed the term 'co-attentiveness' to describe the embodiment of mobile gaming. This term is defined as "the player pays attention to two worlds at one time" (p. 54). It means the player exists in hybrid worlds, that are 'not created in the way one world distracts us from another but in the way the player's attention is held by two worlds simultaneously' (p. 67).

The development of mobile gaming has exposed the 'actual/virtual' circuit, which is often obscured in traditional gaming. It clarifies that game players simultaneously allocate their attention in a hybrid 'virtual-reality' space. Keogh (2018) also emphasised mobile gaming's lower requirement for player attention—in other words, the "video game player is able to perceive the screen space as a virtual world while remaining consciously aware of the

presence of the actual world they are situated in and moving through" (Keogh, 2018, p. 54). This dual awareness leads to a multi-dimensional perception. This collaborative perception should not be seen as one world distracting from the other. Instead, it creates a state where the "player's attention is held by two worlds simultaneously" (p. 67). Keogh (2018) refers to this as 'immersion as distraction'. This seemingly paradoxical term describes the essence of co-attentiveness in mobile games: the player is touching, perceiving and paying full attention to both physical and digital worlds simultaneously. The player's palms touch the screen and interact with the images on it, while the player's 'distraction' is not a prerequisite of mobile gaming but an unavoidable result.

Each world, whether physical or digital, can absorb the player's attention and disrupt the equilibrium, causing the player to lose primary focus on the other world. This creates an unstable equilibrium. In the balancing process, players realise the presence of mixed attention and the assemblage between the actual and virtual worlds and bodies. The game scholars' articles on the embodied experiences created by game-player interactions provide a solid theoretical foundation for this thesis. Building on this foundation, this thesis focuses on the migration of bodily interactions from gaming time to other moments in daily life. In Sudnow's (1983) research, gaming time and space were distinct from other events. While the player was seated in a small room, other guests, the chef and the party continued outside the gaming space, with game and non-game events remaining separate and rarely interfering with each other. However, the popularisation of mobile gaming has increased the extensibility of gaming in both space and time. As Hjorth and Richardson (2009) and Keogh (2018) have shown, mobile gaming allows players to

engage with or leave the game at any moment in their daily lives, not just when sitting in front of a computer or at a desktop. Thus, the link between players and games is no longer spatially and temporally deterministic but has become available at all times in everyday life. Therefore, game researchers need to shift their focus from a slice of determined and separate 'gaming time' to the context of everyday life. The relationship between gamers and games is intertwined with their daily lives; their choices to engage or leave are influenced not only by the game but also by their external environment.

For example, players may engage in mobile gaming between lectures, on the bus, during lunch breaks, or even simultaneously with working, studying and other social activities.<sup>10</sup> As Keogh (2018) mentioned, mobile players construct the 'actual/virtual' circuit and operate in a hybrid 'virtual-reality' space as part of their daily routines. Building on this premise, the contributions of previous studies on embodied experience—such as the repetition to train players' skills, the internalisation of computer logic and the effects of avatars on players—can be applied to how players adapt, arrange and learn to integrate gaming into their daily lives both temporally and spatially.

Building on Keogh's (2018) research, this thesis further develops the rhythmanalysis framework in the context of mobile game studies. This framework refers to the series of physical training produced by the 'player-game' interaction, the skills players develop, their gaming schedules and the physical or mental patterns that evolve to accommodate mobile gaming, all of which contribute to a game-based rhythm. This thesis aims to construct a comprehensive picture of the interaction and mutual influence between game-

<sup>&</sup>lt;sup>10</sup> These examples were derived from participants' descriptions during data collection.

based rhythms and everyday life outside of play. From this perspective, the impact of mobile gaming on players is not a momentary force occurring only in the present. Instead, it represents habits, behaviour patterns and time-use strategies cultivated through countless repetitions over time. These habits and patterns, or the mechanisms of learning and formation, can be understood through Lefebvre's (2013) rhythmanalysis. The introduction of this core theoretical framework effectively explains players' management of game time and space and the recurring and ongoing negotiation of game and non-game events in everyday life. It also extends the impact of games on players more explicitly to the entirety of an individual's social lifetime. In addition, this thesis also addresses game embodiment research from two perspectives.

First, Keogh (2018) discussed everyday synesthesia in creating the game feeling through multi-sensory compounding. The feelings evoked by video games are a combination of visual, auditory and tactile perceptions. Even when players cannot actively interact with the game, such as during a cutscene or another player's turn, they should still be considered actively engaged in the audiovisual play space. He clarified that the engagement between eyes and screens, muscles and input devices and ears and speakers is a form of perceptual engagement that is intertwined with a video game's form. In Chapter 4, I analyse how mobile games' automated mechanisms encourage players to multitask. This state covers the attention engagement between players and games even when they are not physically interacting with the game. Just as the skin's tactile sensations can be subordinated to the fingers, the flow of attention can be understood as originating from the brain. In this context, the perception of time management

and attention allocation related to mobile gaming should be regarded as part of the everyday synesthesia that games create for players. Thus, the everyday synesthesia of mobile gaming may occur in a broader sociotemporal context.

Second, with the development of popular game design, the game mechanics related to daily repetitive gaming experiences have evolved. For example, the experience of avatar death and Save/Load mechanisms address the temporal experience of PC or console video games. Keogh (2018) demonstrated that the repetitive experience led by the player character's constant death embodies the flow of rhythm and time during play. In contemporary mobile gaming, however, few mobile games focus on player death or failure, and they generally do not support Save/Load mechanisms as in single-player games. The passage of time in mobile games is single-threaded and synchronised with the social clock. Players are rarely punished for failing during gameplay. In games that do not encourage direct competition between players, daily tasks or repeated dungeon runs are more common than failure. Compared to the single-player game era, players are less worried about the risk of death or failure and more anxious about losing rare in-game items or currency updated in real-time. Time is still used as a form of punishment, but not in the way Keogh (2018) described single-player games, where it "dismisses the time invested from the previous save point to the character's demise" (p. 153). Instead, it encourages players to complete repetitive content within a specific period or measures how much longer players must spend to acquire precious in-game resources. On the Chapter 4, this thesis introduces contemporary common mobile game mechanics, including daily goal systems,

Battle Passes and daily log-in systems, to analyse the game-based rhythm and its influence on players' daily lives.

The following section elaborates on Lefebvre's (2013) rhythmanalysis theory framework and outlines the necessity of adopting related concepts for a deeper understanding of the research questions in this thesis.

## 2.7 Development of Lefebvre's Rhythmanalysis Theory on Daily Life Research

This section discusses Henri Lefebvre's rhythmanalysis theory to analyse the role of freemium games in daily life. In the early twentieth century, French Marxist philosopher Henri Lefebvre introduced the concept of 'rhythms' to develop his critique of everyday life. Lefebvre's (2013) analysis of rhythm begins with the human body, which he described in terms of "respirations, pulses, circulations, assimilations – durations, and phases of these durations, etc" (p. 5). He employed a method of reasoning that starts with full consciousness of the abstract to arrive at the concrete in order to elucidate the existential nature of rhythm in everyday life.

Rhythm, according to Lefebvre (2013), is a measure of time allocation that illustrates how individuals organise their presence within social practices. He described it as the "most personal, most internal' aspect of life, which is both internal and social" (Lefebvre, 2013, p. 76). In his view, a rhythm points to temporal and spatial repetitions in everyday life, and these repetitions themselves create the rhythm. The back-and-forth pattern is not an entity but a certain immaterial presence that is difficult to perceive objectively by those who are attached to it. To some extent, rhythms are repetitions and can be

defined as "movements and differences within repetitio" (Lefebvre, 2013, p. 90)'. Lefebvre (2013) introduced the concepts of the 'cyclical' and the 'linear' to describe diverse forms of rhythms in everyday life.

First, the 'cyclical' rhythm originates from natural phenomena, such as tides, celestial movements, the alternation of day and night, and the changing of seasons. Lefebvre (2013) described it as:

'The processes and movements, undulations, vibrations, returns, and rotations are innumerable, from the microscopic to the astronomical, from molecules to galaxies, passing through the beatings of the heart, the blinking of the eyelids and breathing, the alternation of days and nights, months and seasons '(Lefebvre, 2013, p. 76).

These cyclical rhythms or periodic repetitions are inspired by the natural world. However, a significant proportion of cyclical rhythms are also artificial, such as the movement of crowds to and from tourist attractions and rush hours for commuting. As Lefebvre (2013) stated:

"These last rhythms (schoolchildren, shoppers, tourists) would be more cyclical, of large and simple intervals, at the heart of livelier, alternating rhythms, at brief intervals, cars, regulars, employees, bistro clients. The interaction of diverse, repetitive, and different rhythms animates, as one says, the street and the neighborhood" (p. 30).

The rhythms of tourists, children and shoppers are cyclical, and their appearance and disappearance are based on a scheduled timetable. The order or repetition produced by the rhythm itself highlights the division of

labour and the distribution of productivity in society. Under this social rhythm, leisure time also becomes part of the production. As Lefebvre (2013) explained, 'leisure activities are also produced (and productive), although they are proclaimed free and even "free time" (p. 32).

The linear rhythm originates more from social practices and represents an 'imposed structure' (Lefebvre, 2013, p. 8), governed by clock time. For Lefebvre, it is a "series of identical facts separated by long or short periods of time" (2013, p. 76). Lyon (2018) described it as "quantified and fragmented time, imposed by technology, industry, and consumption" (p. 35). This research contextualises Lyon's (2018) viewpoint which suggests that a series of in-game repetitive behaviours is designed to encourage more in-game payments.

These two styles of rhythm—cyclical and linear—are not in opposition but rather exist in unity within their opposition, interacting and interfering with each other. They form a relationship of mutual comparison and measurement, where one is the measure of the other and vice versa. Although this study highlights the prevalence of cyclical and linear rhythms in mobile games, the two are inseparable and cannot be completely distinguished. They exhibit inter-containment and cross-referencing within games. The composition and interaction of diverse rhythms from different parts—much like the organs of the human body—construct a cohesive ensemble. The process of rhythmanalysis integrates these rhythms from various sources into an analysable state of being. As Lefebvre (2013) stated, " the act of rhythmanalysis integrates these things—this wall, this table, these trees—in a

dramatic becoming, in an ensemble full of meaning, transforming them no longer into diverse things, but into presences" (p. 23).

Lefebvre (2013)' s concept of rhythmanalysis reflects multiple aspects of daily life. Beyond linear and cyclical rhythms, there are many studies focusing on the notions of polyrhythmia, eurhythmia and arrhythmia. Based on Lefebvre's (2013) original descriptions, rhythmanalysis scholar Dawn Lyon (2019) further clarified these concepts in her book (Table 8).

	Lefebvre's original description	Dawn Lyon's clarification
Polyrhythmia	"It suffices to consult one's body; thus the everyday reveals itself to be a polyrhythmia from the first listening'"(Lefebvre, 2013, p. 16).	"A multitude of rhythms; the effects of different configurations of rhythm"(Dawn Lyon, 2019, p. 25 ).
Eurhythmia	"Rhythms unite with one another in the state of health, in normal (which is to say normed!) everydayness" (Lefebvre, 2013, p.16).	"When rhythms combine smoothly, for example in good health" (Dawn Lyon, 2019, p. 25).
Arrhythmia	"When they are discordant, there is suffering, a pathological state (of which arrhythmia is generally,at	"Discordance between rhythms, being 'out of step'" (Dawn Lyon, 2019, p. 25).

Table 8: The Definition of the Core concept of rhythmanalysis.

	the same time, symptom, cause and	
	effect). The discordance of	
	rhythms brings previously eurhythmic	
	organisations towards fatal	
	disorder" (Lefebvre, 2013, p. 16) .	
Isorhythmia	"isorhythmia (the equality of rhythms)	"An 'equality of rhythms'
	completes this repertoire of	beyond eurhythmia, for
	fundamental concepts. With one	example in the
	reservation: isorhythmia - and	coordination of an
	eurhythmia are mutually exclusive"	orchestra" (Dawn Lyon,
	( Lefebvre, 2013,p. 67).	2019, p. 25).
Dressage	"One can and one must distinguish	"Process of bodily
	between education, learning and	entrainment and
	dressage or training [le dressage]	repetition through which
	Humans break themselves in [se	rhythm is learnt and
	dressent] like animals. They learn to	becomes evident in the
	hold themselves. Dressage can go a	body over time, for
	long way: as far as breathing,	example in military drills"
	movements, sex. It bases itself on	(Dawn Lyon. 2019, p.
	repetition" (Lefebvre, 2013, p.39).	25).

In further research on rhythmanalysis, scholars from different disciplines have applied these concepts to various topics. Geography and History researcher Marcu (2017) focused on the mobility experiences of young Eastern Europeans in Spain. She categorised mobility into three types: arrhythmic, polyrhythmic and eurhythmic mobility. Marcu (2017) described 'arrhythmic' mobility as 'disruptions and loss' (p. 414). 'Polyrhythmic' mobility is defined as "situated between unpredictability and flexibility" (Marcu, 2017, p. 414). 'Eurhythmic' mobility is described as "composed of diverse and equilibrated rhythms" (Marcu, 2017, p. 414).

Specifically, the experience of arrhythmia is linked to young people's loss of employment or temporary work. The disconnect from stable employment leads to a " concatenation of multiple spatio-temporal insecurities" (Marcu, 2017, p. 410). The conflict between the expectation of secure salaried work and the reality of high mobility intensifies the discordant experience of temporal rhythm. Meanwhile, the precarity and flexibility of young people also refer to polyrhythmia in rhythmanalysis. The experience of travelling to different cities creates a multiplicity of rhythms for individuals who experience rapid changes in their life state. Lefebvre's (2013) concept of eurhythmia is evident in interviewees with stable studies or permanent work contracts, reporting regular routines and patterns of social practices. From Marcu's (2017) perspective on rhythmanalysis, everyday life is an interweaving of many different rhythms.

The best-known case of rhythmanalysis is Lefebvre's (2013) balcony observation. This involved observing the interaction between different, repetitive and varying rhythms: the rhythms of cars, regulars, employees and

patrons of the bistro, as well as those of walking people, traffic lights, buses and tourists. For Lefebvre (2006), everyday life is an interconnected whole; human life cannot be understood as a mere sum of daily activities. As he stated:

The word "daily" refers to the set of everyday acts, and especially the fact that they are interlinked, that they form a whole. Implicitly, it is accepted that daily life does not boil down to a sum of isolated acts: eating, drinking, dressing, sleeping, and so on, the sum total of consumer activities. Except when society is defined exclusively by consumption (something increasingly rare), there is an awareness that consideration of these isolated acts does not exhaust daily life, and that we must also attend to their context: the social relations within which they occur. Not only because each action taken separately results from a micro-decision, but because their sequence unfolds in a social space and time bound up (p. 17).

The life of an individual belongs simultaneously to the individual, the social group (family, company, etc.) and society. As he stated, daily life is "simultaneously individual, 'group' (family, colleagues and friends, etc.), and social" (p. 17).

This view can also be understood through Wertheimer's (1938) concept of 'Gestalt'. According to the principle of gestalt, people perceive existing things as wholes rather than as independent parts. Each part has specific functions or attributes, but it can only be defined in relation to the whole. This means that when people think of 'life', they are considering the life experience as a

whole. The independent parts of daily life cannot be completely separated from life as a whole.

The interlinked nature of daily activities implies that 'playing mobile games' is not an isolated activity. First, 'playing' is 'embedded' in personal life and is connected to a wide range of other activities. Second, 'playing mobile games' relates not only to the individual but also to the social context. As Lefebvre (2006) stated:

"Everyone knows how to live in 1981. They know it thanks to a knowledge that does not originate with them, which they have assimilated, and which they apply to their cases, managing their personal affairs – their everyday lives – in accordance with the models developed and diffused for them" (p. 72).

This knowledge is not intrinsic but learned through patterns designed for them. In 1981, the women's magazines provided daily schedules tailored for women, including activities such as buying and selling items, dress codes and cooking menus. Each day's schedule was filled with per-determined tasks. The purpose of these designed life patterns was to produce daily satisfaction. However, their ultimate aim was to produce discontent. As Lefebvre (2006) argued: "It is merely the result and the application to daily life of a management technique and a positive knowledge directed by market research. The economic prevails even in a domain that seemed to elude it: it governs lived experience"(p. 72).

Lefebvre (2006) used the example of the tourism industry to illustrate how the economy influences experience. The leisure industry provides tour plans or group tours, where people buy pre-prepared experiences, including the

discovery of new places. People buy the 'discovery', the change of scene, the departure and escape, which often proves disappointing because they no longer align with the initial wish (not desire) and advertisement. The tourist industry perfects the organisation of leisure and culture, fragmenting it into exchangeable pieces like space. The extraordinary sells well, but it becomes a sad mystification. This way, the image of pseudo-freedom takes shape, practically organised and substituted for "genuine freedom, which has remained abstract" (Lefebvre. 2006, p. 72).

In rethinking gaming, the disassembly of mobile games reveals a series of predetermined plans, similar to tourism plans. These plans can be seen as a series of nested linear or cyclic rhythms that predefine the player's experience. Just as tourist attractions pre-design tourists' viewing locations, video games per-determine the reward content for player actions. Everyone who lives in the mobile game world knows how to live in the game.

In 2011, game culture studies scholar Tom Apperley applied rhythmanalysis to the field of game studies. He believed the rhythmanalysis method is "a conceptual tool that is able to cope with approaching the analysis of the intersection between everyday life and the digital game ecology on a number of different scales" (Apperley, 2011, p. 19) and emphasised the importance of the body in the engagement between daily routines and the in-screen world. He pointed out that the body is the convergence point of in-game and out-ofgame rhythms, reflecting their harmonies, clashes and feedbacks (Apperley, 2011). Rhythmanalysis provides a way to incorporate everyday presence into the digital game ecology. The gaming body, as the centre of polyrhythms, includes a series of intertwined daily rhythms, including but not limited to

eating, working, sleeping, toileting, communicating with others, lighting, digital and control devices, power and software. As he stated, "The bundle of gaming body is a network composed of the elements required to produce play with the literal body of the gamer as its central node" (Apperley, 2011, p. 38).

Apperley (2011) introduced the term 'local rhythms' (p. 84) to understand the interaction between video game rhythms and the player's everyday life rhythms. He contrasted 'local rhythms' with global rhythms to emphasise the life rhythms generated by people in specific geographic settings. According to Apperley (2011), players' gaming rhythms are generally affected by local rhythms. In his study of Venezuelan cybercafés, Apperley (2017) noted:

They attended school in shifts, some in the morning, and others in the afternoon. Students used the café as a meeting place, to gather on their way to and from school, as well as to crossover and socialize with students from the other shift (Apperley, 2017, p. 237).

In his research, Apperley (2011) considered the periods during which specific groups of people entered the café to play digital games as cyclical rhythms influenced by local life rhythms, such as commuting to and from school.

However, Apperley's (2011) research on game rhythms did not explore the impact of specific in-game mechanics on rhythm production, nor did it address mobile games. Nowadays, when people log in to mobile games, they can still experience a well-designed schedule, visit the guild, buy a new outfit for their character, and gain a higher reputation by killing monsters in the field, etc. There is a designed pattern that tells players what they are expected to do. The managing skills and knowledge of the game designer segment the

experience into quantifiable and actionable steps with the fundamental aim of meeting user demand. Pre-designed needs, expectations and satisfaction can be detached from the existence and the experience itself. The 'practically organised time' of exploring experience and freedom is produced in the process of playing.

#### 2.8 Conclusion

This literature chapter revisits the concept of the 'magic circle' (Huizinga, 1938) in video game research history. In early video game development, the game world and the real world were separate. However, in the current gaming context, mobile game mechanisms attempt to incorporate more real-life elements into the virtual world. The evolving game industry makes the game and the gaming process highly compatible with the players' lives, including, but not limited to the merging of in-game environments with out-of-game geographical locations, the synchronisation between in-game and out-of-game time and the exchange of out-of-game currency for in-game items. With the concept of hybrid space discussed in this chapter, people are simultaneously living in the overlap of virtual and real spaces. Digital games are no longer pure, ideal townships: they engage people in real life across multiple dimensions, including space, time and intimate relationships.

These overlaps can lead to potential risks for game players. The exchange of out-of-game currency for in-game items may lead to the phenomenon of free labor in online games, especially in China, where digital labour is widespread but not well-recognised. The number of invisible 'gold farmers' may be greater than currently acknowledged (Liboriussen, 2016). This phenomenon is also

influenced by cultural and historical contexts. The concept of 'immaterial labour' is often used in feminist discussions on unpaid household labor. Due to the lack of feminist discourse in China, mainstream opinions struggle to address female unpaid domestic work. This lack of reflection on labour definitions may make the concept of unpaid digital labour harder for the public to understand.

Additionally, Rettberg (2008) and Zagal et al. (2013) pointed to the risk of video games increasing player engagement to the point of compulsion. Rettberg's (2008) description of endless tasks explains the motivation for players to stay in the gaming world, while Zaga *et al.* (2013) proposed the concept of 'dark game design pattern' to describe how games insert themselves into players' daily lives, even controlling their daily schedules to meet gaming requirements.

In the area of in-game relationships, as Huvila et al (2010) noted, these relationships are not simple mirror images of real-world relationships. Huvila et al. (2010) and Yee (2006) offered optimistic views, while and Williams et al. (2006) took a more pessimistic stance. Relevant research indicates that ingame relationships can be commodified, yet they also have the potential to enhance players' social support levels. At the end of this chapter, Henri Lefebvre's (2013) rhythmanalysis theory and spatial theory are introduced. These theoretical frameworks will be extensively applied in the data analysis chapter of this research.

In addition, the Skinner boxes discussed in this section are also related to the linear and cyclic rhythms in Lefebvre's (2013) rhythmanalysis framework. The

fixed interval schedule, as seen in the experience point system, reflects linear rhythms, while the progression through the Battle Pass system involves a series of cyclic rhythms. A more detailed analysis of these mobile game mechanisms is provided in Chapter 4.

In summary, this chapter presents research on game players' time use, living space and in-game intimacy. Most of the literature focuses on single-player games or MMO games played on personal computers or consoles. Although the mechanisms of mobile games may not differ significantly, the differences in game devices may cause variations in how players interact with the game. These differences caused by the development of game devices, may represent a new gap in current game studies.

#### **Chapter 3 Methodology**

#### 3.1 Introduction

This chapter outlines the research methodology, including the participants recruitment strategy, coding process, and study limitations and challenges. Specifically, section 3.2 detailes characteristics of participants, including the gendered, educational and occupational background, the methods employed to recruit them, the organisation of question sets used in the interviews, and the online interview process. Depending on the type of mobile games the participants had experienced, they received tailored follow-up questions during the interviews in order to encourage them to share their daily gaming experiences..

Section 3.3 describes the coding methods utilised in this study, drawing from Deterding and Waters's (2021) three-step coding process: attribute codes, index codes, and analytic codes. Attribute codes refer to both player characteristics (such as gender, educational background, and professionals background) and mobile game attributes, such as game genre. These codes facilitated the comparison and analysis of the interviewees based on different attributes. Index codes are used to establish cross-interviewee-case connections, in this thesis, they applied for comparison between the daily gaming experiences of different participants. Analytic codes are directly related to the research topic which link the interviewees' responses to the research questions to the theoretical framework and contribute to constructing the analysed chapters.

Section 3.4 describes the researcher's insider status within the mobile gaming society. As a researcher, I am also a natural insider of this society, familiar with its sub-culture, terminology, collective memories and common sense,.This insider status enables me to build trust with my interviewees and influenced my observations during data analysis. My dual identity as both a researcher and a gamer shaped my perspective in this study.

Section 3.5 explores the challenge of context translation. Both the interviewees and the researcher were familiar with in gaming terminology, and specific in-game terms were used throughout the interviews. Hence, individuals unfamiliar with the gaming terminology may find the original interview text confusing. Moreover, the interviews were conducted in Mandarin, and many Chinese gaming terms, rooted in a specific subcultural

history or context, lack direct English<sup>11</sup> equivalents. To avoid confusion, l included research notes for clarification. However, these notes may not fully explain complex game terminology background and history, potentially resulting in information loss during translation.

# 3.2 The Participant Information, Recruitment, and the Development of Rhythmanalysis Method

During the recruitment process between October 2020 and June 2021, the researcher contacted 48 Chinese mobile game players to participate in the study.

Due to the COVID-19 pandemic, the interviews were conducted online. Between October 2020 to June 2021, the interviewees were recruited through three online gaming forums: the NGA game forum, the Zhihu game area, and the Douban game group.

Among these forums, Zhihu and NGA had more male players (Weiboyi and Pangjing,2022, Yingchao, 2019), while Douban had more female players. NGA player's tended to be older and had more gaming experience. The researcher posted recruitment notices on these forums, collected respondents' email addresses, and sent detailed invitations to participate in the study. Upon agreeing to the informed consent form<sup>12</sup>, the participants were offered a selection of available periods and online communication tools (including Skype, Discord, Tencent QQ and Wechat),for the interview. They were

<sup>&</sup>lt;sup>11</sup> For example, "Washing face' means "Begging for good luck', "diamonds' or "stones'refer to the "ingame currency for gamble'

<sup>&</sup>lt;sup>12</sup> With the consent of the Ethics Committee, the interviewee may not use his/her real name for signing the informed consent form. The user ID on social media software or an in-game pseudonym is allowed to signature.

allowed to choose the preferred time slot and conversation tools for the formal interview.

All paticipants in this study were over 18 years old and legally considered adults. Their ages ranged from 20 to 40 years, with most participants between 20 and 25 years old. For privacy reasons, the researcher did not inquire about their exact ages, but rather age ranges. Of the participants, 29 were male, and 19 were female.

Regarding their occupational backgrounds, 17 interviewees were current university students, 29 were employed, and 2 were in a gap or relaxation period. Both professional and non-professional participants were recruited through the same channels. At the beginning of the interviews, the researcher asked the participants whether they had any relevant work experience in the gaming industry, and the those who answered affirmatively were classified as professionals. Nine interviewees had relevant experience in the gaming industry, including one game journalist, one game operations manager, one game reviewer, three game designers (with one being an independent game designer), two online game planners, and one individual who preferred not to specify positions.

Compared with non-professional participants, the professional participants were more inclined to offer more direct negative reviews on mobile game mechanics. For example, one professional participant, who described himself as a 'player of anti-mobile game', explained that most game industry professionals enter the field aiming to 'produce a good game'. However, due to pressures from commercial entities and career prospects, the freemium

model often does not align with the designer's personal intentions. Instead, it is driven by the requirements of the company, the market, and capital investment. The duality of professional requirements and personal wishes was also evident in the interviews with other professional participants. Some provided two views on the same question: one as a professional and the other as a player.

Due to limitations of the sampling method, volunteer interviewees presented features that may not reflect with the broader mobile gamer demographic.

First, the average education level of the participants was higher than the actual Chinese mobile game players' education level. All participants had attained or were pursuing university-level education, including two PhD students and two with international study experience. Second, the participants might have a higher interest in video games than the general population. They likely had a more extensive experience in gaming, including personal computer and console games from an early stage in their lives. Many interviewees mentioned they have at least one game console. In China, the ratio of console gamers to mobile gamers stands at approximately 16.7 million (Nikopartners, 2023) to 657 million (Statista, 2023), with approximately one console gamers in this study is exceeds this figure. Third, the mobile games frequently mentioned by interviewees tended to be more trendy and of higher quality. The pirates or copycat games were hardly mentioned, despite their market presence in China.

Finally, the interview data showed that all middle-aged participants had gaming experience (whether console or personal computer games) at a younger age. This early engagement might explain why they established their identity as gamers in early life. However, due to the mainstream media context in China in the early 21st century, the identity of gamers is often associated with low education levels, low income, unemployment, or oversexualisation (Jiayi Zhang, 2014). It is, therefore, challenging for those with negative perceptions of gaming to identify as mobile game players, even if they engage in gaming on their mobile phones. The digital divide further complicates this issue for older phone users in China who often have access only to a limited range of social media content. Despite this, the games that spread through social networks could still influence this elder group, even if they do not personally identify as gamers. Therefore, this research might only include individuals who self-identify as game players, thus excluding those who play games but do not self-identify as such.

For these reasons, the findings of this research may be limited as the participants of this study may not be representative of the average population in terms of their gaming habits and attitudes. They represent a cohort that is more knowledgeable about video games and gaming culture and more inclined to qualify as 'hardcore' and in-depth gamers than the general population.

### 3.2.1 The Interview Structure and Questions

This thesis used an online semi-structured interview method for the data collection.

This semi-structured method was chosen to provide flexibility in the interview process. First, the mobile gaming market in China is highly dynamic, with players frequently changing their primary games in several months. This variability made it challenging to predict which games interviewees were likely to play. Second, as a kind of subculture, experienced game players often communicate using specific gaming terminology. The diversity of mobile games and the complexity of gaming terminology has led to a rich variety of language texts that people use to describe their in-game experiences. Interviewees might use different language systems to describe the same thing. For these reasons, maintaining a degree of openness during the interviews was essential in the thesis.

In the month leading up to the formal interviews, as part of the recruitment process, I downloaded the top 10 games in each category from the Chinese software application market. I logged in and played these games as a new player to ensure that I could understand most of the popular in-game terms and use these terms effectively in conversations with the interviewees. After that, my supervisors and I developed a basic framework including ten questions. These questions covered aspects of players' daily play time, play space, and their relationships with others.

#### 3.2.2 The Interview Process

According to the interviewees' preferences, the online interviews were held via WeChat, Skype, and Tencent QQ. Forty-three interviewees consented to voice recording, while four refused to be recorded and opted for text communication only. One interviewee agreed to record, but the recording
document was corrupted due to equipment failure; as a workaround, she signed the interview text transcript as an alternative. At the beginning of the interview, I read the brief informed consent form and asked the interviewees whether they were legally adults, had perused the complete informed consent form in advance, and understood their right to have their interview data deleted post-interview by contacting the researcher's email address.

Once the interviewees affirmed that they had read and signed the informed consent form, the interview formally began. The researcher initiated the interview with questions from a pre-structured framework and followed up with specific follow-up questions based on the interviewees' responses. At the beginning, the interviewees were asked three to four questions to identify their main daily mobile game. After that, each interviewee was presented with ten pre-designed main questions. However, the order of these questions and the follow-up questions was flexible. As Adams (2015) discussed:

The agenda for a semi-structured interview is never carved in stone, If a conversation unexpectedly turns from the first to the fourth topic, by all means, reorder the topics on the fly and return later to pick up the ones that were skipped (p. 198)

When an interviewee's answers deviated from the topic of one question to that of another question, the researcher guided them to fully express their thoughts on the diverted topic before steering the conversation back to the original questioning sequence of questions.

The follow-up questions varied based on the responses to the main questions. For example, Main Question 2 asked, 'Could you describe your daily routines

and explain how mobile gaming fits into it? Have you played the mobile game when you are doing the other stuff?'. For interviewees who affirmatively answered this question, possible following follow-up questions included:

'What are the other daily activities you do while playing the game? What times do they usually happen?'

'How do you play mobile games and do other activities at the same time?'

'Do mobile games and other activities ever interfere with each other?'

In contrast, if interviewees indicated they did not engage in other daily activities while playing games, they received different follow-up questions, such as:

'Have you ever been interrupted by other things while playing games?'

'Do you schedule uninterrupted periods in your daily life for gaming?'

For those affirmatively responding to Main Question 8: 'Have you formed any relationship within in-game character.? It was this important to you and why did you choose to make these connections?', a series of follow-up questions about the relationships within the virtual world and with game characters included the following:

'How would you describe a non-player character with whom you had developed a relationship?'

'To what extent, do you believe in the realism of the mobile game world? To what extent do you believe that virtual relationships reflect your real emotions?'

'Do you have real-life relationships that co-exist with virtual relationships? (e.g., having both a game character and a real-life romantic relationship) How do you think in-game relationships differ from real-life relationships?'

These follow-up questions were generally open-ended and relatively flexible. They were also tailored to the interviewees' specific gaming experiences. If a player's main game did not have any non-player character (NPC) with whom they could develop an emotional relationship, related questions were omitted.

During the interview process, follow-up questions underwent a developing and iterative process and evolved with the researcher's growing understanding of the research topic and framework. This iterative approach was profoundly influenced by the interviewees' responses to the follow-up questions. Later follow-up questions were often shaped by insights gained from earlier interviewees' responses to specific questions.

For example, in the early stage of the interview process, the typical follow-up questions about playing times focused on specific periods of playing, such as lunchtime or bedtime. However, after an interview with a participant who answered a series of questions while playing mobile games, it became apparent that gaming could happen in brief, discontinuous fragments throughout the day, rather than in a uninterrupted block of time, it may happen at the intervals of daily routine, in the gaps of daily events. This realisation led to the refinement of follow-up questions to explore how gaming fits into daily

life, the moments players engage with or disengage from gaming, and how these moments interact with, influence, or are influenced by other aspects of their daily routine. Similarly, when an interviewee mentioned that he always played in the taxi because it required minimal attention, the researcher noticed the potential for mobile gaming to represent a re-organisation of daily attention. Thus, I developed follow-up questions regarding how players allocate their attention to mobile gaming within their daily lives.

Further, some follow-up questions about intimate relationships within games were inspired by the experience of an interviewee who maintained a romantic relationship both with an in-game character and an out-game partner. These follow-up questions sought to explore this type of 'virtual-reality' relationships through questions such as, 'What do you think is the difference between ingame and out-of-game relationships?' And 'do you, or to what extent, you believe in the emotions game characters showed to you?' Another interviewee's mention of 'holding hands' with another player within a game prompted the development of more follow-up questions about the in-avatar of out-game self such as, 'Whether, or to what extent, have you consider your ingame avatar as an extension of yourself?' Such follow-up questions, developed early in the interview process, were applied in later interviews to capture more details about players' experiences. The progression of the interview process was also affected by my growing understanding regarding the topic, which evolved through diverse participant narratives about their daily gaming interactions.

Across all 48 interviews, most participants did not limit themselves to a particular game type. They were more likely to download several games on

their smartphones and to allocate their playtime according to personal preferences. While some had periods of dedication to one game before switching to another. A segment of the interviewees had preferences for particular types or styles of mobile games for a long term. However, the majority of player frequently changed games, which led to a focus on the games they were currently or had recently played. If an interviewee had been playing two or three games recently, questions were asked about each game separately to capture a boarder range of experiences.

## 3.2.3 The Build of Rhythmanalysis Method

This research applied rhythmanalysis as the main framework for data analysis. It benefited significantly from conversations in the early stage of the interview process. In these initial interviews, respondents frequently mentioned that their daily mobile gaming experiences were characterised by regular repetition. The updated daily and weekly tasks required them to repeat in-game actions with which they were already familiar. Earlier work in this thesis was built on analyses of the repetition observed in these interviews. Thus, the rhythmanalysis framework was applied to analyse concerns regarding daily repetition. Lefebvre (2013) argued that rhythm is built on the repetition inherent in daily routines. In this research, the concept of rhythm provided a concrete shape to the mobile game players' daily recurring repetitions. It also offered a framework for detailed observation of the life order organised by mobile games and other entities (e.g. companies, families, and social organisations) in daily life.

When an interviewee mentioned having to repeat the same in-game content consecutively for eight hours and continuously click on her smartphone to maximise in-game benefits, I realised this was a linear rhythm occurring in the context of everyday life, as described by Lefebvre (2013). Also, I found that the updated daily and weekly goals could be analysed as in-game cyclical rhythms that the producers aim to impose on the player's body. Coincidentally, mobile game players usually refer to their daily gaming activities as ' $\Pi \mathbb{R}$ ,' a term that denotes all the repetitive tasks they need to perform in the game each day. This term can be directly translated to 'routine,' which is also a key concept in Lefebvre's (2013) framework on everyday life.

In the early stage of the research, the plan was to apply the rhythmanalysis method as one chapter of the analysis framework. However, through in-depth conversations with interviewees and following suggestions from supervisors, the importance of the rhythmanalysis approach in understanding players' everyday lives became evident. Lefebvre's (2013) argument on rhythm, automatisms, or spontaneity became the core focus of the interview process. A large percentage of the interviewees expressed a sense of being forced or obligated, claiming that they felt expected to engage in mobile gaming before the expiration of daily rewards. Despite often feeling physically tired or bored, they tended to log in regularly. However, mobile gaming cannot force anyone to be a 'player'; players can uninstall games from their phones at any time. This blurring of voluntary and involuntary engagement could be strongly interpreted by the rhythmanalysis method. Lefebvre (2013) explained that objects consume time and occupy individuals' time and space, and the rhythm

authentic" (p. 75). Yet, this perception is fragile and illusory. The introduction of new elements to the daily experience can disrupt original rhythms as individuals negotiate a new steady state. This argument explains how mobile games consume players' daily time.

From the view of Lefebvre (2013), daily rhythm is an acquired or trained state. The tendency of individuals to attribute results from external constraints to an essential need highlights the ambiguity of automatisms or spontaneity in daily rhythm. This framework effectively addressed the paradox between players' willingness to engage and their sense of obligation and provided a strong structure to analyse the conflict between the mobile gaming demand for daily time and other daily event requirements.

#### 3.3 Code and Data

The coding process for this research is based on Deterding and Waters's (2021) three-step coding method, in which the code is divided into attribute codes, index codes, and analytic codes. The coding process was performed using NVivo 12 and was later upgraded to NVivo 20. As Weston et al. (2001) highlighted, coding does not precede the study but is an important part of the analysis. A part of the coding system was developed through communication with the interviewees. It also underwent several adjustments throughout the study.

# 3.3.1 Attribute Codes

Deterding and Waters (2021) described that attribute codes are:

"the salient personal characteristics of the interviewees that played a role in the study design. These may be demographic variables such as age, gender, race, occupation, socioeconomic status, or specific contextual data such as experimental group or control group, state, or neighborhood of residence" (p. 724)

In this research, attribute codes include two groups. The first group involves human attributes, including gender, age, educational level, and occupational profile. The second group pertains to mobile game content and mechanisms, Codes in the second group covered three aspects: (1) the main playable content of the game is provided by the environment or other players; (2) the focus of the game's content is on competition or cooperation; (3) the presence of automated mechanisms (detailed further in Chapter 4). In particular, the presence or absence of automated mechanisms significantly influences the analysis of mobile games' impact on players' daily schedules. These attributes helped determine whether players need to devote their full attention during the gaming process.

Players and non-player characters can foster deeper emotional connections in games that emphasise interaction with a computer-controlled environment (Player vs. Environment game). However, this similar emotional connection is nearly absent in competitive (Player vs. Player) games. Games with automated mechanisms do not require players to maintain high concentration throughout. Instead, much of the gameplay unfolds automatically, requiring players to make simple clicks with fingers. Conversely, non-automated mobile games necessitate active player engagement. Players must continually control their game character and interact with the game world. A prime

example is Genshin Impact., an open-world mobile game where players navigate their characters to explore and engage with the environment. Activities such as running, jumping, engaging in dialogue, attacking, dodging, and using skills require specific player inputs on the screen. Similarly, accessing the game's playable content—including challenging dungeons, opening boxes, and interacting with NPCs—requires continuous command inputs from the player. Therefore, the player's physical interaction with the game is unceasing. The mechanisms of mobile games significantly affects how players engage with the game and manage their gaming time within their daily schedules. Thus, the game's mechanisms constitute a critical aspect of the attributes code.

## 3.3.2. Index Codes

The next step in the coding process involves index coding, which is " the inverse of line-by-line coding. Index codes represent large chunks of text, enabling data reduction and retrieval as the analyst proceeds through constructing and documenting their argument" (Deterding and Waters, 2021, p. 726). In this study, this approach enabled comparing the interview texts across several different questions and linking them to broader content areas. In total, six index codes were developed to categorise aspects of how a player's attitude towards mobile gaming influences their daily time, space, and intimacy.

The first index code is 'The Player's Description of the Mobile Game Mechanisms'. This code captures the interviewees' descriptions of in-game

design and mechanisms, including daily tasks (daily goals) system, log-in systems, and stamina systems, etc.

It also includes their responses related to online payments: In mobile games, specific freemium items (e.g. doubling supplies and increasing battle speeds) can significantly reduce the repetitive in-game actions. Additionally, other freemium items (e.g. monthly cards and battle passes) require daily logins or the completion of a certain number of daily in-game actions for players to gain additional rewards after purchasing these items. Hence, the interviewees' willingness to make payments and their views on freemium items are closely tied to their daily in-game behaviours.

The second index code is 'The Player's Description of Their Daily in-Game Action'. This code includes the duration and timing of players' engagement with mobile games. Typically, players have regular playtimes; four key periods were identified as common gaming times: upon waking up, after lunch, after dinner, and before bedtime. This pattern does not imply that every player engages with mobile games during each of these periods: in general terms, most interviewees preferred playing during two or three of these times, with post-dinner being the favoured slot. The daily playtime spanned usually between 1 and 2 hours, with the duration a player spent in the game each day being relatively consistent.

The index code also captures players' feelings about their daily in-game actions. Initially, gaming was often be described as a fun experience: however, over time, the interviewees' aim shifted towards acquiring to 'get new characters or items in the game', or 'gather in-game resources to strengthen

their characters'. Some interviewees used the words 'salary' or similar descriptions to described their motivation for completing daily tasks or daily goals. A significant number of interviewees used the term 'routine' or similar terms, such as 'daily list to do' and 'clearing daily stamina points' to describe their in-game daily activities. Nearly all descriptions of daily in-game repetitive efforts are closely tied to the objectives pre-set by the game's developers, highlighting the influence of designed game goals on player behaviour.

The third index code is '**The Impact of Mobile Gameplay on the Player's Daily Schedules:** This code focuses on the interviewees' descriptions of the interactions between mobile gaming and their other daily activities. These interactions include activities happening simultaneously, one activity interrupting the other or occurring in a seemingly arbitrary sequence. Many interviewees reported being influenced by the mobile game's schedule to varying degrees. They might find themselves subconsciously aligning their daily routines around the game's reward schedule or attempting to schedule other daily activities according to negotiate with their in-game commitments (which is discussed in Chapter 6).

The fourth index code is **The Geography Location of Daily Playing:** This code pertains to the descriptions of a playing spaces. Some interviewees reported playing mobile games exclusively at home or in their residences, while others indicated engaging in gameplay in transit or at work, including on the subway, bus, company shuttle, office, laboratories, and classrooms. This index code also captures interviewees' responses about the differences between playing at home and in public spaces. According to the interview data, the private residence remains a traditional gaming environment. When

interviewees discussed the distinctions between playing at home and in other public spaces, they shared their views regarding various gaming environments.

The fifth index code is '**The Relationships between Human Players**'. This index code covers the interviewees' discussions about their interactions with other human players. It includes three aspects:, the relationship between the player and another player, the game organisation, and the online public discussion group.

According to the interview data, interviewees were more likely to develop ingame relationships with others who had already established a weak social tie with them. Becoming friends with individuals met in mobile games is uncommon. Most people prefer to play games with those they already know in real life. Previous studies on gaming for intimacy, such as those by Wen, Kow, and Chen (2011), Boudreau and Consalvo (2014), and Osmanovic and Pecchioni (2016), have documented cases of gaming fostering family relationships. However, in this thesis, interviewees did not report playing games with family members from previous generations. The development of existing relationships observed in this research typically involved classmates, roommates, colleagues, and neighbors-usually among people within the same age group. Normally, the role of an online friend involves collaboration to complete in-game cooperative content. As Greitemeyer and Cox (2013) noted, "players help other players to attain a common goal" (p. 224), such as defeating monsters by the help of other players, borrowing the game characters from each other, or exchanging in-game materials.

The term 'game organisation' typically refers to a game guild, which can range from a small group of individuals to a large organisation with hundreds of members. Among the interviewees, a common experience was joining a game guild and its associated chat room. However, they often remained silent within these chats. Some noted that other guild members posted non-game-related messages, such as complaints about exams or concerns about work, but only a few admitted to sharing aspects of their own lives. Typically, chat rooms are dominated by a small number of active members, with the majority remaining silent. Some interviewees expressed a desire to avoid joining a guild or engaging in in-game social activities altogether, preferring a more loosely organised connection with the guild. They opted to use the in-game message board for communication rather than social media, which allows for asynchronous and word-limited interactions (the game's message board supports only short messages).

The public discussion group typically refers to online game forums, which host a vast community of mobile game players. It serves as an online gathering space for players of the same game, often manifesting as a website, a social media platform, or an application. Compared with game guilds, game forums are more loosely organised and commonly feature game tips, articles, and player-created fan art. However, discussions in many online game forums are is not strictly confined to gaming topics. Interviewees have reported that they often read posts from other users sharing daily experiences, whether related or unrelated to the game, such as romantic relationships, studies, and friendships. As members of the public discussion group, users can share a common identity as players of specific mobile games. This shared identity

sometimes leads to sharing personal information within this game-related discussion space. The impact of these three types of in-game relationships is further discussed in Chapters 5 and 6.

The sixth index code is '**The Relationships between Human Players and the Non-Player Character**'. This index code was inspired by Waern's (2011) research on players who develop romantic feelings for in-game characters. It includes interviewee descriptions of their their feelings and actions during the interactions with non-player characters. Interviewees reported that they developed emotional relationships with NPCs which could be male, female, gender-neutral, animals, or anthropomorphic objects. They also described their emotional connections and the interactions with these NPCs. Some interviewees mentioned that they would hold a birthday event or send a message to the character in a non-game environment. Those relevant descriptions are also coded under this index code.

#### 3.3.3. Analytic Codes

Analytic codes "represent the concepts to explore in a single paper or book chapter and integrate emergent findings with what is known from the literature" (Deterding, and Waters,2021, p. 722). In this research code, the analytic codes are closely interconnected with Lefebvre's (2013) rhythmanalysis relevant concepts, especially for eurhythmia and arrhythmia.

First, Lefebvre (2013) defined rhythm as 'movements and differences within repetitio' (Lefebvre, 2013, p. 90)'. In this thesis, this code led to the development of an analytic code focusing on '**Repetitive Tasks**'. This code also drew on the research into gold farming and online labour by Dibbell

(2007), Dubbell (2015), Liboriussen (2016), and Rettberg (2008), etc. The distinction between repetitive and non-repetitive playtime produces different feelings in players' daily routines. Players often use the term 'farm' to describe their repeated daily actions in the game. This term can also be used to refer to repetitive playtime. During these periods, interviewees described their daily in-game behaviour from login to the completion of daily gaming, including completing challenges, defeating enemies, collecting items, and interacting with NPCs. These 'repetitive' actions exemplify rhythm performance and production in mobile gaming and show its impact on players' emotions, which is further explored in Chapter 4.

Second, according to Lefebvre (2013), polyrhythmia is "suffices to consult one's body; thus the everyday reveals itself to be a polyrhythmia from the first listening" (Lefebvre, 2013, p. 16) and eurhythmia occurs when "rhythms unite with one another in the state of health, in normal (which is to say normed!) everydayness" (p.16). This research interprets polyrhythmia as the moment of mobile gaming and other daily activity's muti-tasking status, and eurhythmia as the harmonious integration of mobile gaming into the player's daily life rhythm.

In examining the polyrhythmia and eurhythmia within the context of mobile gaming, the concept of '**Multitasking'** emerged as the second analytic code. In games which include automated mechanisms, almost all interviewees admitted that they were used to engage in other activities while gaming. At these moments, the rhythm of mobile gaming intertwines with other daily activities, such as listening to music, watching TV shows or movies, eating lunch or dinner, using the toilet, working in the office, cooking, exercising, and

chatting with friends or family members. These activities are united with one another in a 'normal' (Lefebvre,2013, p. 16) way. According to the interviewee's descriptions, players often place their mobile phones to one side, regularly tap the phone, and simultaneously engage in other tasks. Some individuals concurrently open a mobile game and work software, quickly switching between the two.

Lastly, arrhythmia is the discordance between rhythms, being 'out of step' which Lefebvre (2013) describes as "the suffering, a pathological state" (p. 16). In the context of mobile game rhythmanalysis, arrhythmia indicates that the player's original daily life rhythm is interrupted and distorted by the mobile gaming rhythm. This research identified '**The Negative Feedback on Unfinished Game Tasks'** and '**The Player's Sense of Compulsion'** as relevant analytic codes.

The negative feedback on unfinished game tasks emerged from interviewees' descriptions of their feelings about not completing a task or the time they have not dedicated to gaming. This often includes missing daily or weekly tasks, failing to complete game events, or not logging into the game on time, leading to a loss of in-game rewards. Such negative reactions to disruptions in the playing rhythm are considered part of arrhythmia and rhythm conflict.

The player's sense of compulsion emerged in the interviews where interviewees expressed a sense of compulsion regarding their daily gaming tasks. Despite citing reluctance and reasons such as physical fatigue or appointments, they would inevitably complete the daily mobile gaming tasks.

This sense of compulsion often reflects moments when the rhythm of mobile gaming conflicts with the rhythm of their non-gaming everyday lives.

Interviewees, preoccupied with daily commitments, may miss the deadlines of in-game daily goal systems. Due to this oversight, they may suffer strong negative emotions, typically described as a sense of 'loss' or 'missing' something. While some interviewees expressed that they might not be concerned with the specific in-game items missed, they were more affected by the feeling of missing out. When they discussed the equivalent value of unobtained items, such as in-game currency, they might show a more resigned attitude. However, the emotional impact of an 'unfinished game daily task' tends to be stronger than that of missing out on specific items. The motivation for regular play appears to lean more towards 'avoiding loss' rather than 'gaining reward'. Players may strive to maintain the behavioural inertia of daily play, a tendency that contributes to the production of daily rhythm: this aspect is revisited in Chapter 6.

Furthermore, this research used 'The Influence of Other People/ Non-Human Player Characters on the Player's Behaviour' as an analytic code to examine the rhythms—whether combined or in conflict—in the interviewees' daily lives. During the eurhythmia process, this influence was usually reflected through covenants, shared goals, and the desire to develop or maintain relationships between the interviewee and others. The interviewees reported playing a particular game to spend time with others, share a common identity, or maintain a topic of conversation. This influence extended to common standards of playing time and frequency, the considerations for maximising ingame benefits, or standards to related to in-game character abilities.

Similar to relationships between players, the relationship between humans and non-human characters, controlled by computer algorithms, can also motivate individuals to change their behaviour. Adhering to the requirements of a machine program represented by a human-like character may cause rhythm conflicts between the player's daily routine and the in-game rhythm demands. The interactions between the player and others significantly shape the player's in-game behaviour, potentially leading to rhythm conflicts, a topic that is discussed in Chapter 6. When an interviewee could not follow the rhythm provided by others, they risked experiencing arrhythmia. Using the analytic codes discussed, this thesis explores the impact of mobile gaming rhythm on the player's daily life. The relevant discussion is shown in Chapters 4,5 and 6.

## 3.4 The Researchers' Internalities and Limitations

Sociologist Merton (1972) introduced a structural conception of 'Insiders and Outsiders' to describe the role of social science researchers within the observed groups. According to his view, insiders are " the members of specified groups and collectivizes or occupants of specified social statuses" (p. 21), while outsiders are the non-members. Within this conceptual framework, I am an insider within the mobile game player group. My engagement with mobile games began seven years ago, and my total gaming history extends to 22 years which is almost the same as the longest duration reported by any of my interviewees. Being an insider provided me with many advantages during the interview process. As summarised by Labaree (2002). these advantages include "the value of shared experiences; the value of greater access; the

value of cultural interpretation; and the value of deeper understanding and clarity of thought for the researcher" (p. 103). These advantages of insider status were evident throughout my research process.

As introduced in the first chapter, until the submission of this thesis, I have managed a personal video game review social media account with around 20 thousand followers for seven years and had worked in games journalism prior to attending university. Half of my friends are currently or have been employed in the gaming industry, and my articles have amassed millions of reads within the gaming community. Some of the interviewees were familiar with my work or had read my articles before deciding to participate in the research. Upon receiving the informed consent e-mail, some individuals chose to Google my name, read the results, and consequently decided to trust me. The presence of my articles and social media posts within the gaming subculture has significantly facilitated the establishment of trust and connection with interviewees. They regarded me as a member of their group, readily sharing their personal experiences. Outsiders might find this level of access and truthfulness challenging to achieve.

As a researcher, I am obliged to reflect on my interiority, yet it is undeniable that my research material and data significantly benefitted from this interiority. The interviewees shared their personal feelings based on a shared identity and background. We share a common space of experiences, including playing games in the middle of night, playing games in public, being admonished by elders and parents, and using gaming as an escape from crowds. These insider perspectives allowed me to gain a deeper

understanding of my interviewees' thoughts and to construct analytical logic from their narratives.

However, it is also essential to acknowledge the limitations that this insider status had on my research. Throughout the research process, my identities as both a mobile game player and a researcher were intertwined. For example, when my TAP mentor, Hayley, pointed out the possibility of gaming addiction among my interviewees, I realised that I had been avoiding the related topic. As an insider, I might unconsciously refrain from employing negative stereotypes such as violent, addicted, criminals, or any similar descriptors for players. This 'subconscious avoidance' does not refer to exclusion or denial but indicates a blind spot—a bias in perception that arises from my position as the observer within the group. Similarly, when writing Chapter 5, my supervisor inquired about how mobile game players switch between gaming and reality. This was an aspect I had overlooked: as a gamer, this switching was so routine for me that it escaped my notice. This observer blind spot means I might miss out on research insights that could appear evident to an outsider.

Besides that, my understanding of mobile games might shape or influence the language of my interviewees. On one occasion, after an interviewee had finished his response and I was about to transition to the next topic, he became keenly aware of my familiarity with all the evidence, data, and viewpoints he had mentioned. He then asked, 'Why would you ask me a question to which you already know the answer?' This question made me realise that I have a pre-existing, consensus-based system of understanding as a mobile gamer or video game player. Despite my efforts to control it, this

internalised knowledge invisibly shaped my reflections on the interviewees' answers, influencing my expressions, body movements, and tone of voice. etc. This, in turn, may have led my interviewees to share what they perceived as the consensus among game player—what I supposedly already knew—rather than their own personal daily experiences.

In insider research, the expectations of interviewees become more obvious, as illustrated by the example of Beoku-Betts (1994). As a researcher of African descent, she had an advantage in conversing with African women. However, cultural norms also dictated her avoidance of interactions with local Black men. Her insider status implied that conducting her research without being influenced by the expectations of the observed group was challenging. From the perspective of my research, my interviewees expected that I would not to label them as 'addicts' or 'people controlled by mobile games'. This expectation might differ from that of an outsider researcher. They were also likely to withhold parts of their playing experience from an outsider of the gaming community.

For a social science researcher, claiming complete detachment from the expectations of the observed group is challenging. As Labaree (2002) described his experience with a faculty senate study, "Responsibility to one's community is a particularly vexing issue because any false representations of the phenomenon, either real or perceived, could lead to feelings of betrayal on the part of the participants" (p. 110). Whenever I quote an interviewee's words, I grapple with the challenge of fully conveying the implicit meanings of their statements. This struggle leaves me feeling as though I have betrayed the community of gamers or disappointed them. This emotional burden confirms

that I cannot maintain the stance of a neutral outsider. I am constantly concerned that my academic writing may not accurately reflect the real or perceived realities of the gaming community, always doubting whether there is a significant gap between my descriptions and the truth of the complex, delicate relationship between people and games.

In the data analytical chapter, which explores the combination and conflicts between the non-game daily life rhythm and the mobile game rhythm, I found that the phenomena of combination or conflict were mirrored in my own life. I was actively adjusting my work-life balance and mental health through playing mobile games. In this context, I navigate the boundary between being the researcher and the observed subject of my research. My personal experiences share much in common with those described by my interviewees, leading me to a dilemma highlighted by philosopher Ralph Barton Perry (1910). Perry (1910) suggested that researchers may face an 'ego-centre predicament' when attempting to solve a certain problem. As he posited,

Within my own field of consciousness, I may attempt to define and subtract the cognitive relationship, in order to deal exclusively with the residuum. But after subtracting the cognitive relationship, I must still 'deal with' the residuum; and 'dealing with' is a variety of the very relationship which I sought to banish. Finally, just in so far as I do actually succeed in eliminating every cognitive relationship, I am unable to observe the result. Thus, if I close my eyes I cannot see what happens to the object; if I stop thinking I cannot think what happens to it; and so with every mode of knowledge (Perry, 1910, p. 8).

This implies that a researcher cannot eliminate all ties with the subject of study. In my case, my knowledge of game players, the gaming community, game history, and game design methods served as my 'eyes' to perceive the implicit context of my interviewees' discussions. This knowledge came from my insider identity in the gamer community and is rooted in my long-term subjective gaming experiences. I aimed to analyse and understand the interviewees' discourse from an observationalist perspective. Following Labaree's (2002) argument, 'For the insider, who is already a 'native', the corollary presumption is that true understanding can only be attained by 'going observationalist' (p. 116), I sought to analyse the mobile game rhythm from a position that maintains a distance from the familiar player's daily routine. However, the experience and identity of being an insider might also affect my observation, understanding, and analysis throughout the research process.

For this reason, it is necessary to convey explicitly how my subjectivity might influence the research I conduct. The insider perspective as a game player affords me certain advantages and simultaneously impacts my view of game rhythm. This research is deeply shaped by this insideness.

#### 3.5 Language Translate and Intercultural Context

In this thesis, for the convenience of the reader, all interviewee names are represented by English pseudonyms. The method for assigning these names was as follows: The research used a computer program to randomly generate 50 male and 50 female names. The interview materials were then numbered from 1 to 50, after which a pseudonym was randomly linked to each number.

This research opted for pseudonyms based on English rather than Chinese because Chinese pseudonyms would require phonetic transcription, potentially leading to confusion for the reader.

Further, video gaming, including mobile gaming, is a context-rich domain where players often use specific terms to describe their experiences. The language within gaming communities can be complex and may be challenging for outsiders to understand. For instance, when players of the same game refer to 'crystal' 'diamond 'and''stones', they are actually discussing the same in-game currency, which can be purchased with real-world currency, despite the official name for this in-game currency possibly being different. These terms may originate from early popular mobile games that have since faded from popularity, yet the terminology persists in gamer language. To build trust with interviewees, the researcher also adopted gamer language during interviews. However, in writing this thesis, it was necessary to translate gaming terms into the daily life language. While efforts were made to accurately capture the interviewees' descriptions, some nuances of the original meaning might inevitably be lost due to the translation process.

In addition to translating the Mandarin interviewee's text into English and converting highly contextualised gaming terms into daily expressions, the researcher needed to navigate the cultural contextual differences between different countries. For example, the 'overtime' phenomenon has different cultural contexts in the UK compared with China. In many Chinese companies, a significant proportion does not compensate for legal overtime, and employees are expected to work until 8-9 pm regardless of whether they have completed their tasks. As a result, one strategy for employees might employ is

to deliberately delay their work throughout the day, stretching the 8 hours of expected work into 12 hours. During this extended time, mobile gaming becomes an option to occupy the hours. Similarly to the 'overtime' example, certain aspects of Chinese culture are challenging to translate into brief terms. This hidden context might result in information loss during the transformation between different cultural and social backgrounds.

#### 3.6 Conclusion

This chapter introduces the process of collecting the interview data which might have the following limitations: First, the demographic distribution of the research interviewees does not mirror that of the general mobile game player population. The interviewees in this research were generally more educated, had a longer history of gaming, and most had experience beyond mobile gaming. This leads to the potential for the research interviewees' in-game experiences to differ significantly from those of the general population. This discrepancy may stem from the sampling method, which only included individuals who self-identify as game players, thus excluding those who play games but do not self-identify as such.

Second, the researcher had an insider status in the research process. As a long-term game player, my knowledge and experience in mobile gaming provided advantageous access and trust. However, this insider role also limited the analysis to the researcher's subjectivity.

Third, the interviewees' language texts may have incurred losses through multiple cross-cultural and cross-contextual translations. Fully translating the

game context and the context based on Chinese society and culture into English presented significant challenges.

This chapter details the interviewees' basic information, the process of conducting semi-structured interviews, and the coding methods used to analyse the interview data. The following three chapters explore the analysis of interviewee data based on the rhythmanalysis framework. Chapter 4 discusses how in-game mechanisms produce linear and cyclical rhythms and how these rhythms are integrated into the everyday life of the players. Chapter 5 focuses on the interplay and co-functioning of mobile gaming rhythms and daily life rhythms, while Chapter 6 addresses scenarios where gaming rhythms and non-gaming rhythms conflict.

# Chapter 4 Establishing Gaming Rhythms within Everyday Life: From the in-screen World to the out-screen Physical World

## 4.1 Introduction

The first three chapters of this thesis introduce the background of China's freemium gaming environment and the methodology for data collection. The following three chapters explore polyrhythmia, eurhythmia and arrhythmia between gaming and non-gaming daily life. In this chapter, the terms 'in-game rhythms' and 'out-game rhythms' are used to describe rhythms associated with mobile gaming and those related to other daily activities, respectively. Further discussion on this topic continues in Chapters 5 and 6.

Based on Game scholars Tom Apperley's (2011) argument about online gaming rhythms, this study expands the intertwining of game rhythms and

daily life rhythms from the space in front of the computer/console to a broader time-space social area. This expansion of the time-space gaming area is driven by the 'mobile' nature of gaming devices, particularly with the popularisation of smartphones. However, the shift in gaming trends from stationary home-based play to mobile gaming does not necessarily result in a quantitative increase or decrease in playing time. Instead, it enables players to organise and schedule their gaming activities more freely according to their own preferences. For example, Sabina, an interviewee, detailed how she arranges her daily gaming time:

'An hour in the morning, half an hour at noon, playing in the evening until dinner, and playing after dinner until bedtime'.

——Sabina.

Sabina's gaming time is evenly integrated into different segments of her daily life, such as after waking up at home in the morning, during her lunch break at work at noon and in the evening after dinner at home. Her act of playing is neither continuous temporally nor spatially, as her daily gaming occurs in three periods (morning, noon and evening) and across three locations (the bedroom, office and dining room). This flexibility blurs the boundaries between gaming and non-gaming activities. For players like Sabina, the portability of smartphones allows them to engage with games anytime and anywhere. Gamers can thus constantly reorganise the rhythm of gaming to match the rhythm of their lives at any moment of their daily schedules.

At the same time, it is necessary to consider the impact of mobile games on human behaviour, rather than merely regarding them as leisure devices.

According to interview data, players do not turn on their phones to engage in games solely out of personal desire; they are also motivated by the fear of missing out on rewards pre-designed into the game's timetable. This chapter discusses how in-game rhythms influence players. In general, it explains how gaming rhythms are produced in the digital world and how these constructed rhythms affect the non-gaming world. In this thesis, the 'gaming rhythm' refers to a series of regular and repetitive in-game behaviours that occur due to mobile game mechanisms<sup>13</sup>.

This chapter is divided into five sections. **Section 4.1** introduces this chapter. Section 4.2 discusses how in-game mechanisms generate linear and cyclical rhythms, supported by interview data and descriptions of the interviewees. Linear rhythm refers to the repetition of a player's in-game actions over a short period. For example, players may challenge the same in-game level repeatedly. In the physical world outside the game, this manifests as the player tapping the screen repeatedly with their fingers. Subsection 4.2.1 introduces linear rhythm in the gaming world by describing two in-game mechanisms (the stamina point system and the dual-currency system) to help understand the combined rhythms and rhythm conflicts that are analysed later.

In subsection 4.2.2, the discussion transitions to cyclical rhythm, which involves repetitive actions performed by the player within specific time intervals, such as daily, weekly, monthly or annually. Typical examples include challenging in-game bosses<sup>14</sup> three times a week, collecting materials or sending gifts to friends daily. The cyclical rhythm of mobile games is

<sup>&</sup>lt;sup>13</sup>Such game mechanisms include the login time rewards, guild battle periods, and the rules of in-game events, etc <sup>14</sup>In game basses are relatively powerful in game mensters.

<sup>&</sup>lt;sup>14</sup>In-game bosses are relatively powerful in-game monsters.

usually bound to social time: it serves as a mechanism to link in-game rewards with real-world time. The subsection explores the game content cyclical system and the monthly pass system to help understand how cyclical rhythm functions in mobile game design. Subsequently, Subsection 4.2.3 explains how linear and cyclical rhythms interact within the game. It introduces the daily/weekly task and battle pass systems to explain the combination of multiple in-game rhythms.

Section 4.3 discusses polyrhythmia, multitasking and semi-automatic features of mobile games. Lefebvre (2013) described polyrhythmia as the configuration of multiple rhythms. In mobile games, semi-automatic mechanisms, such as auto-run gaming events, allow players to leave the game running for short periods. This feature means that the way the mobile game fits into players' time is intermittent and encourages multitasking in their daily lives. Furthermore, the semi-automatic and highly repetitive nature of mobile game interactions requires less of players' attention compared with the traditional games. This feature also fosters the occurrence of mobile game behaviour in parallel with other everyday activities.

**Section 4.4** discusses in-game avatars. When players log into the game world for the first time, they are assigned an in-game avatar to represent themselves. This avatar, which can be a model or image, is not merely a blank slate; it typically comes with a designed background story, social relationships, personal identity, motivations and goals. These attributes can only be selected or modified within the limits allowed by in-game mechanisms. In other words, the avatar has an innate set of internal rhythms. The mobile game uses the in-game avatar as a trainer or guide to help players adapt to

the in-game rhythm. The section also details how the rhythms of the digital avatar influence the rhythms of the physical world.

D'Aloia (2009) regarded the avatar as a spiritual bearer for the player in the digital world. Hence, the avatar's pre-engineered rhythms can be transferred to the player's physical body outside the game through the link between the human player and the digital avatar. This connection allows the rhythm of the game to influence the actions of the player outside the game. For instance, when a player takes an action outside the game, the in-game avatar performs a corresponding reaction on the screen. Simultaneously, the avatar's in-game timetable. designed by game developers, integrates into and influences the player's daily life schedule.

To execute a command action for a digital avatar in a mobile game, the player needs to tap, touch or swipe on their smartphone screen. Through predesigned interaction logic between humans and smartphones, the in-game behaviour is translated into the actions of the player's biological body. This connection reflects all in-game rhythms into the player's daily behaviour outside the game. As Gee (2008) described in his argument about embodiment in video games, "the player, acts in the game as if the goals of your surrogate are your goals" (p. 258). By embodying the character, players inherit the goals or needs of the in-game avatar and are be given the powers, limitations and even the perspective of the avatar.

In total, this chapter briefly introduces how mobile games establish rhythm through in-game mechanics. It also explains the role of the in-game avatar and the transmission process between in-game rhythms and daily life rhythms

outside the game. This sets the foundation for the following two chapters, which explore the combination and conflict between mobile gaming rhythms and non-game rhythms.

## 4.2 Linear and Cyclical Rhythm Performance in Mobile game

The following subsections proceed with a rhythmanalysis based on Henri Lefebvre's (2013) theory, using common mechanics in mobile games to explain how the linear and cyclical rhythms are produced within the game environment.

# 4.2.1 Linear Rhythm Performance on Mobile Gaming

The linear rhythm in mobile games is usually expressed as a series of repetitive in-game actions. This section begins by introducing descriptions from interviewees to show how the repetitive behaviour manifests in daily ingame daily playing. The discussion starts with transcriptions of daily tasks (daily goals) from three interviewees: Zachary, Werner and Andrew:

'What is the daily goal? It is like waving to a good friend, opening a door, chasing a light appearing on a particular map, or, for example, lighting some candles or fighting with many crabs to fight.'

——Zachary

'If that is what you are talking about, there should be some specific dungeon you will have some of each day and knock those out.'

-----Werner

'The daily tasks are mainly completing fixed challenges, killing enemies, and something I do not feel sure about, like collecting vegetables or similar activities'

——Andrew.

Regarding of the terms used by the interviewees—whether crabs, candles, dungeons or commissions—their descriptions highlighted a similar repetitive action process. They repeatedly engaged in the same quantifiable standard behaviour within the game.

Andrew's example of challenging dungeons can be used to explain how ingame behaviour is analysed in linear rhythmanalysis. The process of 'completing a challenge' begins when the player clicks on a button to start a specific challenge, waits for a while, clicks on the screen to receive the reward and then clicks the 'start challenge' button again. During this process, the player performs an action equivalent to tapping the phone screen twice. The position of the taps and the timing between the two taps are the same when the player performs this action repeatedly. These repeated and regular clicks are typical performances of linear rhythm in the daily gaming. Next, this section introduces two common mechanisms that foster repeated clicking and produce linear rhythm.

The first mechanism is the stamina points system. Mobile games limit the time a player can engage in daily game activities by using stamina points. These stamina points naturally increase over time, typically by one point every 3 to 8 minutes. Players must spend a specific number of points to participate in ingame activities. For example, in 'Genshin Impact', stamina points are called

'Original Resin'. When it falls below 160 points, the game automatically increases it by one point every 8 minutes (Genshin Impact Wiki, 2023). Each time a player collects an in-game reward, they need to consume Original Resin, typically between 20 and 40 points. Meanwhile, acquiring average ingame equipment (referred to as 'Artifact' in the game) for a character requires several dozen challenge attempts. As the content of the challenges is usually fixed and the amount of stamina points increases over time, the player's behaviour in consuming stamina points becomes endlessly linear and repetitive. These repetitive in-game behaviours establish the linear rhythm produced by mobile game mechanisms imposed on players.

In some interviews, interviewees expressed a desire to use their stamina points in the 'most efficient' way, which greatly intensifies the repetition, as seen in the case of Calvin. Calvin plays a mobile game called Arknights, which allocates 240 stamina points per day. To obtain the most cost-effective rewards, he plans to spend all the stamina points in a precise way.

'We calculated that, if we used the stamina points on the Challenge 1-7, we would get the most rewards ... Nevertheless, that challenge only has a little consumption of stamina, so it takes a long time to operate, I have been playing for a month, and I cannot stand this'.

——Calvin

The challenge he described (named 1-7) consumes six stamina points per attempt. Therefore, Calvin needed to complete the same in-game challenge 40 times a day. As previously mentioned, these 40 times of the in-game commands are entirely repetitive.

The motivation for players to spend stamina points, especially in the 'most cost-effective way', is to acquire materials to reinforce their in-game characters. In order to 'grow' their characters (generally meaning to increase their abilities), players need to collect experience points or specific materials. These are usually obtained by repeatedly spending stamina points to complete challenges. For example, Calvin needs to tap his phone screen in exactly the same manner 40 times. The in-game challenge actions performed by the avatar and the out-of-game tapping performed by the player create the linear rhythm that is a product of the mobile game's design.

Another interviewee, Cornelius, shared a similar experience:

'Sometimes you complete the same challenge several times over for a certain material, or that currency, or experience point., although I do not want to do it a second time'

——Cornelius.

Each time he tries to pass the challenge, the required in-game action remains the same. This theme of repetition is echoed by Rowena, who described her own experience:

'There could be hundreds of times when you start; (when you begin) it could continue like that for the next twenty minutes.'

---- Rowena

For the interviewees, spending daily stamina points on the challenge may involve repeating actions that have just occurred or have occurred repeatedly

over time. These repetitions do not occur naturally but are designed and guided by an artificially designed machine program. Players need to click specific locations on the screen to begin and end the challenge. The content, timing and frequency of these actions are pre-designed by game mechanics. The process through which a player receives rewards from a repeated challenge serves as training, fostering or educating them to engage with a game-based linear rhythm. As described by the interviewees, Cornelius and Rowena, this repetition in game behaviour aligns with the definition of linear rhythm as a "series of identical facts separated by long or short periods of time" (Lefebvre,2013, p. 76). To keep the stamina points increasing, players must collect or spend them at regular intervals. A game's stamina maximisation limit should be viewed as a signal, hint or instruction that influences players to return to playing at regular intervals and should be regarded as guidance for fostering the linear rhythm been designed by the mobile game.

There may be a form of rhetorical trickery within the narrative framework of the mobile game system, where a mobile game allows players to spend stamina points on a specific challenge, suggesting a game-given reward for the player. In contrast, the experience that players might have received—such as a whole and controllable character and a complete story—is broken into countless pieces. This requires players to repeat specific behaviours many times to obtain it. Stamina points are a tool designed for repeated challenges, splitting players' experience and keeping them engaged in endless gaming.

One of the interviewees, Sabina, had set an alarm for midnight to remind her to log in and collect some in-game resources that her character needs. In this

case, Sabina's primary objective is to enhance her in-game character. To achieve this, she has developed a secondary objective: using her stamina points more effectively. For this developed secondary aim, she set this midnight alarm to avoid wasting stamina points. This case illustrates a conversion of demands within the game system, where a player's in-game needs are converted into out-of-game behaviours. The path for this conversion between in-game and out-of-game actions can be dictated by the mobile game's pre-designed system.

In this context, the logic between players' natural desires (to strengthen their characters, pass levels and know the next part of the story) and their actions can be designed to a considerable extent arbitrarily. Another interviewee, Juliana, exemplified this: She repeats the in-game challenge countless times to obtain the game's Christmas reward. Each challenge lasts 15-20 minutes, totalling approximately 60 hours spent during in this specific Christmas game event. In this case, the time spent per challenge, the number of challenges and the number of rewards are all adjustable within the pre-designed game system. The game company may design highly repetitive content to induce players to pay to avoid parts of it, for instance, acquiring strong characters or equipment<sup>15</sup>, rare item or the option of jumping the challenge.

Stamina points are designed to quantitatively assess and limit the highly repetitive content of games. On the one hand, they limit how often a player can engage with the repetitive content within a given period. On the other hand, they encourage players to continuously consume their stamina points, which automatically regenerate over time. This ongoing consumption of

<sup>&</sup>lt;sup>15</sup> These items or characters usually require Gacha.
stamina points eventually becomes a habit, which is converted into a gamebased linear rhythm.

After that, the in-game dual-currency system emerges as the second common mechanism for supporting in-game linear rhythms. The dual-currency system allows players to use both out-of-game currency and in-game virtual currency generated within the system to purchase in-game items. Typically, there is a one-way conversion from out-of-game currency to in-game currency. However, in some specific games, through a player-to-player trading system, players can also exchange virtual currency for real currency. The most common application of items in a dual-currency system is for loot boxes.

Loot boxes are a random reward payout mechanism that links in-game virtual currency with out-of-game currency. On the one hand, players can be rewarded for completing daily or weekly tasks with in-game currency, which they can then use to purchase loot boxes. On the other hand, loot boxes can also be bought with out-of-game currency. Hence, loot boxes are a type of purchasable item for both in-game and out-of-game currencies, making the value of a player's repetitive in-game tasks measurable in real money.

For the interviewees, the moment of opening a loot box may provide a strong positive feeling. The reward items within individual boxes are randomly generated<sup>16</sup>, obtaining a 'Superior Super Rare' reward item is an exciting experience. For instance, Lance expressed that he prefers this moment as the most enjoyable part of his gaming process, as his description as following:

<sup>&</sup>lt;sup>16</sup> There has been some debate about whether the loot box is random. It probably depends on the different mechanisms of each gaming company.

'Opening the Loot box is probably the only moment where I can feel the fun'.

——Lance.

To acquire a loot box, players must either continuously complete repetitive tasks or directly pay for it. Through the loot box mechanism, the linear rhythm within a mobile game is linked to consumption. This repetitive in-game behaviour can be regarded as a "social practice that comes from human activity" (Lefebvre, 2013, p. 8), according to Lefebvre's theory of rhythmanalysis. The repetitive in-game action represents quantified and regulated social time, and players may use this regulated time to exchange for randomly distributed rewards by the game system. On the one hand, to prolong the player's online time, the game imposes repetitive tasks on them; on the other hand, this repetitive work can be significantly reduced through online payments.

Interviewee Wright, who works in mobile game design, supported the positive correlation between playing time and the players' willingness to consume based on his experience:.

'The longer the online time, the longer the player has to play the game, the more likely he is to bring more benefit to the mobile game'.

—— Wright (mobile game designer)

In the digital gaming world, the materials that the players pursue, such as experience points and items for their characters, are almost cost-free in a physical sense. They are composed of pictures, numbers and program logic. The perceived price of an in-game item is not determined by its actual production and distribution costs but by the amount of effort required by players to acquire it, or in other words, by the number of repetitive actions they are required to perform.

Through this method of game design, the value of an in-game item (e.g. characters, weapons, equipment, experience points and enhancement materials,etc) can be measured by the number of times an action needs to be repeated by a player. Powerful or rare items require more repetitions, while weaker or common items require fewer repetitions. Since items usually feature a random reward mechanism, the actual number of repetitions may be uncertain but generally fluctuates within a fixed range. At the same time, the in-game repetitions require the player to invest a corresponding amount of time and attention as a cost. Thus, the player's private time and attention invested in the game become the weight that is exchanged for the value of the game item and the measure of the value of the game's items.

In the case of the interviewee Dora, he spends 30-40 minutes per day in the game and earns a tenth of one mobile game character. In the mobile game he played, some characters are free, while others can only be acquired by paying out-of-game currency or by performing repetitive actions in-game. In the latter scenario, paying with out-of-game currency is equivalent to 300-400 minutes of in-game repetitive actions. The value of the mobile game character, as an exchangeable commodity, is validated by the personal time and attention invested by players such as Dora and becomes the metric for other players who use the out-of-game currency to acquire the same item.

With in-game items as intermediaries, mobile games sell experiences that should inherently belong to the players. A notable phenomenon is that players often need to pay for opportunities to save time. Many mobile games include options to 'skip challenges' or 'speed up challenge animations' in their payment options. These skippable contents typically represent a series of repetitive game enemies or scenes that require players to continuously click the screen or watch the record of the last time they completed the same challenges multiple times. Hence, players are tempted to pay to avoid long, repetitive animations or the need to repeatedly click the screen numerous times. If a player refuses to pay, they would experience a more linear and completely repetitive in-game experience to achieve the same result as a paying player. In this case, the game content that can be paid to skip represents a linear rhythm imposed on the player's body.

Thus, the design goal of 'skip challenges' aligns with Zagal, Björk and Lewis's (2013) concept of dark game design patterns, where the game uses linear repetition to produce negative experiences for players to induce them to pay to reduce these negative experiences. In this context, the in-game linear rhythm is the method or means to create the experience that the player wants to avoid, an artificial schedule occupying the player's leisure time. The function of in-game consumption is to re-purchase of the normal gaming experience and daily leisure time that have been overtaken by the ruled in-game linear rhythm behaviour.

However, in traditional PC or console games, features such as double speed or 'skip challenges' are often included in the 'Options' menu at no extra cost. This contrast between PC or console games and mobile games may suggest

a further commodification of game functions. Through the dual-currency system, a player's repeated in-game actions can be bought with out-of-game currencies, leading to the potential exploitation of the player's attention or personal time.

In conclusion, this section argues that repetitive behaviour within mobile games exists. The game creates a to-do list to consume stamina points and establishes the in-game repetitive time as the standard of value for virtual products. With the intrinsic, inherent meaninglessness of this to-do list, the player's gaming time is typically highly repetitive, which further leads to a linear rhythm in the physical body of the player.

## 4.2.2 Cyclical Rhythms Performance in Mobile Gaming

This section explores the performance of cyclical rhythms within China's freemium mobile games. Cyclical rhythms include multiple actions that are linked head-to-tail and repeat at regular intervals. Compared with linear rhythms, where the same action is continuously repeated, game-based cyclical rhythms more often involve cycles that oscillate back and forth within a specific period. In mobile games, cyclical rhythms are usually performed according to a specific schedule. Specifically, the mobile game produces content or events that correspond to clock time, encouraging players to engage with the game at these specific times. This section focuses on two ingame mechanisms that generate game-based cyclical rhythms.

The first mechanism is the game content cyclical system, which refers to the availability of game content that changes according to the social timetable. The appointment model, which is based on real social time, has become a

renowned design technique, often referred to as the appointment dynamic. As Zagal, Björk and Lewis(2013)'s concept of 'play-by-appointment', a model based on real-world time that " requires players to engage at specific times and dates as defined by the game, rather than by the players" (p. 4). This model is exemplified in the interviewee's case discussed in this section, where the cyclical content and login systems in mobile games effectively mandate that players adhere to a 'play-by-appointment' schedule to engage in gaming.

For example, certain game content may only be accessible on specific days of week, month or year. This mechanism can be explained using the description provided by the interviewee Abner:

'There is the possibility of getting a particularly good reward [from participating in the in-game activity] on Saturday or Sunday, only once a month. I will participate whenever I play that game, which is part of my daily activities'.

——Abner

In Abner's description, the rhythm of mobile games is linked with the social calendar. The time frames he mentioned (Saturday or Sunday, every month) refer to the cyclical periods measured by out-of-game calendar systems. The in-game mechanism links the in-game event with the out-of-game calendar and repeats the cycle according to the days, weeks or months that pass on the calendar.

One common cyclical rhythm created by this link is the daily login activity. With this mechanism, mobile games provide rewards to players who log in during a specific period. Motivated by in-game rewards, players may develop

the behaviour of daily login, which Lefebvre (2013) described as cyclical repetition. The daily restart and rerun of the game each day correspond to the cyclical rhythm. This mechanism can be understood through the descriptions provided by the researcher and the interviewee Rowena:

'Rowena: Most of, or at least half of (Mobile) games, will ask you to log in at a particular times, especially 12:00–14:00, and then at night, there might be 6:00–8:00 pm or 7:00–9:00 pm. These two golden hours, ask you to go online to be able to receive stamina points or receive other benefits like that.

## The researcher: Did you do it?

Rowena: If I am not busy with work, I will. It also depends on the degree of addiction. For example, in some games, when I am in a grinding period, even when I am shopping with friends and I see the time is noon, like 12:00 pm, and I will also go online to receive (the in-game reward).'

The 'logging in' of the interviewee Rowena at a specific time each day has become part of her daily routine. In-game mechanics link the in-game reward to specific times of the day; as Rowena described, she logs into the game from 12:00 pm to 2:00 pm, 6:00 to 8:00 pm or 7:00 to 9:00 pm. This behaviour, which cycles every 24 hours based on established clock time, sets a daily cyclical gaming rhythm for players. Rowena and other players who adhere to this regular login schedule exemplify this daily routine. Through reward mechanisms, a mobile game integrates itself into the daily 24-hour cycle, aligning with the scale of the clock. In addition to the 24-hour system, the mobile game also incorporates mechanisms tied to the weekly cycle. For example, some game challenges are updated every Monday morning, and

challenges to obtain specific materials are available only on certain days of the week.

Other mobile game mechanisms are linked to natural months or country holidays, such as Christmas or summer holidays. Many games update the shop commodities on the first day of each month or hold events on bank holidays each year. For events that regularly occur during specific months of the year, interviewees may prefer to refer to them by the name of the holiday<sup>17</sup> rather than the official in-game event name. For example, in her response, the interviewee Juliana refers to an in-game event in December as a 'Christmas Event' instead of its official in-game name, '闪闪祭'.

In addition to mechanisms that tie game events to specific times, another common in-game mechanism for producing cyclical rhythms is the log-in reward system or the monthly pass system<sup>18</sup>. Both are reward distribution mechanisms within mobile games., The log-in reward system delivers rewards to players who enter the game daily, while the monthly pass is a payment option in the mobile game. Typically, a player pays a relatively low amount (normally 30 Yuan, around 4 to 5 pounds) and receives an instalment return each day for 30 days. Essentially, it distributes the commodity the player has already purchased at regular time intervals. The case of interviewee Elisa could help to explain this mechanism further:

<sup>&</sup>lt;sup>17</sup>Common names include Summer Events, Winter Events, New Year's Events, and Christmas Events, etc.

<sup>&</sup>lt;sup>18</sup>Depending on the effective date, the similar mechanisms also include the season pass, annual pass, and lifetime pass. The monthly pass is the most common one.

'I get nothing from my repetitive actions, but I have to do it for the rewards. Because I need to draw cards or buy other builds, I need stones<sup>19</sup>, so I have to go and do the quests. I could not get stones any other way unless I paid for them, and I did buy monthly cards for three or four months. With the monthly card, I could get 50 stones per day and some acceleration tickets [researcher's note: referring to the in-game time acceleration items]. The monthly card was supposedly the cheapest, costing 30 Yuan (about 4 pounds]. The game gave me 900 free stones per month [As mentioned above, the interviewee needed to perform repetitive actions to get these stones], then add 50 mending stones per day, then that is 1500, plus 30 days for daily tasks, that is 2400, and you only need to pay 30 Yuan [The direct purchase price of 2400 stones is about 240 Yuan, approximately equivalent to 30 pounds].'

——Elisa

By implementing cyclical reward delivery, mobile games encourage players to return at regular time intervals. As Elisa noted, the total reward from a monthly pass usually exceeds that from an instant top-up. However, the revenue a player accumulates is directly proportional to the number of days they log into the mobile game; the more days they log in, the higher the percentage of return becomes. Although the overall benefit of a monthly card is relatively high (1,500 stones), players can only access a limited portion (50 stones) of it in a single day. This suggests that game mechanics create a cyclical rhythm by regularly distributing rewards to players.

<sup>&</sup>lt;sup>19</sup> Stones, or diamonds. tickets,etc, refer to the in-game currencies used to exchange for Gacha chances

Besides the common login rewards, the cyclical mechanic that links game time with real time can also impose penalties on players, a topic previously discussed in research on mobile gaming. For example, Chess (2018) described the casual game FarmVille 2: Country Escape, where players are given a small plot of land to plant crops. These crops require real time to germinate, grow and bear fruit, compelling players to return regularly to tend to their crops. If players do not return after the crops ripen, the crops will wither, resulting in a loss of potential revenue. In this game, the growth cycle of the crops serves as a cyclical rhythm that marks the player's time by periods of productivity enhancement. This in-game cyclical rhythm also fosters a habit in players of returning to the game at regular intervals, as dictated by the game mechanics.

Similarly, research by Lu and Carradini (2019) identified a punishment mechanism in The Lords Mobile<sup>20</sup>, which includes a 'constant risk mechanic.' In this game, if players do not return regularly, their in-game assets are at risk of being attacked by other players. Unlike the case in FarmVille 2, the consequences in The Lords Mobile are more random. A player may leave the game for an extended period without any consequences, or they might be away for just slightly longer than the game's safety period and find their assets plundered by an attacker.

In the current (2023-2024) mobile gaming market, direct punishments, such as reducing players' resource income as described in Chess's (2018) article, are less common. However, mechanisms like monthly passes and login rewards are almost universally applied across mobile games, which can also

<sup>&</sup>lt;sup>20</sup> a mobile multiplayer online real-time strategy game.

be seen as a more subtle form of punishment. As illustrated in Elisa's interviewee case, each login rewards the player with 50 'stones.' Conversely, this mechanism implies a loss of 50 'stones' for each day the player does not log in, subtly penalising inactivity.

According to Zagal, Björk and Lewis (2013), in the 'play-by-appointment' model, the decision regarding when and for how long to play is dictated by the game, not the players. They argue that the game may 'manipulate or take advantage of a player<sup>21</sup>" (p. 7). The interviewees' descriptions of the cyclical mechanisms within their games highlight strong negative experiences, which will be further discussed in the following chapters.

Overall, drawing on the interviews with Abner, Rowena and Elisa, this section explores the relationship between mobile game mechanisms and real time. It introduces the login and monthly pass systems and analyses how these mechanisms establish a cyclical rhythm in the players' gaming experience.

## 4.2.3 The Intertwining and Combination of Cyclical and Linear Rhythms

The previous two subsections respectively described how the mechanics and design of mobile games produce linear and cyclical rhythms within the gaming environment. This subsection introduces two common mobile game mechanisms that combine these rhythms: the daily/weekly task and battle pass systems.

<sup>&</sup>lt;sup>21</sup> According to Zagal, Björk, and Lewis (2013), some games, such as Pokémon and Animal Crossing, incorporate mechanisms related to real-world time as an element, but they do not aim to manipulate or take advantage of players. While these game events may require players to participate at certain real times, players can still have a relatively complete gaming experience without engaging with these elements.

The daily/weekly tasks are a system that measures and calculates player ingame actions and awards points accordingly. This system refreshes the rewards each cycle (usually daily or weekly) and is available to all players at no cost. When a player performs a specific in-game action, the system awards them points. Once a player accumulates a certain number of points, the player receives the corresponding in-game reward. This system, which measures player performance through in-game scores, can be illustrated by the case of the interviewee Zoe, a player of a fashion design and matching game.

'My habit is to sign in, then I will send stamina points to my friends as well as receive stamina points. I will do five commissions for the Alliance and then go to the library to sort out some information. I will go to the Alliance shop to buy 1-2 items, do three commissions at the Dream Weaver<sup>22</sup>, and then draw once on the island... By this time, it is almost halfway at 3/5 of the progress. At this point, I will collect some stamina points and then go and do the Arena. Then I will see [what daily tasks I have not completed]...By this time, your task process is almost to 5/6. I will then see if there are any easy tasks you could do.'

——Zoe

In Zoe's detailed description, she used various phrases to estimate her progress in completing daily tasks, such as 'almost halfway', '3/5 of the progress' and 'almost to 5/6'. This way of quantifying progress suggests that

<sup>22</sup> an in-game building

she is calculating how many bonus points she needs to collect to obtain all the in-game rewards.

First, Zoe's in-game behaviour is cyclical. She begins the introduction of her playing process with the word 'habit'. This word suggests that engaging in daily tasks has become routine for her. During the interview, she indicated that the list comprises tasks she repeats daily to complete. The behaviours she described recur in cycles over her period of playing the game. Second, parts of Zoe's daily in-game behaviour contain linear tasks. For example, she mentioned 'do the Arena' (which involves challenging five opponents) and 'clear stamina points (as previously explained, consuming stamina represents a linear rhythm). These tasks require the completion of repetitive behaviours a specific number of times.

In Zoe's case, the daily task (or daily goal) system coordinates and integrates cyclical and linear rhythms. When the player completes linear tasks, such as using up 100 stamina points, their achievement contributes to the scoring system. This scoring system then distributes rewards based on the score and updates the rewards at a fixed time (normally at 4:00 or 5:00 AM). Since the system updates in a cyclical rhythm, the player's log-ins and engagement in linear rhythmic behaviours are intertwined with the updated cyclical rhythm. The player's in-game behaviour incorporates a series of cyclical and linear rhythms which are guided by the daily task system.

Furthermore, the refresh times for daily tasks in mobile games are also likely to influence people's habitual thinking about games. This is illustrated by the description of the interviewee James regarding his daily quests:

'After 4:00 AM., I suddenly returned to my senses and realised I had forgotten to do the daily task. It is like inertia, making it natural to remember to complete it'.

—James

Like James, other interviewees also showed that they habitually think about the status of their daily tasks before going to bed. James described his thinking about the game's daily tasks as 'natural' and other interviewees, particularly those who regularly participated in timed in-game events, exhibited similar attitudes towards completing their daily tasks. This phenomenon may suggest that, beyond physical engagement, players are likely to develop a mental habit of considering daily tasks at cyclical intervals with the guidance of the game's mechanisms.

Following the daily task system, the discussion explores the battle pass mechanism, which also demonstrates the combination of linear and cyclical rhythms. The battle pass system (players usually refer to it as the 'Big Monthly Pass') is similar to the daily/weekly task system. It consists of a collection of in-game tasks that usually last for 30–40 days or one 'season'<sup>23</sup> of the game. Players earn a certain number of bonus points (independent of the daily task bonus points) upon completing tasks associated with the pass. When a player accumulates a certain number of bonus points (usually 100 or 1,000), the battle pass is upgraded, and the player receives a reward.

Normally, there are two versions of the battle pass: the basic version and the paid version. The basic version is complimentary, while the paid version

<sup>&</sup>lt;sup>23</sup> Season or version refers to the period of major content updates in the game.

requires a 'seasonly' fee. The paid version offers better rewards compared with the free daily/weekly task system. Like the daily goal system, the battle pass guides players through a series of rewarding linear behaviours, and it usually has a more flexible time limit.

Both the daily task and battle pass mechanisms align with the definition of a daily goal provided by Deconstructor of Fun Guest Writer (2016), which stated, 'the player is given the opportunity to do a finite amount of a specific action, with these opportunities refreshing after a specific set of time'.

In this definition, 'a finite amount of a specific action' represents the performance of an in-game linear rhythm, and 'refreshing after a specific set of time' corresponds to the performance of a cyclical rhythm. Hence, the daily goal system can be viewed as an ensemble of cyclical and linear rhythms.

Additionally, the bonus points in both the daily task and battle pass systems resemble the 'symbolic currency' discussed by Christopher A. Paul (2018). In his book, Paul (2018) elaborated on dragon kill points (DKP), where players participating in an in-game activity, such as killing a dragon, have the opportunity to roll dice to determine the winner of an item they desire. Paul (2018) described this as an 'effort-plus-chance system' (p. 113) and argued that the DKP mechanism has evolved into a 'symbolic currency' used for exchanging in-game goods.

The reward system for daily tasks in mobile games represents a developed and more complicated form of DKP, where the player's in-game effort is transformed into bonus points. These bonus points can then be used as counters to reward the player with in-game currency. Typically, this in-game

currency is used to gamble for new characters, weapons or items. Compared with the DKP system in MMORPGs, the mobile game 'effort-plus-chance system' introduces an exchange between in-game currency and real-world currency. This may provide game producers with a stronger motivation to enhance players' desire for in-game currency. Without spending real-world currency, the main method for a player to acquire in-game currency is to follow the game system's guidance, which is a nested and intertwined system of linear and cyclical rhythms.

To keep players engaged with a game, designers often provide a set of aims. Werner, an interviewee who has worked as a mobile game community manager and reviewer, provided the following description using Hearthstone Legend<sup>24</sup> as an example:

'The daily task provides you with 50 gold coins: however, you often find yourself have 90 gold coins, You might want to get up to a hundred gold<sup>25</sup> coins, so you keep playing. This is what all free-to-play games are designed to do, [to make you] either invest time or money. If you have already invested money, you can continue to become stronger.'

——Werner

In Werner's case, when he finishes the in-game daily goal, it also means he has partially achieved another small goal, such as winning three campaigns, which are part of another cycle. Werner's experience demonstrates a threefold in-game rhythm: The first is the daily task rhythm that refreshes

<sup>&</sup>lt;sup>24</sup>A card battle game produced by Blizzard Entertainment, available for both mobile and PC.

<sup>&</sup>lt;sup>25</sup>In Hearthstone Legends, players can buy a card pack for 100 gold coins or \$1.99 . Each pack contains 5 randomly selected cards . When a player wins 3 campaigns, they will receive an extra 10 gold coins.

every 24 hours. The second is the victory reward that grants ten gold coins for every three victorious campaigns. The third is the rhythm of opening a pack, which requires one hundred gold coins to purchase. These nested and intertwined cyclical rhythms occur every day of playing.

In their descriptions, interviewees commonly used the term 'daily routine' to describe all the repetitive actions they perform as part of the daily goal system and the battle pass system. Normally, the 'daily routine' includes the in-game content where players spend most of their time. The term 'routine' likely stems from players' experiences of everyday repetition within the game. According to Becker (2005), the concept of 'routine' involves a recurring interaction pattern—frequent repetition of the everyday. Hence, the players may use this term to describe their daily experiences in the game due to the repeated everyday behaviour patterns and the feelings of monotony that accompany them. Furthermore, they often directly or implicitly describe their 'routine' as repetitive or tiresome. This is evident in the descriptions of three interviewees, Werner, Darlene, and Jack:

'I will give up on the mobile game when I do not do the daily tasks, not the specific day, but if I do not want to do the routine anymore. So, for example, if I spend much time at work someday and forget about doing the daily routine, that is tolerable. However, if I do not want to do the routine and feel like losing something about the routine reward, I will uninstall the game'.

——Werner.

'It depends on the game, and if I do not get bored with it, I will do all the daily routine'.

—— Jack.

'It is almost torture for me, and this mobile game is played without the joy of action. I do my daily goal and get rewards from there, rewards that allow me to strengthen some abilities of my character, but strengthening attributes does not please me. I complete my homework daily tasks, like a primary school student doing his homework.'

——Darlene

In the cases of Werner, Darlene and Jack, they discussed feelings of being ' bored', 'not please[d]' and finding the game merely 'tolerable' with the daily tasks. These feelings can also be seen as their negative experiences of the same rhythm of play repeating over and over again. The interviewees were aware of what behaviours were included in each day of their gaming experience and showed clear dissatisfaction with the linear or cyclical model that was repeated many times.

Another interviewee, Berg, provided a more precise description of the negative experiences of being bored and tired. He described that he gets burned out on the game when he cannot make his in-game character stronger in a short time.

'We know those games; we know that, in the beginning, when I log into the game and complete the daily goal, I will be boosted. For example, my character level is not high enough when my main story missions have not been completed. I do the daily goal; I get the experience points from the daily tasks. Most games are like that: the boost is noticeable in the early

stages of the game, but when it comes to moving to the long term, like many online games, generally online games, there will be long periods where players come online and just do things. It does not feel like anything has changed from yesterday. I will get tired of it'.

— Berg

In his response, Berg described the everyday routine in-game as a method that must be endured to reach a higher level of in-game goals. These in-game goals may include collecting a new game character, acquiring a new item, or enhancing a character they already have. One example is Darlene, who described that the moment of getting a new character is the best part of the game. Another is Lance, who prefers to repeat the same game level over and over again to obtain the item he wants, as he described:

'I will spend some in-game resources to strengthen my character, and I can do that daily as long as I have time until I get the item'.

—— Lance

The descriptions given by these interviewees show that the daily routine in mobile games can be expanded and recycled simply by continually adding new characters or objects to the game. All the work involved in the daily routine could be valued, measured and transformed into bonus points. The game system could then reward players with different types of materials based on the accumulated bonus points during each period. After that, players use these materials to acquire new characters or reinforce their roles.

Albert, an interviewee who has worked in the mobile game industry, noted that a player's devoted time or money is necessary to perform well in a freeto-play mobile game. Within the designed rules of the game, these resources can be converted into each other at a specified rate. Many designs in mobile games compel forcing players to spend one or both of these resources. As Albert stated:

'Stamina systems, building cool downs time<sup>26</sup>, the random card packs.....those designs aim to require time or money'

——Albert

As previously demonstrated, the stamina system embodies a linear rhythm of production, while the building cooldown time can be considered a cyclical rhythm that encourages players to log in at a specific intervals. Another interviewee, Brady, who works as a mobile game designer, stated, 'The only thing we can do is create repetitive work to keep players in the game', the demands or requirements of mobile games affect players' off-game rhythm behaviour.

This process can be explained in three steps: First, the requirements that mobile games place on their players, as described by Albert—either time or money—are transmitted and manifested as in-game materials. For instance, interviewee Werner mentioned that the game's requirements are represented as gold coins, which can be obtained either by winning campaigns or through online payments. In the case of the interviewee Zoe, the requirements are

<sup>&</sup>lt;sup>26</sup>Cooldown is commonly abbreviated as CD. In the context of sentence, it refer to the duration required for an in-game building to produces a resource(such as gold or other materials) again producing it once.

represented as the diamonds, acquired by completing daily goals or via online transactions. In this step, the player's time or money is quantified by the ingame currency.

Second, the rules for acquiring in-game currency or rare materials dictate the player's in-game rhythm. All mechanisms introduced, including stamina points, daily log-in systems, daily tasks and the Battle Pass system, form part of a comprehensive game design system. This system values the in-game currency or materials alongside player behaviour. It establishes rules for the exchange between the player's actions and the in-game materials, subsequently encouraging repetitive actions that manifest as either linear or cyclical rhythms within the game.

Third, the final step involves transforming in-game behaviours into out-ofgame behaviours. This transformation can be illustrated by the experience of the interviewee Elisa, who described repeating daily game activities day after day:

'It has probably changed what I do every night before I go to bed and what I do in the morning. The last thing I do before I go to bed and the first thing I do when I get up in the morning is to open the game. In the past, I might chat with my friends, watch the news on my Weibo account, check the news, and then look at other apps or platforms. But now, after I have said good night to everyone. I open the mobile game and do the task, and It is really like a patterned operation'

——Elisa.

In Elisa's case, the statement 'The last thing I do before I go to bed and the first thing I do when I get up in the morning<sup>27</sup>' implies a strategy to maximise the benefits provided by in-game mechanics, which might not be obvious to non-players. Placing playtime at bedtime and in the morning is not merely a casual choice but a strategic move to avoid wasting stamina points. As mentioned in the previous section, stamina points naturally increase to a maximum limit; therefore, consuming the points before bedtime and after waking up helps to reduce the loss of potential excess.

Elisa's strategy demonstrates how game-based rhythms influence out-ofgame living rhythms. As the interviewee James mentioned, 'making it natural to remember to complete it'. This description highlights that the 'natural' process of mobile gaming rhythm change people's daily routines. Elisa also noted, 'It has probably changed what I do every night'. Players adjust their daily schedules to accommodate gaming, often reorganising their daily routines to align with the game's demands for regular engagement. Game mechanisms subtly encourage players to tap the screen at prescribed times not through compulsion, but subtle induction by the in-game rewards and incentives

In Apperley's (2011) analysis of digital game rhythms, he argued that the effect of game rhythms on human players can be described as 'training', which could be argued as: "They train people in the rhythms (timings, movements, patterns. positions) of digital games, in the gestures, and postures appropriate to play" (p. 48). He also described the aspect of training

<sup>&</sup>lt;sup>27</sup> In the previous interview response, the mobile game which was Elisa playing currently includes a stamina point system.

as "sitting in front of the screen for hours on end" (p. 48). However, based on the data from the interviews conducted for this study, it could be argued that the training imparted by mobile games extends beyond mere contact between the user and the screen, pointing instead to the reorganisation and reshaping of the order of players' daily lives. A game trains people not only to master the gestures, postures and body control necessary for playing the game but also enables them to adjust their daily schedules to facilitate physical entry into the game at the 'right time'. While analysis of computer gaming suggests that the trained playtime refers to the period a player is connected to the computer screen, in mobile gaming, the trained time includes all the player's daily time.

For mobile game players, the act of playing can be regarded as interacting with a smartphone and internalising the logic of mobile games into their consciousness. This notion aligns with Ted Friedman's (1999) argument of consciousness computerisation based on traditional PC games named *SimCity*. This game is based on a series of hypothetical logic about city building. It uses computer logic to simulate urban development, expecting players to grasp this approach. Friedman (1999) described the state of full immersion in the computer as creating a 'symbiotic circuit' between humans and computers. This concept refers to the blurring of lines between the human senses and the computer, where players feel that they become extensions of the computer. As Friedman articulated (1999) :

When a player 'zones' a land area, she or he is less identified with a role than with a process. Furthermore, the reason that the decision, and the continuous series of decisions the gamer makes, can be made so quickly and intuitively is that you have internalized the logic of the program so that

you are always able to anticipate the results of your actions. 'Losing yourself' in a computer game means, in a sense, identifying with the simulation itself (Friedman, 1999, p. 6).

Friedman (1998) also provided a detailed explanation of how human consciousness can internalise computer logic in his other thesis:

To win, you cannot just do whatever you want. You have to figure out what will work within the game's rules. You must learn to predict the consequences of each move and anticipate the computer's response. Eventually, your decisions become intuitive, as smooth and rapid-fire as the computer's machinations...you have to know how to think like the computer because the computer provides artificial Intelligence that determines moves of your rival civilization (Friedman, 1998, p. 136).

In this quote, 'your rival civilization' refers to the character operated by the computer, and the player must learn to think like the computer to understand the logic of the computer-operated characters' actions. As Apperley (2011) stated, this should be understood as a process of reconfiguration or adaptation of human consciousness. Furthermore, it is also an adaptation of the player to the computer-based rhythm.

The repetitive feature of rhythms aids the internalisation of human consciousness into gaming. In today's mobile gaming era, the complexity of the mobile game player actions is much simpler than in the time of Friedman's works, such as SimCity and Civilization. Interviewees from this research reported that most of their in-game behaviours are repetitive. They feel that their time spent in the game involves repeating past actions. Elisa, one of the

interviewees, highlighted the extent of repetition encountered during gameplay:

'How much of the content is duplicated? Probably 90 percent'.

——Elisa.

Due to the repetition of in-game content, the interviewees described finding it less difficult to understand the logic of the computer's performance while playing a game and to successfully predict the outcomes of their in-game behaviour. This familiarity allows their consciousness to more quickly adapt to the computer's logic. Furthermore, after understanding this logic, some interviewees chose to introduce customised computer programs to take over their role in gaming. For example, some interviewees had downloaded preprogrammed software that simulates players' actions within the games. These programs act as a substitute for the players' finger, as exemplified by the interviewee Karin, who uses a mobile phone Clicker<sup>28</sup> to replace himself:

'I got off a mobile phone Clicker, and if I have something to do, I will use it.'

——Karin

This Clicker program is not specifically tailored to any particular mobile game; it requires the users to set the tap positions, number of taps and duration of the clicks. Hence, when users use this tool as a substitute for themselves, it indicates their understanding and prediction of the behavioural model required by the mobile game's computer logic. The functions of the program, as described by Karin, involve tapping the screen continuously and automatically,

<sup>&</sup>lt;sup>28</sup> Or named quick macro. A kind of mobile game software which automatically repeats user-entered commands.

just as a human finger does. It works like a contemporary parable of water dripping through stone—Lefebvre's classic description of linear rhythm (2013). Another interviewee, Marko, explained why players use automatic programs to replace self-play. He believed that limited linear repetitive clicking leads players to prefer Clickers. During the interview, Marko quantitatively described the repetitive actions he needs to perform in the game:

'Let us start with two hundred cycles, twice a year, each for 14 days, an average of six hours per day.'

——Marko

Marko's experience relates to a yearly event in a mobile game. In this event, players typically use stamina items that they have accumulated beforehand to earn more in-game rewards. According to Marko, participating in these events can require up to 70 or 80 hours of completely repetitive steps, which can mechanically be replaced.

Similarly, another interviewee, Juliana, plays the same game as Marko but does not use Clicker. During the event, she spends eight hours per day playing for seven consecutive days each time this in-game event occurs. She stated:

'I bought from the event shop until it was sold out, ten times a day, and I would set a target for myself about how many times I had to buy it all, and the target had to be done before the game event ended'.

—— Juliana

When Juliana entered the scene, she did not feel frustrated. At that moment, she believed that constantly tapping the screen was exactly what she was supposed to do. In terms of outcomes, Juliana's human efforts are just as effective as Marko's Clicker; they both manage to purchase all the event items from the event shop as many times as they wish.

Juliana's case shows the internalisation of a linear rhythm. When she followed the guidance or demands of the mobile game, she engaged in repetitive and regular clicking behaviour, thus developing a game-based rhythm. From the interviews with Marko and Juliana, it is clear that in adopting in-game linear or cyclical rhythms, a human finger is no different from the Clicker. The existence of Clicker was created as a substitute for the human finger, can be seen as a player's revolt against a linear rhythm, but also as a metaphor for an imposed structure. Moreover, this metaphor also signifies the internal consistency between human consciousness and computer logic in daily mobile game playing. The 'symbiotic circuit' argued by Friedman (1999) may indeed be proved by players' use of Clicker to replace their own actions.

After the initial stage, players immersed in the scenario soon realise that they must perform continuous repetitive actions to meet the game's expectations, as explained by the interviewees lke and Link:

'There is a button in the bottom right corner of the screen, and you have to keep tapping it, keep tapping and keep tapping, so many times, that your fingers get tired.'

—— Ike

'It is just clicking, click, and click. There is little difference between the daily tasks, not much change, it is the same.'

——Link

Like in the case of interviewee Marko, for many interviewees, the interaction between the player and the mobile game is more akin to mechanical repetition than to a creative intellectual activity. Brock and Fraser (2018) have shown how game players use interactions with the screen to reflect and improve their skills, and how improved skills can lead to the emergence of creativity. As they stated, "the development of skilled hands also facilitates creative expression" (Brock and Fraser, 2018, p. 1227). However, the simple 'click' of action in mobile games can hardly be regarded as an intellectual or creative expression due to the lack of thinking activity, as the illustrated by the description of the interviewee Erin:

'I already had some almost mechanical movements when I was playing it. I would keep clicking repeatedly and not think about it in the process'.

—— Erin

Brock and Fraser (2018) highlighted 'reflection' as an important element of skill development. However, mobile gamers such as Erin do 'not think about it in the process'. For mobile game players, the primary aim of playing is not to improve skills but to ensure they do not miss the game's rewards. As Marko described, the in-game event was held 'twice a year', and each event lasted '14 days'. To get enough in-game rewards, players must perform a certain number of repetitive actions. Brock and Fraser (2018) showed the variability

and engagement demonstrated by players through mouse clicks while participating in game activities in Dota 2. Their demonstration illustrates how these clicks are a means to express player strategy and skills. In contrast. the interviewees in this thesis displayed strong consistency and repetition in their playing behaviour, where the click is not a tool for expressing thoughts but a repetitive action that must be performed sufficiently to meet the requirements of mobile game design.

The time limit of the mobile game event requires players to complete a specific number of actions within the event duration, and these actions are equivalent to a corresponding number of clicks. According to the descriptions provided by Erin and Ike, players perform clicks mechanically without regard for intrinsic meaning or reflection. Rather, the click is not an intellectually meaningful action but simply an action. As Ike described, 'there is a button in the bottom right corner of the screen, and you have to keep tapping it, keep tapping and keep tapping'. When playing a game over long periods, players often do not perceive a difference between today's and yesterday's content, with repetitive tapping actions dominating a significant portion of their gameplay time. This mechanical operation model became habitual for the interviewees, allowing for extended periods of play without requiring extra attention. As the interviewee Andrew stated:

'The mechanical action [in playing the game] does not require looking at the screen. I do not look at the screen when I am running on my routine'.— Andrew

When Andrew plays the game without looking at the screen, his fingers maintain a dominant, repetitive rhythm. The natural attitude of the interviewees towards these mechanised behavioural patterns suggests the possibility of humans internalising rhythms produced by machinery into their biological instincts. This could be considered a form of human selfprogramming, referring to the process by which people translate the logic of computers into their biological bodies. Alternatively, it could also be regarded as the performance of a player who has completed the online dressage, which is discussed in a further subsection. At the beginning of playing, players follow the guides of the game system to build their in-game behaviour habits. Subsequently, they develop a series of mechanical out-of-game body postures corresponding to the game system, as lke described, 'keep tapping it, keep tapping and keep tapping, so many times'. In these cases, the logic produced by the computer has been transmitted to the player's consciousness and out-of-game body. As Ted Friedman (1999) stated, the 'symbiotic circuit' between the game and the player makes them share a series of logic. In the cases of the interviewees in this study, it also signifies the shared rhythm.

Lastly, players' experiences of game-based linear and cyclical rhythms are not equal but are strongly impacted by their ability to make online payments. The interviewee Scott, who has ten years of work experience in the mobile industry, explained shows how the mobile games train the players into these rhythms through 'timings, movements, patterns. positions' (Apperley, 2011, p. 48):

'For Chinese game designers, the numbers are the most important... planning is the core: our gameplay is not very important. We have a very

clear plan of the numerical growth for each player, about what kind of game fun will be given at which time, precise to the minutes, and the fun comes from which game mechanism. This is the absolute consensus among all mobile gaming companies. In order to guide the new game players, the game process is clearly planned. Players do not necessarily follow the planning strictly, but I have to give this planning about the for player actions'.

----Scott

Scott's response highlights how game designers meticulously plan the experience of mobile game players. This planning of player behaviour is normally extremely precise and detailed, it spans from the first minute a player enters the game to the last minute they quit.

As demonstrated by the interviewee Lance, motivations such as 'spending some in-game resources' and 'strengthening my character' motivate players to continue gaming. This numerical growth is a significant motivator for playing; the point at which players decide to give up a game usually coincides with the point at which numerical growth within the game becomes insignificant. Mobile game designers are acutely aware of this balance and plan accordingly. They understand exactly how much time a player needs to spend in the game and what rewards will be achieved. This planning is both clear and detailed.

From the view of game designers, detailed distinctions are made about player groups. The first criterion for classification is the length of time the player has played, and the second is how much the player has spent on the game, as interviewee Scott stated:

'In terms of experience in game design, we divide games into phases. Through the system design of the game, the numerical growth is connected to behavioural planning'.

——Scott

All of these designs aim to keep players engaged in the rhythm of mobile gaming and encourage them to pay for the game experience. Players can be distinguished by their in-game payments. Based on historical consumption records, the mobile game designers categorise players into different groups, typically referred to as 'small R', 'middle R' and 'big R<sup>29'</sup>. They plan the corresponding game experiences for each group. As previously mentioned in this section, players may acquire items to bypass repetitive tasks through online payments: the more money a player spends on the game, the more privileges they can acquire to skip the standard in-game linear and cyclical rhythms.

The interviewee Brady, who works for a game company, explained the profitability mechanism of the freemium model with the following statement:

'The Gacha system in the freemium game is essentially price discrimination, in traditional video games, the players can, and only need to pay one time, no matter if they are rich or poor, however, in mobile games, if consumers want, they can make unlimited purchases'.

——Brady

<sup>&</sup>lt;sup>29</sup>In the context of Chinese cyber culture, the capital 'R' stands for the 'Renminbi' which is the official currency of China

To explain this opinion, he added: 'The player is not playing the game, but playing the existence beyond the game'.

According to Brady, there are three main reasons why people play and pay for mobile freemium games: First, some players enjoy the feeling of defeating others. Second, people want to discuss or show off their in-game items with friends. Third, some people want to offer or pay something to an object as a mental enjoyment. For these reasons, people continue to play mobile games although they recognise the repetitive nature of the gameplay.

In the cyclical and linear rhythm models of mobile games, each player may have a per-determined role. Even within the same mobile game, players who spend different amounts of money are assigned different rhythm models. As the mobile game designer Scott explained:

'From the view of a designer, the system is the core, and it is based on the numerical value to express. The number is behind the glory: Who is the small R<sup>30</sup>? who is the middle R? Who is the big R? We have to have strong systems to inform the all players who belong to which groups. We have designed this social hierarchy: we have to design the role of each person under the social hierarchy. For example, in a game guild, without the president, there will still be 6-8 characters. For these characters, we need to design their titles, functions and their interactions with other players. All of the in-game mechanics were designed to build a system...how do I generate in-game interaction and social connection based on the social hierarchies? The whole interaction system is the rules of the game, which

<sup>&</sup>lt;sup>30</sup>The term 'RMB player' refer to a person who spends money on a game

were designed in advance, and the game mechanics will strengthen these social hierarchies...From a game design point of view, we never let bottom players flow to the top. The rules of the game world are much colder, the welfare is a bubble, colder than real life'.

——Scott

Scott's description outlines how the environment surrounding different 'player groups' can be differentiated. This variation arises because the game assigns multiple roles to groups of players, each developing distinct rhythmic patterns. For instance, players who purchase specific freemium options may have higher maximum stamina points, allowing them the privilege of exiting the game for extended periods without penalty. These varied and complex freemium options enable players to alter their gaming environment through online payments, thereby diversifying the game's mechanics and affecting both linear and cyclical rhythms.

The preceding three sections introduce common mobile game mechanisms that establish the linear and cyclical rhythms of player in-game behaviour. The following section discusses the features of automation in gaming and the function of in-game avatars.

## 4.3 Polyrhythmia, Multitasking and The Automation of the Mobile Game

As a key concept in Lefebvre's rhythmanalysis theory, polyrhythmia refers to a configurations of multiple rhythms. Lefebvre (2013) described it as follows: "it suffices to consult one's body; thus the everyday reveals itself to be a polyrhythmia from the first listening" (p. 16). The most common form of

polyrhythmia is the multiple rhythms of the organs in the body, where different organs each produce their own rhythm yet simultaneously support the same human body.

In the realm of mobile gaming, gaming time usually overlaps with other daily activities. The interviewees in this research typically display a tendency towards multitasking, engaging in gaming during regular daily events such as dining, working or taking breaks. This blending of rhythms from mobile gaming and non-gaming life can be viewed as polyrhythmia in the context of rhythmanalysis.

To present the concept of polyrhythmia in mobile gaming, this subsection briefly introduces the phenomenon of automation in mobile games. This discussion helps develop an understanding of how players automate the ingame cyclical and linear rhythms and integrate these into their daily lives, and how these automated in-game rhythms contribute to the construction of daily polyrhythmia. It is important to clarify that not all mobile games feature automated mechanisms. Nevertheless, automation is increasingly common, especially in the role-playing genre. Automatic mechanics are predominantly used in player-versus-environment (PvE) content but are less common in player-versus-player (PvP) interactions.

Automation in mobile games involves a two-step process: the player gives instructions, and then the game program automatically carries out the corresponding actions. This usually takes the form of auto-combat (referred to by Chinese players as '扫荡', 'saodang') or auto-collection (commonly termed 'idle').

In some mobile games, auto-combat functions as the game system automatically replicating the commands a player used in their last challenge of the same dungeon. Consequently, players often refer to this feature as 'watching a record'. In these instances, the automatic system takes the same amount of time as the player did originally, and the human player does not need to intervene. During this process, the computer program plays an animation to give the impression that the in-game character is performing these actions.

The essence of the auto-mechanism is akin to a timer or clock, where the player presses the start button, and the game issues a reward after a specified period. The duration of this period may depend on factors such as the strength of the player's character and the time the player took to complete the last pass. Additionally, some mobile games offer options like a double-speed accelerator or the ability to skip the auto phase: players can 'unlock' these features by paying real currency.

With the automation system, players often experience gameplay intermittently rather than continuously. Ruffino's (2021) work on idle games demonstrates fully automated ways of managing game space. However, according to interviewees' responses in this study, the more common pattern of player-game interaction tends to be semi-automated rather than fully automated. Players are required to return to the game at regular intervals and maintain partial attention on the gameplay, but they do not need to manually control specific actions of the game characters.
The automated commands executed by the game system are not support preprogrammable, and the duration of automated commands is usually relatively short, often just a few seconds or minutes. Consequently, players may need to regularly return to the game, input some commands in the short term, and then leave again. This pattern of discontinuous and intermittent command input could be described as the intermittency of the playing experience for the mobile games with automation mechanisms.

Furthermore, the intermittent nature of gameplay encourages players to adopt multitasking strategies. Multitasking involves people 'interleave' two or more tasks by switching back and forth between the tasks prior to finishing any one task' (Carrier *et al.*, 2015). The original metaphor for a multitasking machine is a bread toaster: while waiting for the bread to toast, people can perform other tasks in the kitchen, preparing both bread and other dishes simultaneously. This suggests that when intermediary processes can operate independently, people need only attend to the task at the beginning and the end, similar to how a chef presses the start and stop buttons on a toaster. This strategy allows them to manage multiple tasks simultaneously in the same space (Salvucci & Taatgen, 2008).

In the context of mobile gaming, the automation function allows players to engage in other activities while gaming. With assistance from the automation system, the player clicks the start button and then waits for the program to automatically work and generate in-game resources. Similar to the breadtoaster metaphor, players typically wait for the harvest provided by the automatic system and simultaneously move on to complete other daily tasks outside the game. For most mobile games with automation mechanisms, the

automatic running process, such as combat or collection, cannot be influenced by the players; in other words, it is a period during which the player has nothing to do but wait. This feature is explained by the interviewee Berg in the following statement:

'I am confused as to why I can enter auto-battle and not have to do anything at all; it is a complete waste of my time that they [the game designers] have set the process so long and that I cannot skip it. I do not know what they are designed for, maybe to increase the player's playtime... Do I have to keep an eye on the auto-battle every day? I do not even have to watch it; it does not get better just because I do. It is not like an idle game [a type of game], which, if I am watching, usually has a click to quickly get some kind of resource. Nevertheless, auto-battle does not have that feature, and I cannot change anything by watching it. The logic of this kind of game is that I just have to wait for that result, and even if I quit in the middle of battle, the result is all over, and the data has been uploaded'.

—Berg

Berg's description highlights the player's desire for engagement. However, during the process of automatic running, the player's presence is excluded from the game; they cannot interfere with the in-game character's actions or issue commands to facilitate or assist their character. The only interaction available is 'watching'. Thus, the automatic mechanisms should be regarded as encouraging players to temporarily divert their attention from the game, although it also encourages them to return after a few seconds or minutes. As Chess (2018)'s argument of 'interstitial time' on the social network game,

players' real world and in-game experience constitute the 'push and pull ', the game pulls the players back to the game world but is not allowed to stay continually.

The intermittency of gaming time leads, guides or induces players to allocate their attention back and forth between in-game and out-of-game activities due to the regular operation of automatic mechanisms. This dynamic fosters or provides advantages for player multitasking. The case of interviewee Berg also details how players tend to multitask and demonstrates how players switch their attention between gaming and non-gaming activities.

'I usually do things like watch videos [at the same time as playing], something that I can stop at any time and do not have to spend too much attention. Generally, I watch videos the most while playing the game, like the [in-game] auto-battles, where you cannot stop at all during the and do not have to operate. The time [I] spent watching videos will probably depends on [my] playing game. I watch videos I would not watch a certain kind [slight pause here] and probably also choose more random videos that I can pause—so I do not have a specific time plan. I will watch the video while my phone is over there, and [when] it [the mobile game] tells me to click here. I pause the video, click somewhere, and then return to continue watching.'

——Berg

This excerpt demonstrates the interplay between the rhythms of mobile gaming and the players' daily lives, a topic to be explored in Chapter 5. Furthermore, it shows the polyrhythmia nature of a player's daily life. The

interviewee, seated behind the video player, should be considered as a whole subject experiencing polyrhythms, including the mobile game, online videos, physiological body (such as heart rate and blood pressure), and the external environment in his/her room,etc. Another interviewee, Dave, exemplifies the multifaceted state of polyrhythmia when he returns to his hometown for the Chinese New Year. During family gatherings, while Dave's parents watch TV operas, Dave plays mobile games. In these moments, the rhythms of TV operas, mobile gaming, chatting and other daily activities merge. Dave thus navigates and manages his attention amidst this polyrhythmic environment.

This polyrhythmia in mobile gaming naturally extends beyond the limits of time and space. Players might game with friends in other cities or leave their characters to assist others in overcoming challenges when they are not online. It involves players merging slices of mobile gaming rhythm with other aspects of their daily lives to construct their own polyrhythms. The semi-automatic mechanisms mean that the intrinsic rhythms of mobile games can be spliced and recombined with many events in the out-of-game daily life. Furthermore, with the advantages of mobility, semi-automated mobile games can integrate with daily rhythms occurring in public spaces such as universities, companies and subways, thus going beyond the limits of social space.

The way mobile game semi-automation works is usually quite simple, resembling the action of pressing a knob with just a one-time click. However, the frequency—or rather, the intermittent frequency—of gaming is more significant than the action itself. Engaging in intermittent gaming requires the player to devote attention at regular intervals over a period of time. As the interviewee Andrew described:

'Andrew: It [the daily task] does not require much action, [I will] go in to take the phone and click every one or two minutes.

The Researcher: So you left your phone on the table at the other time?

Andrew: Yes'.

His interview case suggests the formation of an out-of-game bodily rhythm driven by the automatic mechanisms of mobile games. In the scenario he described, he simultaneously performs three different actions: 'eating food, watching videos, and playing Arknights (the name of the mobile game) at the same time' (Andrew). This indicates that his body is interweaving three parallel daily activities at the same time. His description of 'go in to take the phone and click every one or two minutes' implies a regular return to the digital space and context of the mobile game. In this cycle, he would repeat the actions of 'putting down the phone', 'watching a video or eating', 'thinking for the process of gaming', 'picking up the mobile phone and interacting with it' and finally 'putting down the mobile phone'. This cycle is premised on the need for mobile games to require players to pay attention and interact at intervals.

The intervals, as other Arknights players describe to their gaming routine, are referred to as 'playing the video record'. This means that when players use their stamina points on a challenge they have previously won, the game automatically carries out the same actions as when the player last passed the challenge, akin to playing a pre-recorded film. The player taps the screen at the beginning and end of a challenge, and in between, as interviewee Berg mentioned, there is almost nothing for the player to do while the 'video' is

playing. The presence of the 'video record' suggests an interval between the player's attention entering and returning to the game. Although players may not strictly follow the video's playback time, the gap is not very wide. Interviewee Andrew described this interval as 'every one or two minutes', which closely matches the duration of a daily typical video record challenge in Arknights.

When Andrew does not need to actively engage with the mobile game, he naturally shifts his attention to other daily activities. In the interview, he mentioned preferring to play the mobile game during his mealtimes—either breakfast or dinner—so that both activities could occur simultaneously. The 'video record' gameplay does not interfere with his other activities during these times. While he described tapping on the screen every few minutes, he also emphasised that the game 'does not require much action'. This suggests that the impact of the mobile game on him relates more closely to his actions of entering and exiting the game rather than actively playing it.

In this case, the player develops a regular pattern of 'switching in' and 'switching out' of the game contexts based on the automatic mechanisms of the mobile game, demonstrating a spontaneous out-of-game body rhythm.

As discussed in the literature review, Keogh (2018) proposed the concept of 'embodied literacy', which addresses the interaction between gaming input devices and the human body, such as the use of hands. While traditional gamers are trained to use devices to control their characters' movements, shooting, and jumping, mobile gamers, as exemplified by Andrew, develop a different set of body patterns by continuously switching in and out of gaming.

The natural reaction of 'tak[ing] the phone and click[ing] every 1-2 minutes' reflects the negotiation between the player's body and the mechanics of mobile games, as well as the concretisation of human-game polyrhythms. This 'returning action plan' is not a natural behavioural tendency but is 'instilled through the ritualized repetition of behaviours and the sensitisation of the body' (Keogh, 2018, p. 107). Compared to 'embodied literacy' in traditional gaming, mobile game embodied literacy emphasises how players integrate mobile gaming into their daily, non-gaming lives. In Andrew's case, for instance, he regularly switches between eating, watching videos, and playing, developing a cohesive routine that blends these activities into an integrated ensemble.

For further discussion, this thesis explores the interaction between body rhythms and everyday rhythms in the next two chapters. Furthermore, the trained body rhythm, such as clicking the smartphone screen every one or two minutes, integrates into the polyrhythmic life of the interviewee. The intersection and blending of multiple rhythms form the polyrhythmic environment in which the player lives. One rhythm may influence another, and some may facilitate the formation of another part, thus shaping the rhythmic ensemble of the player's daily life.

In summary, this subsection highlights the semi-automated mechanisms in mobile game design and shows how they influence the polyrhythms in a player's daily life. Due to the ruled running time of semi-automated mechanisms, during which the player has nothing to do, there may be

encouragement for multitasking<sup>31</sup>. As demonstrated by interviewee Andrew's case, he enters the game 'every one or two minutes', aligning with the intervals of a linear or cyclical rhythm pre-designed by the system. On the one hand, the semi-automated mechanisms encourage the player to divert their attention away from the game; on the other hand, they encourage returning at regular intervals. This regular return can be understood as a seductive method to guide the player to follow the game's pre-determined timing patterns, leading the player to develop an out-of-game body rhythm. Furthermore, this produced body rhythm, along with other daily out-of-game rhythms, constructs the polyrhythmic experience of the mobile game player in daily life.

This subsection also demonstrates the characteristics of mobile gaming and its demands for player attention, as well as the process of players switching back and forth between the physical world and the game world (this will be discussed in more detail in Chapter Five). It will help the reader understand the combination and conflict of rhythms between the mobile game and daily life in future analysis.

## 4.4 Create 'Your' Character—The Production of In-game Digital Avatar Rhythm

Almost all of the interviewees frequently used 'I, me, myself' rather than 'the character I created in the game' to describe their in-game behaviour or ingame relationships with non-player characters. They made extensive use of

<sup>&</sup>lt;sup>31</sup>Under the research conditions, it is difficult to determine whether the automated design of mobile games leads to players' tendency to multitask of if the digital multitasking habits of players drive game design. In any case, there is likely a two-way facilitation relationship between the two. The more the content of the mobile game supports automation, the more frequently the game player will likely switch between in-game and non-game rhythms.

the rhetoric of 'I kill monsters', 'I participate in the game' and 'I am with other people', while omitting the presence of the game character as a mediator. For example, the interviewee Jessica described her gaming experience as playing a restaurant owner:

'I have to click, to collect the seafood and flowers that I keep in the restaurant [every day], then click on the advertisements, then recruit [in-game characters], then see if there is anything to upgrade [in the restaurant], and reset something [the house layout]'.

——Jessica

In this description, Jessica's in-game actions stemmed from a merger of herself and the protagonist, who was designed as the owner of a restaurant. The restaurant owner avatar is an extension of the player's will, while the daily actions of this character have already been preset by the mobile game. At the same time, the interviewee also tends to perceive other in-game avatars controlled by other people as living beings; the rhetoric used to describe interactions between in-game characters closely mirrors discourse used in daily life. The following paragraph uses interviewee Zachary's description as a reference:

'If you have a friend, you get someone from your friend list, and that friend can hold hands with you, hold hands, and take him along for the ride. It is a feeling that is particularly romantic with the background of the game's graphics, two people holding hands and running together, shoulder to shoulder'.

——Zachary

The phrases 'holding hands and running together, shoulder to shoulder', which Zachary described, do not occur between human physical flesh. Instead, these are two in-game avatars, representing two players, performing the 'hold hands' command within the logic of the game system and reflecting this graphic on the phone screen. As their avatar hands connect, the palms of the two players' physical bodies are in contact with the phone screen—the hand-holding occurs within a media environment. This connection to the digital world is described using the same words as one would use in the physical world. The description of 'holding hands' shows how the interviewee Zachary links his perception with his in-game avatar. This connection has the potential to cause the player to confuse the actions predetermined by the game system for the game character with their own feelings about themselves, considering the game avatar's failures or successes in the digital world as their own. In the case of interviewee Valentina, her in-game behaviour is focused on the achievements and harvests of her in-game avatar as a prefectural magistrate. Her description is as follows:

'I have it set up [an alarm] every 8 hours, and then I will be at my strictest for a while. I will go and click at 12:00 pm to 'start collecting taxes' [an ingame action to collect resource], then get up at 8:00 am to click again, 'start collecting taxes', and then again at 4:00 pm to start collecting taxes. So, that is precisely three 8-hour periods a day. I will follow the time I am awake and can manipulate it. However, if this is disrupted one day, I have a day which I cannot wake up at 4:00 pm to start collecting taxes, it [referring to the in-game time counter] will be delayed into the evening. I will not be able

to click again at 12:00 pm for the next round. I will have one less round, and I will lose money. I would be mad about that'.

—— Valentina

In this case, Valentina's avatar is a prefectural magistrate<sup>32</sup> responsible for managing the city's taxes and buildings, requiring her to collect the taxes at regular intervals (in this case, every 8 hours). All these city management tasks are simulated by the in-game actions of clicking or dragging and dropping. The mobile game constructs a framework for the player's behaviour within a pre-designed range and encourages them to adhere to the in-game rhythm. When Valentina logs into the game, the system presents her with all the in-game tasks that a prefectural magistrate needs (or could) do, as reflected in her task list interface.

The rules of this simulation, namely the process between the action's purpose and the rewarding result received by the player, can be arbitrarily designed. For example, Valentina, who 'collects taxes by clicking on buildings every eight hours', is not bound by any natural reference for this eight-hour interval. A period of six hours, seven hours, or even one minute would not affect the meaning or the result of the act. Similarly, the act of clicking on the building or dragging it once, twice or a hundred times is merely a condition or a number set within the program. Thus, the significance of these rules is in shaping the player's in-game behaviour model; in other words, it constructs the in-game rhythm for the player. In this context, the avatar acts as an invisible guide shaping the player's in-game behaviour. Building on this discussion, this thesis introduces the concept of 'dressage', as proposed by Lefebvre.

<sup>&</sup>lt;sup>32</sup> An ancient Chinese city administrator, similar to a mayor.

In the framework of rhythmanalysis theory, Dawn Lyon (2019) clarifies that dressage is the "process of bodily entrainment and repetition through which rhythm is learnt and becomes evident in the body over time" (p. 25). Lefebvre (2013) describes dressage as a 'natural' rhythmic pattern. In this context, 'natural' means "it conforms perfectly and without apparent effort to accepted models, to the habits valorized by a tradition" (Lefebvre, 2013. p. 38). Dressage refers to the process through which individuals undergo socialisation, enabling them to adapt to the rhythms of a country, group or society by bending and changing themselves. Lefebvre referred to those who are knowledgeable the rhythm as 'breeders and trainers' (2013, p. 40), who assist people in familiarising and understanding established rhythms through long-term repetition. As he stated: "one breaks in another living human being by making them repeat a certain act, a particular gesture or movement" (p. 39). Breeders could "bring about unity by combining the linear and the cyclical" (Lefebvre, 2013. p. 39) to create a series of repetitive instructions and a general organisation of time.

In the context of mobile games, the rhythm of the avatar is predetermined. It consists of a series of pre-produced criteria and predetermined actions, described as "models, the habits valorized by a tradition" (Lefebvre, 2013. p. 38). However, unlike the traditions maintained by humankind, the patterns of the avatar are precisely quantified and measured by computer programs. From the number of monsters killed to the number of taxes collected, the rhythmic patterns of the avatar are measured with exact precision.

One could regard the avatar as a pre-designed set of standard rhythms, with the person playing the role governed by the digital system to complete their

socialisation process in the online game world. The 'breeders and trainers' (2013, p. 40) mentioned by Lefebvre are not specific humans but rather designed industrial digital programs. This process can be related to Lefebvre's (2013) explanation of animal dressage as follows: "Under the imperious direction of the breeder or the trainer, they (the animals) produce their bodies, which are entered into social, which is to say human, practice. The bodies of broken-in animals have a use-value. Their bodies modify themselves, are altered" (2013, p. 40). The in-game avatar acts as an invisible trainer, teaching the game player the rules of the game and reflecting rewards and punishments based on the avatar's performance. When the player exhibits the 'right' behaviour, the avatar will perform stronger; when the player does something 'wrong', the avatar will show a loss, frustration or even death. The avatar serves as an indicator of whether a player is behaving correctly or not; this instructional process could be considered a training period.

Through the pre-designed set of standard rhythms in in-game avatars, the ingame rhythm can be reflected in the player's daily life. For example, Valentina's city buildings automatically generate money every 8 hours, which the prefectural magistrate then uses to construct new buildings. In the physical world, Valentina logs into the game every 8 hours to meet the requirements generated and imposed on her avatar by the game system. In the case of Valentina, she shares the social identity of the prefectural magistrate with her avatar in the mobile game environment. The process of shared identity occurs during the gaming experience when the player interacts with the in-game character or environment. Players are often referred to by the roles of their avatars, such as 'Traveler' in Genshin Impact, 'Master' in

Fate Go or 'Prefectural Magistrate' in The Hundred Scenes of Jiangnan. This title of the avatar also functions outside the game context; the game company uses it to interact with players on social media updates, offline activities, interviews, public events and player forums often refer to users by their ingame titles. Hence, the title of avatars usually carries a dual meaning: firstly, it represents the role of the avatar in the story of the game, and secondly, it refers to the player who plays this game. For the player, the playing experience is also the process of developing a connection with the designed title or identity.

At the same time, it is also possible for the player to inherit or transform the avatar's social relationships with non-player characters. Some interviewees in this research tend to use the in-game relationships between the character and avatars, such as supervisors or subordinates, friends, or romantic partners, to refer to the in-game characters. Gee's (2009) work explained the process of ' inhabit', he claimed that players adopt "beliefs, values, goals, feelings, attitudes, and so forth" (p. 258) from the avatar and use these to interpret the character's actions. This leads to an attunement between the in-game avatar and the player, as described by Gee:

"By inhabit I mean that you, the player, act in the game as if the goals of your surrogate are your goals. Virtual characters have virtual minds and virtual bodies. They become the player's surrogate mind and body" (p. 258).

On the first day, a player enters the mobile game, they are presented with the story of the avatar, which may involve searching for lost relatives, reviving a derelict town, etc. The in-game actions such as killing monsters, passing

challenges or constructing buildings are portrayed as parts of the avatar's story. The player's actions within the game drive the development of the avatar. Through the causal connection between the playing actions and the in-game results, the needs of the in-game avatar or character are converted into a list of actions that the player should (or could) undertake. Furthermore, the repeated in-game behaviour becomes the rhythm of the game environment and may further influence the player's daily life. Below is a communication text between the interviewee Herbert and the researcher, illustrating the process of the in-game rhythm belonging to the avatar transforming into the rhythm in the player's everyday life.

The Researcher: 'So would you say that if it there was a guild war at 8 pm, would you be on the game at 8 pm?'

Herbert: 'Yes'.

The Researcher: 'Okay, do you set an alarm clock for it?'

Herbert: 'I will remember it by myself'.

A game player such as Herbert may internalise the in-game cyclic rhythm into their schedule. The Guild Wars at eight pm represent a typical in-game cyclic rhythm. These events provide in-game materials to the improve the in-game character or acquire a new one. To meet the needs of the in-game character, a player like Herbert would remember the time of an in-game event and incorporate it into their personal schedule.

In summary, this section explains the function of the in-game avatar and considers it as the role of an invisible trainer guiding the player's in-game

dressage. In the mobile game environment, the rhythm associated with the ingame avatar is pre-established. When the player's avatar is created, this character is assigned a series of tasks and goals to accomplish. Through the inheritance relationship between the avatar and the player, the pre-designed rhythms of the mobile game's avatar are translated into the player's in-game rhythms and subsequently migrate into the realm of the player's everyday life.

### 4.5 Conclusion

This chapter discusses the linear and cyclic rhythm performances, including polyrhythmia, multitasking and automation in mobile games, and the function of in-game avatars in shaping the game rhythms.

The first subsection of this chapter explains how regular in-game behaviours that players repeat develop into game-based rhythms during the daily play. According to Lefebvre's (2013) theory, a simple repeated act, such as clicking the screen within a short period, should be considered a linear rhythm. This includes actions such as repeatedly challenging the same boss or dungeon. Acts performed at specific intervals should be classified as cyclical rhythms, such as daily goals or the battle pass system. This subsection outlines how ingame system mechanisms create linear and cyclical rhythms by rewarding the player for their daily routines. This reward system commodifies the player's leisure time and attention. As players acquire in-game currency, enhancement materials, character experience points and other assets 'for free' through repetitive actions, the invisible commodities being traded are human time and attention. The repetitive action, which has become a linear rhythm, should be viewed as a mediator or price tag: it does not generate value—for the game

company, the number of times a player kills a boss is merely data recorded on the computer—but it serves as a measure for other in-game items. With the introduction of buttons like 'skip challenges' or 'speed up challenge animations', the function of 'avoiding repetitive clicking' becomes a freemium commodity that the game attempts to sell to the player. In this process, mobile game design aims to occupy the player's leisure time and attention by encouraging them to follow the game-based linear rhythm and then sell this time and attention back to the player.

As mentioned in this subsection, the Monthly Pass or Battle Pass represents the commodification of natural time. The in-game rules demonstrate the value of waiting time—if players do not want to wait, they may pay 6-7 times as much for the same rewards. These reward systems artificially create a gamebased rhythm, driving players to return regularly and complete predetermined actions to gain rewards on the Pass systems.

Through cyclical events, mobile games create a scale to measure time. Players use in-game events to shape their understanding of time. They wake up early, even set alarm at midnight, log in during the evening and engage with the game at specific times to meet in-game requirements. For example, the mobile game update time (normally 5 am) for daily tasks becomes a measure of time. This indicates that mobile games reassign meaning to specific periods through their in-game reward distribute mechanisms. Players become accustomed to performing daily goals within the game, checking the progress of the Pass system weekly and participating in in-game events for each version. This can be interpreted as a human adaptation to the rhythm of mobile gaming and as a cross-fertilisation of game timetables with real-life

schedules. Thus, the artificial scale of time created by mobile games blends into the human understanding of 'everyday time'.

This subsection further discusses how the rhythm of mobile games reshapes and reorganises the player's everyday life through a three-step process. It explains how the linear and cyclical rhythms produced by mobile games become part of the player's daily schedule. By imitating and predicting the mechanical logic of the game system, players may achieve a form of selfprogramming. According to interviewees' descriptions in this section, players repeatedly perform one or several mechanical actions that could easily be replaced by software such as Clicker. These actions performed by the human players are almost unconscious, unthinking and devoid of any mental engagement. Therefore, this thesis argues that when players perform these highly repetitive actions, they may fall into the 'symbiotic circuit' described by Friedman (1999) and internalise the logic of the program into their selfconsciousness.

Subsection 4.3 of this chapter discusses polyrhythmia, multitasking and the automation of mobile gaming. In traditional gaming, in-game actions controlled by the player are simultaneous and synchronous with real-world actions. However, some mobile game designs allow in-game actions to be completed without the player's presence after receiving instructions. This means that mobile gameplay can be pre-programmed and asynchronous, highlighting the possibility of humans applying both physical and digital rhythms simultaneously. This status refers to polyrhythmia within the framework of rhythmanalysis.

Subsection 4.4 primarily examines the relationship between game avatars and human players. By influencing and changing the avatar, the mobile game can indirectly affect the player's psyche. As interviewees showed, the avatar in a mobile game acts as a mediator and the embodiment of pre-set rhythms within the game. It guides the player's daily actions and serves as a vehicle for the player's digital self. Through the complex emotional connection between the player and the in-game avatar, the game system creates the potential for providing online dressage to the player outside of the screen.

Compared with traditional dressage in the physical environment, machinebased dressage is precise and quantifiable. Each in-game event in a mobile game is valued and given feedback by the game system, awarding bonus points or rare items for corresponding behaviour. This should be seen as a constant surveillance and feedback system. The game system indirectly evaluates the player's behaviour as positive or negative by applying immediate effects—rewards or punishments—to the avatar or related nonplayer characters.

For instance, as discussed previously, the interviewee Herbert used to log into the game at specific times to obtain necessary supplies for his avatar. This avatar-based influence can become internalised as part of his thought process, becoming a form of 'nature' described by Lefebvre (2013) in the context of dressage. Next, this thesis analyses eurhythmia and the ensemble of in-game and out-game daily rhythms.

Chapter 5 Combining Rhythms, Eurhythmia and the Synergy of Multiple Rhythms: How Multiple Rhythms Work Together in a Hybrid World

### 5.1 Introduction

This chapter discusses how mobile game rhythms combine with and influence daily life rhythms. Specifically, it explores how these games interact with other everyday activities. In general, mobile games offer a free and leisurely form of entertainment or relaxation. During the interviews, some participants, as the following three interviewees (Andrew, Berg and Karin) used words such as 'simple', 'convenient' and 'stable' to describe their mobile gaming experience. These characteristics likely explain why these interviewees prefer mobile games over console or PC games during their leisure time. For example, the interviewees stated,

'[I] thought [I] might need some simple pleasure to spice up my life, a simple release point, and when I first played The King's Glory, it was a tune-up for the state I was in at the time when I was preparing for exams. At the time, I had to go and study for over six hours on top of work'.

——Andrew

'The mobile games help to decompress, and the equipment (Smartphone) is more convenient. I play games for fun, most of the time'.

——Berg

'[the mobile game is]more stable access to leisure and entertainment. I will spend 50% of my entertainment time in daily life in mobile games'.

——Karin

For Andrew, the attribute of 'simple' is an essential factor in choosing a game. He had limited leisure time outside of work. Hence, the 'simple' nature of the mobile game experience provided him with quick and effective entertainment to 'tune-up' himself. Compared with traditional console or PC games, mobile games are often less challenging and have a smoother learning curve. This simplicity is partly due to the input precision of mobile devices: touchscreen input is more difficult to control accurately than a mouse. Therefore, mobile games are often designed to be more lenient in difficulty than traditional games and more user-friendly for players such as Andrew.

Other interviewees, Berg and Karin, focused on the convenience and accessibility of mobile gaming. Berg highlighted the convenience of entertainment provided by the portability of mobile devices. The mobility of smartphones supports both Berg and Karin's descriptions of mobile gaming advantages. Digital devices offer instant access to linked entertainment. With the connectivity of digital devices extending beyond social time and space, mobile gaming provides users with a convenient and stable form of engagement in the digital world. Hence, from the interviewees' viewpoints, the 'simple', 'convenient' and 'stable' characteristics are why they choose mobile games for leisure.

The previous chapter discusses the production and performance of linear and cyclical rhythms in freemium mobile games. A rhythm refers to a series of events or behaviours that occur regularly and repeatedly. Daily life includes multiple series of rhythms that usually work together. In the context of this study, out-of-game daily rhythms refer to a mixed state of multiple rhythms. Studying the combination of the interviewees' everyday rhythms and gaming

rhythms reveals a state of rhythmic superposition, where different rhythms work simultaneously on players' bodies.

Apperley (2011) used the term 'local rhythm' to explains out-of-game rhythms, which focused on the globalisation of digital games. Apperley (2011) introduced the term 'local rhythms' to describe variations in gaming rhythms shaped by different geographical and material conditions. As previously discussed (particularly in Section 4.2), this concept is extended in this thesis to include the rhythms generated by the interactions between players and their surrounding environment. The performance of mobile game insertion into their lives is strongly influenced by these local rhythms.

The rhythms of mobile games, including game-based cyclical and linear rhythms discussed in previous chapters, negotiate with the local rhythms of the interviewees. The interviewees typically organise their playtime according to their local rhythms. For instance, employee interviewees often play games during predictable daily routines such as wake-up, lunch, dinner and bedtime. In contrast, student interviewees display more irregular gaming patterns. While employees might play games during their commutes, students and freelancers seldom follow this pattern.

Lefebvre's (2013) concept of eurhythmia can be applied to explain the different states of this rhythmic negotiation. For Lefebvre (2013), eurhythmia refers to the gentle and harmonious interplay of rhythms, akin to the rhythmic integration of the organs within the human body. He described eurhythmia as: 'Rhythms unite with one another in the state of health, in normal (which is to say normed) everydayness '(Lefebvre, 2013, p. 16). In this thesis,

manifestations of eurhythmia refer to the mutual negotiation and adaptation between in-game rhythms and out-of-game local rhythms, where mobile games serve as leisure or entertainment for players or fill gaps in their schedules. This chapter explores how the rhythms of mobile games intertwine harmoniously with the external rhythms experienced by players.

Section 5.1 of this chapter discusses how mobile game rhythms integrate into and become part of everyday life rhythms, focusing on the moments where game-based and everyday rhythms combine. Section 5.2 introduces the term 'switch' to describe the moment when a player's attention moves between the physical and digital game worlds. 'Switch-in' refers to the player entering a mobile game space and its rhythms, while 'switch-out' refers to the player leaving it. Interviewees adopt two strategies to switch between game-based rhythms and local rhythms based on their physical external environment. Some individuals switch when they have completed a pre-determined purpose, while others switch according to the current demands of the external environment or spontaneous thoughts, rather than a pre-plan or purpose.

Section 5.3 examines the performance of eurhythmia between game-based rhythms and everyday life. It categorises the influence of integrating mobile game rhythms with local life rhythms into four aspects: as an entertainment provider, a filler of daily gap time, a tool for linking, coordinating and synchronising with others, and a means for leaving, escaping, or resisting the daily environment. First, mobile gaming serves as an entertainment provider and integrates into the player's leisure life. Interviewees mentioned they use mobile gaming for relaxation or decompression. Compared with traditional console or personal computer games, as previously mentioned in this thesis,

interviewees perceive mobile gaming as 'simple' (according to Andrew), 'convenient' (according to Berg) and 'stable' (according to Karin).

Second, a mobile game can link two daily events that occur regularly and repeatedly in people's lives. It can be inserted into gaps between daily activities and become a mediator of life rhythms. Interviewees often referred to this as 'passing the time'. In this process, a player's attention, which is not consumed in the gaps of daily events, is re-organised and invested in gaming activities. Third, mobile games create a collective rhythm and common schedule for players of the same game. Players of the same game share a collective rhythm, including in-game events, rewards, the banner of new characters and updates of new maps or functions.

Therefore, players of the same game follow a similar digital schedule while living in different physical spaces and having different local rhythms. The common experience of game-related activities promotes the development of stronger social relationships within the player group.

Finally, mobile gaming can be a strategy for players to leave, escape, or negotiate daily repetition and pressures from the external physical environment. These pressures may originate from work or family, which players cannot leave completely. Mobile games provide an alternative space for players to leave a local life rhythm temporarily. However, this 'leaving' typically represents only a pause rather than a disruption of the local life rhythm. After a short gaming period, players generally resume following their local rhythms. Therefore, this short-lived break from daily life should be considered as a form of 'pseudo-disengagement'. During this time, players

simultaneously inhabit both local life rhythms and digital mobile game rhythms, using mobile game rhythms to construct the 'rhythm of the self' (Lefebvre, 2013).

Overall, this chapter explains how the integration of game-based rhythms and local rhythms impacts people's lives. These functions are often rooted in a complicated environment. When this thesis discusses an interviewee's case from a eurhythmia view, it does not imply that the role of a mobile game in this case always works in harmonious forms. Individuals who generally experience a harmonious rhythm between mobile gaming and daily life may also encounter periods of arrhythmia at other times, this thesis will return to further discuss arrhythmia in Chapter 6.

# 5.2 Switches in Mobile Gaming: Translating Between Digital and Physical Worlds

This section focuses on the modes of coordination and switching between game-based rhythms and non-game local rhythms. It lays the groundwork for exploring the effect of mobile game rhythms, eurhythmia and the synergy of multiple rhythms.

To facilitate a better understanding of the subsequent sections of this chapter, this section begins with a case study on how players switch between mobile game rhythms and daily life rhythms. This discussion is greatly informed by the concept of 'hybrid spaces' proposed by Adriana de Souza e Silva (2006) and Brigitte Jordan's (2009) explanations of the blurring boundaries between the 'real' and the 'virtual'. In her article, de Souza e Silva (2006) highlighted the overlap of digital and physical spaces, while Jordan (2009) discussed how

hybrid spaces reshape cultural practices, lifestyles and underlying ideologies as follows:

Work that formerly was tethered to a defined workplace is now routinely done at home, in the car, or in the kinds of public third spaces described by Churchill and Nelson (this volume), As a matter of fact, for many people work activities and related obligations have proliferated into almost all aspects of daily life. (p. 183)

Similar to work, mobile gaming can also be considered a time appointment. Players log into a game, complete tasks as dictated by its daily goal system and receive promised rewards. However, current research on how people switch between in-screen gaming and out-screen activities tends to be unidirectional. It primarily focuses on individuals' need to switch into the virtual world while neglecting the game's requirement for real-time human participation, such as login at specific period or the Monthly Pass. For this thesis, it is essential to consider both the human demand for the game and the game's demand for human involvement simultaneously. Both of them may lead human attention to switch from physical to digital space and vice versa.

When players log into a game, they respond to demands generated by a set of automated algorithms calculated by computer programs. Carmi (2020) noted that business companies, such as Facebook, understand people's rhythms to create experiences aimed at specific economic goals. By understanding data on people's daily behaviours, including the duration, pauses and frequency of their activity, businesses can monetise user attention for profit.

In the mobile gaming sector, companies not only profit from advertisements but also by selling in-game commodities to players. As one game designer interviewee, Brady, stated, 'The free-to-play model, itself, requires its players to stay in the game as much as possible...The nature of the requirement of mobile game design is the desire for players to consume in the game'. Players' in-game behaviour, particularly logins and logouts, can be interpreted as an elaborate rhythm, as previously mentioned in this thesis. For instance, the monthly card and login rewards necessitate that players enter the game within a prescribed interval.

This section revisits the concepts of semi-automation, daily task systems and daily login systems that are discussed in an earlier chapter. The presence of these automated mechanisms and daily task systems in mobile games motivates players to return to the game world periodically. This motivation is not inherent but is stimulated by the design mechanics of mobile games. Chapter 4 explains how players 'naturally' consider returning to the game as part of their daily routine. The present chapter discusses the switch between the on-screen and off-screen worlds, which is influenced both by game-designed incentives and players' spontaneous desire to engage with the game.

This thesis describes 'switching<sup>33</sup>' as a mechanism that redirects players' attention between the in-game and out-of-game environments. The term 'switch in<sup>34</sup>' is used to denote the transition from daily life to mobile games

<sup>&</sup>lt;sup>33</sup> This term was inspired by Atkinson and Willis (2007), as they argued "Multi-user virtual environments also succeed in constructing persistent online communities that cannot be easily switched on or off; they have an emergent quality that partially mirrors the entry and exit of actors in real social settings (Delwiche 2006) (p,825)"

and 'switch out' to indicate the return from games to daily activities. The moment of switching in or out can be viewed as indicative of a rhythmic combination.

The reasons for 'switching in' to mobile games are often complex and influenced by real-life circumstances, such as periods of free time, as well as by game mechanics. Two in-game mechanics that may affect people's decision to switch in are the in-game stamina points mechanics and the ingame log-in system (described in chapter 4).

In the in-game stamina points system, stamina points are automatically restored over time until reaching a maximum limit. As discussed in the linear rhythmanalysis, interviewees prefer to time their return to the game to prevent their stamina points from exceeding this maximum limit. Once they run out of in-game stamina points or complete their daily tasks, they typically switch back to their non-game daily life rhythm.

The second mechanics is the in-game log-in system Many mobile games feature timed login mechanisms at midday and evening, rewarding players for logging in at these specific times (as discussed in the cyclical rhythmanylyse in Chapter 4). The desire to receive these rewards can motivate players to switch-in to mobile games.

'Switch out' from mobile games can also occur for two primary reasons. The first is that players are disturbed by external environmental factors, such as the end of leisure time or the arrival of another event on their daily schedule. For example, the interviewee Abner stated:

'I will be more focused on playing the game... [it] will be interrupted by tweets and phone calls. I get annoyed when I'm interrupted by other people while I am playing'.

——Abner

In this case, when Abner answers a phone call, he is compelled to switch out from gaming. Beyond that, the second reason for switching out is the completion of in-game goals. Once players achieve their in-game goals, they tend to leave the mobile game rhythm and return to their non-game daily life rhythm. These in-game goals are quite flexible and can be adjusted at any time. They may involve winning an in-game battle or completing all daily tasks, which might include a dozen challenges. It may take a few seconds to twenty minutes or more.

During the interviews, the interviewees demonstrated two patterns of rhythm combination methods: the quick switches model and the planned coordination model. First, the quick switches between a mobile game rhythm and a daily life rhythm. A part of interviewees prefer to switch between these two rhythms based on their current needs. When something in the game demands their attention, they focus on the digital world: conversely, when tasks in the physical world require their attention, they prioritise the physical external environment. The switching back and forth between the in-game and the outof-game tasks may occur many times over a period.

The process of switching in and out can be illustrated by the description of the interviewee Berg in the subsection of automation mechanisms (4.3):

'I usually do things like watch videos [at the same time as playing], something that I can stop at any time and don't have to spend too much attention on...The time [I] spent watching videos will probably depends on [my] playing game. I watch videos I wouldn't watch a certain kind [slight pause here]and would probably also choose some more random videos that I could pause—so I don't have a specific time plan. I will watch the video while my phone is over there and [when] it [the mobile game] tells me to click here, I pause the video, click somewhere, and then come back and continue watching'.

—— Berg

According to this description, Berg simultaneously keeps both a mobile game and an online video page open. He interacts with the game when it requires action and, once the action is completed, he switches back to the video. According to his description, he does not follow a specific plan or aim for the process but focuses on the immediate demands of either the game or the video page. When the game captures his attention, he focuses on it; when the video requires interaction, he switches to managing the video page. In Berg's playing process, there is an overlap and mixture of rhythms: the rhythm of the mobile game does not consume all his attention but persists intermittently during his leisure time. In this switch-in-and-out model, players pay partial attention to intermittent participation, with the mobile game rhythm being constantly interrupted and resumed throughout this period.

Second, the planned coordination model can be illustrated by the conversation between the researcher and the interviewee Elisa:

The Researcher: When you play, will you remember the building time of the game? For example, if the building needs 8 hours?

Elisa: I do. When I'm back at home [the interviewee mentioned she had been on a trip earlier] and I have the tablet with me, and I keep an eye on it almost all the time. And then if there's a new building, I remember when it's going to be fixed and I'll go and click on it in time to make sure it's finished and so on.

In this case, the 'building time' represents a specific period for the predesigned in-game goal. Before opening the game, Elisa considers the purpose of playing, such as collecting materials or completing daily tasks. Her gaming is staged: once she completes a stage, she puts the phone down and moves on to another event in her daily life.

Like Elisa, some interviewees focus on playing mobile games until they achieve a planned goal. In this model, players tend to remember the purpose, switch in to the game space, achieve the goal and then switch out. Players of simulation mobile games are more likely to follow this pattern more frequently. In such games, in-game resources are generated in fixed-time units, allowing players to know precisely when to log in.

However, behavioural strategies are not fixed. Individuals may change their strategy depending on the game content. For instance, Berg also claims that he has also plans his behaviour for game rewards.

I even just, in a period. I set up an alarm at a specific time to collect. I forget the name of the game, [set alarm] for the game stamina or something

because many games have similar settings [referring to many games having a mechanism for rewarding players at certain times] to remind myself of that.

– Berg

The interviews revealed that the interviewees' choice of rhythm model is likely influenced by the game's content. When faced with one-off, attentiondemanding, competitive or complex game content, players are less likely to adopt the quick switches behavioural pattern and more likely to devote their full attention to the game. Conversely, as in Berg's case, when the content is straightforward and highly repetitive, they switch more frequently. These differences in strategies also influence how the impact of mobile game rhythm everyday life. In the quick switches model, the game rhythm is constantly interrupted and resumed, while in the planned coordination model, the game rhythm is continuous and occupies a complete segment of daily time.

Overall, this subsection summarises the different patterns of how people switch between in-game and out-of-game rhythms, setting the groundwork for the subsequent discussion.

#### 5.3 The Role and Function of Mobile Game Rhythms in Everyday Life

This section explores three effects of mobile game rhythms related eurhythmia on the players' daily lives. The first subsection explores how mobile gaming filling time 'gaps', which was observed among most interviewees as a common effect. The second subsection focuses on how mobile game rhythms contribute to creating a collective rhythm to describe the impact of mobile gaming on developing social relationships. The third subsection addresses mobile gaming as a means for leaving, escaping, or negotiating daily repetition and pressure from their environment.

### 5.3.1 Gap-filling and Re-use of the Daily attention: How Mobile Gaming Rhythms Can Reorganise Time

This subsection explores the feature of fragmentation in mobile gaming. Mobile games typically involve daily tasks that include a series of linear and cyclical rhythms. However, these in-game rhythms are not continuous: a player can complete part of them in a single log-in and finish the rest later. Each gaming session can be independently inserted into different periods of daily life.

The daily task system can be described as a combination of a series of regular linear repetitions: these linear behaviours lack mutual continuity, which allows players to stop gaming after completing any segment. The interviewee Angela illustrates this feature as follows:

'For most mobile games, I play a few minutes or a little bit more, 15 minutes? And then it is over, I can end it. But console games are different. Although it is said that a console game can also be a small game [the interviewee means short playtimes], most console games are time-consuming in terms of plot or the course of combat. Therefore, [console games] need more complete time, not [like mobile games] laziness time after dinner, play for a while or getting up but want more than ten minutes lying. It is not quite the same'.

——Angela

In Angela's example, when she needs to lie 'in bed for a while', mobile gaming fills this period which allows her to continue playing while resting; she can stop playing whenever she wants to move on to other daily activities. Similarly, the time after lunch can serve as a quick opportunity for playing. Angela's interview also highlights the difference between mobile gaming and PC or console gaming times. Although games on different platforms may have similar elements or mechanics, mobile games offer greater flexibility in playing spaces, such as when lying in bed, as in Angela's case.

Like Angela, many interviewees stated they prefer to play mobile games because they can easily insert gaming into any part of their daily lives, anytime and anywhere. Although mobile games borrow mechanics from PC games and PC games also borrow elements from mobile, the mobile game has more advantages in flexibility in the playing space. This flexibility has become a significant advantage of mobile games compared with console games, which typically require a dedicated time slot. Some interviewees mentioned setting aside console gaming for weekends or days 'with more private free time', as noted by the interviewee Peter. During the interviews, interviewees emphasised the 'completeness' and 'long' duration of the console gaming experience, which requires a significant time commitment. Angela mentioned she would avoid going to the toilet to complete her console gaming session without interruption, while Karin described the aim of console playing as a 'longer period of fulfilment'. Calvin also asserted that console gaming 'requires long periods of free time'. PC and console games deliver a continuous gaming experience that necessitates a dedicated time slot, making

them more difficult to integrate into other everyday life rhythms. However, the semi-automated features of mobile games allow players to swiftly switch back and forth between gaming and life activities.

In this subsection, mobile gaming is portrayed as a mediator that fills and inserts into gaps in people's daily rhythms. At the same time, it can also be described as the re-use of people's attention during the intervals of everyday life. Mobile games act as a mediating reconciliation for specific periods of life. Interviewees used the term 'filler' to describe how mobile gaming is inserted into a gap between two specific activities in their lives and connects two regular daily activities, such as during meals, work or commutes.

In 2016, Game study scholar Isbister highlighted the blending of connectivity in networked games and interactions with people in one's 'real world'. She cited Bruckman's (2013) blog post about Words With Friends (WWF)<sup>35</sup> to illustrate how mobile games integrate into personal life. In this blog post, Bruckman (2013) described that she tended to insert mobile gaming into fragments of her daily life:

Consider the following situation. I'm picking my kids up at aftercare at their elementary school. When I arrive, they are somewhere in a large school building (Doing art in the cafeteria? Out on the playground?) and they are paged to come to the lobby. It usually takes about five minutes for them to stop what they're doing, clean up, travel across the building, find their backpacks and coats, and be ready to go. So it's a perfect time to make my WWF move, right? Perfect except that if I'm playing a couple different games,

<sup>&</sup>lt;sup>35</sup> An online mobile game where players take turns completing the crossword, similar to the multiplayer version of the classic board game Scrabble.

I won't be done when they arrive. So I put away my phone, but part of my brain is still thinking about my move (what words end in 'u'? 'Tofu'? 'Bayou'?) rather than paying full attention to what happened at school today. Until I finish making that move, I won't fully be there. And it's like that through my entire day. The little gaps I have don't match the amount of time it takes to make my WWF moves. The fact that you can play on your phone makes the temptation pervasive (Bruckman, 2013. no page).

During the interview process of this study, several interviewees provided descriptions similar to that of Bruckman. For example, the interviewee Abner mentioned:

'When I'm partying with friends and going out for drinks, I'll play mobile games while I waiting for someone. Mobile games are 3-5 minutes and don't require a lot of attention. I'll play a challenge while waiting for the restaurant to serve me food, or while waiting for my friends to arrive. It won't be as engrossing as playing by myself. It's a bit awkward waiting for someone, give yourself something to do'

—— Abner

Similar situations, such as waiting for someone, during daily events (e.g. a lecture or a meal) or on public transportation (bus or subway), are often referred to as 'time fragments' by the interviewees. These 'fragments' are periods without any specific real-world aim or plan; these periods are usually quite short, lasting from three to ten minutes, and their end may not be precisely known. Interviewees described using mobile gaming as a method or
means to fill these time fragments. For example, as interviewee Jessica stated:

'[I] feel like it is filler...It is an alternative to killing time'.

----Jessica

The interviewees' description of 'time fragments' aligns closely with Chess's (2018) concept of 'interstitial time' in gaming. This concept refers to 'small spaces of time—gaps of time that are not used for anything else. All of our lives are characterised by interstitial time, which is not used for other purposes' (Chess. 2018, p. 111). The 'interstitial time' was applied to the social network game. It is the small space of time between two activities without a defined purpose, like the ten-minute stop of a lecture, or the time waiting for the city bus. Similar to the social network games, mobile games also have the characteristics of utilizing interstitial time. In this research, the interviewee Calvin illustrated why players prefer these time fragments for gaming:

'The smartphone itself can ignore time and occasion, and it can take up pieces of one's time, picking it up and putting it down whenever and wherever. For example, I would not say I like the King's Glory and Knives Out because they take ten minutes to put down, and I cannot abandon my teammates. I do not like mobile games which take up more time [and] I like to access gaming with a few clicks with my head down.'

——Calvin

In this case, Calvin's perspective on 'time fragments' when incorporating mobile gaming into his daily routine highlights two key features. The first is the

short duration of play, typically less than ten minutes, and the second is the ease of entering and exiting the game. As previously discussed, this represents a quick switch method between the mobile game space and the non-game physical space, with the 'picking it up and putting it down' approach illustrating a rhythm that is being intermittently interrupted and resumed during the gaming process.

Many interviewees reported playing mobile games while waiting for public transportation, with the waiting times for buses and subways serving as examples of 'time fragments'. In a player's daily schedule, the 'commute' event has a fixed duration; however, the actions within this event are not continuous. Commuting can be broken down into multiple moments: walking to the stop, waiting for transport, sitting on the bus seat, observing the stop sign and continuing to wait for arrival. These are a series of independent actions rather than a single, indivisible behaviour. This discontinuity creates opportunities for inserting mobile game rhythms.

The case of Abner further explains the discontinuity of daily events. When Abner was waiting to meet someone for drinks, he used mobile gaming as a tool to pass the time. According to his description, the moments of 'the arrival at the agreed location' and 'the arrival of the awaited person' is occur consecutively within short time fragments. These fragments are considered part of the same event of ' partying with friends and going out for drinks', although there is an empty and unscheduled period between them. This fragment represents a gap in time and attention that can be filled by mobile gaming.

In addition to Abner's case, the cases of two more interviewees, Bella and Peter, further explain the role of mobile gaming in daily life. In these cases, the time fragments for gaming correspond to waiting for food or waiting for someone. Bella stated:

'If I'm in a restaurant and order my food, I would like to play mobile games while waiting for it to come. I mean, if I have friends to eat together, I definitely take care of my friends and have conversations with them. But sometimes when I'm eating by myself and have no one to talk to, I go play mobile games'.

——Bella

Similar to Abner's case, in Bella's experience, there is a gap between the moment she orders food and when the 'food is ready to eat'. Another interviewee Peter also mentioned that:

'I go to the canteen with my roommate and classmate. I eat faster and my classmate doesn't finish. While he is still eating, I spend some time playing mobile games'.

——Peter

For Peter, the time fragment between finishing his meal and waiting for his classmate to finish is also sufficient for gaming. Although these moments are typically scheduled as part of a complete routine, fragments of time and attention exist in the subtle breaks or pauses within a series of meaningful actions. During these schedules, parts of the time are set aside for waiting,

and attention is not fully occupied. This is the gap that mobile gaming can fill, integrating seamlessly with people's daily life rhythms.

When mobile game rhythms function as a filler, they usually do not completely occupy a person's entire attention. For instance, while interviewees were playing games at the station, they were still waiting for the bus, and while Abner was playing, he was also waiting for his friend to arrive. This case suggests that the game rhythm acts more as a support-complementary relationship to players' local rhythm. Compared with traditional PC games, mobile games offer more advantages in supporting this function, as illustrated by Abner's experience. Mobile game rhythms allow participation in public spaces and as demonstrated by Calvin's case, feature a lower time cost for switching in and out of games. These attributes make it easier for mobile game rhythms to be conveniently inserted into daily schedules.

As this chapters mentioned, Chess's (2018) concept of 'interstitial time' in gaming describes "small spaces of time—gaps of time that are not used for anything else" (p. 111). However, the research interview data suggest that mobile gaming time cannot be fully considered as 'not used for anything else', as illustrated by the cases of interviewees James, Abner and Berg. These examples show that mobile gaming often coincides with other daily activities, filling small gaps in their day while still allowing users to remain engaged with their surroundings.

Similar to Berg and Abner, the interviewee James stated:

'You go to take a Didi [a taxi-hailing application similar to Uber] and expect an hour and a half when nothing else can be done. Alternatively, say,

commuting by high-speed rail, you know it may take an hour and a half to get home. So you can expect that there may be a longer piece of time, but you have nothing at hand except your phone.'

----James

During their gaming, James was travelling, Berg was watching a video and Abner was waiting for his friend. They did not dedicate their time exclusively to gaming but also engaged in other activities simultaneously. The gaps are more accurately defined by shifts in attention rather than time, with James's case serving as a pertinent illustration of this concept; taking the Didi is a demand on his personal time rather than his attention: while his flesh must remain in the taxi seat, his attention is free to be used on any digital activities. During his travel time, James experiences an excess of attention. However, under the conditions of mobility, the rhythm of the mobile game is employed to fill attention gaps during transportation. It runs parallel to the activity of the player's transportation, effectively occupying these interstitial moments.

The case of Elisa further supports this argument. Elisa plays games during college classes. When a teacher assigns discussion tasks or covers knowledge, she is already familiar with, she briefly engages with her smartphone. Even when the teacher shifts to topics Elisa is interested in, she may still be engaged with her mobile game. Attendance at the lecture is a fixed event in her daily schedule, but the time used for mobile gaming should be considered not as a personal scheduling choice but rather as a reallocation of personal attention within a predetermined daily event.

Like Elisa, many interviewees reported integrating mobile gaming into group activities. For those adhering to the typical schedules of a company or school, inserting mobile games during lunch and dinner times is common. For these individuals, meal times are fixed events, part of a collective rhythm involving hundreds of people, and are orchestrated by a local company, school or other organisation. These cyclical events recur daily. At the same time, individuals might finish their meals quicker than the allocated standard time, thus leaving a small fragment of time for personal use. This time can be filled with activities such as mobile gaming, effectively using these brief fragments within the structure of their day. This concept can be illustrated through the experience of interviewees Jessica:

The Researcher: 'What do you mean by a 3-hours of time fragment?' Jessica: 'Generally, the longer time is the lunch breaks, then when I catch my breath at work, and then on the way from work to home' The Researcher: 'How long do you take your lunch break at noon?' Jessica: 'One-hour lunch break'.

The Researcher: 'Do you play games every lunch break?' Jessica: 'Basically'.

The case of Beatrice also illustrates this concept:

'[I will play] while waiting for a meal, and the waiting time after lunch'.

—Beatrice.

Both cases demonstrate the concept of a 'time fragment' left behind in organised activities. Jessica, a company employee, has a fixed lunch break.

Beatrice, a university student, accesses meals at the canteen at set times. Consequently, their mealtimes are predefined within their daily routines. Jessica and Beatrice repeatedly follow this regular schedule, which is already established in their daily lives. However, they often find themselves with surplus attention during these periods, which can be redirected towards other activities.

As mentioned in the previous chapter, many mobile games distribute stamina points as gifts during lunchtime and dinnertime to encourage players to log in. These extra stamina points are often referred to as 'lunch' and 'dinner' within the game environment and gamer communities. Given this setup, during the short lunch and dinner breaks, individuals may simultaneously experience two concurrent cyclical rhythms: one from the social organisation of the physical environment and another from the mobile game in the digital environment. The latter rhythm inserted into and fills the gaps left by the former.

In this context, playing in the university canteen can be understood as the cooperative interaction between the 'rhythm of the self' and the 'rhythm of the other' (Jaulin, 1973, as cited by Lefebvre, 2013, p. 95). According to Lefebvre (2013), the 'rhythm of the other' refers to the rhythm of public activities, such as students entering the canteen at its opening time in interviewee Beatrice's case, while the 'rhythm of the self' pertains to an individual's reorganisation of private life. This is exemplified by Beatrice's and Jessica's attempts to play games during the ruled lunch collective rhythms.

It is also reasonable to consider that the interviewees are experiencing two collective rhythms at the same time. As Lefebvre (2013) stated, "Every social,

which is to say, collective, rhythm is determined by the forms of alliances that human groups give themselves" (p. 91). The rhythm of logging in at lunchtime is determined by the game company, while the rhythm of the canteen's opening is set by the university. The simultaneous occurrence of these two rhythms implies that the interviewees are present in both digital and physical spaces at the same time. This hybrid space allows for the overlap of the 'rhythm of the other' and the 'rhythm of the self', leading to their mutual insertion. The inserted rhythms can support and complement each other, allowing the player to utilise surplus attention effectively.

In most cases, interviewees use mobile gaming to fill gaps in their schedules, helping them avoid feeling miserable during unscheduled times. Mobile games provide virtual targets that offer immediate positive feedback. Referring back to the concept of the in-game linear rhythm mentioned in the previous chapter, the physical actions required in mobile games to obtain rewards or positive feedback are simple, often involving regular taps on the screen. Thus, the insertion of mobile games into players' everyday lives follows a two-step process: providing actionable goals and offering rewards upon goal completion. During these gap periods, some interviewees tend to avoid complex content<sup>36</sup>. When these interviewees choose mobile games as an intermediary between two daily events, they are more likely to engage in repetitive game segments rather than reading game stories, interacting with in-game characters or tackling difficult in-game challenges. For example, the interviewees Rex stated:

<sup>&</sup>lt;sup>36</sup> It is worth noting that not all interviewees exhibited this trait. Some reported not noticing this behavior, while one interviewee stated having the opposite tendency, purposefully selecting high difficulty levels during fragmented time.

'I will probably be farming content [meaning performing the same game action over and over] on content that does not put too much effort into..... preferring the repetitive content'.

'Like Arknight or the Onmyoji, gaming is not only for finishing in-game quests. The aim of finishing a quest is to develop the story, if it is time to read, I will put it down until I come back home, then read, and the same is true when playing other games.'

—— Rex

And interviewee Zoe also mentioned:

'I am the kind of person who has to play a whole [storyline] at once to finish an ending. Then the process will be long, if I'm at the office, where I am mostly here to study, so I do not open Eternal City in the office...There is no difference in which environment for Love Nikki<sup>37</sup>,

—— Zoe

In these cases, the interviewees' gameplay involves repeating the same ingame content. Whether through linear clicking (described as farming) or the cyclical rhythm of daily goals, the interviewees accept the inherent repetitiveness of game-based rhythms. Another interviewee, Calvin, corroborated this argument by stating:

'The story games need to have an immersive experience'.

—— Calvin.

<sup>&</sup>lt;sup>37</sup> Love Nikki involves fast daily task system, almost all the content is repetitive clicking. while Eternal City has a relatively long storyline need to read

The term 'immersive experience' closely relates to the psychological concept of flow. With an 'immersive experience', gaming occupies all the player's attention. This state means that players do not frequently switch between the virtual and real worlds but remain engaged in the game for extended periods. When players engage with one-off game content—such as in-game character interactions, storylines and activities—they invest all their attention in the mobile game. In this sense, mobile games function similarly to traditional handheld games.

However, this 'immersive' state is something players wish to avoid during brief fragments of free time. The descriptions provided by the interviewees Rex, Zoe and Calvin suggest they try to maintain an unstable connection with the game space. When they have other daily goals to accomplish, such as reading or studying, they prefer to switch in and out of the game conveniently and straightforwardly. Their preference for repetitive and simple content reflects their need to switch between in-game and out-of-game activities at any moment.

This subsection examines the interplay between game-based rhythms and out-of-game rhythms during the daily intervals of life. In this gaming process, the connection between the player and the game is unstable and can be interrupted at any moment. During these unstable periods, people may prefer engaging in straightforward, highly repetitive content to fill the time gaps in their daily lives. Compared to traditional console games, repetitive and divisible mobile games better fit into the segments of people's lives. Mobile game players can complete parts of their daily goals at any time rather than engaging in a relatively longer immersive experience. This preference reflects

the characteristics of the combination of mobile game-based rhythms and everyday rhythms: instability, rapid disconnection and reconnection, and incomplete attention occupation.

In this process, the term 'interstitial time' (Chess, 2018)' also referred to as 'fragment time' or 'free moments', is frequently mentioned by our interviewees. This term describes small pieces of time that players can freely arrange. It also means that the leftover time or attention can be occupied by mobile games when people are subjected to the rhythms of industrialised social organisation. According to the interviewees, most people's work and lives are highly organised and structured. Whether they are office workers who need to clock in in the morning or university students who need to attend lectures and social activities, their daily lives follow a collective, cyclical routine. From morning to night, players' physical bodies are constantly subjected to an imposed industrial rhythm. However, the macroscopic rhythms of industrialised society cannot regulate every aspect of personal lives. During the individual's adherence to the collective schedule, there may arise a surplus of personal attention and time. This surplus becomes the basis for integrating mobile gaming into the rhythm of daily life.

The high degree of fragmentation in mobile games' daily routine systems allows discrete fragments of personal time to be organised and utilised within the mobile game system. This fragmentation characterises the rhythm of mobile games in terms of user time, where the game does not demand continuous, large amounts of attention or personal time but rather focuses on integrating the user's surplus attention. The surplus attention from these daily event fragments can be accumulated and exchanged for in-game items within

the mobile game system. This process can be seen as a form of leisure and a means of commodifying the user's attention.

Additionally, it is important to consider the impact of monetisation mechanics on playing time and attention. As discussed in Chapter 4, game mechanics such as daily quests, monthly passes and battle passes require players to log in daily or at specific times. These demands on players' online time based on game mechanics can be one reason players log in during fragmented time periods.

This subsection argues that mobile games, as a medium beyond social time and limited space, reorganise fragmented personal time and attention, integrating them into the rhythmic time constructed by the systemic environment provided by mobile games. The rhythm of mobile gaming parallels the industrialised societal rhythm imposed on people's physical bodies. This rhythm occurs in both geographical and digital spaces, forming a mutually supportive and complementary relationship.

## 5.3.2 Game Rhythm for Creating the Collective Rhythm

This subsection explores how a mobile game creates a collective rhythm for its players and links groups of people within this created rhythm. Compared with the former subsection, this subsection focuses more on human relationships and interactions and how mobile gaming shapes an everyday social life schedule. By synchronising different people's life rhythms, mobile gaming fosters deeper relationships between individuals.

The concept of collective rhythm was proposed in Lefebvre's (2013) rhythmanalysis of Mediterranean Cities. He defined collective rhythm as the unfolding of alliance interests within "social time that they contribute to producing (or reproducing) by impressing a rhythm upon it...social, which is to say, collective, rhythm determined by the forms of alliances that human groups give themselves" (p. 94). This collective rhythm is built by the city's architecture, daily ceremonies, religion, politics and culture.

This thesis focuses more on an artificially created social schedule shaped by conditional information communication based on digital media devices. In mobile game environments, game companies often design festivals, events or celebrations to update the new commodities<sup>38</sup> or activities. This update cycle for in-game purchasable commodities can be regarded as an artificial social schedule service for the player's online payments. As discussed in Chapter 4, the timing of these updates is fixed and follows a cyclical pattern.

Conditional information communication refers to a game's conditioning of players based on computer logic. If players perform a pre-designed action, they will receive pre-set feedback from the computer program. Through this, players have spontaneously developed a similar in-game playing rhythm based on the conditional feedback given by the mobile game system. This is evidenced by the interviewee Scott, who has extensive experience in mobile game design. As this thesis mentioned in the chapter 4.2.3, Scott stated:

'We have a very clear plan for the numerical growth of each player, what kind of game fun will be given at what time, precise to the minutes, and the

<sup>&</sup>lt;sup>38</sup>These may include new characters, skins, weapons and equipment,etc.

fun comes from which game mechanism. This is an absolute consensus among mobile gaming companies. In order to guide the new game players, the game process is clearly planned. Players do not necessarily follow the plan strictly, but I have to give this planning for player actions'.

——Scott

This description demonstrates that game designers can craft the experiences, items and opponents a player encounters through a detailed plan. In the context of mobile games, this artificially created social schedule is characterised by flexibility within short time frames and fixity over longer periods. Limited flexibility refers to the player's freedom to decide how long and when to play within a relatively short period (e.g. one day). As discussed in the previous section, players can insert mobile gaming into fragments of their daily lives according to their local rhythms. Fixity over longer periods means that a mobile game producer updates content at regular intervals (e.g., every 40 days), and in-game rewards refresh regularly. Players need to obtain these rewards before they are updated or removed, thus their in-game actions corresponding to the rewards are fixed during this period.

To earn in-game rewards, most players of the same game perform nearly identical in-game actions. An in-game activity or event is a series of tasks designed by the game company, with each task corresponding to specific rewards. As discussed in Chapter 4, these events often coincide with festive periods such as summer holidays, Chinese New Year and Christmas, etc. The content of these events creates similar actionable behaviours for players,

allowing them to experience the same story, collect the same items or defeat the same enemies.

Through this shared schedule, players of the same game spontaneously generate a collective rhythm across geographic boundaries. In the mobile game space, the player is part of a collective, not in a physical sense, but transcending geographical limits. The shared identity of being a game player represents an online alliance. The game space functions as a metaphor for a virtual city, where players walk, chat and participate in ceremonies within the digital realm. Online gaming events can be seen as variations of ceremonies held in city squares.

Among all online game events, the moment of opening a new character banner<sup>39</sup> is typically significant for players. Game companies usually preview characters in advance. The 'character/weapon banner' is a cyclical event where a company creates a prize pool filled with different prizes at regular intervals, with the pool being refreshed periodically. Specific prizes (e.g. ingame characters) are available for a limited time. Following the pattern of prize pool updates, players experience the same rhythm: completing daily tasks to accumulate Gacha resources, waiting for the prize pool to update and entering the pool to draw prizes. This process is quite similar for all players of the same mobile game.

A typical daily phenomenon is that mobile games often become hot topics on social media on the day or the day after the in-game content is updated. This

<sup>&</sup>lt;sup>39</sup> This involves making available characters accessible to players through Gacha.

indicates that, during these periods, players are almost spontaneously interested in the new in-game content and eager to discuss it publicly.

The interviewees Eunice and Harold exemplify moments of connecting with the other players of the same game. Eunice stated that:

'[When my friend and I] drew a beautiful character, we would talk together, like two seals<sup>40</sup>'.

——Eunice

And Harold also described:

'I had a classmate with [whom I had] a normal relationship in the high school, but after we graduated college, I suddenly found that we both played as the Fate/Grand Order [a mobile game]. I saw it on his social media and saw him sending posters about his new card<sup>41</sup> [in-game characters].

——Harold

In the cases of Eunice and Harold, they are simultaneously engaged in the game schedule with their friends. The opening of a character banner can be understood as an online ceremony in the mobile game space. By participating in the same series of ceremonies, game players achieve a collective rhythm with others. This participation in a collective rhythm should be regarded as a

<sup>40</sup> In the context of the game, 'seal' refers to 'sunbathing', which means 'to put something you have in the public area', implying showing off.

<sup>&</sup>lt;sup>41</sup> The term 'cards' refers to rare items that users obtain probabilistically in the game. Due to the capabilities of early mobile devices, popular mobile games in China used cards to represent characters and items. Consequently, players began using 'cards' to describe any rare characters or items in the game, regardless of whether they were actually represented as cards.

negotiation between mobile game rhythms and daily life rhythms in a hybrid space.

Mobile games have the unique advantage of building a collective rhythm. First, mobile gaming participation is convenient, fast and requires less attention. Attendance at online ceremonies can occur at any time or place in daily life, such as on buses, subways or during any gap time of the day. Second, mobile games are more persistent than most other cultural products (e.g. books and movies). While consuming a book or movie typically takes only a few hours, some people play mobile games for several months or even longer. Given these premises, the daily task system and cyclical updates of in-game events guide players to use the game regularly and take similar actions in the game. This similar and regular in-game behaviour among players builds a collective rhythm.

Online avatar creates a virtual presence for players to participate in public events organised in digital space. In the earliest stage of online gaming, all players shared the same digital space, challenging the same group of monsters, chatting with the same non-player characters and having to line up to do so. This model soon proved to have significant drawbacks, as the game content could not accommodate a large number of players simultaneously. As a result, game companies invented the design of 'instances'. This term refers to the game system producing an available copy of a space for each group of players who decide to challenge a specific area. Each instance is an exact replica of the original area, allowing players to interact independently.

This complex spatial division means that each player is in their own (or their group's) digital space, but these copied spaces originate from the same area. In mobile gaming, the collective rhythm experienced by players is an extension of the 'instance' concept. Players encounter the same content distributed by the game system, chat with the same non-player characters and exist in nearly identical digital environments, even if they may do not see each other.

Based on the above discussion, the following paragraphs explain how mobile games create connections between players. Mobile gaming can act as a coordinator between people with different life situations. For example, the interviewees Marko described:

'My high school classmate and I now have a good relationship. We just started talking again because both of us play Fate/Grand Order [a mobile game], and [my high school classmate] posted a screenshot of getting a new character. Then we re-chat again'.

——Marko

Harold shared a similar story after discovering that his old friend was playing the same game as him:

'From here, we chatted about more topics, from Fate/Grand Order to talk about our daily lives after graduation'.

——Harold.

For players of the same game, their daily routines within the mobile game are almost identical. The shared daily goal system leads to similar in-game activities. The actions players make in a game are consistent and repetitive, creating a common in-game rhythm. This shared rhythm implies a common communication context or the collective rhythm. The schedule of mobile game updates, including new characters and events, establishes a game-based rhythm. Hence, in the cases of Marko and Harold, when they 'saw their old friends showing off newly acquired in-game characters on social media', it indicates that they are experiencing a similar rhythm through mobile gaming.

Nevertheless, when players live in and negotiate different out-of-game social environments, they may construct unique local rhythms. The interviewee cases in this thesis show that mobile game players from diverse backgrounds try to adapt to the pre-designed rhythms offered by mobile games within their unique social contexts. For instance, during game guild wars occurring at specific times, employees might hide in the restroom to go online with other members. Students might log on to collect stamina points, distributed in the middle of the day and evening, during meal times in the cafeteria.

The rhythms of mobile gaming are interspersed with other daily routines in the office, classroom and laboratory. While their out-of-game daily routines may differ significantly, players of the same mobile game share a relatively consistent set of online schedules. This consistency exists even though they may be located in different parts of the country, with varying occupations and age backgrounds.

The interviewees Max and Darlene were both players of a mobile game called Onmyoji. This game holds online guild activities on certain nights, requiring players in the same guild to work together to defeat in-game bosses. Max, a university student at the time, stated:

'I'm a guild main member<sup>42</sup>, if the night does not happen some special event, I'm in it every time'.

——Max

He has a considerable amount of leisure time and often consistently plays for five hours from 6 pm to 11 pm. In contrast, Darlene, who has been working for many years, often uses his commute time to play but still participates in guild wars. Darlene described his participation as follows:

'As long as there's a reward, I'm in it, and there's a guild event in Onmyoji every day around 6 pm-7 pm. It is the time I have dinner, so I use that time to do those two things together. On weekends, if I have dinner with friends, the schedules will collide, and I will find a way to participate in the guild activities. I do it before dinner or play a game while talking with a friend'.

——Darlene

Max and Darlene have quite different out-of-game daily lives, but their ingame schedules are consistent. They log in during dinner time and fight bosses with other guild members. This consistent game-based rhythm is also part of their private daily routines. As Darlene mentioned, he has a non-game

<sup>&</sup>lt;sup>42</sup> it refers to a player who can deal significant damage to the boss

daily rhythm of 'eating dinner during guild activity time' or 'meeting friends on weekends'. When he attempts to follow both the game events and his out-ofgame daily rhythm, he integrates the two—the rhythms of physical space and digital space, the pre-designed collective rhythm and his personal local rhythm—trying to make them work together.

These interviewee cases also show that mobile games offer the possibility to construct the same rhythm across time and space. The content related to the game-based rhythm is entirely artificial, implying a set of schedules for marking time and understanding timing. Max and Darlene shared the same understanding of nighttime because they participated in the same Guild event. although they might never have contacted or known each other. Similarly, the interviewees Marko and Harold, after leaving campus, built different local schedules dictated by careers, families and companies. However, the mobile game still provided an opportunity to synchronise their rhythms with friends living in different environments.

To return to the beginning of this subsection, mobile games create a collective rhythm that refers to a series of artificial social schedules. Based on the collective rhythm of mobile game mechanisms, players of the same game develop similar rhythms in their out-of-game social groups, whether large or small. This process can be illustrated through the interview cases of Elisa and Werner, Elisa described:

'I would want my best friends around me to play [the same] mobile game with me... It seems to happen in all the games I have played. So I would love for the person I am best connected with to play with me and then we

can talk about it every day, probably for social purposes as well, one more thing in common'.

——Elisa

Similarly, Werner stated, 'My colleague and I are playing the Azur Lane[i], but I cannot say that we are playing together. It is more like we are together. We are playing a game simultaneously'.

The expression 'together' implies that Werner and his colleagues share the same collective game rhythms and develop similar game-based personal life rhythms within their smaller alliances. They acquire new cards simultaneously, complete daily in-game tasks simultaneously and participate in in-game activities at the same time. This overlap of game-based virtual rhythms creates a communal experience. The non-game local rhythm is the result of geospatial and social organisation practices, while the rhythm of mobile games is built on the replicated digital space for each individual or group player. The former implies a simultaneous presence in the physical geospatial space, while the latter points to a shared collective rhythm experience, akin to the parallel universes of science fiction.

The interviewee Elisa's experience also supports this argument. When she returned to her hometown during the Chinese Spring New Year, she needed to socialise with people she did not know well, some of whom were much younger than her. She downloaded a mobile game to play together with these 'kids'. In this case, the personal life of the older Elisa, who lives in another city, is markedly different from that of the younger kids living in her hometown.

However, the collective rhythm provided by the mobile game quickly facilitated familiarity and connection between Elisa and the younger children.

In general, this chapter discusses how an artificially created schedule by a mobile game facilitates players in developing a collective rhythm with each other. Mobile games play a role in creating a collective social life schedule, enabling people to share similar experiences beyond geographical boundaries. This shared experience can generate a group rhythm, which is particularly beneficial for developing weak ties.

## 5.3.3 Leaving, Escape, or Negotiation: The mobile game Rhythm Against Daily Repetition

This section explores how players' switching between mobile gaming and everyday life activities serves as a way to leave their current living environments. It considers mobile games as a tool for switching between digital and physical spaces that helps players negotiate the rhythms of their physical external environment.

Previous research has demonstrated that digital gaming can act as a means for players to negotiate with their external environment (Young, 2009). A typical example is gaming as a coping mechanism for real life. As Young (2009) discussed that:

Gamers immerse themselves into captivating virtual worlds that seem more exciting and interesting than their real lives. This often reinforces addictive behaviour and can be used as a coping mechanism to address missing or

unfulfilled needs. In this way, gaming can allow the gamer to forget his or her problems (p. 367).

For players, the game world contrasts with the real world and provides a safe and relaxed zone, even if this safety and relaxation are generally considered temporary and unsustainable.

Some interviewees in this research also illustrate how they adopt mobile games as a coping mechanism for the pressure or stress of daily life events. They used phrases such as 'changes my identity', 'empty' and 'go relax' to describe the effect of mobile gaming. For example, Bella stated the following:

'It changes my identity and environment. I go inside the game, pretend I am another person, and then live again in this game. It alleviates some of the negative emotions I encountered in real life. It gives me a chance to escape reality'.

—— Bella

Additionally, Elisa described that playing a mobile game 'is a time to empty myself'. Beatrice also mentioned her experience of playing mobile game outside the conference room, as following:

'When I stand in front of those conference rooms, I play inside the building with a self-evident purpose. It is to relax, and it still reminds me that I am about to do something that makes me nervous'.

—Beatrice.

Bella is a freelancer, and Elisa and Beatrice are college students. They have quite different daily lives. However, their descriptions—'alleviate some of the negative emotion' and 'a time to empty myself'—illustrate that mobile gaming helps users change their perception of the environment. By using the digital environment perceived through their avatars, players can partly or wholly replace their perception of the physical external environment.

Beatrice's case can further explain this phenomenon. For Beatrice, the motivation for opening the game seemed to be 'to leave the environment that was stressing her out'. Even though she still felt stressed within the game, the virtual environment created a barrier between her and the natural source of her pressure (the conference room and the university building). This shows that playing the mobile game creates a hybrid space that overlaps her original environment, allowing her to take a brief respite. Similarly, Elisa's 'time for empties me' helps her protect herself from the external environment. Mobile games work as a barrier to create a digital world where the player's spirit can reside. As mentioned in the previous subsection, mobile game rhythms can fill gaps in attention. In Elisa and Beatrice's cases, their 'emptying' and 'relaxing' should be viewed as autonomous allocations of attention.

Further, shifting attention from the physical world to the digital world can be regarded as a way to rebuild one's self rhythm. As mentioned in the previous subsection, Jaulin (1973, as cited by Lefebvre, 2013) differentiates between the 'rhythm of the other' and the 'rhythm of the self'. The former is defined as "the rhythms of activities turned outward" (Lefebvre, 2013, p. 95), while the latter is "organizing a time turned more towards private life" (Lefebvre, 2013, p. 95).

When interviewees use mobile gaming to adapt their self-perception, they negotiate with outward-oriented activities to rebuild their self-organised, private schedules. Before and after Beatrice turns on the mobile game, she must prepare her speeches, walk into the classroom and present her work in front of her lecturer and classmates. This series of actions corresponds to the 'rhythm of the other', built by a specific, regulated schedule. In contrast, the action of opening a mobile game and playing it outside the lecture room is an attempt to reshape the rhythm of the self.

Hence, the player's desire for mobile gaming can be explained as a wish to control or organise personal time and schedules. It is a motivation to rearrange personal time in response to the expectations of social roles or public relationships, a desire to shift from open, collective, prescribed public time to spontaneous, private, casual time. The mobile game serves as a medium or symbol of time for self-control, reflecting a personal inclination toward 'self' rather than 'others' at specific times.

With this argument, mobile gaming is not only a form of entertainment but also a negotiation or 'coping mechanism' (Young, 2009, p. 367) for dealing with a pressurised external environment. The reason for switching from a real-life environment to a virtual digital environment is more complex than merely pursuing entertainment and fun; it also involves negotiating between the rhythms of others and the rhythms of the self.

A rhythm involves repetition. Due to this feature of repetition, a rhythm is predictable by individuals. When a person is in a daily routine, it is reasonable to consider them as having the ability to foresee a series of actions or events

that will occur in chronological order. The series of events that repeatedly occur within a temporal cycle, and the environment or space in which this rhythm occurs, can be regarded as a rhythm-related system. Human subjects can be considered to choose, develop or switch between and negotiate with, various rhythm systems.

At a given point in time, players encounter a series of recurring events within a system by selecting it according to their own foresight. The events of everyday life are subordinated to the schedule of the player's local space, while the events of games are subordinated to the digital schedule. People's choice between virtual and real worlds stems from which set of schedules they prefer to work with at that moment.

Many interviewees mentioned that quickly switching between virtual and real life was one of the essential features of mobile games for them. Even though they describe it as 'escaping' or 'leaving', they are not leaving their original schedules but rather creating parallel rhythms of mobile gaming and everyday life. By creating this parallel rhythm, players can exist simultaneously on two rhythmic tracks. They can pause their perception of the physical environment by entering the track of mobile gaming while still remaining in the rhythm of everyday life. They exist both in the physical, realistic space of everyday life and in the digital space of mobile gaming, with their consciousness (or attention) able to switch between the two exceptionally quickly and interchangeably.

In other words, engaging with mobile games does not mean disengaging from the rhythm established by public social organisations (such as universities,

companies or city spaces) or family bonds. Instead, it creates a hybrid space that allows players to log in and organise their personal time within the gaps of already scheduled time. The development of self-rhythm in this overlapping hybrid space makes the presence of parallel, switchable rhythms possible, or even a common daily state for game players. As mentioned in Chapter 4.3, the automation mechanisms of mobile games significantly facilitate achieving this state.

If the switch from the out-game world to the in-game world can be regarded as a spatial transition, this switch can also be considered a 'pseudodisengagement' from the original rhythm. Unlike physical disengagement from geographical space, this 'pseudo-disengagement' involves freezing or suspending the original out-game rhythm or scheduled events. After gaming, a player can return to the original rhythm and continue following it. This 'pseudo-disengagement' shows the consultation and negotiation between the 'rhythm of the self' and the 'rhythm of the other', between the perception of the physical environment and the digital environment. As the interviewee Bella described, it also involves the identity of the self and the identity of 'another person' who lives in the game. These intersecting perceptions and experiences occur each time the player allocates part of their attention to gaming.

The process of 'pseudo-disengagement' reflects the human desire to break away from prescribed timetables, to experience time that is unplanned and unmeasured, outside the rhythm of cycles and linear progression. Lefebvre (2013) offered a romanticised interpretation of time beyond cyclic and linear rhythms, describing it as 'a time forget time', which is: "self-creation or of a gift

rather than of an obligation or an imposition come from without" (p. 76). The 'pseudo-disengagement' represents a departure from daily obligations, a rebellion against clock time and involuntary repetitive behaviour, a pseudoescape from old rhythms achieved by switching spaces through digital media. It can be described as a spiritual detachment achieved through digital media while the physical body remains within the constraints of a social schedule.

Amy's case illustrates why players choose pseudo-disengagement to negotiate with their daily life rhythm. As described in the introduction chapter of this thesis, Chinese employers, especially in the Internet industry, face a problematic overtime situation. During the interview, Amy described her life as follows:

My working hours are divided into the low season and the peak season. The peak season is recently, and it lasts for about two months. During the peak season, I wake up around 7:15 a.m., leave home at 7:45 a.m., go to work at 8:30 a.m., and finish the morning work at 11:30 a.m. During this period, I usually do not touch the game. Lunch break is from 11:30 to 1:30, and I may open the mobile game and clear the daily tasks for half an hour after eating. Afternoon working hours are from 1:30 to 5:00, then overtime until 8:00 or 11:00. During the evening break, I may check my smartphone after dinner, but not as often as at noon, and not every time I have free time after dinner. After I get home, before I go to bed, from 12:30 to 1:30, I might play a game, or if I am too tired, I will just lie in bed and watch [something] for an hour or ten minutes. When not working overtime, I will get off work before 6–7 pm, arrive home at 8:00 pm, and play mobile games from 9:00 pm to 10:00 pm or 11:00 pm.'

——Amy

Amy has only one rest day per week during her 'peak season'. She considers herself to be following the '996 working hour system', which refers to working from 9:00 am to 9:00 pm, six days a week). Within this working system, 'free times' is severely restricted in both time and space and often overlaps with other daily tasks. During Amy's lunchtime, she has only one hour of free time, and her free space is limited to a reasonably close distance from her company. This leads her to prefer leisure activities that are not constrained by time and space. Furthermore, to ensure she meets the requirements of her daily working hours, it is almost impossible for Amy to engage in energy-consuming leisure activities. Given these limitations, building a self-rhythm in a non-virtual way is challenging, making the parallel rhythm between physical and virtual spaces a necessity. In this parallel form of working and gaming, Amy can achieve more than four hours of 'free time' per day. She explained her motivation for building this parallel 'physical-digital' rhythm as follows:

The reason for playing the mobile game is that I do not want to fill myself up with work...Just makes me feel better if I do not feel like I am 100% occupied with work. This self-feeling feedback gives me a positive impact.

——Amy

For Amy, playing mobile games allows her attention to leave the work environment for short periods. Although the time spent playing mobile games differs from an idealised, unscheduled time, it still provides a break during her working hours. As discussed in Chapter 4, the cyclical or linear rhythm of mobile games is prevalent in daily gaming time. In-game actions such as

completing daily goals or collecting materials for characters are almost entirely repetitive. However, the ability to switch freely between the virtual and real worlds offers a selective 'freedom', reflecting the desire for time outside the imposed rhythm and control over one's own time.

The four hours of 'pseudo-disengagement' playing time allow Amy to feel 'not fully occupied with work' even when her physical body remains in the office. The parallel game-based rhythm provides players with the possibility to choose between the physical and digital worlds. This 'choice' represents a limited freedom, allowing for a self-selected life that temporarily transcends the physical body.

Similar to Amy, when many interviewees were asked about their reasons for continuing mobile gaming, their descriptions often reflected a coping mechanism for their out-of-game environments. When this thesis mentions 'playing' a mobile game, it refers to the interviewee opening the mobile game application and running it for a certain period. In other words, the act of 'playing' is not necessarily for associated with happiness but is a neutral and specific action. Compared to deriving fun from playing, the actual effect of the mobile game is immediate, instant access to the digital world, allowing players to engage in each moment they want.

In many cases, interviewees left their phones under their office or university desks, letting the game run using its auto-mechanism. Thus, mobile gaming is more akin to a symbolic or conceptual presence superimposed on everyday life rather than serving as a genuine source of leisure and relaxation. The method of playing is characterised by 'you can enter and play at any time', not

'you log in and play'. In this context, mobile games must be able to run independently in a simple linear model or near-automatic form to minimise the attention resources needed to enter or leave the game space.

Switching between the parallel rhythms of in-game and out-of-game life appears to be an auxiliary method for people to maintain balance in their lives. However, this balance can be fragile and easily disrupted. When players leave the external environment that drives their mobile gaming habits, their daily gaming rhythm may change to adapt to a new daily status. The interviewee case of Barton illustrates how everyday gaming rhythm can change quickly.

Barton is a university student who spent much of his time playing games during his military education<sup>43</sup>in high school. During this period, he used all his free time for mobile gaming. However, after completing his training, Barton almost immediately reduced his gaming time.

Military education involves group training activities requiring students to complete physical training courses in groups, representing a combination of cyclical and linear rhythms, or a complete and strict set of schedules for social life. In this environment, Barton showed a stronger passion for mobile games during his fragments of free time, as he explained:

"[During the period of military education] except for the time that must be spent completing training, [the rest of the time] is spent on playing games...

<sup>&</sup>lt;sup>43</sup> Military education in a China involves educational training for middle school, high school and college students, typically spanning several weeks. It includes activities such as standing, marching, formation drills ,etc.

What kind of person would play [this mobile game] when they have other things to do? [If] I am not in a closed environment; it is hard for me to play it'.

——Barton

In Barton's case, playing mobile games was a response to his daily life, which passively consisted of much cyclical and linear repetition. In his view, 'mobile gaming' and 'closed military environments' go hand in hand, and he preferred to play mobile games, spending all his free time on them. However, when Barton left his military training location, he immediately stopped playing mobile games. This indicates that the game content or its entertainment value did not interest him. Thus, mobile gaming was more of a means for Barton to actively negotiate with the part of life he had to face.

This case suggests that the stressful rhythm of the external environment may motivate participants to use mobile gaming to compete for control of their private time. Even if the participant is not particularly attracted to the game, mobile gaming can still be used to build a self-rhythm due to its convenience and accessibility.

In this context, players exhibit a paradoxical tension by adopting a series of digital repetition rhythms to negotiate with the repetitive rhythms of the physical world. Chapter 4 of this thesis argues that the rhythms of mobile gaming, characterised by high repetition and constant cycling, are cyclical or linear. This chapter shows that players also tend to use mobile gaming rhythms as a means of leaving the non-gaming environment.

With the information provided by the interviewees, it seems possible to infer that their experience with non-gaming routines—constantly cycling nongaming rhythms—leads to their tendency to choose mobile gaming as a filler for free leisure time. As mentioned, the interviewee Amy's work occupies almost the vast majority of her daily time. Within this strictly operating set of cyclical rhythms based on her company's work timetable, her unoccupied free time also flows in the same cycle as her work rhythms. In this context, the player's private time exists as the residue of the occupied work rhythm and cannot be considered a daily schedule organised and fully controlled by the player.

Another case is that of Morton, who changed his gaming method after becoming a father. When his daughter was born, to manage the menial tasks of childcare and family responsibilities, Morton shifted the majority of his gaming time from console gaming to mobile gaming. On the one hand, this shift demonstrates the flexibility of the gaming rhythm compared to nongaming daily life. On the other hand, it also suggests that although the mobile game rhythm is highly repetitive, it is likely the player's last option for remaining engaged in the game or digital world.

Game studies scholars Crawford and Gosling's (2005) work on the location and use of digital gaming in everyday life may indirectly support this argument. Their data showed that mobile games are more popular with women than men, possibly because women often have more continuous domestic responsibilities and mobile games require less time commitment. This difference in daily responsibilities may lead to different preferences for digital play. The cases of interviewees Barton and Morton demonstrate that their

preference for mobile gaming tends to adapt to the rhythm of the external environment and attempt to create a eurhythmia between the mobile game and the non-game environment.

As the interviewees show, the time they have to cope with stressful environments is often fragmented. For example, Beatrice stands outside the lecture room, Amy plays during her lunch break, and Barton uses his free time outside of training. After their personal lives have been ruled, occupied, and dominated by the rhythm of the other, mobile games fill in the time fragments between external events. These moments allow them to construct their time according to personal preferences, even though game time also involves cyclical and linear repetition.

This subsection analyses mobile gaming as a means for players to negotiate with the rhythm of their external environment. When people open a mobile game during fragmented moments of their day, the digital rhythm represented by the game merges with the rhythm of the physical environment, allowing quick transitions between the two systems. By engaging with the digital rhythms created by mobile games, players can facilitate or foster their selfrhythm by switching between the 'digital-physical' parallel rhythms in a hybrid space.

Participation in the mobile gaming rhythm can be regarded as a 'pseudodisengagement' from the behavioural standards imposed by the external environment. This pseudo-disengagement provides an opportunity to leave, escape, or negotiate with the current daily rhythm. When a player's physical body is present within the rhythm constructed by a company, university, or

public organisation, the pseudo-disengagement achieved through the game avatar demonstrates a spiritual presence in the digital space created by mobile gaming. From this perspective, daily mobile gaming behaviour may conceal a desire for self-control over one's time, especially when the participant's daily time is intensely occupied by repetitive routines.

## 5.4 Conclusion

This chapter focuses on the eurhythmia performance between the rhythm of mobile games and the rhythm of daily life. Mobile games provide a portable playing space, allowing players to access the game anytime, anywhere. This leads to the integration of physical and digital rhythms, as described in the above subsections. The hybrid game-reality space allows players to reorganise their daily lives and establish their self-rhythms within the game environment.

The three subsections of this chapter explain different influential aspects of how mobile game rhythms combine with daily life rhythms. In the subsection on mobile game rhythms as gap fillers, it is noted that people's daily rhythms experience attention surpluses. For instance, when interviewee James travels by taxi between his workplace and home, his physical activity is constrained by the limited space, but his attention is almost completely unspent. In cases like James's, mobile games can be inserted into everyday life during such moments. The rhythms of social life in the traditional sense (e.g. meals and commuting) are imposed only on the human biological body and do not occupy one's full attention, which can be reallocated to the mobile game system.
Due to the flexibility of mobile game time, game time can be inserted, connected, and run parallel to other events in people's daily schedules. In this context, the rhythm of mobile games constitutes a support and complement to daily life.

Therefore, mobile games can reorganise the fragments of daily time and attention. These fragments refer to the unused parts of time and attention during the player's adoption of the local non-game rhythm. A player can complete a series of actions in different intervals of daily life and combine them into an in-game goal, organising fragmented time and attention to the rhythms based on the game system.

The second subsection of this chapter mainly discusses the ability of mobile gaming to create a collective artificial social schedule. The mobile game's periodically updated content can be seen as an artificially created social calendar. By following this common calendar, people in different environments can achieve a collective rhythm. The interview cases of Eunice, Harold and Marko show that people may develop closer social relationships through synchronised playing rhythms. The experience of playing together allows them to mark the passage of time in their daily lives in the same way. They talk about new characters, activities, or events in the game simultaneously, and the overlap of game-based rhythms creates a hypothetical simultaneous presence. This imagined sense of commonality prompts players to develop a collective rhythm and positive relationships.

Compared to previous research, this thesis' interviewees reported experiences similar to those described in Apperley's (2017) study on the free-

to-play game GunBound<sup>44</sup>. In Apperley's study, Nadia, who lives in Melbourne, shared her experience of playing with friends in Malaysia, while another participant, Burgundy, mentioned playing with her brother in Adelaide. Apperley's (2017) work emphasises the globalised nature of video game rhythms, which allow people from different countries to play together. Building on Apperley's contributions, this thesis proposes two key developments: First, while this thesis focuses on interviewees who live in the same country and share the same cultural background, they also exhibit trends of building collective cross-geographic rhythms. Second, and more importantly, this thesis emphasises how mobile game mechanics influence players' collective rhythms. For example, interviewees Eunice and Harold described the significance of card-drawing moments, while Max and Darlene highlighted the importance of guild hours.

Whereas Apperley's (2017) rhythmanalysis of global game rhythms tends to position players as the primary agents constructing these rhythms, this thesis views the player-game ensemble as a mutually negotiated relationship. Gameplay mechanics, such as daily log-ins, daily tasks, and the production of in-game resources, play a significant role in shaping how players organise their daily gaming time. These shared online game rhythms, influenced by deliberate design choices, replace individual social relationships within a limited context and become the foundation for the collective rhythm among players. These players, though geographically dispersed and diverse in occupation, education, and experience, share a common rhythm shaped by the game's mechanics.

<sup>&</sup>lt;sup>44</sup> GunBound is a free multiplayer, turn-based artillery game in which players are divided into two teams and take turns shooting at each other.

The subsection on mobile game rhythm as a means of leaving, escaping, or negotiating discusses how people use mobile games to engage with a digital rhythm that parallels their local rhythm. It argues that the desire for free time beyond the daily routine rhythm may motivate people to play mobile games. The section describes the player's negotiation between the rhythm of the physical environment and the mobile game rhythm as a 'pseudo-detachment'. In this situation, the player does not actually leave their original rhythm of life in a geospatial context but rather suspends their participation in the local rhythm.

As illustrated in the cases of Elisa, Beatrice and Amy, the interviewees used the game to create small periods of time where they could freely choose between 'game' and 'reality'. This 'pseudo-detachment' time provides a positive mental effect for the player. This phenomenon was particularly evident in the case of Amy, for whom playing mobile games provided a sense of freedom from time constraints, meaning she was not 'occupied with work'. That feeling about herself was more important than the content of the game itself.

In this section, the rhythm of mobile gaming is seen as a method of negotiating with the non-game rhythm. The interviewee's action of placing their phones under the table can be regarded as an attempt to build a selfrhythm. In this action, they are not actually engaging with the game—the psychological feeling of being able to play the game replaces the content itself as the main reason for turning it on. By keeping the game open, people use the rhythm of mobile gaming to reshape and re-dominate their local life rhythms.

When people's lives lack a source of positive feedback, they may rely more on the positive feedback generated by mobile gaming. As many studies on game addiction have demonstrated, people are more likely to develop game addiction when they experience frustration and discouragement in real life. When mobile gaming is seen as a channel of negotiation and escape, it also means that people counteract the pressures of the natural world by switching to the virtual world, where positive feelings are readily available. This is precisely what happened in Amy's case, where she needed to use mobile gaming as a medium to briefly escape and counteract her highly regimented and stereotypical real life.

In conclusion, although this chapter examines the combination of mobile game rhythm and daily rhythm, mobile games do not always integrate seamlessly with other parts of daily life. Sometimes, interviewees also face rhythmic disorders. For instance, in one of our cases mentioned earlier, Andrew said that while engrossed in his mobile game, he did not realise that the food on his plate had gone cold. Hence, when Andrew focused entirely on completing the game and pulled his attention away, he ended up with a plate of cold noodles. This illustrates how mobile gaming can disrupt the rhythm of everyday life. However, when people struggle to control the negotiation process between mobile gaming and the rhythm of daily life, they are vulnerable to arrhythmias. The next chapter will discuss how arrhythmias affect people's everyday lives and examine the conflicts these rhythm disruptions create in the player's daily routine.

## Chapter 6 Arrhythmia, Discipline and Industrial Structural Conflicts in Muti-Rhythms of Mobile Gaming and Everyday Lives

### 6.1 Introduction

The previous chapter analyses how gaming and everyday rhythms combine and interact. The present chapter concentrates on how conflicts can arise between these rhythms. In addition to this introductory section and a concluding section, the main section (Section 6.2) of this chapter explores the categories of rhythm conflicts derived from interviews with 48 participants. Based on the interviews, I found that rhythm conflicts often occur in subtle nuances of everyday life. I divided the conflicts among rhythms into three categories: conflicts with biological and psychological rhythms, conflicts with social relationships and social organisational structures and conflicts with geographical space rhythms. These categories were derived from interviewees' descriptions of gaming time as 'unpleasant', 'stressful', 'tense' and associated with 'negative emotions' in their daily lives. I have coded these instances according to the nature of the rhythmic conflict and classified them according to the rhythmic subject that conflicts with mobile gaming.

Subsection 6.2.1 discusses the first category of rhythm conflicts which addresses conflicts between the rhythms of mobile gaming and the biological and psychological rhythms of the players' bodies. Humans naturally adhere to biological rhythms that their bodies produce, including activities such as eating, sleeping and periods of physical fatigue. When players are fully engaged in a game, they may neglect these natural off-screen biological

rhythms. Interviewees reported instances where they ignored their biological needs, such as sleeping less or skipping meals, to continue playing.

This subsection also explores conflicts arising from interviewees' psychological perceptions. As detailed in Chapter 4, daily in-game tasks are updated every 24 hours. Failing to complete them within their limited time means missing rewards, which some interviewees described as a strongly negative experience. Thus, some players developed diverse methods to cope with forgetting, such as setting alarms or developing the habit of checking their in-game task lists before going to sleep. For these interviewees, the perception of getting 'all things had done' in a mobile game can create a cyclical rhythm. When this rhythm is disrupted by other daily routines, it can lead to a rhythm conflict.

Subsection 6.2.2 discusses how the embodiment of immaterial and invisible norms, requirements, or rhythms in mobile games affects the player's physical body. Following Keogh's (2018) argument, this subsection analyses the player and game as an assemblage, where each constructs and influences the other and forms a 'player-game' relationship. This relationship is affected by a series of complicating factors, such as connections with NPCs, social norms within player organisations and interpersonal relationships among players. This subtle and complex relationship reflects various types of arrhythmia, which may be traced to the player's dual presence in both digital and physical worlds. Players must navigate and negotiate multiple rhythms (polyrhythmia), simultaneously managing their physical and digital existences. This negotiation often leads to polyrhythmic conflicts, with mobile games acting as a window through which these conflicts become apparent.

Subsection 6.2.3 investigates conflicts between geographical space rhythms and rhythms of mobile games. This subsection explores spatial contexts in which interviewees engage with mobile games. It mainly describes scenarios where players use mobile games in public spatial environments, such as buses, cafeterias and offices, that contain specific rhythms. These public spaces have their own inherent rhythms that match the characteristics of the location and the facility. For instance, bus stops experience a regular flow of buses, and traffic lights follow a set pattern of colour changes. Conflicts arise when the rhythm of a mobile game does not align with that of a physical location. For example, players absorbed in their game rhythms, may miss a bus, overlook subway announcements, or miss meal service in a canteen.

To some extent, this subsection parallels the previous discussion on mobile gaming in the 'gaps' of everyday life in Chapter 5. In Subsection 6.2.3, I continue to examine how people play games during moments such as subway commutes, breaks between lectures and mealtimes, with a particular focus on instances of arrhythmia.

Through these three categories, this chapter aims to explain the rhythmic conflicts between mobile gaming and out-of-game daily life.

#### 6.2 The Performance and the Reason for Rhythm Conflict

This section explores how interviewees experience different forms of rhythm conflict, categorised into three categories: (1) conflicts with biological and psychological rhythms, (2) conflicts with social relationships and organisational structures and (3) conflicts with geographical space rhythms. In

the following paragraphs, I analyse the performance and reasons for these rhythm conflicts through the experiences described by interviewees.

Some interviewees were aware of these conflicts and attempted to minimise or avoid them, such as quitting the game or describing it as torturous. However, most interviewees used more subtle language to describe their experiences. It is necessary to consider how the interviewees negotiate the sense of these conflicts through their choice of vocabulary during the interviews, as illustrated in the case of Zoe who stated,

'There are times I remember before I go to bed that I do not seem to have the game on today, and then I will insist and turn it on..... Nevertheless, it does not matter'.

—— Zoe

When Zoe used the word 'insist,', she implied a sense of conflict, resistance and a mixture of two rhythms. Her body needs rest; however, her decision was to play the game. Similar descriptions are common among interviewees whose gaming rhythms conflict with other aspects of their lives. They feel to be caught in a dilemma, even as they try to reassure themselves with phrases such as 'it does not matter'. In the cases of Zoe and other interviewees describing rhythm conflicts, it became clear that, as a researcher, I needed to be attentive and precise in identifying the nuanced and sophisticated lexical choices that reflect the context of conflict. Sometimes, these conflicts were also evident in subtle cues such as sudden pauses in the conversation, quiet laughter, or changes in tone of voice.

Demonstrating the implicit meanings in communication, especially when translating between languages, is challenging. Nevertheless, I sought to accurately convey subtle conflicts within the context of Chinese popular culture where mobile gaming is often viewed as a form of leisure. As a result, individuals might unconsciously downplay or avoid expressing conflicts between leisure activities and work or bodily rhythms.

When asked about their daily gaming routines, interviewees typically mentioned periods that did not coincide with work hours. However, their responses often included the workplace as a location for gaming when specifically asked where they played as they frequently mentioned words such as 'company' or 'office'. This discrepancy highlights a subtle contrast between responses to different questions, as illustrated by the following responses of the interviewees Max and Morton:

The Researcher : [A question about daily mobile gaming time]

Max: 'If I am busy, I play for an hour, and if I am not, I can play two to three hours before bedtime'.

Morton: 'On the way to work and on the way home, I will play for almost an hour, and another almost half an hour in the rest of the time'.

The Researcher: [A question about the space of playing mobile gameS]

Max: 'At home, at the company and occasionally on the subway'.

Morton: 'Occasionally, I log into the game at my office, when I am about to finish work or when I am already finished, but I do not play it'.

In Max's second response, the mention of 'company' alongside 'home' and before 'subway' suggests that he might have played games at his workplace

but might have overlooked or subconsciously ignored this scenario in his answer to the first question. Similarly, Morton's use of 'the rest of the time' in his first response subtly avoided specifying office hours, while in his second response, he emphasised three times ('occasionally', 'when I am already finished [with work]' and 'I do not play it') to assert that he did not play games during work hours. The responses of Max and Morton indicate a possible reluctance to acknowledge the rhythmic conflict between mobile gaming and their non-gaming schedules. This suggests that, as a researcher, I need to closely analyse and compare interviewee answers to track and understand these possible rhythm conflicts in detail.

### 6.2.1 Digital Feeding, Bodily Discipline and Disruptions

This subsection examines situations where the bodily rhythms of interviewees conflict with their gaming rhythms. A common theme in the interview data involves interviewees experiencing physiological needs such as eating and sleeping, while the game requires their attention simultaneously. The in-game demands can sometimes overwhelm the needs of the player's physical body. This phenomenon aligns with Apperley's (2011) argument of the body at the screen: "A body that may be ignored during play, as the linear time of gaming subsumes the cyclical demands of the body, causing pain, sleep, food, and bodily functions to be sidelined or even ignored entirely" (p. 37).

As Apperley (2011) described, the cyclical demands of the body off-screen can conflict with on-screen cyclical or linear demands within a game. Typically, the human body follows natural rhythms based on biological needs, where people feel tired at night, hungry or thirsty at certain times of the day, needing

rest after prolonged and repetitive activities. However, gaming rhythms often require players to log in at these exact times, leading to conflicts as illustrated by the experiences of the interviewees James and Sabina. As James stated:

'That is a moment when you really need a break, but you still turn the game on and think about getting the task done and going to bed'.

——James

In this instance, James experiences a contradiction in his desires. On the one hand, he acknowledged that he 'really needs a break'. On the other hand, he described that he still 'turns the game on and thinks about getting the task done'. This deviation between his verbal expression and his behaviours suggests a conflict between his body's natural biological rhythms and the demands of the game-based rhythm.

Similarly, interviewee Sabina provided a more specific description of her personal life:

'[I will] get up at 3 am or 4 am. I clear my stamina points before going to bed, then put on the alarm. [After the alarm rings] I clear my stamina points [again] and then go back to sleep'.

—— Sabina.

In response to another question, Sabina shared another example of rhythm conflict. She had planned a weekend trip to another city with a friend. Despite arriving at their local hotel at 1 a.m after flight, Sabina realised she had not completed her in-game tasks.

Sabina: 'I was still playing the mobile game at 1 a.m. because I was in a hurry to catch a flight, and it was after 1 a.m. when I arrived at the hotel.'

The Researcher: 'Can you give me a detailed description?'

Sabina: 'I went out with a friend on Saturday, took a flight, arrived at the hotel at 1 a.m., and then I played [a mobile game] for two hours.'

The researcher: 'how did your friend feel about that? Did she think it is strange?'

Sabina: 'Two people are clearing their daily tasks together.'[The Researcher's note: this means Sabina's friend was also playing a mobile game.]

In these situations, the interviewees' physiological rhythms conflicted with games' rhythms. James was tired, and his biological clock required rest, yet the game's rhythm did not align with this need and compelled him to continue playing. Similarly, Sabina disrupts her sleep to accommodate the game's schedule. Sabina explained that the game's stamina refills in the early hours of the morning. Thus, she sets an alarm for 3-4 am to maximise her in-game resources without wasting them. When the alarm woke her up, she completed the task and went back to sleep. This behaviour clearly illustrates how the game's rhythm dictates her actions, which is against the body's natural physiological rhythm.

In computer games, players are typically seated in front of a screen. Chesher (2004) presented a nearly cybernetic view of the relationship between video games and players, referring to the connection between the player's eyes and

the game screen as the 'glaze'. The glaze should remain balanced; if it is too tight, the player will be pushed away from the game, and if it is too loose, the player will lose interest. This concept is similar to the challenge difficulty curve in games, where players overcome challenges to achieve satisfaction. Chesher's (2004) argument suggests that video games can control or influence players' behaviour through their engagement with the screen.

Traditional games primarily focus on engaging players during a continuous period of play. In contrast, mobile games are designed to establish a pattern of time commitment. If the relationship between a 'game' and a 'player' in traditional games can be described as a one-way attraction, mobile games resemble a structured schedule that dictates when players engage and disengage. First, the automation of mobile games means that the player-game connection no longer solely depends on eye contact or even constant physical interaction with gaming devices. Second, the mechanics of mobile games are designed not merely to keep players continuously in front of the screen. Instead, these games often create situations of stamina exhaustion and resource depletion to prevent players from playing continuously. This dynamic mechanism encourages players to stimulate players' motivation, attention and engagement with the game and establishes a long-term habit of regular return.

In Sabina's case, after clearing her in-game stamina points, she typically sleeps for several hours until an alarm awakens her. She then logs into the game to complete her tasks. She may not constantly be looking at the screen or continuously clicking, but this routine has likely been repeated dozens of times. At this point, the in-game demands may begin to override her physical

body's needs, replacing an established sleep rhythm with a rhythm dictated by in-game mechanics.

In many instances, the connection between a player and a game system is maintained through in-game cyclical or linear rhythms. These rhythms operate independently of the player's physical body and individual environment, establishing a separate set of schedules. For the digital avatar or identity, players must complete specific actions within a set time frame or face penalties, such as losing in-game rewards. This task system for digital avatars mirrors the feeding process for the human body; players must log in at designated times to 'feed' their avatar or character by completing tasks to strengthen them. In this context, the repetitive nature of a player logging into the game and clicking the mouse is similar to a parent regularly preparing meals for a child.

In my research, I introduce the concept of 'digital feeding' to describe the regular and repetitive actions players perform to sustain or enhance their digital avatars. Unlike physical bodies, digital avatars do not perish if not 'fed', but their status within the game community may decline. This concept may be challenging for readers unfamiliar with online gaming. Specifically, if a player neglects to 'feed' their digital avatar, the avatar's data remains unchanged. However, as other players continue to 'feed' their avatars, the player's relative position of the neglected avatar deteriorates<sup>45</sup>. Consequently, players who do not engage in 'feeding' may permanently lose in-game rewards or content.

<sup>&</sup>lt;sup>45</sup> This deterioration can take a variety of forms. One common scenario is that the mobile game designer publish new content based on the current active players ability. For the players who don't feed their characters for a long period, they may not be able to responding to the new content.

In the material world, 'feeding' involves providing food to living beings to sustain their life functions. By contrast, in the digital world, 'feeding' refers to players supplying data generated by human activity to their avatars to maintain or enhance their character's activity levels. This term was derived from the natural language players use on mobile gaming forums, where they often describe the act of an in-game character consuming materials as 'eating' and the action of strengthening characters by providing materials as 'feeding'. During interviews, my interviewees unconsciously used similar terminology to describe their interactions with in-game characters, despite acknowledging that these characters are inanimate objects.

For example, when James felt the need to relax physically, his digital avatar also needed to be fed. Similarly, while Sabina's physical body needed rest, her digital avatar required feeding at 3 a.m. in real time. Regardless of Sabina's activities in the physical world—whether adhering to her usual routine or travelling with friends—her avatar needs be fed at least once a day according to the game's mechanics. As previously mentioned, mobile games update in-game rewards daily. Therefore, failing to 'feed' the avatar on any given day results in lost opportunities for character growth. This requirement creates a rhythm for the digital avatar that runs parallel to human biological body rhythms. For James and Sabina, this has led to arrhythmia due to the conflict between their natural bodily rhythms and in-game rhythms.

This phenomenon also connects to Lefebvre's (2013) concept of 'automatism or spontaneity'. He argued,

'We attribute what results from external constraints to an essential need. He who rises at six in the morning because he is rhythmed in this way by his work is perhaps still sleepy and in need of sleep. Doesn't this interaction of the repetitive and the rhythmic sooner or later give rise to the dispossession of the body?' (p. 75).

Human activities, shaped through repetitive execution, from rhythmic habits, but these habits often give the illusion of spontaneity. In the cases of James and Sabina, their actions against their body's natural rhythms were influenced by the needs of their digital avatars. However, these game-driven demands were almost invisible in their descriptions. As shown in the previous paragraph, players position themselves as the agents performing gamerelated actions—James states, 'You still turn the game on', while Sabina says, 'I clear my stamina points'. Players subconsciously internalise their voluntary participation. Their adherence to the game's rhythm, shaped by the constraints of its mechanics and coordinated with their personal schedules, is mistaken for spontaneous willingness to engage. This suggests a potential dispossession of the body in players' daily routines and highlights the risk of arrhythmia in the negotiated balance between gaming and life.

Additionally, psychological factors can also contribute to rhythm conflicts, as evident in the case of Valentina mentioned in Chapter 4. Valentina set 3 alarms—for 8 am., 4 p.m, and midnight—to align with her in-game earnings, which are billed every eight hours. She planned her login activities around this strict cycle, aiming for optimal resource collection. However, she was unable to consistently adhere to this schedule and could not log in every day. This inconsistency led to a shortfall in the in-game resources she expected to

collect, causing her negative emotions. This conflict intensified Valentina;s Valentina's experience of arrhythmia.

Valentina eventually moved on to a different phase of her gaming experience, as she said:

'I am almost done with town-building, I am getting to full level, I have four towns and [each town has] pretty much everything laid out, and then [I know] rarely have the energy to care about its [in this case, the in-game town] placement or production'.

——Valentina

At this stage, Valentina's negative feelings towards the game had largely subsided, mainly because she no longer felt a strong need for in-game items and thus did not strictly plan her in-game and out-of-game behaviours. Gamers' negative emotions often stem from losing access to attributes, items or game characters, either now or in the future, due to failing to achieve their planned behavioural goals.

Players often plan their gaming activities over extended periods since games typically set long-term objectives. For example, in Genshin impact, acquiring a newly released character might cost about 16,000 in-game currency. Without bonuses from in-game activities, players earn approximately 60 in-game currency per day by completing daily tasks. With a monthly pass, this increases to 150. Therefore, if players desire a new character but do not want to pay more, they need plan their gaming behaviours for at least a month, if not longer. These long-term goals motivate players to organise their daily in-

game behaviours meticulously. The daily planning process can be considered as a cyclical rhythm of psychological experience. As in Valentina's case, expecting to receive in-game income every eight hours becomes a regular, cyclical and repetitive experience.

In the context of the freemium model, the overwhelming rhythm of the human body may be linked to the monetisation of player performance. The cases of James, Valentina, Sabina and others discussed in Chapter 4 illustrate that daily mobile gaming often resembles economic behaviour involving the exchange of symbolic currency (Paul, 2018). This observation aligns with Voorhees' (2015) analysis of Homo Economicus, which suggests that players tend to train their bodies to "navigate the possibilities enabled by the material and cultural milieu in search of the most advantageous probabilities" (p. 68). He suggested that players weigh the costs and benefits of each potential ingame action, seeking to maximise in-game revenue. For instance, James' desire not to miss his daily tasks and Sabina's wish to use stamina points efficiently may lead them to ignore or temporarily sideline their body's natural rhythms.

This concept also connects with Brock's (2021) argument about the affective power in video games, inspired by Beer's (2016) notion of 'metric power'. Beer (2016) proposed that measurement and calculation influence people's lives by ranking, comparing and evaluating daily behaviours. Brock (2021) extended this idea to video games, arguing that such measurement metrics align players' behaviours with the game's value systems. In the context of mobile gaming, the process whereby players engage daily to exchange their actions for game bonuses can be viewed as a method for mobile games to

link the number system with players' emotions. Gaining reward points through a disciplined, accurate and regular routine (such as logging into the game every eight hours) is perceived as a gain while missing a daily task is perceived as a loss. This system of rewards and penalties enables the degree of compliance with in-game schedules to be reflected in the player's account promptly and clearly through the in-game currency system. It also links the player's physical, out-of-game actions to their game account. Essentially, mobile gaming uses in-game currency to reward or punish players for their out-of-game behaviours, which can be understood as a form of monetising players' everyday physical activities.

Furthermore, the interviewees in this thesis described their reactions to missed daily tasks or events in terms that were more affective than material. According to the interview data, players are particularly more sensitive to the rewards they miss or are at risk of losing, often using phrases such as 'avoid loss', 'not want to lose' or 'deficit' to describe their motivation to maintain the in-game cyclical rhythm. These negative emotions and sensitivity might be attributed to loss aversion (Kahneman and Tversky, 2013), which suggests that people are generally more affected by losses than by equivalent gains. This can be illustrated by respondses from two interviewees, Paddy and Jessica:

'Seems like losing something. It feels like that'

——Paddy

'It would be a waste if the daily tasks were not done, and I have not received the rewards. That is a typical mentality for me. I need to finish my stamina. I am losing out if I don't'.

—Jessica.

According to Kahneman and Tversky (1979), loss aversion means that people are more likely to be affected by a given loss than they are by an equivalent gain. In gaming, missing out on a reward often feels more like a loss than an opportunity for profit to players. Hamari and Järvinen (2011) found in their research on popular social games such as FarmVille that losing crops that had been planted feels significantly worse than the pleasure of gaining an equivalent amount of crops. Players view the effort put into planting and the value of seeds as sunk costs, motivating them to return to the game to avoid losses. The interviewees' descriptions, such as Paddy's 'losing something' and Jessica's 'losing out if I don't', reflect a similar motivation.

However, these descriptions often present the negative experience of loss as a vague and ambiguous feeling. This ambiguity may arise because rewards in mobile gaming are digital items without a clear physical entity or explicit monetary value. Interviewees frequently used terms such as 'as if', 'seemingly' and 'possibly' to qualify their feelings about missing daily tasks, despite not actually losing anything tangible in a mobile game but merely failing to gain a reward for completing a task. In other words, the opportunity to complete daily tasks in exchange for rewards is perceived as a positive resource by the interviewees, a perception possibly influenced by the interaction dynamics between players and mobile games.

Mobile games often provide in-game rewards for players' actions in a more concise, straightforward and quantifiable manner than what is typically found in everyday life. The daily task system presents a list of possible rewards and the actions required to obtain them, which are generally straightforward. These actions may involve tapping on a specific area of the screen or allowing the game to run automatically for a specific time. As discussed in Chapter 4, the daily task system is a collection of repetitive behaviours. Players may perceive these repetitive behaviours as an easy way to earn in-game rewards, viewing them as a guaranteed benefit.

However, the cost in terms of time and attention that players invest in these daily tasks is often less apparent. Unlike the repetitive in-game actions, these less visible costs—such as the time commitment and mental attention required—are closer to what mobile games fundamentally demand from their players. Players may equate the simplicity of completing tasks with the ease of obtaining rewards, overlooking the implicit cost of organising their daily schedules around these tasks. Additionally, mobile gamers often have the option to use out-of-game currency to reduce the time commitment required for gameplay. This involves a conversion mechanism between in-game and out-of-game currencies. Players can earn in-game currencies or virtual items by completing daily in-game goals (a combination of cyclical and linear rhythms, as mentioned in Chapter 4). These in-game currencies and items can be purchased with real-world currencies.

In order to encourage players to make purchases, mobile games often increase the labour and time required to obtain desired items. For interviewees such as Jessica and Sabina, logging into the game is primarily

about acquiring virtual items at specific times rather than 'playing' the game in the traditional sense. These virtual goods act as mediators between the players' actual demands and their repetitive behaviours, serving as a metric to measure the cost of the players' efforts. The perceived value of these items is strongly related to the interviewees' subjective judgments. This means that when they place a higher value on certain game items, the time they invest is more likely to conflict with their real-life rhythms.

Gal's (2006) explanation of psychological inertia offers another perspective on the interviewees' behaviours. Gal viewed loss aversion as merely an auxiliary principle in everyday life. He proposed the 'psychological law of inertia' to explain decision-making behaviours, which he described as: "A person will tend to maintain the status quo unless impelled to alter the status quo by a psychological motive to do so" (p. 25). When people's preferences are not clear, they are more likely to opt for maintaining the status quo.

In the context of mobile gaming, the daily cost of playing is relatively minor—it is not a choice between large blocks of personal time or significant amounts of in-game currency, but rather a series of decisions involving small amounts of time or minor in-game resources. Initially, the time players invest in mobile gaming could be exchanged for substantial amounts of in-game currency. However, as time progresses, the efficiency of this time-to-currency exchange diminishes, eventually stabilising at a relatively low rate. Moreover, because mobile games can be played in short time fragments, the cost of playing time is relatively low for each session. This miniaturisation of decision-making might make the players' preferences less distinct, while the daily gaming habit facilitates the maintenance of this state of mental inertia.

Hence, game players may adapt their daily lives to the rhythms inherent in games to achieve a game-life rhythmic fusion and co-play. Once this fusion occurs, players are likely to maintain this rhythm. Typically, the first week of mobile gaming—often referred to as a 'seven-day event' in many games—serves as an introduction for new players. This period helps them become familiar with the gameplay and teaches them to log in at intervals expected by the game. This introductory phase of playing, teaching or training can be regarded as establishing the in-game rhythm or the norms of the game. Through the provision of relatively abundant in-game rewards, including rare characters or materials, the mobile game sets standards and models for 'how to play' and 'when to play'.

Moreover, the in-game rhythms built during this initial period also lead to the formation of out-of-game daily rhythms that include regular gaming. Furthermore, much of the subsequent daily playtime<sup>46</sup> tends to repeat and extend the content introduced during the initial seven days. Players continue to engage with familiar game content, reinforcing and internalising these behaviours through constant repetition. This process can lead players to become habituated to the rhythm and develop a tendency to maintain it.

The concept of constant rhythm inertia in mobile gaming refers to the phenomenon where players repeatedly perform the same series of in-game actions daily, which can be viewed as a trait of rhythmic self-maintenance or inertia. This suggests that players may prefer to maintain the continuity of this rhythm once it has been established. Several interviewees indicated that they

<sup>&</sup>lt;sup>46</sup> In this context, 'daily play time' refers to the repeat daily or weekly tasks, which update regularly, excluding new events, maps or content.

tended to abandon mobile games when external factors prevented them from completing an in-game task. For instance, the interviewee Werner described his intense negative emotions upon failing to complete a daily task as 'very serious', leading him to say, 'I generally choose just to go ahead and remove that [in my life] at times like this'.

In Werner's case, he discarded the mobile game when his daily gaming rhythm was disrupted by other daily events. As mentioned earlier, the gaming rhythm refers to a set of guidelines for 'how to play' and 'when to play'. When this norm, established by the mobile game's mechanics or design, is disrupted by other factors such as urgent work or social events, some interviewees, like Werner, reported uninstalling the game after missing daily tasks. For example, another interviewee James stated:

'There is a kind of consequence to disconnection. It could be that it is over when it has disconnected. As I mentioned, inertia wears off, and then you may slowly give up and you are not so hooked'.

——James

James described his 'failure to complete the in-game tasks' as a 'disconnection.' According to James, this disconnection signifies a break not only in his connection with the mobile game but also between his current and past actions. Furthermore, this disconnection represents a psychological shift from being 'hooked' to deciding to 'give up.'

This notion of connection and the 'inertia' mentioned earlier collectively suggest a system of self-constructed rhythms by the game player. The

'disconnection' described by James relates to Lefebvre's (2013) argument about the fragile construction of everyday time, which he described as follows:

In everyday life, what is relative to social relations thus appears to every 'subject' as necessary and absolute, as essential and authentic. Were we to introduce a new element into everyday time, this construction might totter and threaten to collapse, so showing that it was neither necessary nor authentic. (p. 75)

Based on this perspective, this thesis examines how interviewees arrange their time around mobile games. For instance, Valentina logs into the game every 'eight-hours', while Werner and James aim to complete the refreshed daily tasks every 24 hours. These patterns, characterised by the regular repetition of specific behaviours within a particular period. form a daily rhythm constructed by the players, which they may perceive as "necessary absolute, essential and authentic" (Lefebvre, 2013, p. 75)

The misperception of the necessity and authenticity regarding mobile gaming daily rhythms may cause dispossession of the bodily and biological rhythms. As players adhere to the time-consuming rhythms established by mobile games' daily tasks, they may internalise these externally constructed needs as regular psychological experiences. Evidently, mobile games do not impose any strict or punitive measures on their players; anyone can uninstall the game from their smartphones at any time. As Lefebvre (2013) mentioned, the collapse of established daily rhythms often occurs when 'a new element [is introduced] into everyday time'. As my interviewee showed, some unexpected daily events can disrupt their habitual gaming routines. This disruption can

lead to the realisation that the original daily gaming rhythm was 'neither necessary nor authentic' (Lefebvre, 2013, p. 75), resulting in a 'disconnection' from the old routine. However, this process often involves strong negative experiences, as described by Werner, who felt it was 'very serious', or by Jessica, who experienced a sense of 'losing out'. These reactions imply a rupture in the cyclical psychological experience, a break in rhythmic continuity and potentially the cost of reconstructing a new daily rhythm.

From this discussion, mobile games may develop a system of measuring and evaluating player participation based on symbolic currency, and the number system turn serves as a motivation for players to maintain the linear and cyclical rhythms associated with the game, whereby players' management of in-game data and performance extends to managing their bodily behaviours outside the game. No matter the desire to 'maintain the status' or an expectation of avoiding loss, a series of rewards and penalties—often in the form of in-game currencies—can play a role in sustaining the player's daily gaming behaviour. At the same time, this pattern of maintaining participation in daily in-game rhythms may lead players to ignore or suppress their own biological rhythms temporarily.

In summary, this section describes the conflict between the biological and psychological rhythms of the human body and the digital rhythms of avatars in the game. The next section explores the conflict between the human in-game and out-game social relationships and the rhythms of the mobile gaming community.

# 6.2.2 Player-Game Assemblage and Dual Presence in Digital-physical Worlds

In this research, social rhythms refer to interviewees' regular rhythms of social activities in daily life, such as going to work and meeting with friends. According to Margraf et al. (2016), a social rhythm "refers to the regularity with which one engages in social activities throughout the week" (p. 1). This subsection explores the dual roles individuals play in both in-game and out-of-game contexts, and both of them produce social rhythms. At the beginning, the data from interviewees' responses indicate that taking an in-game role does not necessarily require them to log into the game. For players, the psychological crossover between their roles can occur at any point in their lives. The out-of-game and in-game roles can coexist in a player's life. The evocation of a role may be triggered by a symbol with a particular meaning or any sentence with an associative meaning. This can be illustrated by the experiences of the interviewees Harold and Gloria:

Harold, an international student, was walking along the Thames River when he saw the Queen moored in the harbour. This sight reminded him of a ship he owned in his mobile game, despite no physical similarity between the two ships. He described it as a 'conceptual resemblance' rather than a 'figurative resemblance,' where the virtual and the real objects intersect in one's perception of the same concept. For a moment, Harold was both a tourist strolling along the Thames and a commander with a considerable fleet group in a mobile game.

Gloria is a player in a love simulation game. Her favourite in-game character is named Li Zeyan. On the birthday of this character, she went to a restaurant, ordered a pudding<sup>47</sup>, and placed a picture of the character beside the pudding before taking a photo. In this case, as Gloria sat in her seat at the restaurant, she was both following her out-of-game social role for interacting with the restaurant staff and also playing her role within the game.

These interview cases align with Keogh's (2018) argument on player embodiment: "I am in this collection seems to flicker between bodies and between worlds. I am both lota and myself; I am both here and there" (p. 2). This statement highlights a complex and dynamic player-game relationship, where players are shaped, even when they are not physically engaged in mobile gaming. It also reveals the player's simultaneous dual existence in both the physical and digital worlds.

In these overlapping and complex relationships, players may need to play dual roles that correspond to different environmental demands. For instance, in the case of Gloria, she exists both as a lover of Li Zeyan and as a customer in a restaurant. In this perceived overlay of reality, the in-game and out-ofgame roles may simultaneously impact a person's behaviour. From the perspective of this thesis, these expectations, demands and inertia from the surrounding environment can be explained as rhythms emanating from both digital and physical worlds. This subsection focuses more on the conflicts arising from the player's dual existence, which can also be understood

<sup>&</sup>lt;sup>47</sup> Li Zeyan used to make pudding for the player's avatar in the storyline of this romance simulation game.

through Lefebvre's (2013) concept of arrhythmia as a "discordance between rhythms, being 'out of step'" (Lyon, 2019, p. 25).

A common example of arrhythmia, or rhythm conflict between the in-game role and the out-of-game role, occurs during players' lunch or dinner times. This is often because mobile game mechanics, such as timed logins and event schedules, overlap with players' daily routines outside the game. For example, the interviewee Werner described that:

'That is a set time for players to free up their time. It [mobile gaming event] might be from 6:00 to 8:00 in the evening, and at 8:00, I might have to work late or have dinner, and there is no way I can actively free up that time for it... I do not like the game design anyway, but I will try to participate if I can because there will be a big reward or more something enticing, although I cannot remember what those rewards were'.

—— Werner

In instances like this, Werner needs to navigate the conflicting rhythms of his mobile gaming and real life simultaneously. For example, from 6:00 to 8:00, his social role as a company employee requires him to stay at work, while his in-game avatar requires him to log in to the game. These two rhythms collide at specific points in his daily life.

Like the beat of a drum, a rhythm usually involves several cyclical moments and sets the expectation that a person will behave in a certain way at a particular time. As previously discussed, mobile games often require players to log in at certain times and perform routine behaviours. This concept can be

linked to Goffman's (1949) dramaturgy theory, which suggests that people adopt different roles on the 'front stage' of everyday life based on the expectations of their audience.

In this analogy, everyday life scenes can be likened to a theatrical front stage. where individuals perform roles according to societal expectations. Similarly, mobile games impose their own set of requirements on the player's avatar. Most mobile games assign specific titles to players, such as Traveller, Master or Captain, each suggesting a distinct code of conduct and behavioural rhythm distinct from real life. At any given time, the different roles of in-game and out-of-game personas may demand different actions from the individual.

Players are likely to experience arrhythmias when the rhythms required for their mobile game avatars conflict with those of their real-life roles. For example, Werner's negative emotions arise when his desire to engage in gaming activities conflicts with the constraints imposed by his out-of-game responsibilities. This dissonance between his dual rhythms motivates his negative emotional responses. Some interviewees chose to prioritise the rhythms of their out-of-game roles over a game's rhythms, as demonstrated by interviewees Alexia and Rex:

'If I cannot get home until eleven o'clock that day, I will probably take a shower and go to bed'.

----- Alexia 'I will never give up any event in reality life for playing games, although there is something I will miss in the game, such as I need to take an exam,

and I will uninstall the game. Of course, this may lead me to miss the new in-game characters'.

—— Rex

According to Alexia, she typically completes in-game tasks unless work or fatigue intervenes, which happens about once or twice a month. Similarly, Rex opts out of gaming during exam periods. These interviewees prefer to adhere to their primary out-of-game roles as workers or students, especially when these roles conflict with their in-game activities.

In addition to those who focus on out-of-game rhythms, some interviewees attempt to achieve a rhythmic accommodation or negotiation; they negotiate and seek a balance between their digital and non-digital identities. For instance, Jessica plays tabletop games with her friends on weekends. While engaging in table games, she also keeps her phone open to participate in activities within the mobile game. She described how she manages this dual engagement as follows:

Jessica: 'There is a situation where it is easier to disturb each other. I sometimes play mah-jong (a traditional Chinese table game) while I open my mobile phone, but this can interfere with my mah-jong. However, there is no choice but to manage both simultaneously, sometimes I might make a mistake or miss something in mah-jong'.

The Researcher: 'Okay, but why do you want to do that?'

Jessica: 'There may be an in-game event that takes me 6-8 hours to get involved that day, or the in-game event may be ending today'.

Jessica tries to accommodate both her social activities with friends and her digital gaming commitments, although this dual engagement may lead to errors or distractions in either activity. In Jessica's case, her engagement with mobile gaming significantly impacted her participation in real-life activities. While playing Mahjong—a game where players wagered a small amount of money as a reward for the winner—Jessica began making mistakes and losing money. Eventually, she decided to invite another friend to take her place at the table so she could focus on completing her tasks within the mobile game.

During the game, Jessica found herself attempting to fulfil two roles simultaneously: her physical, out-of-game role as a friend participating in a table game, and her digital, in-game, and her in-game role as a 'Master (avatar)' leading her 'Servants(no-player characters)' to complete an in-game challenge. While she physically participated in table games with her friends, her digital avatar was on a separate schedule, requiring her attention to complete an in-game event.

Similarly, Harold described his experience in managing dual roles during work hours:

Harold: 'So when I open my smartphone at work, I usually tilt my phone down against my side to keep people from seeing my screen. Furthermore, I am afraid the leader will suddenly spring up behind me'.

The Researcher: 'So you look at your phone, and then play for a few minutes and then look around to see if the leader is looking at you?'

Harold: 'Well. Be ready to quit the game at any time'.

At this point, Harold faces two distinct rhythms: his professional responsibilities at the office, where colleagues expect him to focus on his tasks, and the rhythm of his mobile game, which requires him to log in and complete in-game tasks.

The in-game rhythms and out-of-game rhythms are equally important when analysing a player's daily performance. In the cases of Jessica and Harold, the out-of-game rhythms—such as the turns at the Mahjong table or the watchful eyes of a supervisor—establish expectations based on social roles, while in-game rhythms dictate the expectations of digital roles. This duality corresponds to Lefebvre's (2013) notions of external and internal rhythms. These multi-rhythms—both external and internal, personal (secret) and public—occur simultaneously within the same individual and shape the 'present' experience of the player. This current and present rhythm suggests a series of invisible orders that may include social friendship norms, workplace rules, and the expectations of a 'good player'. These multiple, implied life orders collide at the moment when the digital and physical worlds intersect through mobile gaming.

Returning to the argument in Section 5.3.3 about mobile games and gaming rhythms as a form of resistance to other daily routines, the conflict between ingame and out-of-game rhythms also represents an implied conflict between dual orders or social roles. A player's desire to escape the rhythm of their current physical environment also indicates a reluctance to conform to a persistent social order that extends beyond the constraints of time and space.

The out-of-game rhythm, as a symbol and representation of daily order, becomes a specific rule of action or performance that can be challenged. The collision of in-game and out-of-game rhythms allows for the identification and recognition of the player's desires and motivations.

In this thesis, arrhythmias are considered a result of the conflicting roles that players are expected to perform within dynamic and complex player-game relationships. These involve the development and negotiation of a player's identity and norms within a complex and varied hybrid space. Keogh (2018) presented the idea of a symbiotic relationship between the player and the game, which defines a dynamic and ongoing 'player-game' relationship. This relationship shapes and produces the presence of the player. In the context of mobile gaming, this 'dynamic and ongoing' relationship involves not only interactions between digital mechanisms and the physical body but also encompasses a broader external world, including the player's occupation, social relationships and social status.

First, this symbiotic relationship includes the player's connection to in-game characters, which may be their avatar or other characters depicted as having specific relationships with them. Southerton and Tomlinson's (2005) analysis of interview data from the Health and Lifestyle Survey in the UK suggests that their interviewees experience time disorganisation due to a sense of obligation to others. Their research focused solely on people-to-people and people-to-social-organisations interactions. In other words, this sense of obligation is only relevant to real, living individuals. However, interview data showed that players may also feel an obligation to in-game, inanimate characters. For example, as previously described, an interviewee chose to sit

in a restaurant to celebrate a game character's birthday. Further, some interviewees displayed emotional connections to these game characters or attached some degree of affection to their in-game characters. This emotional connection transforms expectations from in-game characters into significant factors that influence player behaviour.

Many interviewees perceived their characters as existing somewhere between computer programs and living organisms. As characters upgrade and grow, they may learn new skills and undergo changes in appearance, or the towns within the game may evolve over time in response to player behaviour. The game's interactive elements can provide output responses based on the player's input, which can lead players to perceive these characters as more than just non-living entities. This interaction can foster a process of emotional connection, as described by the interviewee Rowena:

'At first, I saw the character, and I just thought his experience was more twisted, bumpy or whatever. And I didn't think much of it, but maybe after a dozen chapters of the plot, there's a sudden breaking, that makes you feel too abusive, too high or too shocked. Then, suddenly, you could not realise where you are anymore: there's no way to realise all of this is just a game'.

— Rowena

The feeling that 'no way to realise all of the this is just a game' may lead players to confuse reality with the game and feel a sense of obligation towards in-game characters. Southerton and Tomlinson (2005) introduced a participant named Ashley to analyse how a sense of obligation is developed. Ashley felt compelled to adjust his schedule to keep an appointment with his

brother. This sense of obligation might stem from two sources: the belief that his brother may be less helpful if he does not show up, and the concern that his actions may hurt his brother's feelings. A similar process for producing a sense of obligation can develop in mobile gaming. For example, players may feel compelled to log in on time to complete tasks to avoid missing in-game rewards. Moreover, some characters might express feelings of longing for the player if they have been offline for a long time. While these game characters and objects are non-human in terms of their compositional material, they are designed to exhibit near-human reactions, which may further fostering this sense of obligation.

In Murdoch's (1997) description of the actor-network theory (ANT), he notes that science and technology are rapidly incorporating non-human objects into relational networks. From the ANT perspective, both humans and non-human objects create a symmetrical network of relationships. This network allows for the establishment of relationships between people and objects that were traditionally considered exclusive to humans. Fuchs (2014) introduced the concept of fictional empathy which is characterised by an 'as-if- consciousness'. The 'as-if-consciousness' emphasises an existential awareness that maintains the boundaries between the real and virtual worlds while maintaining an emotional connection. In this thesis, the 'as-if- consciousness' is evident in the playing process of most interviewees, indicating that players may develop a sense of obligation to the non-human entities of game characters and systems, although they are aware that these are virtual.
For players, game characters and the game world provide predictable interactions. These interactions inherently form a web-like network within the game, involving character upgrades, item acquisitions, story development and other elements. These elements are interconnected with the player to varying degrees within this network of interactions. This network provides a conceptualised simulation of the physical world and establishes the basis for player empathy towards virtual characters.

The interviewed players described how the game system translates its demands into personified expectations conveyed by game characters. One interviewee, Jessica, briefly explained this expectation from the system:

'It [the game character] presumes to make me a tool within the game too'.

—Jessica

This statement reveals that Jessica, as a player, is aware of the expectations the game character sets for her behaviour and understands how to respond to them. However, she feels that complying with the in-game character's wishes makes her feel like a tool of that character. Responding to these expectations may lead the player to change their intended course of action during gameplay, potentially consuming more time than initially planned. Such changes can, under certain conditions, lead to arrhythmias in one's life.

Based on the mechanics of mobile gaming discussed earlier, it can be inferred that mobile game players may perform repetitive actions driven by their emotional connection or empathy towards game characters or in-game relationships. This emotional connection motivates players to meet the

expectations of non-player characters, typically through a series of repetitive in-game actions, which correspond to the linear or cyclical rhythms explored in previous chapters. However, these emotions are often rooted in virtual contexts and conditions. The essence of arrhythmia experienced by players may stem from dissonance in their feelings towards game characters in the dual virtual-physical world. This conflict might arise between perceptual and rational perceptions, or between the player's imagination and the mechanical nature of the game character.

Second, the 'player-game' relationship may also be influenced or shaped by pre-existing community norms among players. Mobile game mechanisms, such as guild wars or team tasks, often lead players to self-organise and establish common standards or norms for cooperation. These norms then create a rhythm that is imposed on the player. Taylor (2006) observed this phenomenon in her study of mods (player-made programs) in World of Warcraft. She described a damage meter used to measure players' performance within the game. Taylor (2006) viewed this measure as a form of 'participatory surveillance' tool. This tool, initiated and adopted by game participants, helps to quantify human in-game behaviour. By measuring and guantifying human behaviour using various tools, a set of criteria for what constitutes a 'reasonable' and 'good player' is established within the game space. In the context of mobile games, the criteria for being a 'good player' extend beyond just the amount of damage dealt or the accuracy of skill deployment. They also include whether players can log in at the right time and participate in battles or other cooperative activities according to the organisational schedule set by player self-organisation management.

According to the interview data, a common method for establishing in-game norms is through game guilds, self-organised groups created by players, usually comprising dozens of members. In many mobile games, certain gameplay elements are directly linked to participation in a guild. Players who do not participate in a guild on these games may miss out on significant game rewards. While a mobile game guild may not influence all players, it can affect the in-game rhythm and potentially lead to arrhythmia for those who choose to join it. Many interviewees expressed reluctance about joining guilds, with some citing reasons such as 'social phobia.' However, they felt compelled to join these player organisations to obtain in-game rewards. To facilitate communication within these guilds, social media chat groups are essential. The in-game communication systems are often inconvenient. Players frequently rely on external gaming communication software to stay connected. In China, this is typically Tencent QQ or WeChat.

Internal rhythms within guilds and the relationships among guild members do not always involve surveillance or oppression. Most interviewees reported that relationships within guilds tend to be superficial. While some guild members occasionally post non-game-related messages—such as complaints about exams or concerns about work—only a few participants share their personal life details with their guild members. The most common situation reported by interviewees involved joining a game guild and its chat room but remaining silent, while still partake in guild activities within the mobile game. Nevertheless, some interviewees reported developing deep relationships with their guild members.

Among the interviewees of this thesis, Clyde is a guild president in a mobile game. The guild he manages ranks in the top 10 in the game. Generally, he communicates with his guild members through social media platforms outside of the game. He mentioned that his main mobile game. Princess Connect, heavily depends on the guilds, specifically on the interplay between different groups of players. The game typically holds guild wars at regularly date, usually starting on the 23<sup>rd</sup> of each month. At the end of these guild wars, guilds are ranked based on their performance, with the game system awarding substantial rewards to the guilds based on how well they performed. Clyde noted that if a player does not participate in guild wars, there is little else to engage with in the game.

The cooperative model in mobile games differs from that in traditional PC games. For example, in World of Warcraft, the player-to-player combat system operates in real-time, with all players engaging simultaneously. In contrast, multiplayer online cooperation in mobile games is often asynchronous due to device limitations. In the guild wars of 'Princess Connect', the member attack sequence in challenges is crucial. If guild members aim for a high ranking, they must challenge bosses in a specific sequence which had been arranged in advance based on each participant's characters ability. The guild's management is responsible for setting a detailed schedule specifying which member challenges which game boss and at what time.

Clyde mentioned that his guild does not strictly manage membership. This guild typically only controls the last person who kills the boss for win the reward time of the next challenge. However, they still find it necessary to send

reminders messages via social media to ensure members log in and participate in the attacks. The following paragraphs show the talking text between the researcher and Clyde:

The Researcher: 'Have you ever been called to fight with the boss at a time which is inconvenient for you?'

Clyde: 'Me? No...as you know, I am the president of this Guild...sometimes the president has a small privilege'.

Clyde's use of the term 'privilege' suggests an awareness of how the guild system can impact a member's daily life. This is further illustrated in another part of the interview where he described a common issue faced by guild members:

'I remember that sometimes I asked some members in the Guild to attack [the in-game boss], and they told me, no, they had used their chance to go to the bathroom in the morning, and they would only have another chance in the afternoon. The attack for each person to the BOSS takes about one or two minutes, but it requires [the players] to be very serious, so that meant they can not be relaxed. It just means they need to find a way to go to the toilet from the company seat, then attack [in the game] in the toilet, and then come back [to the company seat] '.

——Clyde

The rhythm of asking members to log in and participate in the game boss challenge is created by the player organisation itself. It is a self-organised rhythm established from the expectation of being a good player or a dedicated

guild member, as perceived by other members. The act of 'calling players into the game' is typically initiated by Clyde (the president) or other administrative members of the guild, rather than by the mechanisms within the mobile game itself. However, the motivation for guild managers to regularly remind their members to play is rooted in the inherent mechanics of mobile games. Additionally, the desire for better in-game rewards has led to the development of more rigorous membership management strategies.

With this guild management, the rhythm set by player organisations can sometimes conflict with individual members' daily lives at particular times. The guild's rhythm requires members to log in and seriously attack the in-game boss at specific time. Conversely, a member's real-life work role may not permit them to use their phones for one or two minutes of concentrated gameplay. Players often use the excuse of 'going to the bathroom' to manage this role conflict, allowing them to switch between their in-game and out-ofgame roles.

The in-game and out-of-game rhythms can be demanding for the individuals. Guild members sometimes refuse Clyde's requests for assistance by claiming they do not have time even for a quick break, citing 'not having time to go to the toilet'. This indicates that they cannot further take more breaks from their out-of-game roles. At the same time, if a specific guild member does not attack the boss, the guild may not pass the turn to the next player. In the game context, this means losing potential damage output and achieving a lower guild ranking—an outcome some guilds find unacceptable. This situation can also put pressure on the player's online relationships with other guild members.

Despite the potential for conflict and arrhythmia, the relationships between guild members may generally relaxed. My interview with Clyde was conducted outside of the game context. Yet, when mobile games or related elements were discussed, Clyde's interactions with his guild members seemed quite natural. He also noted that when he needed to replace his personal computer, several programmers from the guild offered guidance. This suggests that there may be a close in-group relationship among guild members, which may support their commitment to the spontaneous rhythm of play established within the Guild.

Additionally, online forums can influence players' understanding of in-game repetitive rhythmic behaviours. These forums serve as a virtual space for players of the same game to interact, separate from the game environment. Most of the interviewees in this research were recruited from such gaming forums. Unlike game guilds, these forums are more loosely organised and typically feature recent game topics, tips, articles and player-generated fan art.

These game forums serve not only as an informative resource but also as an potentially emotional support network, where players share a common digital identity receive social recognition and social support. In Apperley's (2017) research, the knowledge, information, life, and culture surrounding video games are referred to as 'game capital'. He builds on Consalvo's (2007) concept of 'paratexts', which refers to artefacts related to gaming, including hardware, software, websites, and more, that contribute to the formation of social relations through digital gaming. Apperley (2017) extended this concept to encompass lived culture and situated practices, further developing the idea of 'game capital'. He also noted that relationships based on game capital exist

independently of the specific times, spaces, and acts of play (p. 73). On this thesis context, the online game forum is a typical place for displaying and conmunation related to the game capital. As Karin, one of the interviewees, explained:

'Both mobile games and forum take their role in my life, and I spend more time on the forums than gaming. It is not just about the game; it includes some insights into life. Someone will share with others when they experience something really sad things in their life'.— Karin

Online gaming forums act as a hub where a wide range of mobile gamers exchange information and inadvertently shape a 'standard rhythm.' In the context of mobile gaming, this term refers to the expected behavioural patterns of a 'normal' player during an action cycle, such as an event or an version. To some extent, mobile game forums establish a standard for what constitutes a good player, establishing social norms among groups.

Further descriptions were provided by the interviewees Diana and Karin:

'[My] references [from the game forum] include daily quests, how to get resources to draw cards, how each stage should be played, and some posts based on probability calculations. People will slowly develop some sort of method to make themselves play more efficiently in the game. The aim is to get the in-game resource without paying for it.'

——Diana

'All the players inside the forum think this is an activity that cannot be missed. They must devote stamina points for that period to this thing to get

in-game resources during this activity. At that time, they would be very strict in calculating, setting a table, and calculating rigorous earnings formulas... I invest much time to look at the forum [to see] how do others do, I subconsciously think I need to do the same. At that time, I thought, I have the time, I want to do so'.

——Karin

During her time adhering to the forum's rhythm, Karin experienced arrhythmias. As a university student, she needed to balance her studies with gaming. Influenced by the best strategies from the game forum, she spent eight hours per day gaming, which left her feeling tired and exhausted. Reflecting on this, she questioned, 'Why did I have to make myself so tired?', and then added, 'I am starting to think it is a weird thing now'. Karin described her former self as a 'good player', a standard she adopted based on her observations of other players. She commented:

'A good player is someone who does the best they can with the resources they can devote to it...What other people do, I do too'

——Karin.

Karin adopted the most effective strategies she observed from other players and repeated these methods to achieve favourable in-game results. Her stress primarily stemmed from competition and comparison with other players. She developed a normative perception of her gaming behaviour and rhythm by imitating the in-game behaviour of other players. Gaming forums, or any discussion platforms where gamers converge and share information, create a

common rhythm or a 'standard rhythm' within the community. These standard rhythms, in turn, are usually created by those members of the forum who are most willing to invest time in mobile gaming in order to gain maximum advantage or obtain the most in-game resources.

This situation implies that if a player tries to adapt to the 'standard rhythm' set by the game guild or forum, there is a high risk of conflict with their local life rhythm. In Karin's case, the rhythm of a 'good player' as defined by the game forum conflicted with her out-of-game rhythm as a university student. This conflict arose from comparing herself with others, highlighting the conflict between personal needs and social norms.

Third, relationships with others can significantly influence a player's motivation for engaging in gaming. Several interviewees, such as Steward, reported that their mobile gaming experience was closely connected to their social circles. However, in the interviews, only friendships and romantic relationships were highlighted, while intergenerational relationships were not found. Steward explained:

'[I] never played handheld games [The researcher note: the handheld games are typically single-player] because I play games for other people, like King's Glory, which I play now, and Genshin Impact because my friends played a lot at that time, and then I also joined to play together... I do not think it should be described as more friends playing, and it is everyone playing.... During the UK epidemic, I had to stay in my dorm, and then I could not do anything. I did not want to face my dissertation, but I could not find any other way to entertain myself. I just watched too many films and I

did not want to watch films at that time. Then I had to find games. Because I was using an Apple computer, I could not play many computer games, so I had to play mobile games and could still play with my friends in my hometown. It was quite good to talk every day'.

—Steward

In this case, the mobile game acts as a medium for integration and synchronisation. It serves as a tool to connect the interviewee, Steward, with his friends through virtual avatars that create a sense of presence in cyberspace. While Steward's friends followed the social rhythms of the Chinese time zone, Steward himself followed the social rhythms of British society during the COVID-19 pandemic. With the mobile game as a medium, Steward's life rhythms merged and synchronised with those of his friends back home.

In Steward's case, there are no significant rhythm arrhythmias, but it highlights that 'playing with friends' can be an important intrinsic motivation for some players. In today's technological environment, video games provide a universal way of creating co-presence with people in different locations. For example, during the COVID-19 pandemic in 2020, the School of Animation and Digital Arts at the Communication University of China used game materials to recreate university buildings within the video game Minecraft for a digital graduation ceremony for students. Sun Guoyu, an associate professor at the School of Animation and Digital Arts, expressed how she felt warmed by the scene in the game where she stood side by side with her students: 'It

does not matter where I am in person, and at least I am together [with my students] at that moment in the game' (Sohu News, 2020).

This shows that people can experience a sense of being present together through digital gaming environments. The interviewee Diana mentioned that one of the reasons she plays a certain mobile game is because 'it is a social topic at the time'. Another interviewee, Abner, explained that one reason for playing is to 'interact with friends in the virtual world'. He added, 'some of whom are classmates and some of whom I met online. I also talk to my classmates about in-game content in real life'. In Chapter 5, the thesis describes this as common and collective rhythm. In this Chapter, it is explored as a potential cause of arrhythmia. Unsatisfactory rhythmic synchronisation can lead to conflict, as demonstrated by the interviewee Lance, whose in-game rhythm could not synchronise with his girlfriend's. He tried to gain a shared experience with his girlfriend through gaming, then he expressed his frustration:

'It was not that I wanted to play. It was that I was being dragged around by other people... I could not take it anymore. I was doing the daily quests and getting certain rewards, which allowed me to enhance some of my character's attributes, and then enhancing the attributes did not bring me any joy. It just felt like I was trying to complete homework for the chance to complete homework every day. Doing homework for completing homework.'

——Lance

After six months, Lance decided to uninstall the game, stating he could no longer endure the torture caused by the highly repetitive task challenges. In

this case, Lance's arrhythmias primarily stemmed from his attempts to synchronise with his girlfriend's life rhythm. He personally lacked interest in mobile gaming, yet the motivation driving him to engage in repetitive experiences daily was his social relationships with others.

As mentioned in Chapter 5, mobile games can serve as a medium for rhythmic coordination, but this does not always prove effective. Lance's game titled 'The Onmyoji', primarily offers single-player challenges. Therefore, Lance and his girlfriend's rhythmic interplay was based on sharing information and experiences. The mobile game creates a replicable information environment that enables player-to-player rhythmic fusion through shared enjoyment of the content. Driven by a desire to align with his girlfriend's rhythm, Lance tried to adapt to the rhythm of life that comes with mobile gaming, but he ultimately failed.

Using games as a medium, players may experience arrhythmia when they cannot reconcile their personal rhythms with those others expect them to maintain. Mobile gaming, therefore, may be a more likely cause of arrhythmia compared with other popular culture products. Unlike other cultural offerings, mobile games are not standalone, one-time engagements. They make periodic and prolonged demands on the user's schedule, often requiring continuous interaction.

In this interviewee's case, it is evident that the relationship between the human player and the digital mobile game is not confined to the moments of player-game interaction. Elements such as guild wars, online forum standards, social relationships and the norms of being a good player arise from mobile

gaming mechanisms and are shaped by the complex interplay between the external world and the game. In 2018, Keogh described video game embodiment as follows: "the player is incorporated into, becomes part of, and is ultimately made by the system of videogame play they instantiate" (p. 22). This thesis extends these cases to demonstrate that the 'system of videogame' is not confined to the virtual world constructed by computer graphics, programs, music and storylines on the screen, but also includes the culture, standards, identities and discussions about what it means to be a gamer. Players exist as an assemblage of two worlds and may experience arrhythmia due to this dual presence, as shown in Clyde's case, where guild members prioritise gaming over the rhythms of their local company employment.

This argument suggests that the feelings associated with mobile gaming are integrated into the body's senses. Players live within the rhythms established by mobile games just as they live within rhythms established by other social organisations. For example, Karin performed 'natural' calculations to make her game characters conform to norms, and the anxiety, regret and other negative feelings discussed earlier are systematically constructed within this framework. These become part of the perceptions of the assemblage between players and games. Mobile games serve as instruments to embody a series of immaterial and invisible norms, requirements or rhythms on the player's body—whether it is company management in Clyde's case, interpersonal relationships in Lance's case, or connections between players and virtual characters. The mechanisms of mobile gaming allow these delicate

arrhythmias to simultaneously occur in the virtual and real worlds, making them visible.

## 6.2.3 City Rhythm and Migration of Spatial Rhythms on the Human Body

According to scholars of rhythmanalysis, urban places also have their intrinsic rhythms. Crang (2003) described the rhythm of urban space as a combination of particular times and places, arguing that: "the urban place or site is composed and characterised through patterns of these multiple beats" (p. 6). The production of this rhythm in individual urban places depends on both their local and imported rhythms, Mulíček et al. (2016) summarised the rhythm study described by Schwanen et al. (2012) as follows: "A particular urban place as a locus of entertainment processes shaping, modulating, and synchronizing rhythms coming from particular sources. The place-based entrainment emerges as an interplay between local settings and incoming rhythms" (p. 116).

Furthermore, this thesis adopts the concept of 'pacemaker' as developed by Mulíček et al. (2016) to analyse city rhythms. This concept originated from " temporal and spatial points serving as sources of reality timing" (Mulíček, et al. 2016. p. 116), described by Parkes and Thrift (1975) and further developed by Hägerstrand (1982). It illustrates how entities such as foundries, schools, railways and churches influence the daily schedules of local residents. In Mulíček et al.'s (2016) research, the term refers to intra-city entities, such as the organisation of public transport or the opening hours of retail shops, which transmit their internal rhythms to other entities within the city. The pacemaker

plays a role similar to that of a metronome, which organises and coordinates the rhythmic development of the city.

The transmission of beats or rhythms between urban places occurs through the movement of the human body. When an individual enters an urban place that generates internal beats, their physical body is also influenced by the rhythms of that space. This interaction highlights the potential for rhythmic conflicts between the rhythms generated in urban spaces, which are imposed on the physical body, and those generated in the mobile gaming space, which are imposed on the digital avatar.

Conflicts often arise when two rhythmic events simultaneously demand the player's attention. For instance, a mobile game may require the players to concentrate for a period; at the same time, the real-world environment may also demand attention. A common scenario is public transport: when mobile game players are on a subway or bus, they are subject to the rhythm of the vehicle's movement. Announcements of stops every few minutes and the movement of passengers around these stops are part of the daily life rhythms of geographic space. However, when players try to synchronise these with the rhythms of a mobile game, they find themselves rapidly switching between the digital environment and the physical space, trying to adapt to the demands of both physical and digital space.

The case of interviewee Tim illustrates this process of rhythm negotiation:

'I missed a bus two days ago because I took a bus I had not taken before, and I forgot that [the bus] was different than usual'.

——Tim

At the time Tim missed his bus, he was playing a mobile game. He had become accustomed to the interplay between the rhythm of his usual bus route (geospatially), the environment around the bus stop and the rhythm of his mobile game. When his regular bus was replaced by a new one, Tim continued to operate under the old rhythm, focused on the mobile game, and did not notice the changes in the bus movements around him.

The rhythm of the bus stop had been imposed on Tim's body through the repetitive nature of the bus ride; he had internalised the artificially set schedule of the bus as a rhythm to follow. This familiarity allowed Tim to pay attention to his mobile game because he knew the usual rhythm of the bus stop well. However, when required to take a new bus route, his body was still operating according to the old rhythm model. At the moment when the mobile game demanded his attention, Tim failed to notice that the rhythm model of the city bus had changed.

In addition to public transport, other public spaces have their intrinsic rhythms, which can be disrupted by mobile gaming. An example is provided by interviewee Beatrice, a university student who often plays games on her phone while waiting for her lunch in the university canteen. This habit once led to an unintended consequence:

The Researcher: 'Have you ever forgotten anything while playing a mobile game?'

Beatrice: 'yes, yes, I have been playing in the canteen before getting my spicy dish, which is a pretty boring jump game, anyway, and I had to play it again and again. After that, when my classmates noticed my spicy dish was half cold'.

The Researcher: 'How long have you been playing?'

Beatrice: 'Maybe 20 or 30 minutes. I do not know how long I have been playing'.

The cafeteria, along with its pedestrians, staff serving meals, tables, chairs and the changing temperature of the food, forms a kind of external rhythm system. When individuals enter a space with a specific function, they often unconsciously adhere to the framing structure that the space imposes. There continue to use Beatrice's answer as the case:

Beatrice referred to her classmate's awareness of her gaming as 'noticing', which in the original Chinese is '发现', a word that can also mean 'find' or 'discover'. In this context, the term suggests that her classmates reminded her or stopped her from doing something. She used the word 'already' to describe her cold dish, indicating that this was an unwanted but unchangeable fact for her. These expressions imply that Beatrice had tacitly agreed to conform to the appropriate rhythm of eating in the canteen, suggesting that upon entering, she subconsciously accepted the space's rhythm. Ordinarily, Beatrice would collect her lunch, sit at a table and begin eating while the meal was still warm. However, by engaging with her mobile game simultaneously, she inadvertently performed these actions while distracted, losing track of time.

Immersed in the game's rhythm (enter the challenge, succeed or fail, try again), her awareness of both time and place was diminished.

Both Tim at the bus stop and Beatrice in the canteen experienced similar situations where the physical space—whether a bus stop or a canteen—imposed a specific rhythm associated with that location. Concurrently, they engaged with the rhythm of the digital space through their avatars, which navigated different in-game events according to the game's rules. In these instances, they operated within a hybrid space, adhering to two sets of rhythmic patterns: one from their geographical location and another from the digital space. The resulting conflict between the geographical rhythm and the mobile game rhythm often led to arrhythmias.

## 6.3 Conclusion

The analyses in the previous section reviewed potential manifestations of arrhythmias in everyday life: the conflicts between mobile game rhythms and biological, social and geographical rhythms

Subsection 6.2.1 on biological rhythms examines how the bodily needs of individuals are ignored or suppressed during gaming. Cases involving interviewees James and Sabina illustrate how they disregard their natural bodily needs to continue gaming. Mobile gaming systems, through their inherent cyclical rhythms, encourage players to return to the game regularly. This frequent return often conflicts with players' biological needs for rest, eating or sleeping.

Additionally, the concept of 'digital feeding' is introduced. This term refers to the regular and repetitive actions players perform to maintain the existence or growth of their digital characters. As discussed in Section 4.4 of this thesis, the player avatar and other in-game characters represent a materialised embodiment of pre-designed cyclical and linear rhythms within the game. The game translates its demands into the needs of in-game characters that require regular attentions, this dynamic is similar to the human behaviours involved in caring for a baby or a pet.

Compared to previous research on game rhythmanalysis, such as that of Apperley (2017), which was mentioned in the literature review, significant contributions have been made in this area. Apperley focused on how video game rhythms influence physical body performance, highlighting issues such as blisters on the fingers, pain, and retinal health. In his analysis, Apperley (2017) examined the physical discomfort experienced by players and introduced Lefebvre's concept of 'dressage' in the context of gaming. He also brought in the idea of 'compulsions' to describe how computer technology drives player behaviour, stating that players are compelled to "act, as in-action on their part will lead to the games cessation" (p. 45). He described this compulsion as throwing "the player in medias res and compelling them to act or be destroyed" (p. 45). Apperley (2017) further characterised game rhythm as a state in which the "player becomes familiar with sitting in front of the screen for hours on end" (p. 48). However, when examining the context of mobile games, the interviewee responses in this thesis indicate a shift compared to Apperley's (2017) findings.

On the one hand, the motivation of interviewees in this research to keep engaging with game rhythms—such as 'avoiding losing rewards'—is similar to Apperley's (2017) concept of 'act or be destroyed'. On the other hand, the interviewees also reveal that prolonged contact between the player and the game world is not the only manifestation of arrhythmia. As Ruffino (2021) argued, the game can continue running without the player's real-time participation; even when the player closes the game, it may still operate in the background. In the context of mobile games, daily gameplay is tied to social time. As discussed in Chapter 4, features such as the Monthly Pass or daily goals are updated based on real-time progression. In the examples presented in this section, the duration of direct contact with mobile games is not the primary cause of arrhythmia. For instance, James spends less than half an hour on daily tasks, while Valentina's 'resource collection' involves only a few clicks. In these cases, the cause of arrhythmia is the game's impact on, and interruption of, the player's daily schedule and routine-such as eating and sleeping. Players respond to the game's demands by performing corresponding actions, which may conflict with their natural biorhythms.

At the same time, due to technological limitations at the time, Apperley's (2017) analysis of gaming spaces was focused on computer rooms and cybercafés. Compared to Apperley's (2017) era, rhythmanalysis in the context of mobile games is more closely connected to the player's daily life outside the game. The physical space in which players engage with games has shifted from bedrooms or cybercafés to nearly all social spaces. Apperley's description of the 'congruence' between digital games and everyday rhythms, particularly regarding the location of play, has become more complex with

mobile games, as seen in examples like James playing in a taxi or Sabina in a hotel room. In Apperley's (2017) analysis, daily arrhythmias could be triggered by environmental factors such as 'a broken toilet or a lack of Coca-Cola' (p. 39). However, these disruptions are less likely to cause the same effect in mobile gaming contexts. Players now negotiate between gameplay and everyday life more flexibly, typically engaging with mobile games in intermittent, discontinuous, and regular bursts, rather than being continuously immersed for extended periods. As a result, mobile game arrhythmias often occur when players fail to engage with the repetitive daily game content or become anxious about doing so. This break in mental inertia, described by James as a feeling of 'disconnect', is a common symptom. Compared to arrhythmias in traditional PC games, which often stem from disruptions in the gaming environment, mobile game arrhythmias are more closely linked to disruptions in the player's daily schedule.

Subsections 6.2.2 and 6.2.3 explore arrhythmia arising from the player's dual existence in both the digital and geographical spaces. Mobile games facilitate a mode of interaction rooted in digital space that does not require the physical presence of either party. According to interviews conducted for this thesis, people engage with interactive objects in digital spaces through symbolic, simulated behaviours represented by computer programs and graphics. This interaction is often described using the same vocabulary as 'human-to-human' interactions in physical space.

In some mobile games, symbolic human beings represented by computer graphics replace actual human functions related to the emotional engagement of game participants. While this phenomenon has been present in traditional

single-player games, in the realm of mobile gaming, it signifies that nonhuman characters can offer new, long-term interactive content. On the one hand, these non-human characters increasingly resemble living humans; on the other hand, they provide long-term companionship that real humans may not be able to offer.

Consequently, this thesis considers the essence of digital mobile game space as a symbolic, conceptualised social space where physical contact is substituted by digital activity and social identity is represented by gaming terms and data. In the context of mobile gaming, this research prefers to describe it as the simultaneous presence of consciousness in two spaces, where the process of entering and exiting a mobile game equates to a redistribution of attention. The 'game' and 'non-game' environments each occupy a portion of the human subject's consciousness.

In terms of intimacy, some interviewees mentioned that they felt compelled to follow the game's rhythm to maintain relationships. It is important to note that the 'others' in this context could refer to real-life intimate relationships, group identities formed through social networking platforms or non-player characters within the game environment.

In these cases, players must navigate social relationships that span both ingame and out-of-game contexts. For example, Jessica's situation illustrates this dynamic: the cyclical rhythm imposed by the mobile game system required her to complete a series of in-game actions within a specific time frame. This obligation forced her to juggle her digital role of adventuring with in-game characters and her non-game role of playing mahjong with friends.

The rhythm conflict can also stem from the relationship between the player and the in-game character. Several interviewees described developing strong emotional ties with their characters. These emotional attachments motivated them to meet the needs of their in-game characters, despite the exhaustion it caused.

Conflicts can also arise from the relationships between players. This is evident in the case of guild wars, as mentioned by interviewee Clyde. In these wars, individual players are required to log in and complete actions at specific times for the benefit of the guild, often conflicting with their local rhythms. For example, some of Clyde's guild members resort to using their toilet breaks to fulfil guild requirements. Another example is interviewee Lance, who attempted to play the same game to find common ground with his girlfriend. To maintain this social relationship outside the game, Lance almost forces himself to adapt to the game's rhythm, leading to a strongly negative experience.

The final subsection of this chapter discusses the conflict between rhythms produced by geographical spaces and mobile games. According to Mulíček (2016), urban spaces shape, modulate and synchronise rhythms based on geographic space. Previous chapters have noted that players often use mobile games to fill gaps in their schedules, inserting playtime into intervals between daily events, typically in public spaces such as restaurants, undergrounds stations or cafeterias. Interviewee Tim's case mentioned earlier serves as an example: he stands at a bus stop trying to follow both the rhythm of the local bus and the game's demands, necessitating a rapid switch in attention between geospatial space and digital spaces. Tim missing his bus

highlights the potential for conflict between the local rhythm based on geospatial space and the game rhythm in the digital space.

For the interviewees, and likely for many mobile game players, turning on a smartphone is not a whim but a steady and regular habit. In line with Lefebvre's theoretical framework, it raise the question of the playing a mobile game constitutes 'automatism' or 'spontaneity'. For mobile game players, the distinction between 'internal need' and 'results from external constraints (game reward system)' is ambiguous in today's context. The former refers to their need for entertainment, while the latter suggests the game's push to engage the player.

The arrhythmia between mobile gaming and everyday life does not represent a conflict between freedom and paradigm; rather, it resembles a clash of one imposed structure against another, one norm against another. In this context, mobile games symbolise an abstraction of social organisation based on the digital environment, overlaying a collective life rhythm with the realities of daily life. Groups with similar occupations share similar life rhythms, and mobile games create virtual communities based on identities in the virtual world, introducing an additional life rhythm superimposed on the local rhythm.

The interview data in this research largely supports Apperley's (2011) analysis of game rhythm, where he suggests that digital games allegorically reflect the power dynamics between individuals and machines. This situation bears resemblance to the reverse side of Alan Turing's (1950) famous thought experiments. In this experiment, when humans and robots communicate through telepresence devices rather than face-to-face, a successful artificial

intelligence could make it impossible for humans to recognise it as a robot. Conversely, if a person is confined within a system that only permits the expression of machine logic, it may also be challenging for others to identify them as a human being.

While Turing's test of artificial intelligence explores the humanisation of machines, the allegory of mobile gaming suggests the opposite: the standardisation, mechanisation and measurability of human leisure and entertainment behaviours. In this research, interviewees have used machine programs to mimic human behaviour and deceive the mobile game's programming to gain rewards. This cycle involves game designers (humans) creating in-game machine logic to monitor player (human) behaviour and players (humans) using machine programs to simulate human behaviour to outwit the mechanical logic.

Discussing the rhythms of everyday gaming in this thesis also involves examining the interactions between on-screen avatars and off-screen bodies, in-game inhuman relationships and off-screen human connections, as well as in-game social standards versus off-screen social norms. The aim of this thesis is not to distinguish between good and bad, or to define what constitutes the real world, but rather to highlight how the world within the screen has become an integral part of everyday life in every sense of the word.

## **Chapter 7 Conclusion**

This thesis focuses on the linear or cyclical rhythms related to mobile games. These rhythms not only train players to navigate the virtual world, but more importantly, enable them to establish patterns of order, regularity, and action based on their own daily lives. These structured in-game rhythms influence players' time management, spatial awareness, and social relationships. Mobile games create a digital life within a virtual space that overlaps with the real-world daily routine, producing a new, hybrid daily rhythm. As De Souza e Silva (2023) suggested, this rhythm develops in the 'hybrid space' where digital and physical realms intersect. In this context, individuals naturally shift between digital and physical rhythms, and this fluid movement becomes an integral part of their daily routine.

Based on the rhythmanalysis theory, this thesis aims to explore the concept of game embodiment in the mobile gaming sector. Within the field of game embodiment research, Keogh (2018) used the term 'literacy' to describe players' ability to engage in gaming activities by manipulating hardware devices. This forms a system of grammar that trains players to understand and interpret in-game events. Building on this idea, the grammar of mobile gaming extends beyond simply interacting with the screen. It also includes how players manage their gaming schedules, maintain long-term relationships with in-game characters, re-shaping their life space, and integrate mobile gaming into their daily routines.

The concept of 'literacy' in mobile gaming thus focuses not only on hardware interaction but also on organizing, coordinating, and shaping time—whether through playtime, long-term engagement with non-player characters, or negotiating game mechanics in relation to non-game activities. These

interactions take place outside the game world while being deeply interconnected with the virtual, in-game environment.

Keogh (2014, 2018) argued for the dual existence of the player and the assemblage state of 'player-game', suggesting that video games and gamers should be viewed as a whole which simultaneously contains 'flesh, hardware, and virtual objects' (2018, p. 15). In the present research, the interviewees' experiences illustrate their frequent and repetitive switching between gaming and non-gaming activities such as eating, entertainment, and work. In other words, players exist simultaneously in both the digital and physical worlds, as well as in the routines of game and non-game life. Players' bodies serve as both carriers of mobile games and other tasks during gameplay. For instance, when a player engages with a mobile game during lunch, their fingers, hand, torso, and arm are occupied with gaming, while they simultaneously use cutlery to eat. This scenario exemplifies what is referred to in this thesis as the polyrhythmia of game and non-game rhythms in daily life. Keogh (2018) emphasises this phenomenon in his book, noting that while players consciously command game characters to perform actions such as running or jumping, they rightly focus their attention on the world within the screen, while the movements of their thumbs and fingers occur with relative unconsciousness. The interviewees in this study expressed similar perceptions of their bodies, noting how mobile gaming intersected with their other daily routines. The rhythms of mobile gaming not only train players to control their characters within the game but also train them to alternate between virtual and real-world bodily actions. This polyrhythmic state is an

acquired skill, much like learning to manipulate a game character in the virtual world, and it is developed and refined through the mobile gaming experience.

Based on this argument, the 'player-game' assemblage in the context of mobile gaming no longer exists within a distinct, separate space, as suggested by the concept of the magic circle. This assemblage must consider not only the interaction between the player and the game but also the influence of non-game daily routines. Mobile gamers exist in a complex hybrid state where in-game elements such as text, gameplay, and music capture the player's attention and shape their digital environment perception. Importantly, the degree and occurrence of this perception depend on the player's physical surroundings. As a result, the experience of mobile gaming becomes intertwined with the player's perception of their daily life. Keogh (2018) introduces the concept of 'everyday synesthesia' to describe players' composite perception during gameplay. This refers to the player's sensory experience of in-game elements—such as crashes, weight, or the flow of water—through interactions with the digital interface. This argument is based on the interplay between everyday perception and the game, where the player is exposed to more complex sensory experiences within the screen through sight, sound, and touch.

However, in this thesis, the interviewees' bodily perceptions of their physical environment and their gameplay perceptions are simultaneous and intertwined. The former includes sensations such as the taste of food, the feel of a bed, or the acoustic environment of an office or classroom, while the latter involves gameplay elements such as scenarios and character movements. These bodily senses interact inseparably, such that the

experience of mobile gaming and the player's perception of other real-world events form a holistic whole. As many interviewees noted, they chose mobile gaming because it fits seamlessly into their daily routines. The necessity of daily gaming, along with its compatibility with everyday life, is a key motivation for players. In this context, the 'player-game' assemblage includes not only the player's body, game hardware, and in-game objects, but also the player's everyday sensory perceptions, which are closely integrated with the gaming experience.

This idea can be linked to the concept of game perception described by Sudnow (1983). Sudnow (1983) detailed how he sat in a family room and immersed himself in a game on a miniature screen to save the digital world, illustrating how human bodily perception converges and transfers into the electronic screen. Contemporary game studies, particularly in the area of mobile gaming, should also explore how player awareness is distributed between the digital screen and the physical environment. Mobile gaming redefines a set of identities, images, and behavioural logic for each player. Individual lives are reshaped and re-organised by the interweaving of multiple rhythms, and the effects of these interweaving rhythms may create either eurhythmia or arrhythmia in the player's personal life.

Chapter 4 explains the process of rhythm production in mobile gaming. This process can be summarised as a three-step transformation from the designer's intentions to the player's out-of-game body behaviour. The first step involves translating the mobile game's demands into in-game materials. These demands include metrics such as player engagement rates, daily or monthly activity percentages, time spent online, and willingness to make in-

game purchases—all of which reflect the gaming company's potential for profit. The in-game materials, such as rare equipment, characters, or rare items are designed to encourage players to meet the game company's expectations. The second step involves transforming these in-game materials into the player's in-game behaviours. This transformation is guided by rules, standards, or action instructions, as discussed in the sections on linear and cyclical rhythms within the thesis. For example, players may repeatedly undertake hundreds of challenges with the same actions or log in daily to earn rewards. This step links the player's rhythmic in-game behaviours to the acquisition of rare in-game materials. The third step involves the transformation of in-game behaviours to out-of-game behaviours, as mentioned at the end of Chapter 4,. The rhythms generated in the digital space through the game's systems are carried over into the player's daily routine, influencing their real-world actions and habits.

Most in-game actions that players can perform are limited and repetitive: the daily gameplay involves the same rhythm repeated over and over again. The rhythm of mobile games is both rewarding and encouraging, yet it also serves as guidance and regulation, and even monitoring and surveillance. The gaming system operates by evaluating and assigning value to player behaviour in the digital space, motivating players to repeat in-game actions through rewards. Unlike abstract language, computer programs use precise numbers and logical processes to assess player behaviour through pre-designed algorithms. The system restricts players to actions that can be simulated within its framework; in other words, all in-game behaviours are predetermined by the designer and are only performed to the extent that the

program allows. As described by Lefebvre (2013), rhythm constitutes the trinity of 'activity-repose-entertainment' (p. 41).

As a vehicle for the player in the digital environment, in-game 'avatars' are often viewed as extensions of the player's self. However, they can also serve as trainers of rhythm imposed on the human body. As discussed in Chapter 4, the avatar operates within a narrowly predefined set of actions aligned with daily goal systems. This logic imposes a predetermined standard rhythm, and the player's agency is reduced to deciding whether to 'do' or 'not do' certain actions in a range of list.

This thesis seeks to extend Lefebvre's (2013) concept of 'breeders, builders and trainers' of rhythm from concrete individuals or social organisations in physical reality to a designed industrial digital program. Although this program is ultimately created by mobile game developers, it can function independently of its human creators to a considerable extent. To aid understanding, an ingame term and case study are introduced: At the start of each day, a player's in-game avatar receives a set of instructions from the daily task system. If players complete most of these tasks, they receive a reward from the system.

The avatar acts as a mediator between the system's instructions and the player's actions, with the player controlling the avatar to complete repetitive tasks. These actions, guided by the system, create a linear or cyclical rhythm that influences the player's body behaviour. In Chapter 6, I introduce the concept of 'digital feeding' to describe the relationship between the player's physical body and the digital avatar, which exists within the game.

In Lefebvre's era, rhythms that regulated and trained individuals typically occurred through human-to-human interaction. In today's digital age, this concept of a 'rhythm trainer' has evolved into digital avatars and systemdriven guidance. Player behaviour is monitored and measured by rubrics built into the game system, with rewards and punishments delivered through the in-game narrative or objects. On the one hand, players 'feed' the digital body to strengthen it; on the other hand, the digital body trains the player, aligning their daily behaviour with the pre-designed rhythm model embedded in the game system.

Lefebvre's (2013) classic rhythmanalysis describes a 'dressage' process where linear, regular, repetitive behaviours train (or teach) humans to adapt to social life. Similarly, the rhythm of mobile games follows a similar process, essentially re-socialising players within digital spaces. Through preplanned interaction modes, players may internalise the logic of mobile gaming into their personal consciousness, leading to a 'computerisation' of human self-awareness. The logic of reward delivery in games is straightforward and predictable, primarily involving repeated detection of specific in-game actions or sequences. Consequently, the player's behaviour in acquiring rewards involves the repetitive execution of these actions, suggesting a machine-like logic that is internalised and repeated by the players themselves—a process that this thesis refers to as the players' self-programming.

The player's daily life in the digital world is shaped by this repetitive selfprogramming. As noted at the end of Chapter 4, players often use the term 'daily routine' to describe the process of following the cyclical and linear rhythms of mobile games. From the initial sense of novelty to feelings of

boredom or habituation, players gradually internalise the rules, guidelines, and events of the mobile game as part of their everyday lives. In this way, the digital routine of the Internet-based world and the routine of the physical world merge, shaping the player's overall day-to-day experience.

Chapters 5 and 6 explore the combination and conflicts between the rhythms of mobile games and everyday non-game activities. This discussion can be linked to the concept of 'hybrid space' (De Souza e Silva, 2006), which refers to the interplay and integration of physical and digital space. This hybrid digital-physical space that emerges from the intersection of virtual and real environments carries the rhythms of multiple spaces simultaneously. According to the interviewees in this study, they often played mobile games while performing other daily tasks, with both activities occurring simultaneously. In these moments, people occupy multiple roles across different spaces. They may be office workers, students in a classroom, passengers at a bus stop, or customers in a restaurant, while also acting as travellers, summoners, commanders, or prefectural magistrates within a mobile game. When people engage with mobile games, they inhabit both physical and digital spaces—moving around with a physical body of flesh and blood and a digital body made of code and programs.

As discussed in Chapter 5, following the rhythms of everyday life in a physical, non-gaming environment often does not consume a person's full attention. This unconsumed attention can be redirected and reorganised by the rules of mobile games, allowing individuals to enter the linear or cyclical rhythms of mobile games while remaining present in the physical world. However, when players are required to fulfil roles in everyday life—such as being a worker (as

required by social organisation), a partner (as required by relationships), or a passenger or customer (as required by geographical space)—the behavioural norms of mobile gaming, such as logging in regularly or spending time on daily tasks, may conflict with these other daily obligations. It is important to clarify that the player's role here is not the same as their in-game avatar. Instead, the player acts as the 'feeder' of the in-game avatar, a concept easily confused with the avatar itself. The mobile gaming system requires the player to assume a hybrid identity—one that exists between the in-game and out-of-game worlds—where their real-world actions support and sustain the in-game character.

Chapter 5 also highlights that players may adopt the rhythm of mobile gaming to regain control over their daily lives and pursue selective freedom. The rhythm of mobile play becomes a tool to counter the rhythms of out-of-game daily life, reflecting a desire to escape from routine and everyday events. Jordan (2009) described an individual living in a hybrid space as having 'a shared consciousness between the online virtual self and the offline physical self. However, in the context of mobile games, this relationship requires careful explanation. As discussed earlier, the in-game avatar in mobile games cannot be regarded as a complete embodiment of the player's self. Instead, it serves as an imaginary bearer for experiencing structured, rule-based adventures. For example, 'completing a task in a game that has been repeated hundreds of times' cannot be equated with a fully realised virtual self, especially in case of some interviewees stated that they completed these tasks unconsciously. However, the act of 'entering the game' itself is an

expression of self-will. In this context, the 'online virtual self' is a deliberately constructed entity by the player, distinct from the 'offline physical self'.

As discussed in Chapter 5, this act of engaging with the virtual self may reflect the player's attempt to regain autonomy in a life dominated by the rhythms of the out-of-game physical world. It represents one form of linear or cyclical repetition being used to break free from another, but it still reflects the player's desire for freedom of choice in their everyday life. In this context, I introduce the concept of 'pseudo-disengagement', which refers to the process by which mobile game players invest their consciousness in digital space without disrupting the rhythms of life in physical space. In this process, the interviewees' physical presence remains within the rhythm of their lives in physical space, but their consciousness is simultaneously engaged with the rhythm of play in the digital space. This concept aids in describing the state of players who are trying to adapt to both the rhythms of digital and physical spaces, as previously mentioned. Referring back to the 'attention gap' described in Chapter 5, 'pseudo-disengagement' fundamentally involves the reallocation of attention while maintaining physical activity. It also points to a human desire to disengage from the physical body and escape from routine.

Both Chapters 5 and 6, whether discussing the combination or conflict between game and non-game rhythms, acknowledge that the rhythms of mobile games and other human activities in modern society are industrial products. They do not originate from human spontaneity but result from a mutual compromise with external social organisations or internal game management systems. The rhythms of digital space, the player's self, and physical space form a tripartite relationship of mutual influence. In this context,
a player's 'voluntary' engagement with the digital rhythm at any given moment can be regarded as resistance to or a means of distancing oneself from the rhythm of physical space. Digital rhythms allow for transient conscious detachment and subjective selectivity, and the individual's choice among multiple industrialised rhythms may foster a perception of subjective freedom.

The rhythms originating in digital space are shaped by the social structure that governs the relationship between game producers and players. This structure is organised through digital, rather than physical space, and is imposed on the player under the guise of being 'voluntary to download' and 'free to play'. As discussed at the beginning of Chapter 6, the digital rhythm often overpowers the physiological rhythm of the human body, disrupting its natural physiological cycles. Through prolonged repetition, training, and external influences, the rhythms of digital space, along with the organisational structures of society, become internalised into one's daily routine, merging with the rhythms of everyday life. While 'playing a role in the digital space' requires corresponding physical actions—such as taking out a phone, tapping the screen, or keeping it within reach—the rhythm of the game demands regular or repetitive behaviours and intermittent attention. This series of actions or attention shifts often conflicts with the roles the player is expected to fulfil in the physical space, as shown by the experiences of many interviewees discussed in Chapter 6.

The discussions in Chapters 4 and 6 highlight how mobile games simultaneously involve computerisation of human behaviour and the personification of electronic objects. The computerisation of human behaviour is illustrated in the examples from Chapter 4, where game players use

automated clickers to simulate their actions and deceive the game system. In such cases, human behaviour becomes indistinguishable from behaviour simulated by a computer program, as the repetitive execution of a linear rhythm by a human can be fully replaced by simulation of computer software. The personification of electronic objects is explored in Chapter 6 through the analysis of intimacy, where interviewees reported forming real emotional connections with in-game characters. These two aspects—point to a common trend towards equating human and non-human entities in virtual spaces. Due to the limited range of behaviours humans can exhibit in a game environment, interactions between players and other entities are confined to the range of actions that the program permits. Under these conditions, there is little difference between the in-game actions performed by human players and those simulated by computers. Moreover, human players can only engage in behaviours dictated by the game's login system, which is controlled by the computer. The interviewees in this thesis perceive themselves as travellers, captains, and prefectural magistrates within their digital lives, yet this perception of 'life' is fundamentally rooted in the simulation of the logic of reality by digital programs. It can be argued that digital games simulate a living environment using computer logic, and players internalise this logic in order to adapt to life within the game.

The discussions in Chapters 4 and 5 lead to the conclusion that mobile games create a structured and rationed system of leisure. Stamina points are limited, the number of daily challenges is capped, and resources are assigned by the mobile game system, all pre-designed by developers in advance. This creates a regulated, quantitative form of leisure, where the number of actions

a player can perform is predetermined. According to players, the daily task system ensures they perform no more and no less today than they did yesterday—a 'daily task' or 'daily goal system' that makes leisure feel repetitive, with linear and cyclical rhythms guiding their actions. This planned digital routine is seamlessly inserted into another set of planned actions based on the physical environment, reflecting the essence of rhythmic combination. As the interview data showed, players lack sufficient leisure time for extended, physically located leisure activities. The time and space of mobile games are intertwined with the time and space of work and study, creating a dual reality where players simultaneously work and relax in their daily time, living in a dual game-reality time and space, where the two are fragments of each other.

As described by the concept of 'pseudo-disengagement', players live in both in-game and out-of-game rhythms simultaneously. Their adaptation to mobile gaming rhythms cannot be seen purely as an expression of freedom or leisure but rather as a form of resistance to everyday life, which is already dominated by the rhythms of industrialisation. However, this resistance itself remains a standard form of domination by industrial society. The rhythm imposed on the player's body—whether through their physical flesh or digital avatar represents a fusion of external guidance and the player's personal desires.

This argument raises concerns about player agency. On the one hand, the complex symbiotic relationship between players and games suggests an underlying power structure, with the rationed system of mobile gaming influencing the subjectivity of human players. However, this influence does not equate to direct control over the player. Aarseth (2017) introduced the concept of the 'implied reader' (Iser, 1974) into game studies. The 'implied

player' refers to the set of expectations and roles designed for the player in advance. Aarseth (2017) also highlighted the potential for 'transgressive play,' where players creatively subvert the game's rules, such as accessing unintended areas or killing unintended characters by exploiting bugs. These moments of transgression demonstrate that players can reclaim control over the gaming process, going beyond pre-designed content and rules. For example, as discussed in Chapter 4, mobile gamers use clickers to simulate human fingers and complete tasks. Although the in-game character strictly follows the pre-designed game content, the use of clickers in the out-of-game world allows the player's physical engagement to deviate from the designer's intended patterns. This phenomenon can be regarded as a form of out-ofscreen transgression, representing the player's deviation from and rebellion against the highly repetitive nature of the game's content. This highlights the autonomy players have in how they engage with games, indicating that they do not fully adhere to the designed methods of balancing mobile gaming with their daily lives. Furthermore, interview data show that players of the same game make subtle distinctions in their playing time and engagement strategies. Although designers may create a model of the 'implied player' with specific expectations for how the game should be played during development, these findings demonstrate the agency players exert in shaping their own engagement processes.

In addition, as Anderson (2018) has shown, the existence of mobile gaming reinforces players' agency over their bodies, space, and time. The smartphone's touch controls offer more physical freedom than a joystick or mouse, as participants in this study reported playing in various settings—on a

bed, at a dining table, in a cafeteria, classroom, or bus stop—while adapting their body positions to these environments. Mobile gaming input methods allow for greater control over posture, including the position of fingers, wrists, and the body, offering more flexibility. This flexibility enhances players' options in terms of both time and space. Compared to traditional devices, this should be seen as a dual reinforcement: it increases players' agency over how they arrange their time and space, while also encouraging engagement with mobile games during fragmented moments of personal time.

Furthermore, the connection between mobile players and games can be fragile. As discussed in Chapter 6, many participants frequently mentioned moments of quitting and uninstalling games, a common experience in their gaming journeys. These moments of disconnection suggest a complex power dynamic between player subjectivity and game structures. On-screen, players follow linear or cyclical rhythms to engage with pre-determined content, but off-screen, their agency is expressed when they disengage at any moment, abandoning their accounts when they grow bored with the game. The video game market, much like a virtual theme park, offers a range of experiences, many of which involve implicitly repetitive content. Players navigate different games in much the same way as they explore various attractions. In this sense, players' negotiations are not just with a specific mobile game but with the broader design logic of mobile gaming and how video games on smartphones interact with their everyday lives. This dimension of player-game interaction has not been adequately addressed in current game studies, a gap this thesis aims to explore.

At the end of this thesis, I am obliged to clarify the limitations of my research. First, from the time I began work on this thesis (2020) to the present (2024), the Chinese mobile gaming market has undergone significant changes. One prominent trend is the increasing connectivity between console, computer, and mobile gaming devices. Beginning with Genshin Impact in 2020, many Chinese game companies now support cross-platform play. A gamer might play on a smartphone during their commute and continue on a personal computer at home. This phenomenon blurs the line between 'mobile games' and 'computer games,' with many mobile games adopting the graphical and gameplay characteristics of computer games. This trend of platform connectivity, which I did not consider during the early stages of my research, may have significant implications for future developments in game studies.

Second, in my research, 'mobile game' refers to any game that can run on a smartphone. The analysis chapters were structured around the general characteristics of the mobile gaming experience, which may have overlooked the specific dynamics of certain genres. For example, location-based games like Pokémon GO<sup>48</sup> may encourage players to explore new physical locations, while dating simulation games may influence players' perceptions of real-life relationships. Although These elements were mentioned by interviewees, they were not included in a structured or thematic analysis. This omission could limit the scope of my findings with respect to the diverse range of mobile game genres.

<sup>&</sup>lt;sup>48</sup> Due to limitations associated with Google's mapping service, Pokémon Go is not available in China. However, the Chinese market offers several games with similar mechanics.

Finally, the recruitment and interviews were conducted during the COVID-19 pandemic, which likely influenced participants' gaming experiences within the context of altered daily routines. After the UK government lifted restrictions on social gatherings, I personally noticed that I had invested more time and emotion in mobile gaming during that period. This subtle shift is a characteristic I did not fully capture during the interview process. In this specific social context, the player-game relationship may vary depending on participants' real-life situations, and their understanding of 'everyday life' is likely shaped by the pandemic and its related social policies. However, relevant questions about the impact of COVID-19 and its social effects were not posed to participants, and this influence was not explored in the analysis.

Additionally, as mentioned in the methodology chapter, I am embedded within the online gaming community, making it difficult to adopt the perspective of an 'outsider.' This pre-existing internal viewpoint may have influenced my research analysis. Furthermore, the intercultural context of this study presents the possibility that some interview information may have been lost in translation, both linguistically and socially.

This thesis offers insights for three key groups of readers. First, for sociologists, it serves as a qualitative study on the relationship between mobile gaming and everyday life, providing valuable information for scholars interested in this area. Second, it is written for mobile game players without an academic background, aiming to help them better understand how games influence their daily lives and decision-making, sometimes in ways they may not be fully aware of. Third, this study seeks to provide a reference for mobile game designers working in the industry. As mentioned in the introductory

chapter, some interviewees included game industry practitioners who noted the lack of ethical standards in current mobile game design processes. Understanding the impact of mobile games on players' lives is crucial knowledge for designers, yet detailed material on this topic is often lacking. This thesis aims to address that gap.

Furthermore, during the later stages of this thesis, artificial intelligence language models such as ChatGPT have been integrated into mobile gaming, enabling free interaction and dialogue between players and non-player characters. The ability to generate personalised, dynamic dialogue may significantly alter the in-game relationships between players and their characters. This thesis explores the potential arrhythmias caused by the evolving relationships between players and non-human characters. The emergence of AI-based interactive systems may profoundly influence the dynamics of these relationships in the near future.

## Reference:

Aarseth, E. (2001). Allegories of space: The question of spatiality in computer games. In Eskelinen, M. and Koskimaa, R. (Eds). Cybertext yearbook 2000. Jyväskylä: University of Jyväskylä, pp. 152–171.

Aarseth, E. (2004). Beyond the frontier: Quest games as post-narrative discourse. In Ryan, M.-L. (Ed.) Narrative across media: The languages of storytelling. Lincoln: University of Nebraska Press, pp. 361-376.

Aarseth, E. (2008). A hollow world: World of Warcraft as spatial practice. In Corneliussen, H. G. and Rettberg, J. W. (Eds). Digital culture, play, and identity. London: MIT Press, pp. 111-122.

Aarseth, E. (2014). I fought the law: Transgressive play and the implied player. In Segal, N. and Koleva, D. (Eds). From literature to cultural literacy. London: Palgrave Macmillan UK, pp. 180-188. Available at:

doi:10.1057/9781137429704\_13.

Adams, W. C. (2015). Conducting semi - structured interviews. In Wholey, J. S., Harty, H. P. and Newcomer, K. E. (Eds). Handbook of practical program evaluation. London: Wiley, pp. 492-505. Available at: doi:10.1002/9781119171386.ch19.

Ahmad, M. A., Keegan, B., Srivastava, J., Williams, D. and Contractor, N. (2009, August). Mining for gold farmers: Automatic detection of deviant players in MMOGs. In 2009 International Conference on Computational Science and Engineering. London: IEEE, Vol. 4, pp. 340-345. Available at: doi:10.1109/CSE.2009.307.

Alha, K., Koskinen, E., Paavilainen, J., Hamari, J. and Kinnunen, J. (2014). Free-to-play games: Professionals' perspectives. In Proceedings of Nordic DiGRA, 2014. Visby: Digital Games Research Association.

Andrejevic, M. (2009). Exploiting YouTube: Contradictions of user-generated labor. In Snickers, P. and Vonderau, P. (Eds). The YouTube reader. Stockholm: National Library of Sweden, pp. 406–423.

Ang, C. S. and Zaphiris, P. (2010). Social roles of players in MMORPG guilds: A social network analytic perspective. Information, Communication & Society, 13(4), 592-614. Available at: doi:10.1080/13691180903266952.

Apperley, T. (2011). Gaming rhythms: Play and counterplay from the situated to the global. Amsterdam: Institute of Network Cultures.

Apperley, T. (2017). Digital gaming, social inclusion and the right to play: A case study of a Venezuelan cybercafe. In Hjorth, L., Horst, H., Galloway, A. and Bell, G. (Eds). The Routledge companion to digital ethnography, pp. 235-243. New York: Routledge.

Appshopper (2023). Appshopper. [Online]. Appshopper. Available at: https://appshopper.com/ [Accessed 10 May 2023].

Ash, J. (2013). Technologies of captivation: Videogames and the attunement of affect. Body & Society, 19(1), 27-51. Available at: doi:10.1177/1357034X11411737.

Atkinson, R. and Willis, P. (2007). Charting the Ludodrome: The mediation of urban and simulated space and rise of the flaneur electronique. Information,

Communication & Society, 10(6), 818-845. Available at: doi:10.1080/13691180701751007.

AudioSurf (2008). AudioSurf. [Online]. AudioSurf. Available at: https://store.steampowered.com/app/12900/AudioSurf/ [Accessed 28 April 2024].

Bartle, R. (1990). Interactive multi-user computer games. Essex: MUSE Ltd.

Baudrillard, J. (1981). Simulacra and simulation. New York: University of Michigan Press.

BBC News (2021). China steps in to regulate brutal '996' work culture. [Online]. BBC News. Last updated: 2 September 2021. Available at: https://www.bbc.co.uk/news/world-asia-china-58381538 [Accessed 1 February 2023].

Becker, M. C. (2005). The concept of routines: some clarifications. Cambridge Journal of Economics, 29(2), 249-262. Available at: doi:10.1093/cje/bei031.

Beer, D. (2016). Metric power. London: Palgrave Macmillan. Available at: doi:10.1057/978-1-137-55649-3.

Bengtsson, T. T., Bom, L. H. and Fynbo, L. (2021). Playing apart together: Young people's online gaming during the COVID-19 lockdown. Young, 29(4\_suppl), S65-S80. Available at: doi:10.1177/11033088211032018.

Bennerstedt, U. and Ivarsson, J. (2010). Knowing the way: Managing epistemic topologies in virtual game worlds. Computer Supported Cooperative Work (CSCW), 19(2), 201-230. Available at: doi:10.1007/s10606-010-9109-8.

Beoku-Betts, J. (1994). When black is not enough: Doing field research among Gullah women. NWSA Journal, 6(3), 413-433.

Berger, A. A. (2002). Video games: A popular culture phenomenon. London: Transaction Publishers.

Bissell, D. (2007). Animating suspension: Waiting for mobilities. Mobilities, 2(2), 277-298. Available at: doi:10.1080/17450100701381581.

Boellstorff, T. (2010). Placing the virtual body: Avatar, chora, cypherg. In Mascia-Lees, F. E. (Ed). A companion to the anthropology of the body and embodiment, pp. 504-520. Hoboken: John Wiley & Sons.

Boellstorff, T., Nardi, B., Pearce, C. and Taylor, T. L. (2012). Ethnography and virtual worlds: A handbook of method. Princeton: Princeton University Press. Available at: doi:10.2307/j.cttq9s20.

Bojin, N. (2013). Exploring the notion of 'grinding' in massively multiplayer online role playing gamer discourse: The case of 'Guild Wars'. Simon Fraser University. PhD.

Boudreau, K. and Consalvo, M. (2014). Families and social network games. Information, Communication & Society, 17(9), 1118-1130. Available at: doi:10.1080/1369118X.2014.882964.

Boudreau, K. and Consalvo, M. (2016). The sociality of asynchronous gameplay: Social network games, dead-time and family bonding. In Willson, M. and Leaver, T. (Eds). Social, casual and mobile games: The changing

gaming landscape. London: Academic Press, pp. 77-133. Available at: doi:10.5040/9781501310591.ch-006.

Brock, T. (2021). Counting clicks: Esports, neoliberalism and the affective power of gameplay metrics. In Jin, D. Y. (Ed.) Global esports. London: Bloomsbury Publishing, pp. 115-131. Available at:

doi:10.5040/9781501368745.0013.

Brock, T. and Fraser, E. (2018). Is gaming a craft? Prehension, practice and puzzle-solving in gaming labour. Information, Communication & Society, 21(9), 1219-1233. Available at: doi:10.1080/1369118X.2018.1468920.

Brock, T. and Johnson, M. (2021). The gamblification of digital games. Journal of Consumer Culture, 21(1), 3-13. Available at: doi:10.1177/1469540521993904.

Brock, T. and Johnson, M. (2022). Videogaming as craft consumption. Journal of Consumer Culture, 22(3), 598-614. Available at: doi:10.1177/14695405211016085.

Brooks, P. (1984). Reading for the plot: Design and intention in narrative. Cambridge: Harvard University Press.

Bruckman, A. (1992). Identity workshops: Emergent social and psychological phenomena in text-based virtual reality. MIT Media Laboratory. Master's Thesis.

Bruckman, A. (2013). A great experience that must stop: Words with Friends and the mindful use of technology. [Online]. The Next Bison: Social

Computing and Culture. Last updated: 6 April 2013. Available at: https://nextbison.wordpress.com/2013/04/06/a-great-experience-that-muststop-words-with-friends-and-the-mindful-use-of-technology/ [Accessed 21 February 2024].

Bruns, A. (2008). Blogs, Wikipedia, Second Life, and beyond: From production to produsage. New York: Peter Lang.

Bull, M. (2005). No dead air! The iPod and the culture of mobile listening.

Leisure Studies, 24(4), 343-355. Available at:

doi:10.1080/0261436052000330447.

Caillois, R. (1961). Man, play, and games. Urbana: University of Illinois Press.

Campbell, C. (2005). The craft consumer: Culture, craft and consumption in a postmodern society. Journal of Consumer Culture, 5(1), 23-42. Available at: doi:10.1177/1469540505049843.

Cao, Y. (2007). System. [Online]. Jianshu. Available at: https://www.jianshu.com/p/281d77133938 [Accessed 1 February 2023].

Cao, Y. and Downing, J. D. (2008). The realities of virtual play: Video games and their industry in China. Media, Culture & Society, 30(4), 515-529. Available at: doi:10.1177/0163443708091180.

Carmichael, S. (2013). What it means to be a whale and why social gamers are just gamers. [Online]. Vulturebeat. Last updated: 14 March 2013. Available at: http://venturebeat.com/2013/03/14/whales-and-why-socialgamers-are-just-gamers/ [Accessed 10 September 2024]. Caroux, L., Isbister, K., Le Bigot, L. and Vibert, N. (2015). Player-video game interaction: A systematic review of current concepts. Computers in Human Behavior, 48, 366-381. Available at: doi:10.1016/j.chb.2015.01.066.

Carrier, L. M., Rosen, L. D., Cheever, N. A. and Lim, A. F. (2015). Causes, effects, and practicalities of everyday multitasking. Developmental Review, 35, 64-78. Available at: doi:10.1016/j.dr.2014.12.005.

Castronova, E. (2008). Synthetic worlds: The business and culture of online games. London: University of Chicago Press.

Cheok, A. D., Devlin, K. and Levy, D. (Eds.). (2017). Love and sex with robots: Second International Conference, LSR 2016, London, UK, December 19-20, 2016, revised selected papers. London: Springer.

Chesher, C. (2004). Neither gaze nor glance, but glaze: Relating to console game screens. SCAN: Journal of Media Arts Culture, 1(1), 98-117.

Chess, S. (2005). Playing the bad guy: Grand Theft Auto in the panopticon. In Garrelts, N. (Ed). Digital gameplay: Essays on the nexus of game and gamer. London: McFarland and Company, pp. 80-90.

Chess, S. (2018). A time for play: Interstitial time, Invest/Express games, and feminine leisure style. New Media & Society, 20(1), 105-121. Available at: doi:10.1177/1461444816660729.

China Audio-visual and Digital Publishing Association (2022). China gaming industry report (January - June 2022). [Online]. ycwb.com. Last updated: 21

July 2022. Available at: https://news.ycwb.com/attachment/2022-

07/21/74fee951-02d6-4a92-8bf1-5b87e03c723c.pdf [Accessed 10 May 2023].

China Game Publishing Committee (2016). 2016 China's gaming industry report. [Online]. CGI Constructors. Available at:

http://www.cgigc.com.cn/gamedata/6237.html [Accessed 10 May 2023].

China Game Publishing Committee (2017). 2017 China's gaming industry report. [Online]. CGI Constructors. Available at:

http://www.cgigc.com.cn/industrydt/16925.html [Accessed 10 May 2023].

China Game Publishing Committee (2018). 2018 China's gaming industry report. [Online]. baidu.com. Available at:

https://pan.baidu.com/s/1i8phE1eYZ4ECgpNvE4FEXg [Accessed 10 May 2023].

China Game Publishing Committee (2019). 2019 China's gaming industry report. [Online]. CGI Constructors. Available at:

http://www.cgigc.com.cn/details.html?id=08da81c4-ceb3-4631-89a4-2b1b4618cfdc&tp=report [Accessed 10 May 2023].

China Internet Network Information Center (2012). The survey report of mobile game users in China. [Online]. China Internet Network Information Center. Available at:

http://www.cnnic.cn/hlwfzyj/hlwxzbg/ydhlwbg/201303/P020130325581009590 187.pdf [Accessed 10 May 2023].

China Internet Network Information Center (2013). The survey report of mobile game users in China. [Online]. China Internet Network Information

Center. Available at:

http://www.cnnic.cn/hlwfzyj/hlwxzbg/wybg/201409/P02014091153872027915 8.pdf [Accessed 10 May 2023].

China Internet Network Information Center (2015). 2014–2015 China mobile game user survey report. [Online]. China Internet Network Information Center. Available at: http://www.cac.gov.cn/files/pdf/cnnic/20142015phonegame.pdf [Accessed 10 May 2023].

China Ministry of Culture (2000). The notice of the special management method on the electronic games premises. [Online]. China Ministry of Culture. Available at: http://www.gov.cn/gongbao/content/2000/content\_60240.htm [Accessed 10 May 2023].

China Ministry of Culture (2015). The notice of the Ministry of Culture allowing domestic and foreign enterprises to engage in the production and sale of games and entertainment equipment. [Online]. China Ministry of Culture. Available at:

https://zwgk.mct.gov.cn/zfxxgkml/zcfg/gfxwj/202012/t20201204\_906274.html [Accessed 10 May 2023].

China National Press and Publication Administration (2022). [Online]. China National Press and Publication Administration. Available at: https://www.nppa.gov.cn/nppa/channels/718.shtml [Accessed 1 February 2023].

China National Press and Publication Agency (2021). The notice prevent minors from being addicted to online games. [Online]. China National Press

and Publication Agency. Available at:

http://www.gov.cn/zhengce/zhengceku/2021-09/01/content\_5634661.htm [Accessed 1 February 2023].

China State Council (2014). The documents of General Office of the State Council of the People's Republic of China No.51. [Online]. China State Council. Available at: http://www.gov.cn/zhengce/content/2014-01/06/content\_7548.htm [Accessed 1 February 2023].

Christiansen, S. L. and Gebauer, M. (2019). Lefebvre and rhythms today. In Christiansen, S. L. and Gebauer, M. (Eds). Rhythms now: Henri Lefebvre's rhythmanalysis revisited. Aalborg: Aalborg Universitetsforlag, pp. 5-16.

Chua, P. H., Jung, Y., Lwin, M. O. and Theng, Y. L. (2013). Let's play together: Effects of video-game play on intergenerational perceptions among youth and elderly participants. Computers in Human Behavior, 29(6), 2303-2311. Available at: doi:10.1016/j.chb.2013.04.037.

Colley, A., Thebault-Spieker, J., Lin, A. Y., Degraen, D., Fischman, B., Häkkilä, J., Kuehl, K., Nisi, V., Nunes, N. J., Wenig, N. and Wenig, D. (2017). The geography of Pokémon GO: beneficial and problematic effects on places and movement. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. New York: ACM, pp. 1179-1192. Available at: doi:10.1145/3025453.3025495.

Consalvo, M. (2007). Cheating: Gaining Advantage in Videogames. Cambridge: MIT Press: p. 1 Copier, M. (2003). The other game researcher: Participating in and watching the construction of boundaries in game studies. In Raessens, J. and Copier, M. (Eds). Proceedings of DiGRA Level Up Conference 2003. Utrecht: Utrecht University Press, pp. 404-419.

Copier, M. (2005). Connecting worlds: Fantasy role-playing games, ritual acts, and the magic circle. In Proceedings of DiGRA 2005 Conference. Vancouver: Digital Games Research Association. Available at: http://www.digra.org/wpcontent/uploads/digital-library/06278.50594.pdf [Accessed 10 May 2023].

Crang, M. (2003). Rhythms of the city: Temporalised space and motion. In May, J. and Thrift, N. (Eds). Timespace: Geographies of temporality. New York: Routledge, pp. 187-207. Available at: doi:10.4324/9780203360675-14.

Crawford, C. (1984). The art of computer game design. Washington: McGraw-Hill / Osborne Media.

Crawford, G. (2012). Video gamers. London: Routledge. Available at: doi:10.4324/9780203863374.

Crawford, G. (2015). Is it in the game? Reconsidering play spaces, game definitions, theming, and sports video games. Games and Culture, 10(6), 571-592. Available at: doi:10.1177/1555412014566235.

Crawford, G. and Gosling, V. (2005). Toys for boys? Women's marginalization and participation as digital gamers. Sociological Research Online, 10(1), 46-56. Available at: doi:10.5153/sro.1024. Crawford, G. and Rutter, J. (2006). Digital games and cultural studies. In Rutter, J. and Bryce, J. (Eds). Understanding digital games. London: SAGE Publications Ltd, pp. 148-165. Available at: doi:10.4135/9781446211397.n9.

Curtis, P. (1992). Mudding: Social phenomena in text-based virtual realities. In Ludlow, P. (Ed). High noon on the electronic frontier: Conceptual issues in cyberspace. Cambridge: The MIT Press, pp. 347-374.

Curtis, P. and Nichols, D. A. (1994). MUDs grow up: Social virtual reality in the real world. In Proceedings of COMPCON'94. San Francisco: IEEE, pp. 193-200.

Cutting, J., Gundry, D. and Cairns, P. (2019). Busy doing nothing? What do players do in idle games? International Journal of Human-Computer Studies, 122, 133-144. Available at: doi:10.1016/j.ijhcs.2018.09.006.

D'Aloia, A. (2009). Adamant bodies: The avatar-body and the problem of autoempathy. E/C, 3(ns 5), 51-58.

Dayan, P. (2019). How musical is Henri Lefebvre's rhythmanalysis? In Christiansen, S. L. and Gebauer, M. (Eds). Rhythms now: Henri Lefebvre's rhythmanalysis revisited. Aalborg: Aalborg Universitetsforlag, pp. 5-16.

De Grove, F. (2014). Youth, friendship, and gaming: A network perspective. Cyberpsychology, Behaviour, and Social Networking, 17(9), 603-608. Available at: doi:10.1089/cyber.2014.0088. De Souza e Silva, A. (2006). From cyber to hybrid: Mobile technologies as interfaces of hybrid spaces. Space and Culture, 9(3), 261-278. Available at: doi:10.1177/1206331206289022.

De Souza e Silva, A. (2009). Hybrid reality and location-based gaming: Redefining mobility and game spaces in urban environments. Simulation & Gaming, 40(3), 404-424. Available at: doi:10.1177/1046878108314643.

De Souza e Silva, A. (2023). Hybrid spaces 2.0: Connecting networked urbanism, uneven mobilities, and creativity, in a (post) pandemic world. Mobile Media & Communication, 11(1), 59-65. Available at: doi:10.1177/20501579221132118.

Denham, J. and Spokes, M. (2020). The right to the virtual city: Rural retreatism in open-world video games. New Media & Society, 23(6), 1567-1583. Available at: doi:10.1177/1461444820917114.

Deterding, N. M. and Waters, M. C. (2021). Flexible coding of in-depth interviews: A twenty-first-century approach. Sociological Methods & Research, 50(2), 708-739. Available at: doi:10.1177/0049124118799377.

Dibbell, J. (2007). The life of the Chinese gold farmer. The New York Times. [Online]. 9 December 2007. Available at:

http://www.nytimes.com/2005/12/08/technology/08iht-gaming.html [Accessed 21 May 2023].

Domahidi, E. and Quandt, T. (2015). 'And all of a sudden my life was gone...': A biographical analysis of highly engaged adult gamers. New Media & Society, 17(7), 1154-1169. Available at: doi:10.1177/1461444814521791. Dominick, J. R. (1984). Videogames, television violence, and aggression in teenagers. Journal of Communication, 34(2), 136-147. Available at: doi:10.1111/j.1460-2466.1984.tb02165.x.

Drachen, A., El-Nasr, M. S. and Canossa, A. (2013). Game analytics: The basics. In Seif El-Nasr, M., Drachen, A. and Canossa, A. (Eds). Game analytics: Maximizing the value of player data. London: Springer, pp. 13-40. https://doi.org/10.1007/978-1-4471-4769-5\_2

Drummond, A. and Sauer, J. D. (2018). Video game loot boxes are psychologically akin to gambling. Nature Human Behaviour, 2(8), 530-532. Available at: doi:10.1038/s41562-018-0360-1.

Dubbell, J. (2015). Invisible labor, invisible play: Online gold farming and the boundary between jobs and games. Vanderbilt Journal of Entertainment and Technology Law, 18, 419.

Duggan, G. B., Johnson, H. and Sørli, P. (2013). Interleaving tasks to improve performance: Users maximize the marginal rate of return. International Journal of Human-Computer Studies, 71(5), 533-550. Available at: doi:10.1016/j.ijhcs.2013.01.001.

Dyer-Witheford, N. and De Peuter, G. (2009). Games of empire: Global capitalism and video games. Minneapolis: University of Minnesota Press.

Edirisinghe, C. and Cheok, A. D. (2016). Robots and intimacies: A preliminary study of perceptions, and intimacies with robots. In International Conference on Love and Sex with Robots. Cham: Springer, pp. 137-147. Available at: doi:10.1007/978-3-319-57738-8\_13.

Erikson, E. H. (1993). Childhood and society. London: WW Norton & Company.

Farman, J. (2009). Locative life: Geocaching, mobile gaming, and embodiment. Body & Society, 19(1), 27-51.

Fields, T. and Cotton, B. (2011). Social game design: Monetization methods and mechanics. London: CRC Press. Available at: doi:10.1201/9780240817675.

Fiske, A. P. (1998). Human sociality. International Society for the Study of Personal Relationships Bulletin, 14(2), 4-9.

Friedman, T. (1998). Civilization and its discontents: Simulation, subjectivity, and space. In Smith, G. (Ed). On a silver platter. New York: New York University Press, pp. 132-150. Available at:

doi:10.18574/nyu/9780814786734.003.0008.

Friedman, T. (1999). The semiotics of SimCity. First Monday, 4(4). Available at: doi:10.5210/fm.v4i4.660.

Frith, J. (2013). Turning life into a game: Foursquare, gamification, and personal mobility. Mobile Media & Communication, 1(2), 248-262. Available at: doi:10.1177/2050157912474811.

Fuchs, T. (2014). The virtual other: Empathy in the age of virtuality. Journal of Consciousness Studies, 21(5-6), 152-173.

Gal, D. (2006). A psychological law of inertia and the illusion of loss aversion. Judgment and Decision Making, 1(1), 23-32. Available at: doi:10.1017/S1930297500000322.

GameSoftware (1995). Crow, corw, caw. [Online]. GameSoftware. Available at: https://games.qq.com/a/20111108/000189.htm [Accessed 1 February 2023].

Ganzon, S. C. (2019). Investing time for your in-game boyfriends and BFFs: Time as commodity and the simulation of emotional labor in Mystic Messenger. Games and Culture, 14(2), 139-153. Available at: doi:10.1177/1555412018793068.

Gao, L., Gan, Y., Whittal, A., Yan, S. and Lippke, S. (2020). The mediator roles of problematic internet use and perceived stress between health behaviours and work-life balance among internet users in Germany and China: Web-based cross-sectional study. Journal of Medical Internet Research, 22(5), e16468. Available at: doi:10.2196/16468.

Gee, J. P. (2008). Video games and embodiment. Games and Culture, 3(3-4), 253-263. Available at: doi:10.1177/1555412008317309.

Genshin Impact Wiki (2023). Artifact. [Online]. Genshin Impact Wiki. Available at: https://genshin-impact.fandom.com/wiki/Artifact/Sets [Accessed 29 December 2023].

Genshin Impact Wiki (2023). Original resin. [Online]. Genshin Impact Wiki. Available at: https://genshin-impact.fandom.com/wiki/Original\_Resin [Accessed 29 December 2023].

Giddings, S. (2007). Playing with non-humans: Digital games as technocultural form. In de Castell, S. and Jenson, J. (Eds). Worlds in play: International perspectives on digital games research. New York: Peter Lang, pp. 115-128.

Goffman, E. (1949). Presentation of self in everyday life. American Journal of Sociology, 55, 6-7.

Goggin, J. (2011). Playbour, farming and leisure. Ephemera: Theory & Politics in Organization, 11(4), 357-368.

Gong, W., Lim, E. P. and Zhu, F. (2015). Characterizing silent users in social media communities. Proceedings of the International AAAI Conference on Web and Social Media, 9(1), 140-149. Available at: doi:10.1609/icwsm.v9i1.14582.

Gotham, K. F. (2005). Theorizing urban spectacles: Festivals, tourism and the transformation of urban space. City, 9, 225-246. Available at: doi:10.1080/13604810500197020.

Greitemeyer, T. and Cox, C. (2013). There's no "I" in team: Effects of cooperative video games on cooperative behavior. European Journal of Social Psychology, 43(3), 224-228. Available at: doi:10.1002/ejsp.1940.

Guest Writer of Deconstructor of Fun (2016). The making of a mechanic: Daily goals. [Online]. Guest Writer of Deconstructor of Fun. Available at: https://www.deconstructoroffun.com/blog//2016/07/the-making-of-mechanic-daily-goals.html [Accessed 31 January 2023].

Guo, Q., Wang, Y. and Yang, H. (2019). Analysis of user engagement under freemium business model -- an empirical study based on online games.

Hägerstrand, T. (1982). Diorama, path and project. Tijdschrift voor Economische En Sociale Geografie, 73(6), 323-339. Available at: doi:10.1111/j.1467-9663.1982.tb01647.x.

Hamari, J. and Järvinen, A. (2011). Building customer relationship through game mechanics in social games. In Cruz-Cunha, M., Varvalho, V. and Tavares, P. (Eds). Business, technological, and social dimensions of computer games: Multidisciplinary developments. London: IGI Global, pp. 348-365. Available at: doi:10.4018/978-1-60960-567-4.ch021.

Haniff, N. Z. (1985). Toward a native anthropology: Methodological notes on a study of successful Caribbean women by an insider. Anthropology and Humanism Quarterly, 10(4), 107-113. Available at: doi:10.1525/ahu.1985.10.4.107.

Haraway, D. (1985). A manifesto for cyborgs: Science, technology, and socialist feminism in the 1980s. Socialist Review, 15(2), 65–107.

Hassard, J. (Ed.). (2016). The sociology of time. London: Springer.

Heidegger, M. (1996). Being and time: A translation of Sein und Zeit. London: SUNY Press.

Heimo, O. I., Harviainen, J. T., Kimppa, K. K. and Mäkilä, T. (2018). Virtual to virtuous money: A virtue ethics perspective on video game business logic.

Journal of Business Ethics, 153(1), 95-103. Available at: doi:10.1007/s10551-016-3408-z.

Hjorth, L. and Richardson, I. (2009). The waiting game: Complicating notions of (tele)presence and gendered distraction in casual mobile gaming. Australian Journal of Communication, 36(1), 23-35.

Hommadova Lu, A. and Carradini, S. (2020). Work-game balance: Work interference, social capital, and tactical play in a mobile massively multiplayer online real-time strategy game. New Media & Society, 22(12), 2257-2280. Available at: doi:10.1177/1461444819889957.

Hommadova Lu, A. and Carradini, S. (2020). Work–game balance: Work interference, social capital, and tactical play in a mobile massively multiplayer online real-time strategy game. New Media & Society, 22(12), 2257-2280. Available at: doi:10.25384/SAGE.c.4763117.v1.

Hong, Y. (2008). Distinctive characteristics of China's path of ICT development: a critical analysis of Chinese developmental strategies in light of the Eastern Asian model. International Journal of Communication, 2, 456-471.

Huizinga, J. (1938/2016). Homo ludens. London: Angelico Press.

Huvila, I., Holmberg, K., Ek, S. and Widén - Wulff, G. (2010). Social capital in second life. Online Information Review, 34(2), 295-316. Available at: doi:10.1108/14684521011037007.

Interactive Software Federation of Europe (2021). Key facts. [Online]. Interactive Software Federation of Europe. Available at:

https://www.isfe.eu/data-key-facts/key-facts-from-2021-europe-video-gamessector/ [Accessed 10 May 2023].

Isbister, K. (2016). How games move us. London: MIT Press. Available at: doi:10.7551/mitpress/9267.001.0001.

Isbister, K. C. (1998). Reading personality in onscreen interactive characters: an examination of social psychological principles of consistency, personality match, and situational attribution applied to interaction with characters. Unpublished: Stanford University. PhD.

Iser, Wolfgang (1974). The implied reader: Patterns of communication in prose fiction from Bunyan to Beckett. Baltimore: Johns Hopkins University Press.

Jarrett, J. (2021). Gaming the gift: The affective economy of League of Legends 'fair' free-to-play model. Journal of Consumer Culture, 21(1), 102-119. Available at: doi:10.1177/1469540521993932.

Jinghan, L. (2020). Research on media image of online game players--A case study of People's Daily Online report (2012-2020).

Johan, H. (1955). Homo ludens: A study of the play-element in culture. Boston: Beacon.

Johnson, M. R. and Brock, T. (2019). How are video games and gambling converging. [Online]. Guelph: Gambling Research Exchange Ontario. Available at: https://www.greo.ca/Modules/EvidenceCentre/files/Johnson%20and%20Brock, 20.

Johnson, M. R. and Brock, T. (2020). The 'gambling turn' in digital game monetization. Journal of Gaming & Virtual Worlds, 12(2), 145-163. Available at: doi:10.1386/jgvw\_00011\_1.

Jordan, B. (2009). Blurring boundaries: The 'real' and the 'virtual' in hybrid spaces. Human Organization, 68(2), 181-193. Available at: doi:10.17730/humo.68.2.7x4406g270801284.

Joseph, D. (2021). Battle Pass Capitalism. Journal of Consumer Culture, 21(1), 68-83. Available at: doi:10.1177/1469540521993930.

Juul, J. (2005). Half-real: Video games between real rules and fictional worlds. London: MIT Press.

Juul, J. (2008). The magic circle and the puzzle piece. In Günzel, S., Liebe, M. and Mersch, D. (Eds). Conference proceedings of the Philosophy of Computer Games 2008. Potsdam: Potsdam University Press, pp. 56-67. Available at: http://pub.ub.uni-potsdam.de/volltexte/2008/2455/.

Juul, J. (2010). A casual revolution: Reinventing video games and their players. Cambridge: MIT Press.

Kahneman, D. and Tversky, A. (1977). Intuitive prediction: biases and corrective procedures (Technical Report PTR-1042-77.6). Defense Technical Information Center. [Online]. Available at:

http://www.dtic.mil/dtic/tr/fulltext/u2/a047747.pdf [Accessed 10 March 2023].

Kahneman, D. and Tversky, A. (1979). Prospect theory: An analysis of decision under risk. Econometrica, 47(2), 263-292. Available at: doi:10.2307/1914185.

Kahneman, D. and Tversky, A. (2013). Prospect theory: An analysis of decision under risk. In MacLean, L. C. and Ziemb, W. T. (Eds). Handbook of the fundamentals of financial decision making: Part I. London: World Scientific, pp. 99-127. Available at: doi:10.1142/9789814417358\_0006.

Karlsen, F. (2018). Exploited or engaged? Dark game design patterns in Clicker Heroes, FarmVille 2 and World of Warcraft. In Jørgensen, K. and Karlsen, F. (Eds). Transgression in games and play. London: MIT Press. Available at: doi:10.7551/mitpress/11550.003.0019.

Karpashevich, P., Hornecker, E., Dankwa, N. K., Hanafy, M. and Fietkau, J. (2016). Blurring boundaries between everyday life and pervasive gaming: An interview study of Ingress. In Proceedings of the 15th international conference on mobile and ubiquitous multimedia. London: ACM, pp. 217-228. Available at: doi:10.1145/3012709.3012716.

Keogh, B. (2014). Across worlds and bodies: Criticism in the age of video games. Journal of Games Criticism, 1(1), 1-26.

King, C. A. P. and Kirschenbaum, D. S. (1992). Helping young children develop social skills: The social growth program. Pacific Grove: Brooks/Cole Publishing Company.

Klimmt, C., Hefner, D. and Vorderer, P. (2009). The video game experience as 'true' identification: A theory of enjoyable alterations of players' self-

perception. Communication Theory, 19(4), 351-373. Available at: doi:10.1111/j.1468-2885.2009.01347.x.

Koeder, M. J. and Tanaka, E. (2017). Game of chance elements in free-toplay mobile games: A freemium business model monetization tool in need of self-regulation? 28th European Regional ITS Conference, Passau 2017 169473, International Telecommunications Society (ITS).

Korczynski, M., Pickering, M. and Robertson, E. (2013). Rhythms of labour: Music at work in Britain. Cambridge: Cambridge University Press. Available at: doi:10.1017/CBO9781139030311.

Kowert, R., Festl, R. and Quandt, T. (2014). Unpopular, overweight, and socially inept: Reconsidering the stereotype of online gamers. Cyberpsychology, Behaviour, and Social Networking, 17(3), 141-146. Available at: doi:10.1089/cyber.2013.0118.

Kücklich, J. (2005). Precarious playbour: Modders and the digital games industry. Fibreculture, 5(1), 1-5.

Labaree, R. V. (2002). The risk of 'going observationalist': Negotiating the hidden dilemmas of being an insider participant observer. Qualitative Research, 2(1), 97-122. Available at: doi:10.1177/1468794102002001641.

Lai, G. and Fung, K. Y. (2020). From online strangers to offline friends: A qualitative study of video game players in Hong Kong. Media, Culture & Society, 42(4), 483-501. Available at: doi:10.1177/0163443719853505.

Lammes, S. (2008). Spatial regimes of the digital playground: Cultural functions of spatial practices in computer games. Space and Culture, 11(3), 260-272. Available at: doi:10.1177/1206331208319150.

Lazzarato, M. (1996). Immaterial labor. In Virno, P. and Hardt, M. (Eds), Radical thought in Italy: A potential politics. Minneapolis: University of Minnesota Press, pp. 142-157.

Lee, E. H., Kang, J., Kim, J., Ahn, J., Kang, K. G. Y., Kim, J., Lee, S., Jo, S. and Nam, K. (2020). The effects of video games on aggression, sociality, and affect: A meta-analytic study. Science of Emotion and Sensibility, 23(4), 41-60. Available at: doi:10.14695/KJSOS.2020.23.4.41.

Lefebvre, H. (2006). Critique of everyday life: From modernity to modernism (towards a metaphilosophy of daily life). New York: Verso Books.

Lefebvre, H. (2013). Rhythmanalysis: Space, time and everyday life. London: Bloomsbury Publishing. Available at: doi:10.5040/9781350284838.

Levy, D. (2009). Love and sex with robots: The evolution of human-robot relationships. New York: Harper Collins.

Li, W., Mills, D. and Nower, L. (2019). The relationship of loot box purchases to problem video gaming and problem gambling. Addictive Behaviors, 97, 27-34. Available at: doi:10.1016/j.addbeh.2019.05.016.

Liboriussen, B. (2016). Amateur gold farming in China: 'Chinese ingenuity,' independence, and critique. Games and Culture, 11(3), 316-331. Available at: doi:10.1177/1555412015598603.

Liebe, M. (2008). There is no magic circle: On the difference between computer games and traditional games. In Günzel, S., Liebe, M. and Mersch, D. (Eds). Conference proceedings of the philosophy of computer games 2008. Potsdam, Germany: Potsdam University Press, pp. 324-341. Available at: http://pub.ub.uni-potsdam.de/volltexte/2008/2459/.

Lindstrom, C. (2019). Mod money, mod problems: A critique of copyright restrictions on video game modifications and an evaluation of associated monetization regimes. William & Mary Business Law Review, 11, 811.

Lizardi, R. (2012). DLC: Perpetual commodification of the video game. Democratic Communiqué, 25(1), 33-45. Available at: doi:10.7275/democraticcommunique.278.

Lukacs, A., Embrick, D. G. and Wright, T. (2010). The managed hearthstone: Labor and emotional work in the online community of World of Warcraft. In Facets of virtual environments: First International Conference, FaVE 2009, Berlin, Germany, July 27-29, 2009, Revised Selected Papers 1. Berlin: Springer Berlin Heidelberg, pp. 165-177. Available at: doi:10.1007/978-3-642-11743-5 13.

Lyon, D. (2018). What is rhythmanalysis? London: Bloomsbury Publishing. Available at: doi:10.5040/9781350018310.

Malliet, S. (2006). An exploration of adolescents' perceptions of video game realism. Learning, Media and Technology, 31(4), 377-394. Available at: doi:10.1080/17439880601021983.

Margraf, J., Lavallee, K., Zhang, X. and Schneider, S. (2016). Social rhythm and mental health: A cross-cultural comparison. PloS One, 11(3), e0150312. Available at: doi:10.1371/journal.pone.0150312.

Matsuo, Y., Tajima, S., Nohara, S. and Sakamoto, A. (2007). Video games and the training of sociality: A survey of video game players. In DiGRA Conference. Vancouver: Digital Games Research Association. Available at: doi:10.9762/digraj.2.1\_23.

Mäyrä, F. (2019). The player as a hybrid: Agency in digital game cultures. GAME: The Italian Journal of Game Studies, 2019(8), 29–45.

McCrea, C. (2011). We play in public: The nature and context of portable gaming systems. Convergence, 17(4), 389-403. Available at: doi:10.1177/1354856511414987.

Merleau-Ponty, M. (1962). Phenomenology of perception. New York: Humanities Press.

Merleau-Ponty, M. (2002). Phenomenology of perception. London: Routledge Classics. Available at: doi:10.4324/9780203994610.

Merton, R. K. (1972). Insiders and outsiders: A chapter in the sociology of knowledge. American Journal of Sociology, 78(1), 9-47. Available at: doi:10.1086/225294.

MobTech (2018). The China mobile game market research report, Q1 of 2018. [Online]. Sohu. Available at: https://www.sohu.com/a/230714471\_99975453 [Accessed 10 May 2023]. Molesworth, M. (2009). Adults' consumption of videogames as imaginative escape from routine. Advances in Consumer Research, 36, 378-383.

Mulíček, O., Osman, R. and Seidenglanz, D. (2016). Time-space rhythms of the city: The industrial and postindustrial Brno. Environment and Planning A, 48(1), 115-131. Available at: doi:10.1177/0308518X15594809.

Muramatsu, J. and Ackerman, M. S. (1998). Computing, social activity, and entertainment: A field study of a game MUD. Computer Supported Cooperative Work (CSCW), 7(1-2), 87-122. Available at: doi:10.1023/A:1008636204963.

Murdoch, J. (1997). Inhuman/nonhuman/human: Actor-network theory and the prospects for a nondualistic and symmetrical perspective on nature and society. Environment and Planning D: Society and Space, 15(6), 731-756. Available at: doi:10.1068/d150731.

Musick, G., Freeman, G. and McNeese, N. J. (2021). Gaming as family time: Digital game co-play in modern parent-child relationships. Proceedings of the ACM on Human-Computer Interaction, 5(CHI PLAY), 1-25. Available at: doi:10.1145/3474678.

Newzoo (2018). China games market 2018. [Online]. Newzoo. Available at: https://newzoo.com/insights/infographics/china-games-market-2018/ [Accessed 10 May 2023].

Newzoo (2019). Global games market report. [Online]. Newzoo. Available at: https://newzoo.com/solutions/standard/market-forecasts/global-gamesmarket-report/ [Accessed 10 May 2023].

Nieborg, D. B. (2015). Crushing candy: The free-to-play game in its connective commodity form. Social Media + Society, 1(2), 2056305115621932. Available at: doi:10.1177/2056305115621932.

Nikopartners (2019). China mobile games. [Online]. Nikopartners. Available at: https://nikopartners.com/china-mobile-games/ [Accessed 10 May 2023].

Nikopartners (2019). Chinese gamer segmentation study. [Online]. Nikopartners. Available at: https://nikopartners.com/chinese-gamersegmentation-analysis-study/ [Accessed 10 May 2023].

Nikopartners (2023). China console games & 5 year forecast report. [Online]. Nikopartners. Available at: https://nikopartners.com/china-console-games/ [Accessed 3 April 2024].

Nitsche, M. (2008). Video game spaces: Image, play, and structure in 3D game worlds. Cambridge: MIT Press. Available at: doi:10.7551/mitpress/9780262141017.001.0001.

Nowotny, H. (2018). Time: The modern and postmodern experience. London: John Wiley & Sons.

Oldenburg, R. (1999). The great good place: Cafes, coffee shops, bookstores, bars, hair salons, and other hangouts at the heart of a community. London: Da Capo Press.

Orland, K. (2018). Video game loot boxes are now considered criminal gambling in Belgium. [Online]. Ars Technica. Available at:
https://arstechnica.com/gaming/2018/04/video-game-loot-boxes-are-nowconsidered-criminal-gambling-in-belgium/ [Accessed 10 May 2023].

Osborne, H. (2012). Performing self, performing character: Exploring gender performativity in online role-playing games. Transformative Works and Cultures, 11. Available at: doi:10.3983/twc.2012.0411.

Osmanovic, S. and Pecchioni, L. (2016). Beyond entertainment: Motivations and outcomes of video game playing by older adults and their younger family members. Games and Culture, 11(1-2), 130-149. Available at: doi:10.1177/1555412015602819.

Parkes, D. and Thrift, N. (1975). Timing space and spacing time. Environment and Planning A, 7(6), 651-670. Available at: doi:10.1068/a070651.

Paul, C. A. (2018). The toxic meritocracy of video games: Why gaming culture is the worst. Minneapolis: University of Minnesota Press. Available at: doi:10.5749/j.ctt2204rbz.

Pearce, K. E., Yip, J. C., Lee, J. H., Martinez, J. J., Windleharth, T. W., Bhattacharya, A. and Li, Q. (2022). Families playing animal crossing together: Coping with video games during the COVID-19 pandemic. Games and Culture, 17(5), 773-794. Available at: doi:10.1177/15554120211056125.

Pecchioni, L. L. and Osmanovic, S. (2018). Play it again, grandma: Effect of intergenerational video gaming on family closeness. In Human aspects of IT for the aged population. Acceptance, communication and participation: 4th international conference, ITAP 2018, held as part of HCI International 2018, Las Vegas, NV, USA, July 15-20, 2018, proceedings, part I. London: Springer

International Publishing, pp. 518-531. Available at: doi:10.1007/978-3-319-92034-4 39.

Peng, W. and Hsieh, G. (2012). The influence of competition, cooperation, and player relationship in a motor performance centered computer game. Computers in Human Behavior, 28(6), 2100-2106. Available at: doi:10.1016/j.chb.2012.06.014.

People's Daily Online (2019). The anniversary of the game licence restoration approval: The industry ecology changed. [Online]. People's Daily Online. Available at:

https://baijiahao.baidu.com/s?id=1653215571864026188&wfr=spider&for=pc [Accessed 23 September 2023].

Perry, R. B. (1910). The ego-centric predicament. The Journal of Philosophy, Psychology and Scientific Methods, 7(1), 5-14. Available at: doi:10.2307/2011576.

Peter, B. (1984). Reading for the plot: Design and intention in narrative. New York: A.A. Knopf.

Prey, R. (2015). Henri Lefebvre and the production of music streaming spaces. Sociologica, 9(3), 1-22. Available at: doi:10.2383/82481.

Priebatsch, S. (2010). Welcome to the decade of games. [Online]. HBR Blog Network, Harvard Business Review. Last updated: September 2010. Available at: https://hbr.org/2010/09/welcome-to-the-decade-ofgames [Accessed 1 February 2023]. Reid, E. (1994). Cultural formations in text-based virtual realities. University of Melbourne. M.A. Thesis.

Rettberg, J. W. (2008). Quests in World of Warcraft: Deferral and repetition. London: MIT Press.

Rettberg, S. (2008). Corporate ideology in World of Warcraft. In Corneliussen, H. G. and Rettberg, J. W. (Eds). Digital culture, play, and identity: A World of Warcraft reader. London: MIT Press, pp. 19-38.

Ribbens, W. and Malliet, S. (2010). Perceived digital game realism: A quantitative exploration of its structure. Presence, 19(6), 585-600. Available at: doi:10.1162/pres\_a\_00024.

Rideout, V. J., Foehr, U. G. and Roberts, D. F. (2010). Generation M2: Media in the lives of 8-to 18-year-olds. London: Henry J. Kaiser Family Foundation.

Ritzer, G. and Jurgenson, N. (2010). Production, consumption, prosumption: The nature of capitalism in the age of the digital 'prosumer'. Journal of Consumer Culture, 10(1), 13-36. Available at:

doi:10.1177/1469540509354673.

Rose, M. (2013). Chasing the whale: Examining the ethics of free-to-play games. [Online]. Gamasutra. Available:

https://www.gamedeveloper.com/business/chasing-the-whale-examining-theethics-of-free-to-play-games [Accessed 10 September 2024]. Ross, A. and Nieborg, D. (2021). Spinning is winning: Social casino apps and the platformization of gamble-play. Journal of Consumer Culture, 21(1), 84-101. Available at: doi:10.1177/1469540521993931.

Ruffino, P. (2021). The end of capitalism: Disengaging from the economic imaginary of incremental games. Games and Culture, 16(2), 208-227. Available at: doi:10.1177/1555412019886242.

Salen, K., Tekinbaş, K. S. and Zimmerman, E. (2004). Rules of play: Game design fundamentals. London: MIT Press.

Salvucci, D. D. and Taatgen, N. A. (2008). Threaded cognition: An integrated theory of concurrent multitasking. Psychological Review, 115(1), 101. Available at: doi:10.1037/0033-295X.115.1.101.

Sasson, S. (2015). How to build a smart game economy. [Online]. TechCrunch. Available at: https://social.techcrunch.com/2015/08/05/how-tobuild-a-smart-game-economy/ [Reviewed 10 February 2017].

Schwanen, T., Van Aalst, I., Brands, J. and Timan, T. (2012). Rhythms of the night: Spatiotemporal inequalities in the nighttime economy. Environment and Planning A, 44(9), 2064-2085. Available at: doi:10.1068/a44494.

Schwiddessen, S. and Karius, P. (2018). Watch your loot boxes! Recent developments and legal assessment in selected key jurisdictions from a gambling law perspective. Interactive Entertainment Law Review, 1(1), 17-43. Available at: doi:10.4337/ielr.2018.01.02.

Shen, C., Monge, P. and Williams, D. (2014). The evolution of social ties online: A longitudinal study in a massively multiplayer online game. Journal of the Association for Information Science and Technology, 65(10), 2127-2137. Available at: doi:10.1002/asi.23129.

Shokrizade, R. (2013). The top F2P monetization tricks. [Online]. Gamasutra. Available at:

http://www.gamasutra.com/blogs/RaminShokrizade/20130626/194933/The\_T op\_F2P\_Monetization\_Tricks.php [Accessed 1 February 2023].

Sina Game (2011). A ten-year history of gaming in South Korea with an average annual growth rate of more than 20 percent. [Online]. Sina Game. Available at: https://games.sina.com.cn/o/n/2011-03-18/1147483079.shtml [Accessed 1 February 2023].

Sina Game (2012). The seven year itch of in-game items selling, analyze the China game industry charging model. [Online]. Sina Game. Available at: http://games.sina.com.cn/y/n/2012-07-02/1625631574.shtml [Accessed 1 February 2023].

Skinner, B. F. (1972). A case history in scientific method. In Skinner, B. F. (Ed). Cumulative record: A selection of papers. New York: MeredithCorporation, pp. 101-124.

Smythe, D. W. (1981). Dependency road: Communications, capitalism, consciousness, and Canada. Norwood: Ablex.

Snodgrass, J. G., Lacy, M. G., Dengah, H. F., Batchelder, G., Eisenhower, S. and Thompson, R. S. (2016). Culture and the jitters: Guild affiliation and

online gaming eustress/distress. Ethos, 44(1), 50-78. Available at: doi:10.1111/etho.12108.

Sohu News (2020). Communication University of China creates 'online graduation ceremony', restores 'campus' with gaming platform. [Online]. Sohu News. Available at:

https://k.sina.com.cn/article\_1644114654\_61ff32de020010h7y.html?from=edu [Accessed 1 February 2023].

Sommerseth, H. (2007, January). "Gamic realism": Player, perception and action in video game play. In Proceedings of DiGRA 2007 Conference: Situated Play. Vancouver: Digital Games Research Association.

Song, W. and Fox, J. (2016). Playing for love in a romantic video game: Avatar identification, parasocial relationships, and Chinese women's romantic beliefs. Mass Communication and Society, 19(2), 197-215. Available at: doi:10.1080/15205436.2015.1077972.

Southerton, D. and Tomlinson, M. (2005). 'Pressed for time': The differential impacts of a 'time squeeze.' The Sociological Review, 53(2), 215-239. Available at: doi:10.1111/j.1467-954X.2005.00511.x.

Stanney, K. M. and Salvendy, G. (1998). Aftereffects and sense of presence in virtual environments: Formulation of a research and development agenda. International Journal of Human-Computer Interaction, 10, 135-187. Available at: doi:10.1207/s15327590ijhc1002 2.

Statista (2019). Cashing in on free-to-play games. [Online]. Statista. Available at: https://www.statista.com/chart/20395/worldwide-revenue-generated-with-free-to-play-video-games/ [Accessed 23 September 2023].

Statista (2021a). Number of mobile gaming users worldwide in 2021, by region. [Online]. Statista. Available at:

https://www.statista.com/statistics/512112/number-mobile-gamers-world-byregion/ [Accessed 23 September 2023].

Statista (2021b). Number of mobile gamers in China from 2012 to 2022.

[Online]. Statista. Available at:

https://www.statista.com/statistics/310208/china-number-of-mobile-gamers/ [Accessed 12 August 2023].

Statista (2023). Number of mobile gamers in China from 2013 to 2023.

[Online]. Statista. Available at:

https://www.statista.com/statistics/310208/china-number-of-mobile-gamers/ [Accessed 3 April 2024].

Stegner, B. (2021). What is 'pay-to-win' in video games? [Online]. MakeUseOf. Available at: https://www.makeuseof.com/what-is-pay-to-win-in-video-games/ [Accessed 15 March 2024].

Steinkuehler, C. A. and Williams, D. (2006). Where everybody knows your (screen) name: Online games as 'third places'. Journal of Computer-Mediated Communication, 11(4), 885-909. Available at: doi:10.1111/j.1083-6101.2006.00300.x.

Stenros, J. (2014). In defence of a magic circle: The social, mental and cultural boundaries of play. Transactions of the Digital Games Research Association, 1(2), 147–185. Available at: doi:10.26503/todigra.v1i2.10.

Stenros, J., Waern, A. and Montola, M. (2012). Studying the elusive experience in pervasive games. Simulation & Gaming, 43(3), 339-355. Available at: doi:10.1177/1046878111422532.

Sullins, J. P. (2012). Robots, love, and sex: The ethics of building a love machine. IEEE Transactions on Affective Computing, 3(4), 398-409. Available at: doi:10.1109/T-AFFC.2012.31.

Sun, X. and Lin, H. (2023, November). Research on the gameplay evolution based on Warcraft 3 mod platform. In Ciancarini, P., Di Iorio, A., Hlavacs, H. and Poggi, F. (Eds). Entertainment computing – ICEC 2023. ICEC 2023. Lecture notes in computer science, vol 14455. Singapore: Springer Nature Singapore, pp. 306-314. Available at: doi:10.1007/978-981-99-8248-6 25.

SuperData (2019). 2019 year in review: Digital games and interactive media. [Online]. SuperData. Available at: https://www.superdataresearch.com/2019year-in-review/ [Accessed 23 September 2023].

Swink, S. (2008). Game feel: A game designer's guide to virtual sensation. London: CRC Press. Available at: doi:10.1201/9781482267334.

Tai, Z. and Hu, F. (2018). Play between love and labor: The practice of gold farming in China. New Media & Society, 20(7), 2370-2390. Available at: doi:10.1177/1461444817717326.

Takahashi, D. (2014). Only 0.15 percent of mobile gamers account for 50 percent of all in-game revenue (exclusive). [Online]. VentureBeat, Inc. Available at: https://venturebeat.com/games/only-0-15-of-mobile-gamers-account-for-50-percent-of-all-in-game-revenue-exclusive/ [Accessed 23 September 2023].

Talking Data (2017). The report of Arena of Valor. [Online]. TalkingData. Available at: http://mi.talkingdata.com/report-detail.html?id=529 [Accessed 1 February 2023].

Taylor, T. L. (2006). Does WoW change everything? How a PvP server, multinational player base, and surveillance mod scene caused me pause. Games and Culture, 1(4), 318-337. Available at: doi:10.1177/1555412006292615.

Taylor, T. L. (2009). Play between worlds: Exploring online game culture. London: MIT Press.

Terranova, T. (2000). Free labor: Producing culture for the digital economy. Social Text, 18(2), 33-58. Available at: doi:10.1215/01642472-18-2\_63-33.

Thompson, C. (2008). Back to the grind in WoW - and loving every tedious minute. [Online]. Wired. Last updated: 28 July 2008. Available at: https://www.wired.com/2008/07/gamesfrontiers-0728/ [Accessed 10 May 2023].

Thompson, E. P. (1967). Time, work-discipline, and industrial capitalism. Past & Present, 38, 56-97. Available at: doi:10.1093/past/38.1.56.

Trepte, S., Reinecke, L. and Juechems, K. (2012). The social side of gaming: How playing online computer games creates online and offline social support. Computers in Human Behavior, 28(3), 832-839. Available at: doi:10.1016/j.chb.2011.12.003.

Turing, A. M. (1950). Computing machinery and intelligence. Mind, 59, 433– 460. Available at: doi:10.1093/mind/LIX.236.433.

Turkle, S. (1994). Constructions and reconstructions of self in virtual reality: Playing in the MUDs. Mind, Culture, and Activity, 1(3), 158-167. Available at: doi:10.1080/10749039409524667.

Van Rooij, A. J., Schoenmakers, T. M., Vermulst, A. A., Van Den Eijnden, R. J. and Van De Mheen, D. (2011). Online video game addiction: Identification of addicted adolescent gamers. Addiction, 106(1), 205-212. Available at: doi:10.1111/j.1360-0443.2010.03104.x.

Voorhees, G. (2015). Neoliberal masculinity: The government of play and masculinity in e-sports. In Brookey, R. A. and Oates, T. P. (Eds). Playing to win: Sports, video games, and the culture of play. Bloomington: Indiana University Press, pp. 63-91.

Waern, A. (2011). 'I'm in love with someone that doesn't exist!' Bleed in the context of a computer game. Journal of Gaming & Virtual Worlds, 3(3), 239-257. Available at: doi:10.1386/jgvw.3.3.239\_1.

Wagner, T. M., Benlian, A. and Hess, T. (2014). Converting freemium customers from free to premium: The role of the perceived premium fit in the

case of music as a service. Electronic Markets, 24, 259-268. Available at: doi:10.1007/s12525-014-0168-4.

Wajcman, J., Bittman, M., Jones, P., Johnstone, L. and Brown, J. (2007). The impact of the mobile phone on work/life balance. Canberra: Australian Mobile Telecommunications Association & Australian National University.

Wang, B., Taylor, L. and Sun, Q. (2018). Families that play together stay together: Investigating family bonding through video games. New Media & Society, 20(11), 4074-4094. Available at: doi:10.1177/1461444818767667.

Wang, J. J. (2020). How managers use culture and controls to impose a '996' work regime in China that constitutes modern slavery. Accounting & Finance, 60(4), 4331-4359. Available at: doi:10.1111/acfi.12682.

Warde, A. (1999). Convenience food: Space and timing. British Food Journal, 101(7), 518-527. Available at: doi:10.1108/00070709910279018.

Warde, A., Shove, E. and Southerton, D. (1998). Convenience, schedules and sustainability. Lancaster: The European Science Foundation Workshop on Consumption, Everyday Life and Sustainability, Lancaster University.

Weiboyi and Pangjing (2022). Trend Insight Report on Mainstream Social Media Platforms. [Online]. Weiboyi and Pangjing. Available at: https://www.199it.com/archives/1435001.html [Accessed 19 December 2023].

Wen, J., Kow, Y. M. and Chen, Y. (2011, September). Online games and family ties: Influences of social networking game on family relationship. In IFIP

Conference on Human-Computer Interaction. Springer, Berlin, pp. 250-264. Available at: doi:10.1007/978-3-642-23765-2 18.

Wertheimer, M. (1938). Gestalt theory. In Ellis, W. D. (Ed). A source book of Gestalt psychology. London: Kegan, Trench, Trubner, pp. 1-11. Available at: doi:10.1037/11496-001.

Whitson, J. R. (2013). Gaming the quantified self. Surveillance & Society, 11(1/2), 163-176. Available at: doi:10.24908/ss.v11i1/2.4454.

Williams, D., Ducheneaut, N., Xiong, L., Zhang, Y., Yee, N. and Nickell, E. (2006). From tree house to barracks: The social life of guilds in World of Warcraft. Games and Culture, 1(4), 338-361. Available at: doi:10.1177/1555412006292616.

Williams, D., Ducheneaut, N., Xiong, L., Zhang, Y., Yee, N. and Nickell, E. (2006). From tree house to barracks. Games and Culture, 1(4), 338-361. Available at: doi:10.1177/1555412006292616.

Williams, D., Martins, N., Consalvo, M. and Ivory, J. D. (2009). The virtual census: Representations of gender, race, and age in video games. New Media & Society, 11(5), 815-834. Available at:

doi:10.1177/1461444809105354.

Wilson, F. (2006a). My favorite business model. [Online]. A VC Blog. Last updated: March 2006. Available at:

http://avc.blogs.com/a\_vc/2006/03/my\_favorite\_bus.html [Accessed 19 December 2023].

Wilson, F. (2006b). The freemium business model. [Online]. A VC Blog. Last updated: 23 March 2006. Available at:

https://avc.com/2006/03/the\_freemium\_bu/ [Accessed 19 December 2023].

Witmer, B. G. and Singer, M. J. (1998). Measuring presence in virtual environments: A presence questionnaire. Presence, 7(3), 225-240. Available at: doi:10.1162/105474698565686.

Yang, F. (2019). The narrative study of video games: Using the example of The Legend of Zelda: Breath of the Wild.

Yang, L. (2018). Freemium mobile game profit model analysis: Take the example of Arena of Valor.

Yang, R. and Zhang, J. (Eds.). (2012). Southern weekend deep report. Nan fang ri bao chu ban she.

Yee, N. (2006). The Daedalus Project. [Online]. Available at: http://www.nickyee.com/daedalus [Accessed 1 February 2023].

Yee, N. (2006). The labor of fun: How video games blur the boundaries of work and play. Games and Culture, 1(1), 68-71. Available at: doi:10.1177/1555412005281819.

Yee, N. (n.d.). The virtual skinner box. [Online]. Available at: http://nickyee.com/eqt/skinner.html [Accessed 28 July 2023].

Yingchao (2019). NGA player community app product analysis. [Online]. Yingchao. Available at: https://www.jianshu.com/p/a085136e9975 [Accessed 19 December 2023]. Young, K. (2009). Understanding online gaming addiction and treatment issues for adolescents. The American Journal of Family Therapy, 37(5), 355-372. Available at: doi:10.1080/01926180902942191.

Zagal, J. P., Björk, S. and Lewis, C. (2013). Dark patterns in the design of games. In Yannakakis, G. N., Aarseth, E., Jorgensen, K. and Lester, J. C. (Eds). Proceedings of the 8th International Conference on the Foundations of Digital Games (FDG 2013). Society for the Advancement of the Science of Digital Games, pp. 39-46.

Zanescu, A., French, M. and Lajeunesse, M. (2020). Betting on DOTA 2's battle pass: Gamblification and productivity in play. New Media & Society, 23(10), 2882–2901. Available at: doi:10.1177/1461444820941381.

Zhang, J. (2014). The demonized world--reproduction of media images of online game players. Soochow University. Master's Thesis. Available at: https://www.ckcest.cn/default/es3/detail/1003/dw\_thesis\_copy/d0e7ade8a12e 629cc91b2bc86f882a94 [Accessed 10 May 2023].

Zhang, L. and Fung, A. Y. (2014). Working as playing? Consumer labor, guild, and the secondary industry of online gaming in China. New Media & Society, 16(1), 38-54. Available at: doi:10.1177/1461444813477077.

Zhihu (2014). How does the people in the game industry think of Zhengtu? [Online]. Zhihu. Available at: https://www.zhihu.com/question/24037357 [Accessed 28 September 2023].

Zimmerman, E. (2012). Jerked around by the magic circle: Clearing the air ten years later. [Online]. Gamasutra. Available at:

https://www.gamedeveloper.com/design/jerked-around-by-the-magic-circle--clearing-the-air-ten-years-later [Accessed 1 February 2023].

Zwick, D., Bonsu, S. K. and Darmody, A. (2008). Putting consumers to work: Co-creation and new marketing governmentality. Journal of Consumer Culture, 8(2), 163-196. Available at: doi:10.1177/1469540508090089.